

1

UNITED STATES NUCLEAR REGULATORY COMMISSION REGION III 799 ROOSEVELT ROAD GLEN ELLYN, ILLINOIS 60137

SEP 1 7 1982.

MEMORANDUM FOR:	James G. Keppler, Regional Administrator
THRU:	R. F. Warnick, Acting Director, Office of Special Cases
FROM:	W. D. Shafer, Chief, Midland Section
SUBJECT :	GAP COMMUNICATIONS (MS. BILLIE GARDE)

On September 17, 1982, I was requested to contact Ms. Billie Garde to answer some general questions about the Midland project. To the best of my recall, the following was discussed:

(1) She asked about the status of the six GAP affidavits.

I explained that the OI investigation was progressing and that some of the people had been contacted. I stated that when the investigations were completed that OI would turn the information over to our staff for technical review and inspection.

(2) She asked about the status of the Zack investigation.

I informed her that the investigation was progressing and that Midland had priority after LaSalle. I also told her that CPCo had a copy of the Zack affidavit. She said they did not get it from GAP.

She stated that she was very concerned that we have not pursued the issue as to whether CPCo should have reported the Zack problem under 10 CFR 50.55(e). I explained that this issue would be addressed in our investigation and inspection effort.

(3) We discussed several current issues at the site as follows:

- (a) Investigations
  - I stated that the investigation into the March 10, 1982 meeting where Messrs. Cook and Landsman alleged they had been lied to was nearing completion and that a final report would be forthcoming.

8406050016 840517 PDR FOIA RICE84-96 PDR 1

- I informed her that a request for an investigation into the potential violation of the board order had been forwarded to OI in HQ. I told her I did not know if an investigation had commenced.
- (b) Discussed the development of the Work Authorization Procedure

I stated that RIII had determined that a formal communications mechanism was needed to ensure that all work authorizations would be in writing.

(c) Pipe Support and Restraint Problems

I discussed Isa Yin's inspection report and CPCo's subsequent inspection findings in this area. I stated that we have informed CPCo that we want a 100% reinspection of all supports and restraints installed prior to 1981.

(d) Misrouted Electrical Cables

I stated that we had informed CPCo that a reinspection of all SR cable was mandatory.

(e) Midland Section

I identified the Midland Section personnel and stated that the remedial soils work interface was the highest priority we had. I also stated that we were waiting for CPCo's commitments for improving their program and that you would not allow any major soils work to proceed until the Midland Section was satisfied that the program was acceptable.

(4) After discussing item 3(e) above, Ms. Garde stated she was disappointed that GAP input was not solicited during the formation of the Midland Section. I stated that this was a management decision and could not comment further. However, I stated that she was welcome to contact me at anytime in order to ensure good communications. Ms. Garde stated she would like to meet with the Midland Section and would get back to me regarding when. I encouraged her to do so. She stated that open communications were very important in that when she made a press release she would be able to discuss what the NRC was doing. James G. Keppler

1

(5) Ms. Garde discussed the SSER and wanted to know if R. Landsman's concern about the board violation would be addressed in it. I said I doubted if it would. The SSER document would extensively identify the design elements for the remedial soil underpinning activities. We expected the SSER to be issued on October 4, 1982.

- 3 -

- (6) Ms. Garde stated that she was preparing to meet with D. Saunders and was trying to obtain his affidavit. I wished her good luck and stated that we would be reviewing the relevant allegations we have obtained from Mr. Saunders.
- (7) I informed Ms. Garde that our section was developing a monthly status report which would indicate the status of RIII's effort at Midland. I told her the report would be docketed and if she wanted access to it she would have to request it through formal channels. She said she would do that.

I believe that this summary was the extent of our conversation. It was not necessarily in the order I have described above, but I do believe I have covered the most salient issues. I intend to send Ms. Garde a copy of this summary.

Should you have any questions regarding this communication, I will be happy to discuss them with you.

Midland Section

cc: A. B. Davis



Your The Council fel



UNITED STATES NUCLEAR REGULATORY COMMISSION REGION III 799 ROOSEVELT ROAD GLEN ELLYN, ILLINOIS 60137

TIUN 2 8 1983

Lone Tree Council ATTN: Mr. Leo R. Romo Corresponding Secretary P. O. Box 421 Essexville, MI 48732

Dear Mr. Romo:

\$307050184

This is in response to your letter dated May 31, 1983, in which you expressed your thoughts about Consumers Power Company's independent third party design and construction verification program (IDCVP) and the construction completion program (CCP) including the independent third party construction implementation overview (CIO).

The IDCVP at Midland is a detailed examination of design adequacy and construction quality using as a basis three safety-related systems. A copy of TERA's plan for the IDCVP is attached for your information (Attachment 1). The IDCVP proposed for Midland is similar to, but more extensive than, independent reviews conducted at other plants. TERA selected the auxiliary feedwater system using the selection criteria on pages 14 and 15 of Attachment 1. The NRC agrees that these are appropriate selection criteria. The NRC selected a second system, standby electrical power, for the IDCVP from the three candidate systems identified by Consumers Power Company. The three candidate systems were selected by Consumers Power Company based on systems important to probabilistic risk assessment analysis. The third system, control room heating, ventilation, and air conditioning, was selected independently by the NRC, not from the candidate systems. Our choice of the second and third systems was made in part after considering suggestions made by members of the public. A copy of TERA's first status report (Attachment 2), is also attached for your information and as you can see, the TERA team already has identified and confirmed items on the auxiliary feedwater system which may become significant findings. The extent to which TERA will examine construction is detailed in Section 3.2 of Attachment 1. The program scope for the three selected systems will concentrate on the criteria of the as-built conditions versus the design criteria. Additional sampling and verification, however, will be conducted on other systems if discrepancies or problems are found allowing program flexibility as discussed in Section 3.2.7 of Attachment 1.

#### Lone Tree Council

## JUN 2 8 19831

30

You have also raised a concern regarding the NRC's permitting Stone and Webster Engineering Corporation (SWEC) to start the CIO prior to the NRC having completed its review of the Consumers Power Company proposal for the third party overview. Region III felt it was desirable to have the overview program begin in order to assure that Consumers Power Company CCP systems are working properly. The SWEC overview presently involves nine people onsite performing the CIO. These activities started on April 28, 1983. The CIO is not a 100% inspection, rather it is an audit of Consumers Power Company's implementation of the CCP. In addition, the NRC Midland team will be reviewing the quality of the CPCo CCP and the SWEC CIO to assure adequate implementation. It should also be noted that the CCP has not been approved to date by the NRC. CPCo CCP activities have been limited to program preparation and some training.

- 2 -

We believe the actions being taken should provide assurance to the community that the plant has been constructed safely. We believe the intent of the three actions you suggest will be accomplished by the CCP and the third party programs in place.

We hope this is responsive to your concerns. If you have further questions, please do not hesitate to contact us.

Sincerely,

Original signed by A. Bert Davis

James G. Keppler Regional Administrator

Actachment: As stated

ce w/attachment and ltr dtd 5/31/83: See attached distribution list

			1				Der felecon D. Hord JA 6/25/83 4
OFFICE	RIII	JAH.	NH In	101	111.		٥
SURNAME	Gardner/1s	Harrison	Warnick	Strama	Stille"		~
BATE	6/27/83	6/27/83	6/21/83	6/27/35	6/27	L/27	Keppler
NRC FORM	318 110. 801 NRCM	8240	OFFICIAI	PECOPO	CUBA		* 1107 1920 - 200

- 3 -

Lone Tree Council

cc w/attachment and ltr dtd 5/31/83: DMB/Document Control Desk (RIDS) Resident Inspector, RIII The Honorable Charles Bechhoefer, ASLB The Honorable Jerry Harbour, ASLB The Honorable Frederick P. Cowan, ASLB The Honorable Ralph S. Decker, ASLB William Paton, ELD Michael Miller Ronald Callen, Michigan Public Service Commission Myron M. Cherry Barbara Stamiris Mary Sinclair Wendell Marshall Colonel Steve J. Gadler (P.E.) Howard Levin, TERA Billie P. Garde, Government Accountability Project Lynne Bernabei, Government Accountability Project

ENGINEERING PROGRAM PLAN PROJECT INSTRUCTION PI-3201-009 MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM PROJECT 3201

MAY 18, 1983 REVISION: 2

(

017 COPY NO.

158pp

8345244567

ATTACHMENT 1

SEC	TION					PAGE
1.0	GEN	ERAL				1
	1.1 1.2 1.3 1.4	BACK OVER SYSTE	GROUND VIEW OF EMS SELE PENDENC	AND PUR IDCV SCOP CTION CRI E REQUIRI	POSE - PE - ITERIA EMENTS	1 3 14 15
2.0	ORC	GANIZA	TION AND	CONTRO	L	17
	2.1 2.7 2.3	PROJE AUTH ADMIN	ECT ORG	ANIZATION ND RESPON VE CONTR	N NSIBILITY OL	17 19 21
3.0	ENC	21				
	3.1	3.1 INDEPENDENT DESIGN VERIFICATION				
		3.1.1	CATEG	ORIES OF I	REVIEW: THE DESIGN	24
			3.1.1.1	Review of Commitm	f Design Criteria and ents	25
			3.1.1.2	Review of Document	f Implementing	25
			3.1.1.3	Check of Evaluation	Calculations and	26
			3.1.1.4	Confirma Evaluation	tory Calculations or ns	27
			3.1.1.5	Check of Specificat	Drawings and tions	28
		3.1.2	BASES I	FOR SAMP	LE SELECTION	28
		3.1.3	DEFINIT AFW SY	TION OF RI	EVIEW SCOPE FOR THE	30
			3.1.3.1	AFW Syst	em Performance Criteria	33
				3.1.3.1.1	System Operating Limits - Topic I.I-I	33
				3.1.3.1.2	Accident Analysis Considerations – Topic 1.2–1	• 33
				3.1.3.1.3	Single Failure - Topic I.3-1	34
				3.1.3.1.4	Technical Specifica- tions - Topic 1.4-1	34

( :

i

#### (continued)

SECTION

3.1.3.1.5	System Alignment/ Switchover - Topic 1.5-1	35
3.1.3.1.6	Remote Operation and Shutdown - Topic 1.6-1	35
3.1.3.1.7	System Isolation/ Interlocks - Topic 1.7-1	35
3.1.3.1.8	Overpressure Protec- tion -Topic 1.8-1	36
3.1.3.1.9	Component Functional Requirements – Topic 1.9–1	36
3.1.3.1.10	System Hydraulic Design -Topic 1.10-1	37
3.1.3.1.11	System Heat Removal Capability - Topic 1.11-1	37
3.1.3.1.12	Cooling Requirements - Topic 1.12-1	37
3.1.3.1.13	Water Supplies - Topic 1.13-1	37
3.1.3.1.14	Preservice Testing and Capability for Operational Testing – Topic 1.14–1	38
3.1.3.1.15	Power Supplies - Topic 1.15-1	38
3.1.3.1.16	Electrical Character- istics - Topic 1.16-1	38
3.1.3.1.17	Protective Devices/ Settings - Topic 1.17-1	39
3.1.3.1.18	Instrumentation - Topic 1.18-1	39
3.1.3.1.19	Control Systems - Topic 1.19-1	40
3.1.3.1.20	Actuation Systems - Topic 1.20-1	40
3.1.3.1.21	Nondestructive Exam- ination Commitments - Topic 1.21-1	. 41
3.1.3.1.22	Materials Selection - Topic 1.22-1	41
3.1.3.1.23	Failure Modes and Effects - Topic 1.23-1	41

### (continued)

# SECTION

1 ...

AFW Syst	em Protection Features	42
3.1.3.2.1	Seismic Design - Topic II.1-1	42
3.1.3.2.2	Seismic Design/Pressure Boundary Integrity – Topic II.2–1	43
3.1.3.2.3	Seismic Design/Pipe and Equipment Support – Topic II.3–1	43
3.1.3.2.4	Seismic Design/Equip- ment Qualification – Topic II.4–1	44
3.1.3.2.5	High Energy Line Break Accidents -Topic 11.5-1	44
3.1.3.2.6	HELBA/Pipe Whip - Topic II.6-1	45
3.1.3.2.7	HELBA/Jet Impingement - Topic II.7-1	45
3.1.3.2.8	Environmental Protec- tion - Topic II.8-1	45
3.1.3.2.9	Environmental Enve- lopes - Topic 11.9-1	46
3.1.3.2.10	Environmental/Equip- ment Qualification - Topic II.10-1	46
3.1.3.2.11	HVAC Design - Topic II.II-1	47
3.1.3.2.12	Fire Protection - Topic II.12-1	47
3.1.3.2.13	Missile Protection - Topic II.13-1	48
3.1.3.2.14	Systems Interaction - Topic II.14-1	48
Structures System	that House the AFW	48
3.1.3.3.1	Seismic Design/Input to Equipment - Topic III.I-I	49
3.1.3.3.2	Wind and Tornado Design/Missile Protection - Topic III.2-1	49
	AFW Syst 3.1.3.2.1 3.1.3.2.2 3.1.3.2.2 3.1.3.2.3 3.1.3.2.4 3.1.3.2.4 3.1.3.2.5 3.1.3.2.6 3.1.3.2.7 3.1.3.2.7 3.1.3.2.7 3.1.3.2.10 3.1.3.2.10 3.1.3.2.11 3.1.3.2.12 3.1.3.2.12 3.1.3.2.13 3.1.3.2.14 Structures System 3.1.3.3.1 3.1.3.3.2	<ul> <li>AFW System Protection Features</li> <li>3.1.3.2.1 Seismic Design - Topic II.1-1</li> <li>3.1.3.2.2 Seismic Design/Pressure Boundary Integrity - Topic II.2-1</li> <li>3.1.3.2.3 Seismic Design/Pipe and Equipment Support - Topic II.3-1</li> <li>3.1.3.2.4 Seismic Design/Equip- ment Qualification - Topic II.4-1</li> <li>3.1.3.2.5 High Energy Line Break Accidents -Topic II.5-1</li> <li>3.1.3.2.6 HELBA/Pipe Whip - Topic II.6-1</li> <li>3.1.3.2.7 HELBA/Jet Impingement - Topic II.7-1</li> <li>3.1.3.2.8 Environmental Protec- tion - Topic II.8-1</li> <li>3.1.3.2.9 Environmental Enve- lopes - Topic II.9-1</li> <li>3.1.3.2.10 Environmental/Equip- ment Qualification - Topic II.10-1</li> <li>3.1.3.2.11 HVAC Design - Topic II.10-1</li> <li>3.1.3.2.12 Fire Protection - Topic II.11-1</li> <li>3.1.3.2.14 Systems Interaction - Topic II.13-1</li> <li>3.1.3.3.1 Seismic Design/Input to Equipment - Topic III.1-1</li> <li>3.1.3.3.2 Wind and Tornado Design/Missile Protection - Topic III.2-1</li> </ul>

ſ

#### (continued)

SECTION

3.1.4

	3.1.3.3.3	Flood Protection - Topic III.3-1	50
	3.1.3.3.4	HELBA Loads - Topic III.4-1	50
	3.1.3.3.5	Civil/Structural Design Considerations – Topic III.5–1	50
	3.1.3.3.6	Foundations - Topic III.6-1	51
	3.1.3.3.7	Concrete/Steel Design - Topic III.7-1	51
DEFINI	TION OF R	EVIEW SCOPE FOR THE RIC POWER SYSTEM	52
3.1.4.1	SEP Syste	m Performance Requirements	54
	3.1.4.1.1	System Operating Limits (DG) - Topic 1.1-2	55
	3.1.4.1.2	Accident Analysis Considerations (DG, AC, DC) - Topic 1.2-2	55
	3.1.4.1.3	Single Failure (DG, PDS, AC, DC) - Topic 1.3-2	56
	3.1.4.1.4	Technical Specifications (DG, DC) - Topic 1.4-2	56
	3.1.4.1.5	Load Operation (DG) - Topic 1.6-2	56
	3.1.4.1.6	System Interlocks (DG) - Topic 1.7-2	56
	3.1.4.1.7	Component Functional Re- quirements (DG, PDS, AC, DC) - Topic 1.9-2	57
	3.1.4.1.8	Cooling/Heating Require- ments (DG) - Topic 1.12-2	57
	3.1.4.1.9	Preservice Testing and Capability for Operational Testing (DG) - Topic 1.14-2	58
	3.1.4.1.10	Electrical Characteristics (DG, PDS, AC, DC) - Topic 1.16-2	58
	3.1.4.1.11	Protective Devices/Settings (DG, PDS) - Topic 1.17-2	58

#### (continued)

SECTION

( :

	3.1.4.1.12	Instrumentation (DG, AC, DC) - Topic 1.18-2	59
	3.1.4.1.13	Control Systems (DG) - Topic 1.19-2	59
	3.1.4.1.14	Actuation Systems (DG) – Topic 1.20–2	60
	3.1.4.1.15	Failure Modes and Effects (DG, PDS, AC, DC) - Topic 1.23-2	60
N	3.1.4.1.16	Electrical Load Capacity (DG, PDS, AC, DC) - Topic 1.24-2	61
	3.1.4.1.17	Electrical Load Sequencing (DG, PDS) - Topic 1.25-2	61
	3.1.4.1.18	Electrical Load Shedding (DG, PDS) - Topic 1.26-2	61
	3.1.4.1.19	Fuel Oil System (DG) - Topic I.27-2	62
	3.1.4.1.20	Lube Oil System (DG) - Topic I.28-2	62
	3.1.4.1.21	Starting Mechanism and Air Supply System (DG) - Topic 1.29–2	62
	3.1.4.1.22	Combustion Air Supply (DG) - Topic 1.30-2	63
	3.1.4.1.23	Independence (DG, PDS, AC, DC) - Topic 1.31-2	63
	3.1.4.1.24	Cable Sizing/Routing/ Separation (PDS) – Topic 1.32–2	64
3.1.4.2	SEP Syste	m Protection Features	64
	3.1.4.2.1	Seismic Design - Topic II.I-2	64
	3.1.4.2.2	Seismic Design - Pressure Boundary (DG) - Topic II.2-2	65
-	3.1.4.2.3	Seismic Design — Pipe/ Equipment Support (DG, PDS) - Topic 11.3-2	65
1	3.1.4.2.4	Seismic Design – Equipment Qualification (DG, PDS) –	66
	3	i obic III 4-2	

#### (continued)

SECTION

	3.1.4.2.5	High Energy Line Break Accidents - Topic 11.5-2	66
	3.1.4.2.6	HELBA/Pipe Whip (PDS, AC, DC) - Topic II.6-2	66
	3.1.4.2.7	HELBA/Jet Impingement (PDS, AC, DC) - Topic II.7-2	66
	3.1.4.2.8	Environmental Protection - Topic II.8-2	67
	3.1.4.2.9	Environmental Envelopes (DG, PDS) - Topic II.9-2	67
	3.1.4.2.10	Environmental/Equipment Qualification (DG, PDS) - Topic II.10-2	67
	3.1.4.2.11	HVAC Design (DG) - Topic II.II-2	67
	3.1.4.2.12	Fire Protection (DG) - Topic II.12-2	68
	3.1.4.2.13	Missile Protection (DG) - Topic II.!3-2	68
	3.1.4.2.14	Systems Interaction (DG, PDS, AC, DC) - Topic 11.14-2	68
3.1.4.3	Structures	that House the SEP Systems	69
	3.1.4.3.1	Seismic Design/Input to Equipment (DGB) - Topic III.1-2	69
	3.1.4.3.2	Wind and Tornado Design/Missile Protection (DGB) - Topic III.2-2	70
	3.1.4.3.3	Flood Protection (DGB) - Topic III.3-2	70
	3.1.4.3.4	HELBA Loads (DGB) - Topic III.4-2	71
	3.1.4.3.5	Civil/Structural Design Considerations - Topic III.5-2	71
	3.1.4.3.6	Foundations (DGB) - Topic III.6-2	71
	3.1.4.3.7	Concrete and Steel Design (DGB) - Topic 111.7-2	72
	3.1.4.3.8	Tanks - Topic III.8-2	72

vi

#### (continued)

# SECTION

3.1.5	DEFINITION IF REVIEW SCOPE FOR THE CONTROL ROOM HVAC SYSTEM					
	3.1.5.1	CR-HVAC Requirem	C System Performance lents	75		
		3.1.5.1.1	System Operating Limits - Topic I.I-13	75		
		3.1.5.1.2	Accident Analysis Consi- derations - Topic 1.2-3	75		
		3.1.5.1.3	Single Failure - Topic 1.3-3	76		
		3.1.5.1.4	Technical Specifications – Topic 1.4–3	76		
		3.1.5.1.5	System Alignment/Switchover - Topic 1.5-3	76		
		3.1.5.1.6	System Isolation/Interlocks - Topic 1.7-3	76		
		3.1.5.1.7	Component Functional Require- ments - Topic 1.9-3	77		
		3.1.5.1.8	System Pneumatic Design – Topic II.10–3	77		
		3.1.5.1.9	Cooling/Heating Requirements - Topic 1.12-3	77		
		3.1.5.1.10	Preservice Testing and Capability for Operational Testing - Topic 1.14-3	78		
		3.1.5.1.11	Power Supplies - Topic 1.15-3	78		
		3.1.5.1.12	Instrumentation/Detection - Topic 1.18-3	78		
		3.1.5.1.13	Control Systems - Topic 1.19-3	78		
		3.1.5.1.14	Actuation Systems - Topic 1.20-3	79		
		3.1.5.1.15	Nondestructive Examination Commitments - Topics 1.21-3	79		
		3.1.5.1.16	Materials Selection Topic 1.22-3	80		
		3.1.5.1.17	Failure Modes and Effects - Topic 1.23-3	80		
		3.1.5.1.18	Filtration - Topic 1.33-3	80		

## TABLE OF CONTENTS (continued)

# SECTION

Ć

# PAGE

	3.1.5.1.19	Pressurization - Topic 1.34-3	80
	3.1.5.1.20	Ventilation - Topic 1.35-3	81
3.1.5.2	CR-HVAC	System Protection Features	81
	3.1.5.2.1	Seismic Design - Topic II.1-3	82
	3.1.5.2.2	Seismic Design – Pressure Boundary - Topic II.2-3	82
	3.1.5.2.3	Seismic Design Duct/Pipe/ Equipment Support - Topic II.3-3	82
	3.1.5.2.4	Seismic Design – Equipment Qualification – Topic II.4–3	83
	3.1.5.2.5	High Energy Line Break Accidents - Topic II.5-3	83
	3.1.5.2.6	HELBA/Pipe Whip - Topic II.6-3	83
	3.1.5.2.7	HELBA/Jet Impingement	84
	3.1.5.2.8	Environmental Protection - Topic II.8-3	84
	3.1.5.2.9	Environmental Envelopes - Topic II.9-3	84
	3.1.5.2.10	Environmental Equipment Qualification - Topic II.10-3	85
	3.1.5.2.11	Fire Protection - Topic II.12-3	85
	3.1.5.2.12	Missile Protection - Topic II.13-3	85
	3.1.5.2.13	Systems Interaction - Topic II.14-3	86
3.1.5.3	Structures	that House the CR-HVAC System	86
	3.1.5.3.1	Seismic Design/Input to Equipment - Topic III.1-3	86
	3.1.5.3.2	Civil/Structural Design Considerations-Topic 111.5-3	87
	3.1.5.3.3	Concrete and Steel Design Topic III.7-3	87
	3.1.5.3.4	Leak Tightness - Topic III.9-3	88

viii

#### (continued)

# SECTION

	3.1.6	DEVELOPMENT OF IDV PROGRAM CHECKLISTS					
		3.1.6.1	Developm of Design	nent of Checklists for Review Criteria and Commitments	89		
		3.1.6.2	Developm of Implen	nent of Checklists for Reviews nenting Documents	90		
		3.1.6.3	Developm of Calcul	nent of Checklists for Checks ations and Evaluations	91		
		3.1.6.4	Developm of Drawin	nent of Checklists for Checks ngs and Specifications	92		
	3.1.7	PLAN PLAN	OR ADDIT	IONAL SAMPLING AND	93		
3.2	METH	INDEPENDENT CONSTRUCTION VERIFICATION METHDOLOGY					
	3.2.1	CATEG	ORIES OF REVIEW: THE CON-				
		3.2.1.1	Review of tion	f Fabrication Documenta-	96		
		3.2.1.2	Review of Document	Review of Storage and Maintenance Documentation			
		3.2.1.3	Review of Construction/ Installation Documentation				
		3.2.1.4	Review of Activities	f Selected Verification	97		
		3.2.1.5	Verificati Configura	on of Physical Ition	98		
	3.2.2	BASES	FOR SAMP	LE SELECTION	98		
	3.2.3	DEFINI AFW SY	TION OF R	EVIEW SCOPE FOR THE	99		
		3.2.3.1	Mechanic	al Systems and Components	100		
			3.2.3.1.1	Mechanical Equipment -	100		
				Topic I.I-Ic			
			3.2.3.1.2	Piping -	101		
				Topic I.2-lc			
			3.2.3.1.3	Pipe Supports -	101		
				Topic I.3-Ic			
		3.2.3.2	Electrical	Systems and Components	102		
			3.2.3.2.1	Electrical Equipment - Topic II.I-Ic	102		

#### (continued)

SECTION

		3.2.3.2.2	Cable Trays and Supports -	103
		3.2.3.2.3	Conduits and Supports -	104
			Topic II.3-Ic	
		3.2.3.2.4	Cable -	104
			Topic II.4-Ic	
	3.2.3.3	and Com	ntation and Control Systems ponents	105
		3.2.3.3.1	Instruments -	105
			Topic III.I-Ic	
		3.2.3.3.2	Piping/Tubing -	105
			Topic III.2-Ic	105
		3.2.3.3.3	Cable -	104
			Topic III.3-Ic	100
	3.2.3.4	HVAC Sy	stems and Components	104
		3.2.3.4.1	HVAC Equipment	100
			Topic IV.I-IC	107
		3.2.3.4.2	HVAC Ducts and Supports	107
			Topic IV.2-Ic	107
	3.2.3.5	Structura	Components	100
		3.2.3.5.1	Foundations -	108
			Topic V.I-Ic	108
		3.2.3.5.2	Concrete Componente	100
			Topic V 2-la	108
		3.2.3.5.3	Structural Steel	
			Components	109
			Topic V 2 Is	
	3.2.3.6	NDE/Mat	Topic V.S-IC	
224	0.2.0.0	Topic VI.I	-le	109
3.2.4	STAND	BY ELECTR	EVIEW SCOPE FOR THE RIC POWER SYSTEM	Ш
	3.2.4.1	Mechanica	al Systems and Components	112
		3.2.4.1.1	Mechanical Equipment - Topic 1.1-2c	. 112
		3.2.4.1.2	Piping - Topic 1.2-2c	112
		3.2.4.1.3	Pipe Supports - Topic 1.3-2c	113
	3.2.4.2	Electrical	Systems and Components	112
		3.2.4.2.1	Electrical Equipment - Topic II.1-2c	113

DC-82-13

#### (continued)

SECTION

3.2.5

( .

	3.2.4.2.2	Cable Trays/Conduits and Supports - Topics II.2-2c and II.3-2c	114
	3.2.4.2.3	Cable - Topic II.4-2c	114
3.2.4.3	Instrumen and Comp	tation and Control Systems onents	114
	3.2.4.3.1	Instruments - Topic III.1-2c	115
	3.2.4.3.2	Piping/Tubing - Topic III.2-2c	115
	3.2.4.3.3	Cable - Topic III.3-2c	116
3.2.4.4	HVAC Sys	items and Components	116
	3.2.4.4.1	HVAC Equipment - Topic IV.I-2c	116
	3.2.4.4.2	HVAC Ducts and Supports - Topic IV.2-2c	117
3.2.4.5	Structural	Components	117
	3.2.4.5.1	Foundations - Topic V.1-2c	117
	3.2.4.5.2	Concrete Components - Topic V.2-2c	118
	3.2.4.5.3	Structural Steel Components - Topic V.3-2c	118
DEFINI	DL ROOM	EVIEW SCOPE FOR THE HVAC SYSTEM	118
3.2.5.1	Mechanica	al Systems and Components	119
	3.2.5.1.1	Mechanical Equipment - Topic 1.1-3c	119
	3.2.5.1:2	Piping - Topic 1.2-3c	120
	3.2.5.1.3	Pipe Supports - Topic 1.3-3c	120
3.2.5.2	Electrical	Systems and Components	120
	3.2.5.2.1	Electrical Equipment - Topic II.1-3c	121
	3.2.5.2.2	Trays/Conduit and Supports - Topics 11.2-3c and 11.3-3c	121
	3.2.5.2.3	Cable - Topic II.4-3c	122
3.2.5.3	Instrument	tation and Control Systems and .	122
	3.2.5.3.1	Instruments - Topic III.I-3c	123
	3.2.5.3.2	Piping and Tubing - Topic III.2-3c	123
	3.2.5.3.3	Cable - Topic III.3-3c	123

(continued)

F	~	T	10	AI	
<pre>C</pre>	5	1	i U	14	

5

PAGE

		3.2.5.4	HVAC		124
			3.2.5.4.1	HVAC Ducts and Supports - Topic IV.2-3c	124
		3.2.5.5	Structura	l Components	124
			3.2.5.5.1	Concrete Components - Topic V.2-3c	125
			3.2.5.5.2	Structural Steel Components - Topic V.3-3c	125
		3-2.5.6	NDE/Mate Topic VI.I	erial Testing Program – –3c	125
	3.2.6	CHECK	DPMENT OF	F ICV PROGRAM	125
		3.2.6.1	Developm of Supplie	ent of Checklists for Review r Documentation	127
		3.2.6.2	Developm of Storage tation	ent of Checklists for Review and Maintenance Documen-	128
		3.2.6.3	Developm of Constru Document	ent of Checklists for Review action and Installation ration	129
		3.2.6.4	Developm of Selecte	ent of Checklists for Review d Verification Activities	130
		3.2.6.5	Developm of Verific	ent of Checklists for Review ation of Physical Configuration	131
	3.2.7	PLAN F	OR ADDIT	IONAL SAMPLING, ND TESTING	132
DOC	UMENT	ATION A	ND PROTO	COL	134
4.1	DOCU TIONS TION F	MENITAT	ION OF EN	GINEERING EVALUA- AND FIELD VERIFICA-	134
4.2	DOCU	MENTAT	ION AND P	ROTOCOL FOR EXTERNAL	135

4.0

#### (continued)

#### SECTION PAGE 5.0 PROGRAM REPORTING 136 5.1 TYPES OF REPORTS 5.2 REPORTING PROCESS 136 138 5.2.1 REPORTING SYSTEM 5.2.2 REPORT PREPARATION AND DISTRIBUTION 138 142 5.2.3 INTERCHANGE OF INFORMATION 5.3 IDENTIFICATION AND EVALUATION OF DESIGN 143 OR CONSTRUCTION DEFICIENCIES 143 6.0 QUALITY ASSURANCE 144 6.1 APPLICABLE REQUIREMENTS 6.2 VERIFICATION OF COMPUTER CODES 144 6.3 AUDITS 144 145

P	ROJECT INSTRUCTIO	N
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineerin Midland Inder Construction	ng Program Plan pendent Design and Verification Program
PAGE _1 of _145	PREPARED BY: 141-1000	APPROVED BY:

#### 1.0 GENERAL

#### 1.1 BACKGROUND AND PURPOSE

The Nuclear Regulatory Commission (NRC) issued a letter on July 9, 1982 which requested that Consumers Power Company (CPC) provide for an independent assessment of the design adequacy of the Midland plant. CPC responded to this request on October 5, 1982 by submitting an outline of the scope of a proposed independent review program. A public meeting was held on October 25, 1982 at the NRC's Bethesda, Maryland offices to discuss details of the proposed program, the scope of which included an evaluation of the Midland Unit 2 Auxiliary Feedwater (AFW) system. During this meeting, the NRC requested that the scope of the independent design assessment program be expanded, including an assessment of the quality of construction. The NRC requested that CPC identify three candidate systems for scope expansion based upon their contribution to plant risk, from which one system would be selected.

CPC responded to NRC by a letter dated December 3, 1982 which identified the Standby Electric Power system (diesel generator), Safeguards Chilled Water system and Containment Isolation system as candidate systems. A public meeting was held on February 8, 1983 at Midland, Michigan to discuss details of the program related to the evaluation of the AFW system and to provide status.

On March 22, 1983 the NRC selected the Standby Electric Power system and the Control Room HVAC system for scope expansion. Proposed elements of the scope of evaluation for these systems as well as the AFW system were discussed at another public meeting held on April 13, 1983 at the NRC's Bethesda, Maryland offices.

	P	ROJECT IN	STRUCTION	
PI REV.: 2	DATE: 5/18/83	SUBJECT:	Engineering Program P Midland Independent Desir	lan gn and
PAGE _2	of 145	PREPARED BY:	APPROVED I	Y:

TERA Corporation has been selected by CPC to scope, manage, and implement the Midland Independent Design and Construction Verification (IDCV) Program. By a letter dated May 3, 1983, the NRC approved the selection of TERA. The selection is based upon the firm's technical qualifications, experience, and independence from the Midland project. Such independence includes all individuals who may contribute to the IDCV Program.

This project instruction, or Engineering Program Plan (the Plan), has been established to outline the scope, philosophy of review, methodology, independence requirements, organization, control, documentation, reporting, and quality assurance requirements for the Midland IDCV Program.

The IDCV approach selected is a review and evaluation of a detailed "vertical slice" of the Midland project with a focus on providing an overall assessment of the quality of the design and the constructed plant. Therefore, the primary emphasis of the IDCV evaluation is on the end results of the design and construction process and not on an evaluation of the process itself which is typical of the more common quality assurance audit. The "vertical slice" constitutes a carefully selected sample of three safety systems from which the results of the IDCV may be extrapolated to other similarly designed and constructed systems. Thus, the IDCV is intended to provide the necessary assurance to CPC, NRC, and the public that the Midland Plant is designed and constructed such that it is capable of functioning in accordance with its safety design bases and that applicable licensing commitments have been properly implemented.

P	ROJECT INS	TRUCTION
PI	SUBJECT:	Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE _ 3 of _145	PREPARED BY:	ILT 1 - APPROVED BY: Bert

#### 1.2 OVERVIEW OF IDCV SCOPE

The Midland IDCV consists of two major components: the Independent Design Verification (IDV) Program and the Independent Construction Verification (ICV) Program. The Unit 2 Auxiliary Feedwater (AFW) system, the Standby Electric Power (SEP) system and the Control Room Heating, Ventilating and Air Conditioning (CR-HVAC) system related to control room habitability have been selected as applicable samples of the design engineering and construction efforts at the Midland plant. The AFW system was selected by TERA based upon the system selection criteria discussed in Section 1.3 of this Plan. The SEP and control room HVAC systems selected by the CPC and NRC have a sufficiently high profile for each of these criteria to justify their selection.

The scope of review corresponds directly to the design and construction chains, addressing major activities and outputs of the various contributing engineering and construction disciplines. Accordingly, the products of the design and construction process, from concept to installation, hydros, functional and preoperational testing and turnover will be evaluated. Interfaces between CPC, Babcock and Wilcox (B&W), the nuclear steam system supplier (NSSS) vendor, Bechtel, the architect-engineer (A-E), and other contractors will be identified and evaluated relative to such items as the proper transfer and interpretation of design or construction information.

Figure 1.2-1 shows the inter-relationship between the Midland design and construction process and the Midland IDCV program. The following figures present the IDCV scope for the AFW, SEP and CR-HVAC systems in the form of matrices which identify the initial level of review and evaluation in each design or construction area respectively:

# INTER-RELATIONSHIP BETWEEN THE MIDLAND DESIGN AND CONSTRUCTION PROCESS AND THE MIDLAND IDCV PROGRAM



DESIGN AND CONSTRUCTION PROCESS

MIDLAND IDCV PROGRAM

FIGURE 1.2-1

# INITIAL SAMPLE REVIEW MATRIX FOR THE AUXILIARY FEEDWATER SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM

			1	sc	OPE OF	REVIEW	1
TOPIC MIL	DESIGN AREA	REVIEW OF DEC	REVIEW OF	CHECK OF C.	CONFIRMATIONS ONS	CHECK OF CALCULATION	SPECIFICATIONS AND
	AFW SYSTEM PERFORMANCE REQUIREMENTS						
1.1-1	SYSTEM OPERATING LIMITS	×	×	×	inter i		
1.2-1	ACCIDENT ANALYSIS CONSIDERATIONS	X	•				
1.3-1	SINGLE FAILURE	×	×	×	•	1.1.1	
1.4-1	TECHNICAL SPECIFICATIONS	×	×				
1.5-1	SYSTEM ALIGNMENT/SWITCHOVER	×	×			1.1.1.1	
1.6-1	REMOTE OPERATION AND SHUTDOWN	x					
1.7-1	SYSTEM ISOLATION/INTERLOCKS	×	X.		1.11		
1-8-1	OVERPRESSURE PROTECTION	×				1.1.1	
1.9-1	COMPONENT FUNCTIONAL REQUIREMENTS						
1.10-1	SYSTEM HYDRAULIC DESIGN	1 ÷	1 Ç	10		1	
1.11-1	SYSTEM HEAT REMOVAL CAPABILITY	1 x	1 ÷	2			
1.12-1	COOLING REQUIREMENTS	x		1			
1.13-1	WATER SUPPLIES	×				1.0	
1.14-1	PRESERVICE TESTING/CAPABILITY FOR						
	OPERATIONAL TESTING	×	•	•	1.0	1.1	
1.13-1	POWER SUPPLIES	×	×		2.22	•	/
1.17.1	PROTECTIVE DEVICES	×		•	10.55	1.1	
1+1 /=1	PROTECTIVE DEVICES/SETTINGS	×	×	12.53		×	
1.18-1	INSTRUMENTATION	×	x	X	14	×	
1.19-1	CONTROL SYSTEMS	×	×	×			
1.20-1	ACTUATION SYSTEMS	×			1.1.1.1.1.1		1
1.21-1	NDE COMMITMENTS	×					
1.22-1	MATERIALS SELECTION	×	×	1.1.1.1			
1.23-1	FAILURE MODES AND EFFECTS						/

#### KEY

#### NOTE

X - INITIAL SCOPE OF REVIEW

DELETED SCOPE OF REVIEW
 - ADDED SCOPE OF REVIEW

I. INITIAL SAMPLE DOCUMENTED IN REV. 0 AND I OF THIS PLAN HAS BEEN MODIFIED EFFECTIVE 4/13/83

FIGURE 1.2-2a

# 'NITIAL SAMPLE REVIEW MATRIX FOR THE AUXILIARY FEEDWATER SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM (CONTINUED)

		and the second second	0		1
x	1997				
x	x	x	x	x	
x	×	x	x	X	
x	x	x		x	
×					
x	x	x	1. S.	x	
x					
x					
×	×	x	×	x	
x	×	x		x	
×			1.00		
×	×	x			
×					
x	×	x			
×	×	×		×	
x					
×	1	1.1.1	14 (de 14)		
x		1.1			1
~	1.15				1
2	~	~			
2	2	~			1
â	â	â		. *	1
0	0	0			1
	×××× ×××⊗	× × × × × × × × × × × × × × × × × × ×	x x x x x x x x x x x x x x x x x x x	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

. DELETED SCOPE OF REVIEW

. ADDED SCOPE OF REVIEW

INITIAL SAMPLE DOCUMENTED IN REV. 0 AND 1 OF THIS PLAN HAS BEEN MODIFIED EFFECTIVE 4/13/83

## INITIAL SAMPLE REVIEW MATRIX FOR THE AUXILIARY FEEDWATER SYSTEM MIDLAND INDEPENDENT CONSTRUCTION VERIFICATION PROGRAM<sup>1</sup>

Min	SYSTEM, COMPONENT	EVIEW	UCUMENTATION	EW OF COUNTENTATION OF	EVIEW OD CUMENTION	FICATION SELECTED A STINITIES A SILATION ACTIVITIES A SILATION OF PHYSICAL
1001 1001	MECHANICAL • EQUIPMENT • PIPING	×××	× MANVIE	× × INUSTALL	X X	× ×
1.3-1c	PIPE SUPPORTS  ELECTRICAL	×		×	×	×
II.1-1c II.2-1c II.3-1c II.4-1c	EQUIPMENT     TRAYS AND SUPPORTS     CONDUIT AND SUPPORTS     CABLE	× × × × ×	× ×	× • • ×	× • • ×	x x x x
	INSTRUMENTATION AND CONTROL					
.1-1c    .2-1c    .3-1c	INSTRUMENTS     PIPING/TUBING     CABLE	× × ×	× .	× •	× •	x x x
	HVAC					
V.1-le V.2-le	EQUIPMENT     DUCTS AND SUPPORTS	××	×	×	×	x
	STRUCTURAL	1.0		15		
V.I-Ic	· FOUNDATIONS	×		x		
V.2-1e	CONCRETE     STRUCTURAL STEEL	X		×	2	×
A Day L Co	. STRUCTURAL STEEL	X		X		×

OELETED SCOPE OF REVIEW
 ADDED SCOPE OF REVIEW

INITIAL SAMPLE DOCUMENTED IN REV. 0 AND I OF THIS PLAN HAS BEEN MODIFIED EFFECTIVE 4/13/83

FIGURE 1.2-3

#### INITIAL SAMPLE REVIEW MATRIX FOR THE STANDBY ELECTRIC POWER SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM

/	DESIGN AREA		MITMENTS MITMENTS MITMENTS	MENTS MING	OPE OF	Review Notion Alion	CATIONS AND
TOPIC MIN		REVIEW OF DI	REVIEW OF	CHECK OF C	CONFIRMATOR	CHECK OF	1
	STANDBY ELECTRIC POWER SYSTEM PERFORMANCE REGUIREMENTS						
.1-2	SYSTEM OPERATING LIMITS - DG ACCIDENT ANALYSIS CONSIDERATIONS	××	××	×			
.3-2	SINGLE FAILURE - DG, PDS, AC, DC	x	x	×	x		
.4-2	TECHNICAL SPECIFICATIONS - DG, DC	X	X				
.6-2	LOCAL OPERATION - DG	X		1.11	1.1.1.1.1		
.7-2	SYSTEM INTERLOCKS - DG	X	X		1		
.9-2	COMPONENT FUNCTIONAL REQUIREMENTS - DG, PDS, AC, DC	×	×	×		×	
.12-2	COOLING/HEATING REQUIREMENTS - DG	X	X	×			
.14-2	PRESERVICE TESTING/CAPABILITY FOR OPERATIONAL TESTING - DG	×	×	×		×	
.16-2	ELECTRICAL CHARACTERISTICS - DG, PDS, AC, DC	×	×	×	19.11	×	
.17-2	PROTECTIVE DEVICES/SETTINGS - DG, PDS	X	X	X	1.1.1		
.18-2	INSTRUMENTATION - DG, AC, DC	X	X	X	1.16	X	
.19-2	CONTROL SYSTEMS - DG	X	X	X	1. 1. 1. 1.	X	
.20-2	ACTUATION SYSTEMS - DG	X	X	X	101/12	X	
.23-2	FAILURE MODES AND EFFECTS - DG, PDS, AC, DC	X	×	×			
.24-2	ELECTRICAL LOAD CAPACITY - DG, PDS, AC, DC	×	×	×	×		
.25-2	ELECTRICAL LOADS SEQUENCING - DG, PDS	X	X	X		X	
.26-2	ELECTRICAL LOAD SHEDDING - DG, PDS	X	X	X	10.4		
.27-2	FUEL OIL SYSTEM - DG	X	X	X			1
.28-2	LUBE OIL SYSTEM - DG	X	X.	1.1	1000		
.29-2	STARTING MECHANISM AND AIR SUPPLY SYSTEM - DG	×	×	×		×	1
.30-2	COMBUSTION AIR SUPPLY - DG	X	X	X	1000		1
	INDEPENDENCE - DG. PDS. AC. DC	X	X	X			1
1.31-2	to the first of the second of the second sec						

DG -DGB -PDS -AC -

- DIESEL GENERATOR DIESEL GENERATOR BUILDING FOWER DISTRIBUTION SYSTEM PREFERRED 120V AC POWER SYSTEM SERVICING AFW SYSTEM 125V DC POWER SYSTEM SERVICING AFW SYSTEM DC -

FIGURE 1.2-4a

# INITIAL SAMPLE REVIEW MATRIX FOR THE STANDBY ELECTRIC POWER SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM (CONTINUED)

Topica	DESIGN AREA	REVIEW OF DES.	REVIEW OF	CHECK OF	CONFIRMATIONS ONS A	CHECK OF THON ATION ATION	SPECIFICATIONS AND
	STANDBY ELECTRIC POWER SYSTEM						
1.1-2	SEISMIC DESIGN	×					
1.2-2	· PRESSURE BOUNDARY - DG	x	x	x			
1.3-2	PIPE/EQUIPMENT SUPPORT - DG, PDS	×	X	×	×	×	
.4-2	EQUIPMENT QUALIFICATION - DG, PDS	×	×	×	1.1.1	×	
.5-2	HIGH ENERGY LINE BREAK ACCIDENTS	×			0.00		
.6-2	FIPE WHIP - PDS, AC, DC	X		-	1.1		
0.7	ENURONING THE PROTECTION	×			1.		
0.2	ENVIRONMENTAL PROTECTION	X	1.1.1			6.14	
.10-2	EQUIPMENT QUALIFICATION - DC. PDS	×	~				
.11-2	. HVAC DESIGN - DG	â	^	^		~	
.12-2	FIRE PROTI CTION - DG	x	×	×	Sec.		
.13-2	MISSILE PROTECTION - DG	x	^	1 ^	120 3		
14-2	SYSTEMS INTERACTION - DG. PDS. AC. DC	x	×				
	STRUCTI IPES THAT HOUSE THE STANDAY				1.00		
	ELECTRIC POWER SYSTEM	1.1		1.00		12.1	
1-1-2	SEISMIC DESIGN/INPUT TO FOURPMENT - DCP	~	~		- 11-1		
1.2-2	WIND & TORNADO DESIGN/MISSILE PROTECTION	Ŷ	Ŷ	1 û		×	
	- DGB	^	^	<b>^</b>		~	
1.3-2	FLOOD PROTECTION - DGB	×	x	×	1.20		
.4-2	HELBA LOADS - DGB	×				100	
1.5-2	CIVIL/STRUCTURAL DESIGN CONSIDERATIONS					1.1	
.6-2	· FOUNDATIONS - DGB	×	x	x	1.11.19		
.1=2	CONCRETE/STEEL DESIGN - DGB	×	×	×		×	
18.7						100 March 100 Ma	

DGB -PDS -AC -

DIESEL GENERATOR DIESEL GENERATOR BUILDING POWER DISTRIBUTION SYSTEM PREFERRED 120V AC POWER SYSTEM SERVICING AFW SYSTEM 125V DC POWER SYSTEM SERVICING AFW SYSTEM DC -

FIGURE 1.2-4b

# INITIAL SAMPLE REVIEW MATRIX FOR THE STANDBY ELECTRIC POWER SYSTEM MIDLAND INDEPENDENT CONSTRUCTION VERIFICATION PROGRAM

TOPIC NUMOR	SYSTEM/COMPONENT	REVIEW	MAINTEVIEW OF JUNE	REVIEW OF CLUMENTATION &	VEEVIEW DOCUMENTATION	VERIFICATION ACTIVITIES AMAIN	DA ICUPATION SICAL
	MECHANICAL						
1.1-2c	. EQUIPMENT - DG	×	x	×	×		
1.2-2c	PIPING - DG	x		x		x	
1.3-2c	· PIPE SUPPORTS - DG	x		x		x	
	ELECTRICAL						
11.1-2c	. EQUIPMENT - DG. PDS. AC. DC	X	x	x	x		
11.2-2c	. TRAYS AND SUPPORTS - PDS	x	-	x	x	x I	
11.3-2c	. CONDUIT AND SUPPORTS - PDS	X		x	x	x	
11.4-2c	· CABLE - PDS	×	x	x	x	x	
	INSTRUMENTATION AND CONTROL						
111.1-2e	INSTRUMENTS - DG	×	×	×	×		
111.2-2c	· PIPING/TUBING - DG	X		x	1 ^	Î X	
111.3-2c	· CABLE - DG, PDS	x	X	x	×	x	
	HVAC		-				/
V.1-2c	. EQUIPMENT - DG	x					
V.2-2c	. DUCTS AND SUPPORTS - DG	×				x	
	STRUCTURAL						1
V.1-2c	. FOUNDATIONS - DC			×	2.73		1
V.2-2c	. CONCRETE - DG	Îx		Ŷ			
V.3-2e	• STRUCTURAL STEEL - DG	×		×			1
VEY.							1
DG - DG8 - POS - AC -	DIESEL GENERATOR DIESEL GENERATOR BUILDING POWER DISTRIBUTION SYSTEM PREFERRED 120V AC POWER SYSTEM						

DC - 125V DC POWER SYSTEM SERVICING

FIGURE 1.2-5

# INITIAL SAMPLE REVIEW MATRIX FOR THE CONTROL ROOM HVAC SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM

		SCOPE OF REVIEW					
TOPICAL	DESIGN AREA	REVIEW OF DES.	REVIEW OF	CHECK OF CLIMENTS	CONFIRMATIONS ONS	CHECK OF CALCULATION	SPECIFICATIONS AND
	CONTROL ROOM HVAC SYSTEM PERFORMANCE						
1.1-3	SYSTEM OPERATING LIMITS	x	x	x			
1.2-3	ACCIDENT ANALYSIS CONSIDERATIONS	x	x				
1.3-3	SINGLE FAILURE	x	×	x			
1.4-3	TECHNICAL SPECIFICATIONS	x	x				
1.5-3	SYSTEM ALIGNMENT/SWITCHOVER	x	x				
1.7-3	SYSTEM ISOLATION/INTERLOCKS		x	x		x	
1.9-3	COMPONENT FUNCTIONAL REQUIREMENTS	x	x	x		x	
1.10-3	SYSTE & PNEUMATIC DESIGN	x	x	x	x	x	
1.12-3	COOLING/HEATING REQUIREMENTS	x	x	x			
1.14-3	PRESERVICE TESTING/CAPABILITY FOR OPERATIONAL TESTING	×	×				
1.15-3	POWER SUPPLIES	×	x				
1.18-3	INSTRUMENTATION/DETECTION	x	x	x		x	
1.19-3	CONTROL SYSTEMS	x	x			x	
1.20-3	ACTUATION SYSTEMS	x	x	x		x	
1.21-3	NOE COMMITMENTS	x	x	×			
1.22-3	MATERIALS SELECTION	x	x	×		x	
1.23-3	FAILURE MODES AND EFFECTS	x	x	x			
1.33-3	FILTRATION	x	x	x		x	
1.34-3	PRESSURIZATION	x	x	x		x	
1.35-3	VENTILATION	×	×	×	×	٠x	

# INITIAL SAMPLE REVIEW MATRIX FOR THE CONTROL ROOM HVAC SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM (CONTINUED)

		SCOPE OF REVIEW					
TOPICAL	DESIGN AREA	REVIEW OF DEL	REVIEW OC.	DOCUMENTS CHECK OF	CONFIRMATIONS ONS	CHECK OF CALCULATION	SPECIFICATIONS AND
	CONTROL ROOM HVAC SYSTEM PROTECTION						
11.1-3	SEISMIC DESIGN	×					
11.2-3	· PRESSURE BOUNDARY	Îx	x	X	12.0		
11.3-3	. DUCT/PIPE/EQUIPMENT SUPPORT	x	x	x		×	
11.4-3	. EQUIPMENT QUALIFICATION	X	x	x		x	
11.5-3	HIGH ENERGY LINE BREAK ACCIDENTS						
11.6-3	· PIPE WHIP	1 û			141.13	1	
11.7-3	· JET IMPINGEMENT	x	1.1.1				
11.8.2					26.54		
11.9.3	. ENVIRONMENTAL ENVELOPES	×					
11.10-3	EQUIPMENT QUALIFICATION	1 Č	×	X	×	×	
		^		-		×	
11.12-3	FIRE PROTECTION	X	x				
11.13-3	MISSILE PROTECTION	×		1.0			
11.14-3	SYSTEMS INTERACTIONS	X					
	STRUCTURES THAT HOUSE THE CONTROL						
111.1-3	SEISMIC DESIGN/INPUT TO EQUIPMENT	×	×	×			
111.5-3	CIVIL/STRUCTURAL DESIGN CONSIDERATIONS	1 x	~	^		1. 1. 6.	
111.7-3	· CONCRETE/STEEL DESIGN	Ŷ	×		1.1.1	1.1	
111.9-3	. LEAK TIGHTNESS	x	×	x		1280	
						•	

FIGURE 1.2-6b

# INITIAL SAMPLE REVIEW MATRIX FOR THE CONTROL ROOM HVAC SYSTEM MIDLAND INDEPENDENT CONSTRUCTION VERIFICATION PROGRAM

/	SYSTEM/COMPONENT	/	ENTATION E	DOCUMENTAND	DOCUMENTATION	WACIENED	N OF PHYSICAL
Topicas	Bennou	HEVIEW	MAINTEVIEW OC	INGEVIEW OF	NE REVIEW	VERIFICATIO	Covers
1.1-3c 1.2-3c 1.3-3c	MECHANICAL • EQUIPMENT • PIPING • PIPE SUPPORTS	x x x	×	× × ×	×	x	
II.1-3c II.2-3c II.3-3c II.4-3c	ELECTRICAL • EQUIPMENT • TRAYS AND SUPPORTS • CONDUIT AND SUPPORTS • CABLE	* * * *		* * * *	x	x x x x	
111.1-3c 111.2-3c 111.3-3c	INSTRUMENTATION AND CONTROL INSTRUMENTS/DETECTORS PIPING/TUBING CABLE	x	×	* * *	×	* * *	
IV.2-3c	HVAC • DUCTS AND SUPPORTS STRUCTURAL	×	×	×		×	
V.2-3c V.3-3c V1.1-3c	CONCRETE     STRUCTURAL STEEL     NDE/MATERIALS TESTING PROGRAM	××		××		× × ×	

FIGURE 1.2-7

	P	ROJECT INS	TRUCTIO	N
PI- <u>3201</u> -0 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE	of 145_	PREPARED BY:	61.1	APPROVED BY:

	- Igores			
System	Design Verification	Construction Verification		
AFW SEP CR-HVAC	1.2-2a, 2b 1.2-4a, 4b 1.2-6a, 6b	1.2-3 1.2-5 1.2-7		

Figures

It should be noted that the scope of technical review is dynamic and subject to change as more emphasis will be given to specific review areas that meet prescribed criteria. These criteria are documented in Section 3.1.2 of this Plan. Accordingly, the review matrices generally represent the initial IDCV "sample". On April 13, 1983, a change of the AFW system initial sample was effected. This change is noted on Figures 1.2-2a, 1.2-2b and 1.2-3.

#### 1.3 SYSTEMS SELECTION CRITERIA

The selection of the auxiliary feedwater system was based upon the following six criteria:

- Importance to Safety The system should have a relatively high level of importance to the overall safety of the Midland Plant.
- Inclusion of Design and Construction Interfaces The system should be one which involves multiple interfaces among engineering and construction disciplines as well as design and construction organizations, such as the NSSS vendor, architect engineer, constructor, and subtier contractors. The system should also be one where design or construction changes have occurred and thus provide the ability to test the effectiveness of the design and construction process exercised by principal internal and external organizations or disciplines in areas of design or construction change.

	P	ROJECT IN	STRUCTIO	N		
PI00	9	SUBJECT:	ECT: Engineering Program Plan Midland Independent Design and			
REV.: 2	DATE: 5/18/83	Construction Verification Program				
PAGE 15 of 145		PREPARED BY:	LIN-1 -	APPROVED BY:		

- <u>Ability to Extrapolate Results</u> The system should be sufficiently representative of other safety systems such that the design criteria, design and construction control and change processes are similar so that extrapolation of findings to other systems can be undertaken with confidence.
- <u>Diverse in Content</u> The major engineering and construction disciplines should all have input to the design of the system.
- <u>Sensitive to Previous Experience</u> The system should be one which includes design or construction disciplines or interfaces which have previously exhibited problems and thus a test of the system should be indicative of any generic condition.
- <u>Ability to Test As-Built Installation</u> The system configuration should be sufficiently completed that the as-built configuration can be verified against design.

The AFW system was selected after consideration of a number of other candidate systems. The Midland Plant probabilistic risk assessment (PRA) was utilized as a tool to assess the relative importance to safety of plant systems on the basis of their contribution to overall plant risk. The profile for this criterion as well as each of the other five criteria is sufficiently high for the AFW, SEP and CR-HVAC systems to justify their selection.

#### 1.4 INDEPENDENCE REQUIREMENTS

The Midland IDCV program will be conducted in accordance with the "independence" criteria documented in a letter from Nunzio J. Palladino, Chairman, NRC, to the Honorable John D. Dingell, Chairman, Committee on Energy and Commerce, United States House of Representatives, dated February I, 1982. This letter was originally written as applicable to Pacific Gas
Р	ROJECT INSTRUCTION	
PI	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE 16 of 145	PREPARED BY: APPROVED BY:	1

and Electric Company's Diablo Canyon project; however, it is being applied to the Midland IDCV program and the reader should interpret the words PG&E or Diablo Canyon to mean CPC or Midland, respectively. The following criteria are excerpted from Enclosure 3 of this letter:

"The competence of the individuals or companies is the most important factor in the selection of an auditor. Also, the companies or individuals may not have had any direct previous involvement with the activities at Diablo Canyon (Midland) that they will be reviewing.

In addition, the following factors will be considered in evaluating the question of independence:

- Whether the individuals or companies involved had been previously hired by PG&E (CPC) to do similar seismic (delete seismic) design work.
- Whether any individual involved had been previously employed by PG&E (CPC) (and the nature of the employment).
- Whether the individual owns or controls significant amounts of PG&E (CPC), stock.
- Whether members of the present household of individuals involved are employed by PG&E (CPC).
- Whether any relatives are employed by PG&E (CPC) in a management capacity.

In addition to the above considerations, the following procedural guidelines will be used to assure independence:

- An auditable record will be provided of all comments on draft or final reports, any changes made as a result of such comments, and the reasons for such changes; or the consultant will issue only a final report (without prior licensee comment).
- NRC will assume and exercise the responsibility for serving the report on all parties."

DC-82-13

	P	ROJECT INST	RUCTION
PI- 3201 - 00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineering Program Plan idland Independent Design and astruction Verification Program
PAGE	of 145	PREPARED BY:	APPROVED BY

The individuals taking part in the Midland IDCV program meet the preceding criteria and have signed a statement attesting to this fact.

TERA Corporation is under contract to CPC to provide the engineering services necessary to complete the Midland IDCV program. Prior to this contract, TERA has never been under contract to CPC.

The contract requires TERA to maintain an auditable record to document the process leading to Findings as well as meetings to discuss Findings. Section 4.0 of this Plan addresses documentation and protocol requirements governing external communications which have been developed to meet obligations of the contract and NRC requirements. The protocol will be conducted in accordance with the requirements documented in a letter from James G. Keppler, Administrator, NRC Region III to James W. Cook, Vice President, CPC dated March 28, 1983, as interpreted by TERA.

Section 5.0 of this Plan addresses the report generation process, during the IDCV program to report Findings and at its conclusion as a final report. TERA will maintain an auditable record of all comments on the draft final report.

### 2.0 ORGANIZATION AND CONTROL

#### 2.1 PROJECT ORGANIZATION

The project organization is addressed in Section 2.' of the Project Quality Assurance Plan (PQAP), Midland Independent Design and Construction Verification Program, Project 3201. Figure 2.1-1 provides the project organization chart. Technical and administrative personnel (not shown) receive assignments directly from the Project Manager (PM). The PM serves as the principal point of contact



FIGURE 2.1-1 PROJECT ORGANIZATION MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION

	P	ROJECT IN	STRUCTIC	N
PI 0	09 DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE	of 145	PREPARED BY:	112-	APPROVED BY

with CPC. The Project Quality Assurance Engineers report directly to the Executive Vice President, TERA. They will identify internal quality assurance deficiencies, work with the PM in providing clarification relative to identified deficiencies and any recommendations made by them for resolution.

#### 2.2 AUTHORITY AND RESPONSIBILITY

The project authority and responsibility is addressed in Section 2.2 of the PQAP, Project 3201, as augmented by various project instructions and engineering control procedures which are referenced in the PQAP.

The Principal-in-Charge (PIC) is responsible for helping establish the general philosophy of review, setting forth guidance to the Project Manager and the Lead Technical Reviewers (LTR), assisting as an interface with the Senior Review Team (SRT), NRC and CPC and reviewing/concurring in reports issued to CPC, NRC and other outside parties.

The Project Manager is responsible for overall planning and direct supervision of all in-house activities undertaken to fulfill the contract requirements. All documentation, correspondence, reports, calculations, etc., issued to CPC, NRC and other outside parties are to be issued under his signature or otherwise receive his approval as required by the applicable Engineering Control Procedure or Project Instruction.

The Project Manager is responsible for overall planning and management of all outside activities performed by subcontractors or Associates, but may delegate responsibility for supervision to other individuals within the project.

P	ROJECT INS	TRUCTION
PI 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE _ 20 of 145	PREPARED BY:	APPROVED BY:

Documentation may be issued to the subcontractor or Associate under the signature of the designated individual.

The Senior Review Team (SRT) is responsible for the review of Open, Confirmed or Resolved (OCR) Item Reports, as requested by the PIC, Finding Reports, Finding Resolution Reports, as well as Interim Technical Reports and Final Reports. The SRT may at any time recommend to the PIC that the PM expand the scope of review, provide clarification or reassess elements of the review to assess the technical validity and significance of project team conclusions and the proper classification of OCRs and Findings. (These reports are defined in Section 5.0 of this Plan). The SRT is also responsible for the review of Monthly Status Reports, OCRs as directed by the SRT Chairman, and any Draft Interim Technical Reports to maintain current awareness and assure a high level of technical quality. They will also provide recommendations to resolve differing technical views which may arise among project team membes. The SRT Chairman is responsible for coordination and direction of SRT activities.

The Lead Technical Reviewers (LTR) are responsible for management and implementation of all review activities within their discipline of review, including supervision of individuals on the project and outside activities performed by Associates. The LTRs report to the Project Manager. The LTRs are responsible for the classification of OCRs and Findings, the preparation of Finding Reports and Finding Resolution Reports.

The Project Quality Assurance Engineer is responsible for verification of the implementation of the PQAP and will perform audits evaluating the implementation of applicable procedures and instructions in accordance with Section 6.3 and ECP-5.6.

	P	ROJECT IN	STRUCTIC	N
PI- <u>3201</u> 00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ing Program Plan pendent Design and Verification Program
PAGE	of 145	PREPARED BY:	111-1 -	APPROVED BY:

### 2.3 ADMINISTRATIVE CONTROL

The project administrative control is addressed in Section 4.0 of the PQAP, Project 3201, as augmented by various project instructions and engineering control procedures which are referenced in the PQAP.

Procedures and instructions are addressed which will be implemented to control documentation generated on the Midland IDCV project which is subject to quality assurance and control measures or is required to provide an auditable record of the IDCV review process leading to Findings. The following documents are controlled: engineering evaluations, documents such as Monthly Status Reports, Draft and Final Interim Technical Reports and Draft and Final Reports, calculations, analyses, computer analyses, PQAP, quality assurance documents, personnel qualifications, correspondence, Open, Confirmed and Resolved Item Reports, Observations, Finding Reports, Finding Resolution Reports, the Engineering Program Plan and records documenting external communications and meetings.

### 3.0 ENGINEERING PROGRAM PLAN METHODOLOGY

This section provides the overall method of approach for the IDV and ICV portions of the IDCV with particular emphasis on those features of the methodology which are common to both. Specific details of the methodology for the IDV and ICV are addressed below in Sections 3.1 and 3.2, respectively.

The initial review step includes the identification and review of pertinent documents to permit an understanding of the design and construction chains

	P	ROJECT INS	TRUCTIC	N
PI 0 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineer Midland Inde	ing Program Plan ependent Design and Verification Program
PAGE 22	of 145	PREPARED BY:	ルビー	APPROVED BY:

including the interrelationships between the organizations and suborganizations participating in the Midland project. Next, the design bases in the form of regulatory requirements and design criteria are identified and reviewed in parallel with a review of project design and construction related experience. The design bases review provides an overall understanding of the plant and system design. The project design and construction experience review ensures that the IDCV program encompasses previously identified problem areas to verify that these have been adequately addressed and that they do not exist elsewhere in the same or similar form.

For the systems, components, and structures identified in Sections 3.1.3 through 3.1.5 and 3.2.3 through 3.2.5, detailed information which documents the implementation of the design and construction commitments will be identified, reviewed, and evaluated. The IDCV review and evaluation process will be documented in accordance with the procedures addressed in Section 4.0 of this Plan. The reporting of Findings including the disposition of items potentially leading to Findings will be reported in accordance with the procedures addressed in accordance with accordance with the procedures addressed in accordance with the procedures addressed in Section 5.0 of this Plan. The IDCV will be conducted in accordance with applicable provisions of 10 CFR 50, Appendix B, which are addressed in Section 6.0 of this Plan.

### 3.1 INDEPENDENT DESIGN VERIFICATION METHODOLOGY

ANSI N45.2.11 defines design verification as the "process of reviewing, conforming, or substantiating the design by one or more methods to provide assurance that the design meets specified inputs." Design inputs include design bases or criteria, regulatory requirements, codes and standards, and other design commi.-

	P	ROJECT IN	STRUCTIO	N	
PI00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerir Midland Indep Construction V	ng Program Plan pendent Design and /erification Program	
PAGE 23	of _145_	PREPARED BY:	lik.	APPROVED BY:	

ments. The IDV includes a determination of the design inputs; an evaluation of their accuracy, consistency, and adequacy; and an evaluation of the implementation of these commitments. The emphasis will be on making a determination of the overall quality of the design and an assessment of its compliance with licensing commitments. The review approach has been designed to be introspective in making this overall quality assessment by integrating the many design inputs and licensing commitments. This integrated assessment will ensure that all parameters have been considered which are important for the system in meeting its functional requirements.

The IDV methodology will utilize the applicable guidelines of ANSI N45.2.11. The methodology will include diverse approaches such as checking original calculations, conducting alternative confirmatory calculations, or checking design ouputs including drawings or specifications. Where independent calculations are utilized, they may incorporate methods which are either similar to or different from the original design. In certain instances these independent calculations will be "blind," in that the original design calculations will be compared to the independent calculations upon their completion, without prior review by the IDV analyst.

The categories to be reviewed for certain design areas include review of design criteria and commitments, review of implementing documents, checks of calculations and evaluations, confirmatory calculations or evaluations, and checks of drawings and specifications. These categories are defined in Section 3.1.1. As a rule, all design areas will not be reviewed in each of the preceding categories. For example, a design area for the AFW system is "heat removal capability." This item would not typically have drawings and specifications associated with it

	PROJECT INSTRUCTION	N
PI	SUBJECT: Engineer Midland Inc Construction	ring Program Plan dependent Design and Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

as a direct output. In other instances, it may be the judgment of the review team based upon experience that emphasis is not needed in certain categories for each design area.

The bases for sample selection are presented in Section 3.1.2. The definition of the scope of review is provided in the following sections:

System	Section
AFW	3.1.3
SEP	3.1.4
CR-HVAC	3.1.5

The IDV will be conducted utilizing detailed checklists which are described in Section 3.1.6. Additional sampling and verification that may be conducted as a result of the IDV are discussed in Section 3.1.7.

### 3.1.1 CATEGORIES OF REVIEW: THE DESIGN CHAIN

The categories of review selected include the major design activities identified in the design chain. The IDV review categories included are:

- Review of design criteria and commitments
- Review of implementing documents
- Check of calculations and evaluations
- Confirmatory calculations or evaluations

P	ROJECT INSTRUCTION	-
PI	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE _25 of _145	PREPARED BY:	7

#### Check of drawings and specifications

Each of these categories is described in detail in sections 3.1.1.1 through 3.1.1.5 respectively. Checklists have been prepared for each of these categories to aid IDCV reviewers in the implementation of their review. These checklists are discussed in section 3.1.6.

### 3.1.1.1 Review of Design Criteria and Commitments

An identification and review of the design criteria and commitments concerning each specific design area will be performed. This review catego y provides the assurance that all necessary design inputs are considered in the IDV. The results of this review of design criteria and commitments are then used in subsequent stages where appropriate. The review of design criteria and commitments begins with an identification of appropriate criteria for the system. Such criteria may be determined from sources such as the FSAR, the docket file, 10 CFR 50, Appendix A, criteria supplied by the NSSS vendor, industry codes and standards, and other documents which provide criteria for system design.

### 3.1.1.2 Review of Implementing Documents

Implementing documents are those design documents which translate the design inputs into working level documentation. Typically, implementing documents include design criteria documents, project procedures, standard design practices, specific plant design basis documents, drawings, and calculations. Most frequently, implementing documents are intermediate steps in the design process

P 3201 000	ROJECT INSTRUC	CTION
REV.: 2 DATE: 5/18/83	SUBJECT: Engi Midland	ineering Program Plan d Independent Design and
PAGE _26 Of _145	PREPARED BY:	APPROVED BY

which are subsequently used to produce design outputs. It is important that design inputs are properly interpreted and documented in implementing documents. Therefore, the objective of the review is to determine the existence and general reasonableness of the documentation and whether the documentation correctly reflects the design inputs. Design outputs are defined as documents such as drawings, specifications, and similar materials defining technical requirements for the fabrication, installation, or construction of the system. The design output documents are reviewed for the application of the design criteria and commitments as part of the check of drawings and specifications.

# 3.1.1.3 Check of Calculations and Evaluations

When specified, a detailed check of calculations and evaluations is made (i.e. inputs, assumptions, methodology, outputs, etc.). This activity follows the review of design criteria and commitments and the review of implementing documents. The check may take several forms, ranging from a number-by-number detailed mathematical check to a review and evaluation of outputs for reasonableness. The overall presentation of the sampled calculations and evaluations will also be reviewed to verify that all steps are clearly presented and consistent throughout. The IDV reviewer may, at his discretion, choose to conduct an alternative calculation or evaluation. Where computer programs were used in the analysis, the reviewer will verify that appropriate inputs have been used in the calculation, and that the appropriate outputs have been programs used have been verified in accordance with appropriate verification procedures.

	ROJECT INSTRUC	TION
PI-3201 009 REV: 2 DATE: 5/18/83	SUBJECT: Engi Midland Construc	neering Program Plan I Independent Design and tion Verification Program
PAGE _ 27 of _ 145	PREPARED BY:	APPROVED BY:

### 3.1.1.4 Confirmatory Calculations or Evaluations

For selected oreas, confirmatory calculations or evaluations will be performed. Generally, these evaluations will be made to confirm judgements relative to the review of areas which are suspect to the IDCV reviewer; however, "blind" confirmatory calculations will be undertaken in pre-selected areas to independently verify the original design calculations. Such confirmatory calculations will be performed by obtaining the necessary input data and independent specification of calculation or evaluation objective. The reviewer will select and apply the appropriate techniques to achieve the end results. Such calculation methods will be performed without benefit of first reviewing the existing design calculational method. In order to preserve the "blind" nature of this approach, it will be necessary that a person other than the reviewer of the implementing documents perform the confirmatory calculation or evaluation. The confirmatory calculation or evaluation will be performed under procedures appropriate for the type of calculation or evaluation being performed. To the extent appropriate, the calculation or evaluation will be equivalent to that initially performed. After completion of the confirmatory calculation or evaluation, a comparison between the original calculation and the confirmatory methods will be made to determine whether differences exist. If differences occur, a determination will be made to assess whether these differences are due to the inherent nature of the calculation methods chosen or due to errors.

For example, differences may result due to the selection by the originator of simplifying or conservative assumptions. In the event that the original calculation is more conservative than the confirmatory calculation and meets design basis acceptance criteria, no further action will be necessary. On the other

	P	ROJECT IN	STRUCTION
PI 0	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE 28	of _145	PREPARED BY:	LILL APPROVED BY:

hand, if the confirmatory calculation uses more conservative methods, a check of the original calculation will be made to determine whether the difference in degree of conservatism is appropriate.

### 3.1.1.5 Check of Drawings and Specifications

Where appropriate, design outputs such as drawings and specifications will be reviewed and checked to assure that they accurately and consistently reflect that which has been called for in design documents such as calculations or engineering evaluations. Drawings and specifications will also be reviewed to determine whether design change notices and field change notices have been incorporated. In cases where several related drawings exist, a cross-comparison among drawings will be made. Additionally, a review will be made of correspondence with vendors to determine the existence of deviations from the specifications and the approval by the design organization of such changes.

### 3.1.2 BASES FOR SAMPLE SELECTION

The systems selection criteria which are documented in Section 1.3 of this Plan also apply to the selection of specific structures or components to be reviewed within each design area of the IDV, including the depth of review in each design area. As a rule, the selection is based upon engineering judgment, as statistical techniques are considered to be largely inappropriate for a design verification program. Senior members of the project team with requisite experience are responsible for selecting the sample and determining its size. This process provides greater assurance than a random sampling plan since the initial IDV sample is purposely biased towards typical problem areas. Furthermore, the

P	ROJECT INST	RUCTION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Aidland Independent Design and Instruction Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

initial sample is considered broad enough to ensure that significant deficiencies could not propagate through the systems under evaluation without being detected.

In the course of designing a nuclear power plant, numerous reviews and evaluations are typically performed. These reviews and evaluations may result in the identification of areas requiring additional work. These reviews and evaluations reflect the project's design experience and are a valuable input to the refinement of the IDV scope and sample selection. In order to make use of this information, a review is made of the ongoing CPC inspection programs, 50.55e reports, CPC Safety Concern and Reportability Evaluation (SCRE) reports, Bechtel Management Corrective Action Reports (MCAR), NRC inspection reports, audit reports, and similar documentation. Three principal criteria are used to modify the technical review scope and the initio' sample, providing more emphasis or a higher frequency of sampling to:

- Criterion 1 Areas experiencing repeated problems within the industry or specifically on the Midland Project, to verify that these do not exist in the same or similar form,
- Criterion 2 Areas previously receiving a lower level of review to achieve a sufficient level of assurance, and
- Criterion 3 Areas where suspect items or Findings have been identified to provide further confirmation, close out outstanding issues and fully assess the extent and root cause.

	P	ROJECT INS	TRUCTION
PI-3201 - 00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and Instruction Verification Program
PAGE	of	PREPARED BY:	APPROVED BY:

### 3.1.3 DEFINITION OF REVIEW SCOPE FOR THE AFW SYSTEM

Section 3.1.1 identified the categories of review which essentially correspond to major activities of the design chain. When combined with a listing of each of the design areas (or topics), a matrix is formed which can be utilized to direct the conduct of the IDV effort for each system in the program. This matrix is shown on Figures 1.2-2a and 1.2-2b for the AFW system. A set of "X" marks are shown which indicate the review scope applicable to each design area. The criteria discussed in Sections 1.3 and 3.1.2 of this Plan were incorporated to develop the initial matrix. The design areas of the IDV review matrix for the AFW system are divided into three major divisions: AFW System Performance Requirements, AFW System Protection Features, and Structures that House the AFW System. The design areas addressed within each of these major divisions are discussed in Sections 3.1.3.1, 3.1.3.2, and 3.1.3.3 of this Plan, respectively. The initial review scope as documented in Revisions 0 and 1 of this Plan has been modified based upon meeting one or more of the criteria documented in Section 3.1.2. This change is indicated on Figures 1.2-2a and 1.2-2b by a circled "X" for items deleted from scope and by the symbol "\*" for items added to scope.

Because the AFW system sample selection interfaces with other systems, it is necessary to define the boundaries for items within the scope of the IDCV. In general for the AFW system, the selection was made to include all components identified as being part of the AFW system on Bechtel P&ID drawing M439 sheets 3A, revision 9, and 3B, revision 10. Specific interface points are as follows:

P	ROJECT INST	RUCTION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Aidland Independent Design and Instruction Verification Program
PAGE of	PREPARED BY:	APPROVED BY

### AFW SYSTEM SAMPLE SELECTION BOUNDARIES

#### Interfacing System

Main Steam NSSS Service Water A Service Water B Unit 2 Condensate Tank (from) Condenser Hotwells Unit 1 Condensate Tank (return) Cooling Pond (return) ac/dc Power System 2

ESFAS Main FW Loop A Vents and Drains HVAC Interface Point (component included in AFW)

Valves 074 and 077 1 Steam Generator Nozzles Valve 283 Valve 282 Valve 008 Valve 006 Valve 019 Valve 017 Breaker or fuse interfacing AFW components with power source AFW actuation system and FOGG Valve 303 First Valve AFW pump room fan coolers and associated ductwork and supports

#### NOTES:

1. P&ID M-432, Sheet IA, Revision 5

2. Power supplies dedicated to AFW system are within sample selection boundaries.

P	ROJECT IN	STRUCTIC	N
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE _32 of _145	PREPARED BY	山江二	APPROVED BY:

In view of the fact that the design process involves a great number of individuals and organizations who may have contributed to the project engineering activities, it is necessary to define a reasonable set of limits on the scope of the IDV. Criteria were established by the project team to define the end points of the design chain applicable to this project. The majority of the design was performed by Bechtel. However, portions of the design may have been performed or affected by work performed by other organizations including, but not limited to, Babcock & Wilcox (B&W), engineering contractors, and equipment vendors. For the purposes of the verification program, the following limitations were applied. The information obtained by Bechtel from B&W does not receive, as part of the IDV program, an independent evaluation of the process by which B&W developed its data. The verification program verifies that data obtained from B&W are consistent and reasonable based upon engineering judgment. If the B&W data are suspect, additional investigation into the causes may be warranted. Equipment vendors are reviewed to verify that the documents with which they were supplied are accurate and current and that the results of their design efforts conform with the specified requirements given to them by Bechtel or CPC. Vendor documentation will be reviewed to determine that his product does, in fact, meet applicable requirements of the specifications. In the event that deviations are determined to exist, the appropriate IDCV Program reporting procedures will be applied. For engineering contractors, the scopes of work applicable to these contractors will be determined and, in general, they will be treated as if they were part of the Bechtel design organization. That is, they ... ill not be treated like a vendor who is given a specification and is expected to deliver a product in conformance with that specification. They will be treated as part of a design organization which has similar responsibilities to other parts of the Bechtel project organization.

P	ROJECT INSTRUCTION	
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Ind-pendent Design and Construction Verification Program	
PAGE of	PREPARED BY:	

The following sections discuss the initial scope of review for each of the design areas.

### 3.1.3.1 AFW System Performance Requirements

The AFW system will be reviewed to assess its capability to perform as required by the design criteria and commitments. Included in the scope of this portion of review are design areas such as system operating limits, single failure, component functional requirements, electrical, instrumentation and control, and hydraulic design.

### 3.1.3.1.1 System Operating Limits - Topic 1.1-1

The specified system operating-limits will be reviewed to determine whether they have been appropriately specified in consideration of functional performance requirements during normal (startup and shutdown), transient and accident conditions. These performance requirements will be generally based upon NSSS considerations. Specified limits such as heat removal requirements, pressure requirements, time constraints, and system logic will be reviewed. To accomplish the preceding, the review will consist of a design criteria and commitments review, a review of implementing documents, and a check of calculations and evaluations.

### 3.1.3.1.2 Accident Analysis Considerations - Topic 1.2-1

The FSAR accident analyses will be reviewed to identify those accidents in which the AFW may be involved either as a contributor or as an engineered

P	ROJECT INSTRUC	TION
PI 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engir Midland Construct	neering Program Plan Independent Design and tion Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

safety system which helps mitigate the consequences of an accident. An evaluation will be made to determine if the system has been appropriately considered in these analyses and also to provide feedback into Topic I.1-1 to assure that system operating limits appropriately reflect accident analysis considerations. In addition, CPC/Bechtel actions in response to the B&W Anticipated Transient Operator Guidelines (ATOG) will be reviewed to assure that the design and operator guidelines are in compliance.

# 3.1.3.1.3 Single Failure - Topic 1.3-1

All "active" components (e.g. pumps, motor-operated valves etc.) of the AFW system will be reviewed to determine whether the failure of one component can incapacitate the system or whether the system has sufficient redundancy, including power supplies, to withstand a single failure. (This will include a review of the flow logic "matrix" (FOGG system - Feed Only Good Generator) that is designed to prevent AFW flow to a depressurized steam generator, and provide steam flow to the turbine-driven pump only from the "good" generator). Automatic and manual initiation of the system will be reviewed. To accomplish the preceding, the review will consist of a design criteria and commitments review, a review of implementing documents, and a check of design evaluations.

# 3.1.3.1.4 Technical Specifications - Topic 1.4-1

The technical specifications will be reviewed to assure that important plant operating limits associated with the AFW system are appropriately and accurately specified, consistent with the intent of the NRC's Standard Technical Specifications.

	P	ROJECT INSTR	UCTION
PI- <u>3201</u> 0	DATE: 5/18/83	SUBJECT: Mi Con	Engineering Program Plan dland Independent Design and struction Verification Program
PAGE _35	of 145	PREPARED BY:	APPROVED BY:

### 3.1.3.1.5 System Alignment/Switchover - Topic 1.5-1

System alignment criteria and commitments under all modes of operation will be reviewed along with P&IDs and other implementing documents. Additionally, since the AFW system incorporates substantial switchover capability between Units I and 2 available water sources, all switchovers and potential alignments will be reviewed against applicable procedures (if available) to determine whether the system can meet design objectives. Any switchovers designed to occur automatically will be reviewed against single failure criteria as discussed previously. Switchovers requiring manual activities will be reviewed by determining time required versus time available to accomplish necessary actions.

### 3.1.3.1.6 Remote Operation and Shutdown - Topic 1.6-1

The criteria and commitments for safe shutdown from outside the control room will be identified and reviewed. Selected components employed to meet the remote operation requirements will be reviewed as described under Topic 1.9-1, Component Functional Requirements. Other design features applicable to remote operation will be reviewed under Topic 1.16-1, Electrical Characteristics and Topic 1.18-1, Instrumentation.

### 3.1.3.1.7 System Isolation/Interlocks - Topic 1.7-1

The AFW system criteria, commitments, and implementing documents will be reviewed to determine the adequacy of all isolation requirements and interlocks which have been designed to implement system performance requirements. The single failure review in Topic 1.3-1 will address these items as well.

	PROJECT INSTRUCTION
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE of	PREPARED BY:

### 3.1.3.1.8 Overpressure Protection - Topic 1.8-1

The AFW system criteria and commitments will be reviewed to assess the need for and incorporation of protective devices which may be required to prevent system overpressurization for modes of operation. This review will serve as input into Topic I.10-I, System Hydraulic Design.

# 3.1.3.1.9 Component Functional Requirements - Topic 1.9-1

Selected mechanical, electrical, instrumentation and control (E,I&C) componen's specified and used in the AFW system will be reviewed for compliance to their functional requirements. The development of the functional requirements will be traced from the AFW system design criteria as dictated by licensing commitments, industry codes and standards, plant environmental conditions, and system performance requirements for the intended operating modes. The design criteria and commitments used for the AFW system will be checked to ensure the inclusion of all required design inputs. Component functional requirements design criteria include factors such as flow rate, allowable pressure drops, NPSH, unitage, device settings, and similar characteristics. The design process (caiculations or analyses) used to translate the overall system design criteria include specific component specifications will also be reviewed. Finally, the validated component functional requirements will be compared to the component procurement specifications. Equipment seismic and environmental qualification will be considered separately.

	P	ROJECT INSTRUC	CTION
PI- <u>3201</u> 0 REV.: 2	DATE: 5/18/83	SUBJECT: Eng Midlan	ineering Program Plan d Independent Design and ction Verification Program
PAGE	of 145	PREPARED BY:	APPROVED BY:

### 3.1.3.1.10 System Hydraulic Design - Topic 1.10-1

A review of criteria and commitments and implementing documents will be made for the system hydraulic design. The system hydraulic design review will also include a detailed check of calculations and evaluations of the system hydraulic parameters. This activity will incorporate results obtained from the configuration verification effort which is part of the ICV. For example, line sizes, lengths of pipe, and numbers of pipe fittings will be checked in the ICV effort. These quantities will then be compared against the basis for calculations of pressure drop in various portions of the AFW system.

# 3.1.3.1.11 System Heat Removal Capability - Topic 1.11-1

Calculations and evaluations performed to demonstrate the adequccy of the system's heat removal capability will be checked. The scope includes a comparison between the results of the hydraulic design evaluation and the system requirements for heat removal.

### 3.1.3.1.12 Cooling Requirements - Topic 1.12-1

Cooling requirements for AFW mechanical and electrical components will be checked and a determination made that these heat loads have been considered as design criteria for the interfacing systems.

### 3.1.3.1.13 Water Supplies - Topic 1.13-1

The criteria established for water supply, from both safety and nonsafety sources, will be identified. A review will be made of implementing documents for proper use of these criteria.

DC-82-13

	P	ROJECT INS	STRUCTIO	N
PI00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde Construction V	ng Program Plan pendent Design and Verification Program
PAGE 38	of 145	PREPARED BY:		APPROVED BY:

### 3.1.3.1.14 Preservice Testing and Capability for Operational Testing -Topic 1.14-1

A determination will be made of the design criteria and commitments which exist for preservice testing and the capability for operational testing, including a review of implementing documents and associated evaluations that support the testing programs. The results of this review will be used in the ICV portion of the IDCV, which will verify that the system has been constructed such that it can function in accordance with its design criteria and commitments.

### 3.1.3.1.15 Power Supplies - Topic 1.15-1

The applicable design criteria for AFW power supplies will be identified from NSSS vendor, regulatory and industry requirements. The documents implementing the design criteria will be checked to verify the proper consideration of the applicable criteria determined from the criteria review. The AFW logic system and schematic diagrams will be reviewed to ensure that requirements relative to the quality of power supplies (diversity and redundancy) are met. In particular the review will include the assurance that the AFW system will meet design criteria including the loss of offsite power and station blackout.

# 3.1.3.1.16 Electrical Characteristics - Topic 1.16-1

The AFW system electrical characteristics as determined by design criteria and commitments will be reviewed to verify that all required commitments and

Р	ROJECT INSTRUCTIC	N
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE of	PREPARED BY:	APPROVED BY

criteria have been addressed. The areas of criteria and commitment review shall include cable physical separation, system electrical separation, cable and raceway sizing and terminal voltage on power circuits. Cable sizing calculations will be reviewed and applied to selected power circuits in the AFW system.

# 3.1.3.1.17 Protective Devices/Settings - Topic 1.17-1

Implementing documents will be reviewed to check for the inclusion of applicable design criteria and commitments for protective devices/settings. AFW schematic diagrams and component specifications will be reviewed to ensure that protective device/settings criteria have been properly implemented for motors and penetration assemblies. Protective device bypass features for thermal overload and opening torque switch will be verified for motor operated valves that are safety related.

### 3.1.3.1.18 Instrumentation - Topic 1.18-1

The instrumentation and alarms required to operate, monitor and protect the AFW system, as determined by design criteria, commitments and expected plant operations, will be reviewed against that specified for the AFW system to verify the adequacy of the instrumentation. Selected instrument accuracies under applicable plant operating conditions will be reviewed and evaluated. Selected instrument logic diagrams will be reviewed for proper circuit electrical design. Calculations for alarm set points or time delays for selected representative devices (e.g. steam generator water level trip point) will be reviewed for compliance with design criteria. The implementing specifications or lists documenting the consideration of the applicable design criteria will be reviewed to verify that the criteria are reflected in the devices shown for review.

P	ROJECT INSTRUCTIO	N
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineeri Midland Inde Construction	ing Program Plan pendent Design and Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

#### 3.1.3.1.19 Control Systems - Topic 1.19-1

Design criteria and commitments governing the steam generator water level and AFW turbine control systems will be checked to verify the inclusion of necessary regulatory, industry, and system performance requirements. Design specifications or other implementing documentation will be reviewed to verify that the necessary requirements were used as input to the control system design. This review will include a check of calculations or evaluations relative to control system performance, time response, component characteristics, and separation from actuation systems. Failure Modes Effects Analyses (FMEA) will be reviewed in conjunction with Topic 1.23-1, Failure Mode and Effects, to verify that system failures are in the safe direction. Control system circuitry design (voltages, currents, polarity) will be reviewed to verify that selected components will function as intended in the control circuit. The circuitry design review will include a review of applicable instrument loop diagrams, logic diagrams, and valve and motor schematic diagrams.

3.1.3.1.20 Actuation Systems - Topic 1.20-1

The auxiliary feedwater actuation system (AFWAS - which includes FOGG, feed only good generator) design criteria and commitments will be reviewed to verify the proper consideration of regulatory commitments, industry codes and standards, plant operational requirements and operator actions. AFW system logic diagrams for selected motor operated valves and pump motors will be

	P	ROJECT IN	ISTRUCTIC	N
PI	DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ing Program Plan ependent Design and Verification Program
PAGE 41	of 145	PREPARED BY:	山上二	APPROVED BY:

reviewed against design commitments as will the applicable schematic diagrams. In addition, the AFWAS procurement specification will be reviewed against the design criteria and commitments.

### 3.1.3.1.21 Nondestructive Examination Commitments - Topic 1.21-1

A determination will be made of the design criteria, commitments and implementing documents (e.g. testing plans, manuals and procedures) which exist for NDE of AFW system piping, components, and structures. The results of this determination will serve as input to the ICV portion of the IDCV which will review NDE records to verify the quality of construction.

### 3.1.3.1.22 Materials Selection - Topic 1.22-1

This activity will include the review of criteria and implementing documents related to establishing the bases for the material specification process of selected structural elements, components, and a portion of the AFW piping system. Included will be a review of material selection requirements related to such factors as strength, toughness, hardness, compatability, electrical insulation properties, protective coatings, corrosion resistance, fire protection, and other chemical and physical requirements appropriate to the particular structure, component, or system.

### 3.1.3.1.23 Failure Modes and Effects - Topic 1.23-1

The Midland 'SAR contains a synopsis of a failure modes and effects evaluation performed to assess the capability of the AFW system to accomplish its intended

P	ROJECT INSTRUCTION	
PI 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE 42 of 145	PREPARED BY:	_

safety functions. The IDCV FSAR evaluation will serve as the starting point of the review process for this topic, followed by a review of any implementing evaluations that support the FSAR FMEA. An independent confirmatory FMEA is under consideration, the scope of which will be based upon results of the initial review efforts. It is contemplated that this analysis will concentrate on electrical, instrumentation, and control systems components and the effects of their failure.

### 3.1.3.2 AFW System Protection Features

In addition to the review of the capability of the AFW system to perform its required functions, a review will be made of external factors which could affect the capability of the system to achieve these functions. Included in the scope of this portion of the review are factors such as seismic design, high energy line break accidents (HELBA), environmental protection, fire protection, missile protection, and systems interaction. The following sections address these and other design areas related to system protection.

#### 3.1.3.2.1 Seismic Design - Topic II.1-1

Seismic design criteria and associated commitments related to the AFW system will be reviewed, and the establishment of the proper basis for the associated design process will be confirmed. Included will be the review of seismic design parameters and methodologies which were utilized in the seismic design process for structures, systems, and components associated with the AFW system.

P	ROJECT INST	RUCTION
PI 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Nidland Independent Design and Instruction Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

# 3.1.3.2.2 Seismic Design--Pressure Boundary - Topic 11.2-1

This activity will include a review of the commitments, implementing documents, calculations, drawings, and specifications associated with the seismic design of a selected portion of the AFW piping system. The utilization of the proper design input, such as response spectra, piping and component weights, and other piping characteristics, will be verified. The ASME code evaluations will be reviewed to verify that pertinent acceptance criteria are met. Drawings and specifications will be reviewed for consistency with design calculations. Included will be an independent confirmatory seismic analysis of a selected portion of the piping system based upon independently verified as-built dimensions utilizing a verified computer program. Pipe stresses and support loads will be calculated. To preserve the "blind" nature of the confirmatory calculation, the individuals who, perform the calculation will not have prior benefit or knowledge of the specific calculational approach followed by the original analysts. Upon completion, a comparison will be made between the original design and IDV calculated forces and stresses at key locations. Any discrepancies will be identified and their cause determined.

3.1.3.2.3 Seismic Design--Pipe/Equipment Support - Topic 11.3-1

A review of a selected portion of the AFW system will be conducted to verify that selected pipe supports have been designed and specified in accordance with criteria and commitments. Included will be the review of design loads, load combinations, and the methods of analysis utilized. The associated design drawings and specifications will be reviewed for consistency. The support loads calculated during the confirmatory piping analysis of Topic II.2-1 will be

Р	ROJECT INSTRUCT	ION
PI	SUBJECT: Engine Midland Ir Constructio	ering Program Plan Independent Design and Inverification Program
PAGE _44 of _145	PREPARED BY:	APPROVED BY:

compared to the design loads for all supports in the selected portion of the AFW system. Several support types (e.g., snubber, rigid restraint, anchor, spring hanger, etc.) will then be sampled, and an independent confirmatory analysis will be made to verify the capability of the original design organization to properly design and size these supports given the design loads. This analysis will be based upon independently verified as-built dimensions. In addition, the design calculations, drawings and specifications associated with the anchorage and support of selected AFW system equipment will be reviewed for conformance to requirements.

### 3.1.3.2.4 Seismic Design--Equipment Qualification - Topic II.4-1

This activity will include the review of commitments, implementing documents, calculations, drawings, and specifications associated with the seismic qualification of selected equipment. Qualification requirements including response spectra, load combinations, and equipment functional criteria will be reviewed. The review will include various types of AFW system equipment of representative complexity such as the following: electrical-motor control center, motor-operated valve, and electrical panel; mechanical-AFW pump, motor-operated valve and heat exchanger.

### 3.1.3.2.5 High Energy Line Break Accidents - Topic 11.5-1

HELBA criteria and associated commitments related to the AFW system will be reviewed, and the establishment of the proper basis for the associated design process will be confirmed. Included will be a review of HELBA design parameters and the methodologies which have been utilized in the HELBA

Р	ROJECT IN	STRUCTIC	N
PI	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE of	PREPARED BY:	14/1:	APPROVED BY:

design process for structures, systems, and components associated with the AFW system.

3.1.3.2.6 HELBA/Pipe Whip - Topic II.6-1

Design criteria, implementing documents, calculations, drawings, and specifications associated with pipe whip resulting from postulated high energy line breaks will be reviewed. Included will be the review of the definition of the methodology employed in determining postulated pipe break locations, the magnitude of associated pipe whip loads, and the techniques utilized for pipe restraint design. In addition, calculations for selected AFW system pipe rupture restraints will be reviewed, including the associated drawings and specifications for consistency with these calculations.

3.1.3.2.7 HELBA--Jet Impingement - Topic II.7-1

The design criteria and commitments applicable to preventive protective measures taken to assure acceptable consequences due to postulated jets will be reviewed. This topic will be reviewed in conjunction with Topic II.6-I, Pipe Whip, and Topic III.4-I, HELBA Loads, and will be considered in the evaluation of Topic III.7-I, Concrete/Steel Design.

3.1.3.2.8 Environmental Protection - Topic II.8-1

The design criteria and commitments applicable to all issues related to the plant's environmental protection will be reviewed. The environmental protection review will consist of a determination of the appropriate environmental

P	ROJECT INSTR	UCTION
PI-3201 - 009 REV.: 2 DATE: 5/18/83	SUBJECT: Mic Cons	Engineering Program Plan Iland Independent Design and truction Verification Program
PAGE 46 of 145	PREPARED BY:	APPROVED BY:

envelopes, the qualification requirements for equipment to these envelopes, and the HVAC design criteria which are necessary to assure that the environmental envelopes will not be exceeded.

### 3.1.3.2.9 Environmental Envelopes - Topic 11.9-1

The environmental envelope design criteria will be determined by a review of existing criteria and commitments and a review of the system arrangement. These environmental envelopes will be verified by a review of implementing documents and a check of calculations and evaluations which were used to determine the environmental parameters. Drawings and specifications for AFW equipment will be checked for consistency with the environmental envelope specified. In addition, a confirmatory calculation or evaluation will be performed to verify the environmental envelope specification for one portion of the AFW system.

To preserve the "blind" nature of the confirmatory calculation, the individuals who perform the calculation will not have prior benefit or knowledge of the specific calculational approach followed by the original analysts. Upon completion, a comparison will be made between the original design and IDV environmental envelopes at key locations. Any significant discrepancies will be identified and their cause determined.

3.1.3.2.10 Environmental/Equipment Qualification - Topic II.10-1

Equipment qualification requirements will be reviewed to determine whether the correct environmental envelopes were specified and, given these envelopes,

P	ROJECT INSTRUC	TION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engin Midland Construct	neering Program Plan Independent Design and tion Verification Program
PAGE 47 of 145	PREPARED BY:	APPROVED BY:

whether the qualification methods specified were adequate to demonstrate that the component would meet its functional requirements. The review will include various types of AFW system equipment of representative complexity such as electrical cable, connectors, transmitters and motor-operated valves.

# 3.1.3.2.11 HVAC Design - Topic 11.11-1

Requirements imposed upon the HVAC system design as a result of the need to meet environmental envelope or equipment qualification parameters will be checked. This will be achieved by a verification of the design interface between the AFW system design and the HVAC system design.

# 3.1.3.2.12 Fire Protection - Topic II.12-1

The applicable fire protection criteria will be determined for the AFW system. A review will be made of fire protection evaluations to determine whether the fire protection system meets the necessary requirements for the AFW system. Included in this review will be the designation and location of minimum AFW safe shutdown components, physical separation, designation of fire zones, ratings of fire barriers, construction and qualification of fire barrier penetration seals, associated circuits analysis and protection, fire hazards analysis, remote shutdown transfer switch design, fire detection and suppression systems, and emergency lighting pertaining to AFW system components. The scope of this review includes fires within the AFW pump rooms and fires external to the rooms which would affect redundant safe shutdown AFW components.

Ρ	ROJECT INSTRUCTION
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Varification Proc
PAGE of	PREPARED BY:

# 3.1.3.2.13 Missile Protection - Topic II.13-1

A review of criteria and commitments will be made of the potential missiles which could affect the AFW system and the protection provided for system components. The review includes missiles external to the AFW system and those that could be generated within the AFW system and will serve as input to Topic III.7-I, Concrete/Steel Design.

# 3.1.3.2.14 Systems Interaction - Topic II.14-1

As part of the overall systems review, the potential for systems interaction and means of prevention thereof will be reviewed. The review will include an examination of criteria utilized to analyze potential systems interactions, whether they be physical (electrical, mechanical, hydraulic), or spatial (thermal, fluid, mechanical, radiation). The procedures and results for the Midland systems interaction walkdowns will also be reviewed and, if possible, ongoing walkdowns will be observed. Human factors or inherent failure modes (common manufacturer, similar technology, equal aging or wear) will not be considered a part of the systems review.

### 3.1.3.3 Structures that House the AFW System

Many safety-related plant structures such as the containment, auxiliary and diesel generator buildings, and the intake structure support the functioning of the AFW system or its support systems. The overall criteria and commitments applicable to the design of these safety related structures will be reviewed and evaluated. Selected features and design areas from one or more of these structures will be isolated for a more in-depth review in the following topics.

P	ROJECT INS	TRUCTION
PI-3201 009 REV: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and
PAGE 49 of 145	PREPARED BY:	APPROVED BY:

# 3.1.3.3.1 Seismic Design/Input to Equipment - Topic III.1-1

This activity will include the review of commitments, implementing documents, calculations, drawings, and specifications related to the development of seismic design input for a portion of the AFW system and components in the auxiliary building. Included will be a review of seismic input parameters such as seismic design spectra, damping, material properties, and boundary conditions, including soil-structure interaction. The methodology utilized for the location of the mass points and the computation of masses and equivalent member properties will be reviewed. Parameter variation studies will also be reviewed to verify that the variance of important input parameters and modeling assumptions has been appropriately considered. The scope of this activity will include the review of the dynamic analysis of the building, the time history analysis and the generation of floor response spectra for both horizontal directions and the vertical direction. The utilization of proper floor response spectra for the system will be verified.

3.1.3.3.2 Wind and Tornado Design/Missile Protection - Topic III.2-1

Criteria and commitments for wind loading, tornado effects, and missile protection will be reviewed to verify the proper basis is established for the design process. Included will be the review of the criteria associated with wind pressure loading, tornado wind loading, tornado depressurization effects, tornado missiles, and other related requirements. Loading combinations, methodologies of analysis, associated allowable stresses or conditions, and other specified criteria will be included in this review activity. The results of this review will be considered in evaluation of Topic III.7-1, Concrete and Steel Design.

P	ROJECT INSTRUCTIO	N
PI	SUBJECT: Engineer Midland Inde	ing Program Plan ependent Design and Varification Program
PAGE _50 of _145	PREPARED BY: 11+ K+	APPROVED BY:

#### 3.1.3.3.3 Flood Protection - Topic III.3-1

This activity will include the review of criteria and commitments related to establishing the basis for flood protection from sources both external and internal to the plant. The criteria associated with the specification of the design flood level and the methods to be utilized to provide the necessary flood protection will be reviewed. Included will be the review of the criteria associated with the determination of postulated pipe break locations, the methodologies to be utilized in determining flow rates and resulting water levels, loading combinations, cllowable stresses or conditions, and other related criteria. The results of this review will be considered in evaluation of Topic III.7-1, Concrete and Steel Design.

#### 3.1.3.3.4 HELBA Loads - Topic III.4-1

Criteria and commitments for high energy line break accident loads will be reviewed to verify that the proper basis is established for the design process. Included will be the review of the criteria for jet impingement and pipe whip loading on structures and components related to the AFW system. The review will address loading combinations, methodologies of analysis, associated allowable stresses or conditions, and other related criteria. The results of this review will be considered in evaluation of Topic III.7-I, Concrete and Steel Design.

#### 3.1.3.3.5 Civil/Structural Design Considerations - Topic III.5-1

Civil/structural design criteria and associated commitments related to the AFW system will be reviewed, and the establishment of the proper basis for the

	P	PROJECT INSTRUCTION	
PI-3201 - 00 REV.: 2	DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE 51	of _145	PREPARED BY:	

associated design process will be confirmed. Included will be the review of design parameters and the methodologies utilized in the design process for structures and affected systems and components associated with the AFW system.

### 3.1.3.3.6 Foundations - Topic III.6-1

Included in this activity will be the review of criteria, implementing documents, and calculations associated with the design of selected foundations associated with structures housing the AFW system. The review will address design criteria, methodologies of analysis and calculations associated with each type of foundation loading including dead, live, tornado and seismic loadings.

3.1.3.3.7 Concrete and Steel Design - Topic III.7-1

This activity will include the review of criteria, implementing documents, calculations, drawings, and specifications associated with the reinforced concrete and structural steel design of selected structural elements associated with the AFW system. Structural elements, including a major load bearing shear wall and a floor diaphram will be selected that require consideration of a broad spectrum of loadings such as dead, live, wind, tornado, seismic, flood, and HELBA loads. The review will address design criteria, methodologies of analysis and calculations associated with each type of loading with emphasis on a verification that these items have been considered in a realistic manner. Loading combinations, allowable stresses or conditions, and other applicable criteria will be reviewed. Drawings and specifications for the selected structural elements will be reviewed against design calculations for consistency.
P	ROJECT IN	STRUCTIC	N
PI	SUBJECT:	Engineeri Midland Inde	ng Program Plan pendent Design and
PAGE 52 of 145	PREPARED BY:	+ 11/2 in	APPROVED BY:

# 3.1.4 DEFINITION OF REVIEW SCOPE FOR THE STANDBY ELECTRIC POWER SYSTEM

The categories of review identified in section 3.1.1 are also applicable to the review of the Standby Electric Power (SEP) system. Similarly, the criteria discussed in sections 1.3 and 3.1.2 were incorporated to develop the initial design verification matrix shown on Figures 1.2-4a and 1.2-4b. The design areas (or topics) of the IDV review matrix for the SEP system are somewhat different from those for the AFW system, consistent with the differences in the functions and physical configuration of these systems. The review philosophy, matrix concepts and organization remain the same. The design areas for the SEP system review matrix are divided into three major divisions: SEP System Performance Requirements, SEP System Protection Features, and Structures that House the SEP System. The design areas addressed within each of these major divisions are discussed in sections 3.1.4.1, 3.1.4.2 and 3.1.4.3 of this Plan, respectively. As previously mentioned, the identified initial review scope is subject to change depending upon whether the criteria documented in section 3.1.2 have been met.

Because the SEP system sample selection interfaces with other systems, it is necessary to define boundaries for items within the scope of the IDCV.

The SEP system as defined in the IDCV program includes four major elements: the diesel generator (DG) and its support systems; the power distribution system (PDS); the preferred I20V ac power system (AC) and the I25V dc power system (DC). Continuity with the AFW system review is emphasized by drawing the boundaries of evaluation for the PDS and the two low voltage AC and DC systems as they service the AFW system. The PDS boundaries are drawn at

F	PROJECT INSTRUCTION
REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and
PAGE 53 of 145	PREPARED BY:

breakers interfacing with the 480 V buses. The DG and all of its support systems are included within the sample selection boundaries of the SEP system.

Specific interface points are as follows:

STANDBY ELECTRIC POWER SYSTEM SAMPLE SELECTION BOUNDARIES

Interfacing System	Interfacing Point		
Station Power and Offsite Power	Breakers connecting 4160 V Class IE and non-IE buses: 2305 1200A 2405 1200A		
Non-Class IE 4160 Volt Bus Loads	Breakers at 4160 V Class IE Buses: 2599 1200A 2A0612 1200A		
Class IE Electrical Distribution	Includes distribution system from 4160 V buses to the breaker at the 480 V buses: 2766 1600A 2866 1600A and distribution to loads fed directly at 4160 volts.		
Aux. Feedwater System (AFW)	Include all portions of the Class IE power supply which feed essential components in the AFW, includ- ing the 480 Vac, 120 Vac, and 125 Vdc loads.		
Diesel Generator	Include all portions of the Class IE power supply which feed essential components for the diesel generator and supporting systems discussed below, including the 480 Vac, 120 Vac, and 125 Vdc loads.		
Control Room HVAC System (CR-HVAC)	Include all portions of the Class IE power supply which feed essential components for the CR-HVAC including the 480 Vac, 120 Vac and 125 Vdc loads.		

	P	ROJECT INS	TRUCTION
PI 0	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and
PAGE 54	of 145	PREPARED BY:	APPROVED BY:

Interfacing System	Interfacing Point		
Class IE Loads	For loads other than AFW, DG, and CR-HVAC, the review will be limited to confirming that all Class IE loads have been included in establishing the system design electrical loads.		
DG Fuel Oil Storage and Transfer System	System (FSAR Figure 9.5-25) is included. Interface with Demineralized Water Supply is at DeLavel interface.		
DG Cooling Water System	System integral to diesel is included (FSAR Figure 9.5–26). Service Water boundary is at DeLavel interface.		
DG Starting System	System (FSAR Figure 9.5-27) is included.		
DG Lubrication System	System (FSAR Figure 9.5-28) is included.		
DG Combustion Air Intake and Exhaust System	System (FSAR Figure 1.2-27) is included.		
Structures	DG building and foundations, and foundation for fuel oil storage tank.		

The following sections discuss the initial scope of review for each of the design areas.

# 3.1.4.1 SEP System Performance Requirements

The SEP system will be reviewed to assess its capability to perform as required by the design criteria and commitments. Included in the scope of this portion of review are design areas such as system operating requirements, single failure, component functional requirements, diesel electrical load sequencing/shedding, diesel starting mechanism, independence, cable sizing, routing and separation.

PROJECT INSTRUCTION				
PI-3201 009 REV: 2 DATE: 5/18/83	SUBJECT: Engineering Midland Indepe Construction Ver	Program Plan Indent Design and Dification Program		
PAGE of	PREPARED BY:	PPROVED BY:		

### 3.1.4.1.1 System Operating Limits (DG) - Topic 1.1-2

The objective of this topic is to assemble and review the salient operating parameters of the total diesel/generator "system". This will include: the fuel system; the cooling system; the building heating, ventilation, and air conditioning system; the starting system; the lubrication system; and the combustion air intake and exhaust system.

Each of the parameters will be "tracked" through the design chain to assure consistency. Applicable calculations and evaluations will be reviewed to determine whether important parameters can be kept within limits.

The key components and applicable parameters determined from this review will in turn be reviewed in greater depth in other topics related to the mechanical and electrical components.

3.1.4.1.2 Accident Analysis Considerations (DG, AC, DC) - Topic 1.2-2

FSAR Safety Analyses will be reviewed to determine accident and post-accident operability requirements. This will include such DG features as the various trippping devices and the automatic reset to 4.16 kV on LOCA start. The DG building HVAC system also will be reviewed to determine the effect of its failure on the DG.

As part of the review, the potential for DG, AC, and DC systems' failures to exacerbate accident conditions will be examined.

	P	ROJECT IN	ISTRUCTIC	N
PI 0	DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 56	of 145	PREPARED BY:	JILKY	APPROVED BY:

# 3.1.4.1.3 Single Failure (DG, PDS, AC, DC) - Topic 1.3-2

A confirmatory evaluation of a portion of the SEP system will be conducted to determine its ability to perform required functions, during both normal and accident conditions, in the event of a single failure. This review will also include, but not necessarily be limited to, an evaluation of the FSAR single failure analyses and supporting documentation.

# 3.1.4.1.4 Technical Specifications (DG, DC) - Topic 1.4-2

The Technical Specifications will be reviewed against existing regulatory criteria to assure that important operating limits are appropriately and accurately specified. An effort will also be made to assess whether additional specifications might be necessary, based upon knowledge gained from the detailed review of the DG and DC systems.

# 3.1.4.1.5 Local Operation (DG) - Topic 1.6-2

The local operability of the various DG system components, including the diesel, the generator, the lube oil system, starting mechanism and air supply system, and HVAC system, will be reviewed to determine the capability for local operation and the ability of the system to properly start and operate if controls are left in the local or manual position.

# 3.1.4.1.6 System Interlocks (DG) - Topic 1.7-2

The substantial number of interlocks of the DG system will be reviewed to determine their adequacy in implementing system performance requirements.

P	ROJECT IN	ISTRUCTION
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE 57 of 145	PREPARED BY:	LINE APPROVED BY:

This will include an evaluation of protective devices to prevent placing the DGs in parallel and of several devices associated with system components such as the fuel oil transfer pumps and the jacket water cooler.

3.1.4.1.7 Component Functional Requirements (DG, PDS, AC, DC) - Topic 1.9-2

Selected mechanical, electrical, instrumentation, and control (El&C) components specified and used in the SEP system will be reviewed for compliance with their functional requirements. The development of the functional requirements will be traced from the system design criteria as dictated by licensing commitments, industry codes and standards, plant environmental conditions, and system performance requirements for the intended operating modes. The design criteria and commitments used for the system will be checked to ensure the inclusion of all required design inputs. Component functional requirement criteria for support systems include factors such as engine starting time, pump flow rate, allowable pressure drops, NPSH, and similar characteristics. The electrical system functional requirements are both steady state and transient limits on electrical parameters, such as voltage, AC frequency, impedance, etc. In reviewing the development of functional requirements, operating plant experience will be assessed to evaluate the adequacy of system design criteria.

3.1.4.1.8 Cooling/Heating Requirements (DG) - Topic 1.12-2

Cooling requirements for the diesel-generator will be checked and a determination made that the design interface with the cooling water system is adequate to handle the heat load. Heat exchangers in the cooling system will be reviewed for proper sizing and other design considerations such as fouling. Any

Р	ROJECT INSTRUCTIO	N
PI 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

heating requirements for the engine during its standby mode will be determined and supporting calculations checked.

3.1.4.1.9 Preservice Testing and Capability for Operational Testing (DG) -Topic 1.14-2

A review will be made of the criteria and commitments and implementing documents which exist for preservice testing and the capability for operational testing for a selected diesel generator. Included will be the review of associated evaluations to ensure adequate provision for preservice and operational testing of the equipment.

3.1.4.1.10 Electrical Characteristics (DG, PDS, AC, DC) - Topic 1.16-2

Design criteria and commitments will be surveyed relative to system electrical characteristics including voltage profiles for 4160 V and 48 Vac, and 125 Vdc buses, short circuit currents on the same buses and terminal voltage available to power components from these buses. Voltage profile calculations for the 4160 V and 480 Vac buses will be reviewed. An independent determination will be made of terminal voltage available to power selected components from the 4160 V and 480 Vac buses. Short circuit calculations to determine the momentary and interrupting currents on the 4160 V system will also be reviewed.

3.1.4.1.11 Protective Devices/Settings (DG, PDS) - Topic 1.17-2

The design criteria and commitments, and implementing documentation for protective devices/settings will be reviewed for the diesel generator and power

P	ROJECT INSTRUCTION
PI	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE _59 of _145	PREPARED BY:

distribution system (4160 and 480 Vac buses). The protective devices intended to meet the identified design criteria will be checked on system schematics. Device setting criteria will be reviewed to ensure coordination of overcurrent devices associated with the 4160V and 480V buses. The interrupting rating of 4160V and 480V circuit breakers will be reviewed against the short circuit currents reviewed under Topic 1.16-2 Electrical Characteristics.

# 3.1.4.1.12 instrumentation (DG, AC, DC) - Topic 1.18-2

The adequacy of the instrumentation provided for monitoring and alarm functions associated with the dirsel generator, 120V ac preferred power system and 125 Vdc systems will be evaluated by reviewing design criteria and commitments, implementing documents, schematic diagrams and one line diagrams. Design input will be taken from industry codes and standards, regulatory requirements and guidance, and NSSS vendor input. Monitoring requirements for operational status information and surveillance instrumentation will be checked. Selected calculations of instrument setpoints will be reviewed against regulatory requirements.

# 3.1.4.1.13 Control Systems (DG) - Topic 1.19-2

Design criteria and commitments, implementing documents, check of evaluations and drawings will be made to review the adequacy of the diesel generator control systems for automatic and manual operation. In addition, the controls associated with diesel generator support systems will be reviewed. The support systems include diesel fuel, cooling water and lube oil.

	P	ROJECT INS	STRUCTIO	N
PI-3201 009 REV.: 2 DATE:	5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE of	5	PREPARED BY:	1114 :-	APPROVED BY: Reel

The design criteria compilation will include a consideration of codes and standards, regulatory requirements, NSSS vendor input and industry experience. Control system logic diagrams and schematic diagrams will be reviewed to ensure that design requirements are met in issued design drawings.

#### 3.1.4.1.14 Actuation Systems (DG) - Topic 1.20-2

The diesel generator actuation system and hardware is part of the Engineered Safety Feature Actuation System (ESFAS) which is partially reviewed during the AFW Actuation System (AFWAS) review. The actuation system design, including inputs, logic and actuation devices, will be reviewed against IEEE Standard-279 and Regulatory Guide 1.9 criteria. Actuation system logic diagrams and applicable schematic diagrams will be reviewed against design commitments. The ESFAS procurement specification will be reviewed for compliance with design commitments.

# 3.1.4.1.15 Failure Modes and Effects (DG, PDS, AC, DC) - Topic 1.23-2

The various modes of failure, and the effects of such failures, will be reviewed, using as a basis the analyses presented in summary form in the FSAR. The basis of this review will be to determine if there is any potential failure that has been previously unanalyzed or that might be common-mode, thus affecting both trains of emergency power. It is presently envisioned that the review will stop at the component level and will not entail a review of failure modes and effects of subcomponents (e.g. resistors, switches, etc. that make up specific electrical devices).

	P	ROJECT INS	RUCTION
F:- 3201 - 0	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and Instruction Verification Program
PAGE _61	of _145_	PREPARED BY:	APPROVED BY:

### 3.1.4.1.16 Electrical Load Capacity (DG, PDS, AC, DC) - Topic 1.24-2

Diesel generator, 4160 Vac bus, 120 Vac preferred power and 125 vdc battery loads will be independently tabulated for design basis conditions to determine steady state and transient SEP system loads. The loads will be reviewed against design criteria and commitments. The diesel generator loading will reflect the load sequence information from Topic 1.25-2. The battery load cycle will be determined for loss of offsite power and station blackout conditions. The resulting battery capacity requirements will be reviewed against regulatory sizing criteria. Diesel generator 4160 and 120 Vac bus loads will be compared ro design ratings and preoperational test data (Topic 1.14-2) to determine the suitability of the selected hardware.

# 3.1.4.1.17 Electrical Load Sequencing (DG, PDS) - Topic 1.25-2

A review of design criteria and commitments will be performed for electrical load sequencing provisions for the diesel generator and its associated 4160 Vac bus. Sequence logic will be reviewed against design commitments as will the schematic diagrams implementing sequencing logic. The verified loading sequence will be used as input to the load capacity review (Topic 1.24-2) and the preservice testing review (Topic 1.14-2).

### 3.1.4.1.18 Electrical Load Shedding (DG, PDS) - Topic 1.26-2

Electrical load shedding design criteria and commitments, implementing documents and logic diagrams will be reviewed against regulatory requirements and industry codes and standards. The review will be performed for the 4160 Vac

P	ROJECT INSTRUCTION	
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE 62 of 145	PREPARED BY: +1+1++ APPROVED BY:	

bus and will include those design provisions for load shedding coordination with load sequencing (Topic 1.25-2) and interlocks/permissives (Topic 1.7-2) associated with load shedding.

### 3.1.4.1.19 Fuel Oil System (DG) - Tupic 1.27-2

The Fuel Oil System provides storage and transfer for 7 days of Engineered Safety Feature (ESF) load. Criteria for system design will be reviewed against industry standards, FSAR commitments, and sound engineering practice based on operating plant experience. The review will further examine the documents which implement the criteria, including a review of portions of calculations which implement the criteria. Technical areas to be evaluated include: pump performance, fuel consumption vs. tank capacity, control logic, buried tank flooding, piping design (including vibration considerations), and unique fire protection considerations.

# 3.1.4.1.20 Lube Oil System (DG) - Topic 1.28-2

The performance criteria and implementation for lubrication and lubrication oil cooling will be reviewed with particular attention to the interface requirements established by the engine manufacturer. The key issue is to examine the design and maintenance requirements necessitated by engine design, and to review the documents used to implement such requirements.

# 3.1.4.1.21 Starting Mechanism and Air Supply System (DG) - Topic 1.29-2

The starting system includes both the pressurized air supply and distribution system, and the electric automatic start provisions. Criteria for the design will be reviewed against industry standards, FSAR commitments, and typical design

	P	ROJECT INS	TRUCTIO	N
PI0	DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde onstruction \	ng Program Plan pendent Design and /erification Program
PAGE _63	of _145	PREPARED BY:	バーニ	APPROVED BY

practice in response to operating experience. Implementation will be examined by reviewing technical specifications for system components, logic diagrams, piping and instrument drawings, and calculations. Interface with engine design requirements will receive particular attention. Technical areas under review include: diesel engine starting air specifications, pneumatic design and compressor sizing, control logic, moisture entrainment, air filter clogging, and automatic start power supply and field flashing.

# 3.1.4.1.22 Combustion Air Supply (DG) - Topic 1.30-2

Combustion air supply and engine exhaust for the diesel engine will be reviewed from criteria to implementation, including a check of selected design calculations. The criteria will be reviewed against FSAR commitments, e.g., seismic criteria and interface requirements of the engine design. Specific areas of technical review include: tornado, missile and debris considerations, seismic design, exhaust back pressure, intake flow rate, exhaust recycle, and effects of offsite gas release.

# 3.i.4.1.23 Independence (DG, PDS, AC, DC) - Topic 1.31-2

Diesel generator, power distribution system, 120 Vac preferred power and 125 Vdc battery systems will be reviewed for compliance with design criteria governing electrical independence. This review will include a survey of design criteria and commitments, implementing documents and a check of plant single-line drawings, selected schematic diagrams and circuit schedules.

P	ROJECT INSTRU	ICTION
PI	SUBJECT: E Midl Const	ngineering Program Plan and Independent Design and ruction Verification Program
PAGE of	PREPARED BY:	APPROVED BY.

#### 3.1.4.1.24 Cable Sizing/Routing/Separation (PDS) - Topic 1.32-2

This activity will include the review of criteria, implementing documents, calculations, drawings and specifications associated with SEP system cable sizing, routing and separation. The review will include a check of the sizing calculations applicable to the 4160 V cable, as well as other selected cable sizes. The specified routing of these cables will be reviewed for compliance with requirements, including separation criteria. In addition, an independent calculational check will be made of selected cable sizes and the results reviewed against the output of the design process.

### 3.1.4.2 SEP System Protection Features

In addition to the review of the capability of the SEP system to perform its required functions, a review will be made of external factors which could affect the capability of the system to achieve these functions. Included in the scope of this portion of the review are factors such as seismic design, high energy line break accidents (HELBA), environmental protection, fire protection, missile protection, and systems interaction. The following sections address these and other design areas related to system protection.

#### 3.1.4.2.1 Seismic Design - Topic II.1-2

The seismic design criteria and associated commitments related to the SEP system will be reviewed to ensure that an adequate basis for the design process is established. The review will include seismic design parameters and methodologies specified for the design of structures, systems and components associated with the SEP system.

	P	ROJECT INSTRUCTI	ON
PI_ 3201 _ 00 REV.: 2	DATE: 5/18/83	SUBJECT: Enginer Midland In Construction	ering Program Plan adependent Design and on Verification Program
PAGE 65	of 145	PREPARED BY:	APPROVED BY:

# 3.1.4.2.2 Seismic Design - Pressure Boundary (DG) - Topic 11.2-2

This activity will include a review of the commitments, implementing documents, calculations, drawings and specifications associated with seismic design of a selected portion of the diesel generator fuel oil piping system. The utilization of the proper design input, such as response spectra, piping and component weights, and other piping characteristics, will be verified. The ASME code evaluations will be reviewed to verify that pertinent acceptance criteria are met. Drawings and specifications will be reviewed for consistency with design calculations.

# 3.1.4.2.3 Seismic Design - Pipe/Equipment Support (DG, PDS) - Topic 11.3-2

A review of a selected portion of the diesel generator fuel oil system will be conducted to verify that selected supports have been designed and specified in accordance with criteria and commitments. Included will be the review of design loads, load combinations, and the methods of analysis utilized. The associated design drawings and specifications will be reviewed for consistency. Several support types will then be sampled, and an independent confirmatory analysis will be made to verify the capability of the original design organization to properly design and size these supports given the design loads. In addition, the design calculations, drawings and specifications associated with the anchorage and support of a diesel generator and selected other SEP system equipment will be reviewed for conformance to requirements.

	Р	ROJECT IN	STRUCTIO	N
PI REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde Construction V	ng Program Plan pendent Design and /erification Program
PAGE66	of _145_	PREPARED BY:	initel	APPROVED BY:

3.1.4.2.4 Seismic Design -- Equipment Qualification (DG, PDS) - Topic 11.4-2

This activity will include the review of commitments, implementing documents, calculations, drawings, and specifications associated with the seismic qualification of selected equipment. Qualification requirements including response spectra, load combinations, and equipment functional criteria will be reviewed. The review will include the following types of equipment of representative complexity such as: diesel generator, motor control center, electrical panel, fuel oil pump, motor-operated valve and heat exchanger.

3.1.4.2.5 High Energy Line Break Accidents - Topic 11.5-2

Criteria for postulation and evaluation of high energy line breaks will be reviewed to the extent that there are aspects unique to the SEP system which were not incorporated in the AFW review.

3.1.4.2.6 HELBA/Pipe Whip (PDS, AC, DC) - Topic II.6-2

Pipe whip evaluation criteria will be reviewed with an emphasis on interactions with electrical equipment, cable, controls, etc.

3.1.4.2.7 HELBA/Jet Impingement (PDS, AC, DC) - Topic II.7-2

Jet impingement evaluation criteria will be reviewed to assess the manner in which an interaction within the SEP system has been addressed.

1

化的规范	P	ROJECT IN	STRUCTIO	N
PI00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerir Midland Indep Construction V	ng Program Plan bendent Design and /erification Program
PAGE	of 145	PREPARED BY:	LIL	APPROVED BY:

#### 3.1.4.2.8 Environmental Protection - Topic II.8-2

The criteria review conducted for the AFW system will be extended to include considerations unique to the SEP system. Particular emphasis will be given to electrical circuit components, e.g., breakers.

### 3.1.4.2.9 Environmental Envelopes (DG, PDS) - Topic 11.9-2

The criteria review conducted for the AFW system will form a basis for the review of the SEP system. Concerns unique to this system will be included in the review. A check will be made to ensure that the system is included in the scope of the criteria.

# 3.1.4.2.10 Environmental/Equipment Qualification (DG, PDS) - Topic 11.10-2

Equipment qualification requirements will be reviewed to determine whether the correct environmental envelopes were specified and, given these envelopes, whether the qualification methods specified were adequate to demonstrate that the component would meet its functional requirements. The review will include the following types of equipment of representative complexity: electrical insulation, connectors, transmitters, and circuit breakers. Documents reviewed include technical specifications which are part of the procurement package, and vendor qualification reports or test reports.

### 3.I.4.2.II HVAC Design (DG) - Topic II.II-2

Requirements imposed upon specific portions of the HVAC systems that limit environmental conditions for the SEP system equipment will be reviewed. This

P	ROJECT IN	STRUCTIO	N
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineerir Midland Inder Construction V	ng Program Plan Dendent Design and Verification Program
PAGE of	PREPARED BY:	1111=1:	APPROVED BY:

will be achieved by a verification of the design interface between the two systems.

#### 3.1.4.2.12 Fire Protection (DG) - Topic II.: 2-2

The unique fire protection concerns associated with diesel fuel will be addressed in a review of the Diesel Generator Building and underground storage area. The extent of that review will include criteria development to a check of calculations. In particular, malfunctions which result in spillage of fuel or lube oil will be reviewed for potential to cause loss of redundant power generation. The function of detection and suppression capabilities under most severe incidents will be reviewed. Included in the review will be adequacy of fire barriers separating redundant diesel-generator units, potential effects on the diesel reliability due to fire protection systems, and potential interactions between diesel-generator rooms in a fire.

# 3.1.4.2.13 Missile Protection (DG) - Topic II.13-2

The criteria for postulating events which result in internal missile generation will be reviewed against current licensing practice and operating experience. Criteria for assessing damage due to postulated missiles will be reviewed as an extension of the AFW review.

# 3.1.4.2.14 Systems Interaction (DG, PDS, AC, DC) - Topic 11.14-2

The potential for systems interaction and means for prevention thereof will be reviewed. The review will include an examination of criteria utilized to analyze

P	ROJECT INSTRUC	TION
PI REV.: 2 DATE: 5/18/83	SUBJECT: Engin Midland Construct	neering Program Plan Independent Design and tion Verification Program
PAGE of	PREPARED BY:	APPROVED BY: Aut

potential physical or spatial interactions. The procedures and results for the Midland systems interaction walkdowns will be reviewed and, if possible, ongoing walkdowns will be observed.

#### 3.1.4.3 Structures that House the SEP Systems

Many safety-related plant structures support the functioning of the SEP system or its support systems. The overall criteria and commitments applicable to the design of these safety related structures will be reviewed and evaluated. Selected features and design areas of the diesel generator building structure will be isolated for a more in-depth review in the following topics.

# 3.1.4.3.1 Seismic Design/Input to Equipment (DGB) - Topic III.1-2

This activity will include the review of commitments, implementing documents, calculations, drawings, and specifications related to the development of seismic design input for a portion of the SEP system and associated components in the diesel generator building. Included, will be a review of seismic input parameters such as seismic design spectra, damping, material properties, and boundary conditions, including soil-structure interaction. The methodology utilized for the location of the mass points and the computation of masses and equivalent member properties will be reviewed. Parameter variation studies will also be reviewed to verify that the variance of important input parameters and modeling assumptions has been appropriately considered. The scope of this activity will include the review of the dynamic analysis of the diesel generator building, the time history analysis and the generation of floor response spectra for both horizontal directions and the vertical direction. The utiliziation of

	P	ROJECT INS	STRUCTIC	N
PI- <u>3201</u> -0 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ing Program Plan pendent Design and Verification Program
PAGE 70	of _145	PREPARED BY:	山山	APPROVED BY:

proper floor response spectra for the specification of selected SEP system components will be verified.

#### 3.1.4.3.2 Wind and Tornado Design/Missile Protection (DGB) - Topic 111.2-2

Criteria and commitments, implementing documents, and calculations for a portion of the diesel generator building will be reviewed to verify proper incorporation into the design. Included will be the review of wind pressure loading, tornado wind loading, tornado depressurization effects, tornado missile, and other related requirements. Loading combinations, methodologies of analysis, associated allowable stresses or conditions, and other specified criteria will be included in this review activity. In addition, drawings and specifications associated with the selected portion of the diesel generator building will be reviewed for consistency with criteria and calculations.

#### 3.1.4.3.3 Flood Protection (DGB) - Topic III.3-2

This activity will include the review of criteria, implementing documents and calculations related to flood protection from sources both external and internal to the diesel generator building. The specification of the design flood level and the methods utilized to provide the necessary flood protection will be reviewed. Included will be the review of the calculations associated with the determination of postulated pipe break locations and the methodologies utilized in determining flow rates and resulting water levels. The results of this review will be considered in the evaluation of Topic III.7-2, Concrete and Steel Design of the diesel generator building.

P	ROJECT INSTRUCTIO	N
PI-3201 -009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineerin Midland Indep Construction	ng Program Plan pendent Design and /erification Program
PAGE of	PREPARED BY: LILLE	APPROVED BY:
		and the second s

#### 3.1.4.3.4 HELBA Loads (DGB) - Topic III.4-2

Criteria and commitments for high energy line break accident loads in the diesel generator building will be reviewed to verify that the proper basis is established for the design process. Included will be the review of the criteria for jet impingement and pipe whip loading on structures and components related to the SEP system. The review will address loading combinations, methodologies of analysis, associated allowable stresses or conditions, and other related criteria. The results of this review will be considered in evaluation of Topic III.7-2, Concrete and Steel Design.

#### 3.1.4.3.5 Civil/Structural Design Considerations - Topic 111.5-2

Civil/structural design criteria and associated commitments related to the diesel generator building will be reviewed, and the establishment of the proper basis for the associated design process will be confirmed. Included will be the review of design parameters and the methodologies utilized in the design process for structures and affected systems and components associated with the SEP system.

#### 3.1.4.3.6 Foundations (DGB) - Topic III.6-2

Included in this activity will be the review of criteria, implementing documents, and calculations associated with the design of selected foundations of the diesel generator building. The review will address design criteria, methodologies of analysis and calculations associated with each type of foundation loading including dead, live, wind and seismic loadings.

P	ROJECT INSTRUCTION	
PI	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE of	PREPARED BY:	

#### 3.1.4.3.7 Concrete and Steel Design (DGB) - Topic 111.7-2

This activity will include the review of criteria, implementing documents, calculations, drawings, and specifications associated with the reinforced concrete and structural steel design of selected structural elements associated with the diesel generator building. Structural elements, including a major load bearing shear wall and a flood diaphragm will be selected that require consideration of a broad spectrum of loadings such as dead, live, wind, tornado, seismic, flood and HELBA loads. The review will address design criteria, methodologies of analysis and calculations associated with each type of loading with emphasis on a verification that these items have been considered in a realistic manner. Loading combinations, allowable stresses or conditions, and other applicable criteria will be reviewed. Drawings and specifications for the selected structural elements will be reviewed against design calculations for consistency.

#### 3.1.4.3.8 Tanks - Topic 111.8-2

This activity will include the review of criteria, implementing documents, and calculations associated with the design of a selected tank associated with the diesel generator. A review of the calculations for the underground emergency diesel oil storage tank and foundation will be made. All applicable loadings will be reviewed, such as dead, live, wind, tornado, seismic (including fluid dynamics effects and flooding, as applicable. The review will address tank design criteria and methodologies of analysis, and will include loading combinations, allowable stresses or conditions, and other applicable criteria. Drawings and specifications will be checked for consistency with calculations. In addition, an independent

Р	ROJECT INSTRUC	CTION
PI_3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Eng Midlan Construct	ineering Program Plan d Independent Design and ction Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

confirmatory analysis of the diesel oil storage day tank integrity including its support system will be conducted and the results compared with the output of the design process.

# 3.1.5 DEFINITION OF REVIEW SCOPE FOR THE CONTROL ROOM HVAC SYSTEM

The categories of review identified in section 3.1.1 are also applicable to the review of the Control Room HVAC (CR-HVAC) system. Similarly, the criteria discussed in section 1.3 and 3.1.2 were incorporated to develop the initial design verification matrix shown on Figures 1.2-6a and 1.2-6b. The design areas (or topics) of the IDV review matrix for the CR-HVAC system are somewhat different from those for the AFW or SEP systems, consistent with the differences in the functions and physical configuration of these systems. The review philosophy, matrix concepts and organization remain the same. The design areas for the CR-HVAC system Performance Requirements, CR-HVAC System. The design areas addressed within each of these major divisions are discussed in sections 3.1.5.1, 3.1.5.2 and 3.1.5.3 of this Plan, respectively. As previously mentioned, the identified initial review scope is subject to change depending upon whether the criteria documented in section 3.1.2 have been met.

Because the CR-HVAC system sample selection interfaces with other systems, it is necessary to define boundaries for items within the scope of the IDCV. In general the CR-HVAC system sample selection boundaries include the Control Room Area Ventilation System (CRAVS), its support systems and components important to control room isolation and habitability during accident conditions; either radiological or chemical.

1.	P	ROJECT IN	STRUCTIO	N
PI 00	DATE: 5/18/83	SUBJECT:	Engineerir Midland Indep Construction V	ng Program Plan pendent Design and /erification Program
PAGE 74	of _145	PREPARED BY:	山北	APPROVED BY:

Specific interface points are as follows:

# CONTROL ROOM HVAC SYSTEM SAMPLE SELECTION BOUNDARIES

Interfacing System	Interfacing Point	
ac/dc Power System	All portions of Class IE electric system serving the CR HVAC are included in the Standby Electric Power (SEP) System review (see Section 3.1.4 for SEP sample selection boundaries).	
Plant HVAC	Portion of the Control Room Area Ventilation System (CRAVS) (FSAR Figures 9.4–1 and 9.4–2) up to and including:	
	Valves OMO 6545A OM 6557 OMO 6545B OM 6549 OMO 6543A OM 6547A OMO 6543B OM 6547B OMO 6554	
Equip. & Piping Supports	Includes all supports incorporated in the seismic qualification of the Control Room portion of the CRAVS as defined above.	
ESFAS	Includes Control Room Isolation System (CRIS) subsystem, FSAR Figure 7.3-5.	
Accident Monitoring Inst.	Portions essential for isolation of Control Room and operation of CRAVS, e.g. - intake duct radioactivity - charcoal filter temperature - hazardous gas concentration See FSAR Tables 7.5-1 and 7.5-3:	
Plant I&C	Portions essential for isolation of Control Room and CRAVS operation.	

Control Room Structure Portions required for pressure boundary including penetrations and doors.

PROJECT INSTRUCTION				
PI3201 _ 009		SUBJECT: Eng Midlan	ineering Program Plan d Independent Design and	
PAGE 75	of 145	PREPARED BY:	APPROVED BY:	

The following sections discuss the initial scope of review for each of the design areas.

#### 3.1.5.1 CR-HVAC System Performance Requirements

The CR-HVAC system will be reviewed to assess its capability to perform as required by the design criteria and commitments. Included in the scope of this portion of review are design areas such as system operating requirements, single failure, component functional requirements, instrumentation, detection, failure modes and effects, filtration, and ventilation.

#### 3.1.5.1.1 System Operating Limits - Topic 1.1-13

This topic will involve review of available documentation to extract various system parameters and their values. These will include temperature to be maintained, assumed leakage rates, system flow rates, required control room pressure, etc. Results of this review involving key components and their parameters will be reviewed further in related mechanical and electrical topics.

Ancillary systems such as Safeguards Chilled Water and Essential Service Water will be reviewed only from the standpoint of ascertaining that parameters dependent on their operability can indeed be met.

#### 3.1.5.1.2 Accident Analysis Considerations - Topic 1.2-3

FSAR Safety Analyses will be reviewed to determine accident and post-accident operability requirements involving the control room HVAC system (smoke from an external fire, chemical release, radiological accident, and main steam line break). This will include a review of detection systems and assumptions regarding their capabilities.

P	ROJECT INST	RUCTION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: M Cor	Engineering Program Plan idland Independent Design and astruction Verification Program
PAGE of145	PREPARED BY:	

#### 3.1.5.1.3 Single Failure - Topic 1.3-3

Single-failure-proof design aspects of the system will be reviewed, commencing with information in the FSAR and proceeding to a check of calculations and evaluations, if available.

# 3.1.5.1.4 Technical Specifications - Topic 1.4-3

The Technical Specifications will be reviewed against existing regulatory criteria to assure that important operating limits are appropriately and accurately specified. An effort will also be made to assess whether additional specifications might be necessary, based upon knowledge gained from the detailed review of the system.

# 3.1.5.1.5 System Alignment/Switchover - Topic 1.5-3

This review will encompass such diverse topics as the local handswitches for the various fans, the interlocks associated with these switches and fans, the required actions of the Control Room Isolation System (CRIS), the requirements for post-accident operation resulting from the various external initiating events, and any assumptions regarding required manual operation of the system.

# 3.1.5.1.6 System Isolation/Interlocks - Topic 1.7-3

The interlocks of the control room HVAC system will be reviewed to determine their adequacy in implementing system performance requirements. This will include any interlocks associated with CRIS, fan and isolation damper interlocks, pressurization system interlocks, etc.

P	ROJECT IN	STRUCTIC	N
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE of5	PREPARED BY:	LIK	APPROVED BY:

# 3.1.5.1.7 Component Functional Requirements - Topic 1.9-3

Functional requirements for selected components will be reviewed against requirements imposed directly by licensing commitments and industry codes and standards. Additionally, system design imposes performance requirements. The development of such requirements will be traced from their source to statement in system design criteria, and the validated component functional requirements will be compared to technical specifications in the procurement documents. Component functional requirements include fan curve data, valve/damper leak rates, load rating, instrument sensor detection threshold, and similar characteristics. Equipment seismic and environmental qualifications will be considered separately.

# 3.1.5.1.8 System Pneumatic Design - Topic 11.10-3

The calculations of system pressure drop through ducts, dampers, filters, etc., for safety-related modes will be checked against design criteria. A confirmatory calculation for a critical system alignment will be performed to establish functional requirements which will be compared to those specified.

# 3.1.5.1.9 Cooling/Heating Requirements - Topic 1.12-3

System heating and cooling loads for the control room will be reviewed if they are safety related. The review will examine stability of control room temperature in an acceptable range under isolated conditions.

	P	ROJECT INSTRUCT	NON
PI3201 _ 00 REV.: 2	DATE: 5/18/83	SUBJECT: Engin Midland Construct	eering Program Plan Independent Design and ion Verification Program
PAGE 78	of _145_	PREPARED BY:	

3.1.5.1.10 Preservice Testing and Capability for Operational Testing - Topic 1.14-3

A review will be made of the criteria and commitments and implementing documents which exist for preservice testing and the capability for operational testing of a selected portion of the CR-HVAC system.

3.1.5.1.11 Power Supplies - Topic 1.15-3

Design criteria and commitments will be reviewed for control room HVAC power supplies. The aspects to be reviewed include power supply separation, redundancy and diversity. The FSAR commitments regarding power supply design features will be compared to applicable design criteria.

3.1.5.1.12 Instrumentation/Detection - Topic 1.18-3

Control Room HVAC instrumentation will be reviewed for its adequacy to monitor the operational status of the system and to detect adverse environmental conditions. The design basis, drawings and specifications for make-up air radiation and toxic gas detectors will be reviewed. The review will include a review of the basis for selected gas concentration limits and setpoints for the Hazardous Gas Monitoring System (HGMS).

3.1.5.1.13 Control Systems - Topic 1.19-3

Design criteria and commitments will be reviewed against FSAR commitments regarding control systems associated with CR-HVAC system. The criteria and

	P	ROJECT INSTRUCT	TION
P! REV.: 2	9 DATE: 5/18/83	SUBJECT: Engin Midland Construct	neering Program Plan Independent Design and tion Verification Program
PAGE 79	of 145	PREPARED BY:	APPROVED BY:

FSAR commitments will be reviewed against system logic diagrams and schematic diagrams or applicable loop diagrams to verify implementation of the design criteria.

#### 3.1.5.1.14 Actuation Systems - Topic 1.20-3

The CR-HVAC actuation system will be reviewed against criteria governing the design of protection systems for nuclear power plants (IEEE Standard-279). The review will cover design criteria and commitments, a review of implementing documents and a check of evaluations and drawings governing or documenting actuation system design and implementation. The review will include a check of schematic diagrams for actuation system input and a check of the actuation system procurement specification for compliance to the design criteria.

3.1.5.1.15 Nondestructive Examination Commitments - Topic 1.21-3

A determination will be made of the design criteria and commitments which exist for NDE for the CR-HVAC system components. Included will be the review of associated evaluations to ensure adequate provision for NDE. The results of this determination will be used in the ICV portion of the IDCV, which will verify that the system has been constructed such that it can function in accordance with its design criteria and commitments.

	P	ROJECT IN	ISTRUCTIO	N
PI- <u>3201</u> - 00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde Construction	ng Program Plan pendent Design and /erification Program
PAGE80	of 145	PREPARED BY: 111- APPROVED BY:		APPROVED BY:

#### 3.1.5.1.16 Materials Selection - Topic 1.22-3

This activity will include the review of criteria implementing documents and evaluations related to establishing the bases for the material specification process of selected components and elements of the CR-HVAC system. Specifications will be reviewed to verify that these are in compliance with design engineering requirements.

#### 3.1.5.1.17 Failure Modes and Effects - Topic 1.23-3

The various modes of failure, and the effects of such failure will be reviewed, using as a basis any analyses presented in the FSAR. This review will determine if there is any potential failure that has not been previously analyzed.

#### 3.1.5.1.18 Filtration - Topic 1.33-3

Licensing commitments will form the basis for reviewing the filtering capability for airborne radioactive materials and toxic gases. Calculation of airborne dose rates will be checked. Specifications of critical filter parameters will be checked, and interface requirements between system design and filter performance will be reviewed.

#### 3.1.5.1.19 Pressurization - Topic 1.34-3

Control room pressurization to ensure exfiltration under post-accident and offsite hazardous gas releases will be reviewed. The basis for establishing quantitative values for leakage area will be checked. System flow

P	ROJECT INS	TRUCTION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and onstruction Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

characteristics validated in the pneumatic design review and the fan curves will be used to check global pressure differential. Local effects and relevant meteorological assumptions will be reviewed. The calculations for sizing of the bottled air supply will be checked. Analyses considering the capability of the self-contained breathing apparatus will be evaluated.

#### 3.1.5.1.20 Ventilation - Topic 1.35-3

The overall HVAC ventilation system design will be reviewed to determine its capability to maintain control room habitability parameters, i.e., temperature, oxygen content, toxic gas levels, and airborne radiation. Criteria for acceptable levels will be checked against licensing commitments and industry standards for both survival and minimal conditions under which reasonable decision making is expected. Calculations to predict equilibrium temperature, oxygen level, and airborne contamination will be checked in critical modes. Synergistic effects of toxic gases will also be considered. A confirmatory calculation of control room habitability parameters will be performed for one critical mode of operation.

# 3.1.5.2 CR-HVAC System Protection Features

In addition to the review of the capability of the CR-HVAC system to perform its required functions, a review will be made of external factors which could affect the capability of the system to achieve these functions. Included in the scope of this portion of the review are factors such as seismic design, high energy line break accidents (HELBA), environmental protection, fire protection, missile protection, and systems interaction. The following sections address these and other design areas related to system protection.

	P	ROJECT INSTRUCTION	ON
PI	DATE: 5/18/83	SUBJECT: Enginee Midland Inc Construction	ring Program Plan dependent Design and Verification Program
PAGE 82	of 145	PREPARED BY:	APPROVED BY:

# 3.1.5.2.1 Seisrnic Design - Topic II.1-3

The seismic design criteria and associated commitments related to the CR-HVAC system will be reviewed to ensure that an adequate basis for the design process is established. The review will include seismic design parameters and methodologies specified for the design of ductwork and supports, equipment and components associated with the CR-HVAC system.

# 3.1.5.2.2 Seismic Design -- Pressure Boundary - Topic II.2-3

This activity will include a review of the commitments, implementing documents, calculations and evaluations associated with seismic design of the pressure boundary of a selected portion of the CR-HVAC system. The utilization of the proper design input, such as response spectra, ductwork and component weights, and other ductwork characteristics, will be verified. The calculations and evaluations will be reviewed to verify that pertinent acceptance criteria are met.

3.1.5.2.3 Seismic Design - Duct/Pipe/Equipment Support - Topic 11.3-3

A review of a selected portion of the CR-HVAC system will be conducted to verify that selected supports have been designed and specified in accordance with criteria and commitments. Included will be the review of design loads, load combinations, and the methods of analysis utilized. The associated design drawings and specifications will be reviewed for consistency. The review will encompass representative duct and pipe supports associated with the selected portion of the system. In addition, the design calculations, drawings

	P	ROJECT INSTRUC	TION
PI	DATE: 5/18/83	SUBJECT: Engir Midland	neering Program Plan Independent Design and tion Verification Program
PAGE 83	of 145	PREPARED BY:	

and specifications associated with the anchorage and support of selected CR-HVAC system equipment will be reviewed for conformance to requirements.

3.1.5.2.4 Seismic Design - Equipment Qualification - Topic II.4-3

This activity will include the review of commitments, implementing documents, calculations, drawings, and specifications associated with the seismic qualification of selected CR-HVAC equipment. Qualification requirements including response spectra, load combinations, and equipment functional criteria will be reviewed. The review will include various types of HVAC equipment of representative complexity such as: fans, motors, coolers, dampers, filters and associated instrumentation.

### 3.1.5.2.5 High Energy Line Break Accidents - Topic 11.5-3

Criteria for postulation and evaluation of high energy line breaks will be reviewed to the extent that there are aspects unique to the CR-HVAC system which were not incorporated in the AFW review.

#### 3.1.5.2.6 HELBA/Pipe Whip - Topic II.6-3

A review will be made to determine the criteria applicable to high energy lines routed near the control room. To the extent that the CR-HVAC system is subject to criteria for HELBA/Pipe Whip which differ from those applicable to the AFW system, those criteria will be evaluated.

	P	ROJECT INST	RUCTION
PI 00	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Aidland Independent Design and
PAGE	of _145_	PREPARED BY:	

#### 3.1.5.2.7 HELBA/Jet Impingement

A review will be made to determine the criteria applicable to preventive protective measures taken to assure acceptable consequences due to postulated jets that may impact CR-HVAC components important to the functioning of this system. This topic will be reviewed in conjunction with Topic II.6-3, Pipe Whip.

# 3.1.5.2.8 Environmental Protection - Topic II.8-3

The criteria review conducted for the AFW and SEP systems will be extended to include considerations unique to the CR-HVAC system.

### 3.1.5.2.9 Environmental Envelopes - Topic 11.9-3

All major equipment and representative electrical components in the system will be reviewed to verify that they can perform their required function under the most severe conditions resulting from the pipe breaks postulated in accordance with the HELBA criteria. Implementation of criteria to select break locations will be reviewed. The calculations to determine sub-compartment temperatures and pressures will be reviewed to trace the application of criteria and the validity of the technical approach. Validated conditions will be compared with the specifications in procurement documents for the equipment. One environmental envelope applicable to the CR-HVAC system will be subject to a confirmatory calculation, provided that the envelope used to qualify the HVAC equipment is different from that used to qualify AFW components.

P	ROJECT INSTRUCTION
PI-3201 009 REV: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE 85 of 145	PREPARED BY: 111- APPROVED BY: Que Bud

#### 3.1.5.2.10 Environmental Equipment Qualification - Topic 11.10-3

Based on the environmental envelope review, equipment and components will be selected to examine their qualification for the most severe environment. Various types of equipment will be reviewed including valve operators, dampers, instrumentation, and fan motors. The review will focus on two separate aspects: the vendor calculations and testing, and CPC's program for technically confirming vendor compliance to procurement specifications. The depth of review for vendor calculations will be determine based on an assessment of the comprehensiveness of the CPC's program.

### 3.1.5.2.11 Fire Protection - Topic 11.12-3

The CR-HVAC system will be reviewed to determine its ability to control smoke due to small fires in the control room, smoke from fires outside the building, and smoke from fires in areas adjacent to the control room. This review includes an evaluation of criteria and implementing documents. The potential for interaction between the control room and the remote shutdown panel area with regard to smoke control and fire suppression will be reviewed.

#### 3.1.5.2.12 Missile Protection - Topic II.13-3

The criteria for postulating events which result in internal missile generation will be reviewed against current licensing practice and operating experience. Criteria for assessing damage due to postulated missiles will be reviewed as an extension of the AFW and SEP review.

	P	ROJECT INST	RUCTION	
PI		SUBJECT: M Cor	Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE	of 145	PREPARED BY: 1-1	APPROVED BY:	

# 3.1.5.2.13 Systems Interaction - Topic II.14-3

The potential for systems interaction and means for prevention thereof will be reviewed. The review will include an examination of criteria utilized to analyze potential physical or spatial interactions. The procedures for, and results of, the Midland systems interaction walkdowns will be reviewed and, if possible, ongoing walkdowns will be observed.

# 3.1.5.3 Structures that House the CR-HVAC System

The auxiliary/control building structure supports the functioning of the CR-HVAC system. The overall criteria and commitments applicable to the design of this structure will be reviewed and evaluated in conjunction and as an extension of the AFW system evaluation. Selected additional features from this structure will be isolated for more in-depth review in the following topics.

# 3.1.5.3.1 Seismic Design/Input to Equipment - Topic III.I-3

This activity will include the review of commitments, implementing documents, calculations and evaluations related to the development of seismic design input for the CR-HVAC system and associated components in the auxiliary building. Included will be a review of seismic input parameters such as seismic design spectra, damping, material properties, and boundary conditions, including soil-structure interaction. The methodology utilized for the location of the mass points and the computation of masses and equivalent member properties will be reviewed. Parameter variation studies will also be reviewed to verify that the variance of important input parameters and modeling assumptions has been

	P	ROJECT IN	STRUCTIO	N
PI		SUBJECT:	ECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE 87	of 145	PREPARED BY:	141:4	APPROVED BY:

appropriately considered. The scope of this activity will include the review of the dynamic analysis of the auxiliary building, the time history analysis and the generation of floor response spectra for both horizontal directions and the vertical direction. The utilization of proper floor response spectra for the specification of the selected CR-HVAC system components will be verified.

# 3.1.5.3.2 Civil/Structural Design Considerations - Topic III.5-3

Civil/structural design criteria and associated commitments related to the CR-HVAC system will be reviewed, and the establishment of the proper basis for the associated design process will be confirmed. Included will be the review of design parameters and the methodologies utilized in the design process for structures and affected systems and components associated with the system.

3.1.5.3.3 Concrete and Steel Design - Topic 111.7-3

This activity will include the review of criteria and implementing documents associated with the reinforced concrete and structural steel design of selected structural elements associated with the CR-HVAC system. The review will address design criteria and specified methodologies of analysis associated with each type of loading with emphasis on a verification that these items are considered in a realistic manner. Loading combinations, allowable stresses or conditions, and other applicable criteria will be reviewed.
P	ROJECT INS	TRUCTION
PI	SUBJECT:	Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE of	PREPARED BY:	IIII APPROVED BY:

#### 3.1.5.3.4 Leak Tightness - Topic 111.9-3

This activity will include the review of criteria, implementing documents and calculations associated with the leak tightness of the control room. The review will address the provision for component specifications required to ensure control room leak tightness criteria are met.

#### 3.1.6 DEVELOPMENT OF IDV PROGRAM CHECKLISTS

Generic checklists exist for each of the review scope categories discussed in section 3.1.1, utilizing guidance contained in ANSI N45.2.11. These checklists have been developed to assist in documenting engineering evaluations, and providing uniformity of review from topic to topic and between reviewers. For each of the design areas within the scope shown in Figures 1.2-2a and 2b, 1.2-4a and 4b and 1.2-6a and 6b, the reviewer may utilize the generic checklists or develop a specific checklist incorporating portions of the generic checklists as appropriate. In most cases, the specific checklist will be derived from the generic checklist by the addition of specific requirements applicable to the design area being reviewed. In some cases, it may be appropriate to use only a portion of the generic checklist or to develop a unique checklist. The specific instructions for the use of design verification checklists are contained in Project Instruction PI-320I-006, Use of Design Verification Checklists.

In each case that a generic checklist has been modified, the checklist prepared by the reviewer will be checked by the Lead Technical Reviewer (LTR) for the area. (Note that if the LTR prepares a checklist, it is permissible for him to both originate and check the contents of the checklist). During their review process, the LTRs examine the checklist for interfaces with other IDV areas and perform a general review of the completeness and adequacy of the

DC-82-13

	P	ROJECT INST	RUCTION
PI 0	DATE: 5/18/83	SUBJECT:	Engineering Program Plan idland Independent Design and instruction Verification Program
PAGE 89	of 145	PREPARED BY:	APPROVED BY

proposed checklist. The LTR's review is to be coordinated with the Project Manager as necessary to resolve questions which cut across discipline lines. In the event that the Project Manager or the LTR has comments on the checklist, the checklist preparer and those having comments will discuss the comments and reach an appropriate resolution. After reaching concurrence in the adequacy of the checklist, the LTR will indicate his approval and the checklist will be available for use by the reviewer.

The reviewer, having an approved checklist, can then proceed into the review process for this specified area, in accordance with Project Instruction PI-3201-001, Engineering Evaluation Preparation and Control. In performing the engineering evaluation, the reviewer will document the information which he used in order to complete the checklist. Such information will include the data or revision number of the document, the document number, an indication of the source of the document (e.g., whether the document was obtained from an individual, a file, or the records center).

# 3.1.6.1 Development of Checklists for Review of Design Criteria and Commitments

The generic checklist for review of design criteria and commitments was developed considering questions such as:

- What are the design inputs for the design area under review?
- Do any of these design inputs affect other design areas?
- Do any of these design inputs affect interfacing systems outside the scope of AFW or vice versa?

	P	ROJECT IN	ISTRUCTIC	DN
PI320100 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineer Midland Inde Construction	ing Program Plan ependent Design and Verification Program
PAGE 90	of 145	PREPARED BY:	Lipton	APPROVED BY:

- Are the design inputs for this design area complete?
- Are the identified design inputs for this design area consistent?
- Are the design inputs adequately defined to allow implementation for the design area?

For each design area the lead technical reviewer will supplement the generic checklist with appropriate additional questions.

#### 3.1.6.2 Development of Checklists for Reviews of Implementing Documents

The generic checklist for reviews of implementing documents was developed considering questions such as:

- What is the identity of the implementing document being reviewed? (List document identification such as title, revision number, date, etc.)
- For the design inputs being reviewed, is the document complete and internally consistent?
- Are design interface requirements specified?
- Have the design inputs been correctly interpreted and incorporated in this implementing document?
- Is this implementing document consistent with other implementing documents being reviewed for this area?
- Are assumptions and limitations on the use of the document adequately defined?
- Where appropriate, are quality assurance requirements specified?

		P	ROJECT IN	STRUCTIC	N
PI_3201 00 REV: 2	DATE:	5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 91	of 145		PREPARED BY:	LILICA	APPROVED BY:

For each design area the lead technical reviewer will supplement the generic checklist with appropriate additional questions for each implementing document.

#### 3.1.6.3 Development of Checklists for Checks of Calculations and Evaluations

The generic checklist for checks of calculations and evaluations was developed considering questions such as:

- What is the identity of the calculation or evaluation being checked?
- What is the purpose of the calculation or evaluation?
- Are the data sources identified?
- Are the assumptions listed?
- Are the assumptions reasonable and valid?
- Was the calculation or evaluation checked and approved within the originating organization?
- Are the equations and methods specified?
- Are the equations and methods appropriate for the intended purpose?
- If computer programs were used, were such programs verified?
- Are the calculation or evaluation results reasonable?
- Have design outputs been compared to the acceptance criteria to allow verification that design requirements . have been satisfactorily accomplished?

1. 1. S. C. .

San and

**.** .

1. 1. Sax

ું કે છે. '

P	ROJECT INSTRUC	TION
PI-320: 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engir Midland Construct	neering Program Plan Independent Design and tion Verification Program
PAGE 92 of 145	PREPARED BY:	APPROVED BY:

For each design area the lead technical reviewer will supplement the generic checklist with appropriate questions for each calculation or evaluation checked.

#### 3.1.6.4 Development of Checklists for Checks of Drawings and Specifications

The generic checklists for checks of drawings and specifications were developed considering questions such as:

- What is the identity of the drawing or specification (e.g. number, revision number, date)?
- Does the drawing or specification reflect the selected design inputs?
- Is the drawing or specification consistent with related calculations or evaluations?
- Has this drawing or specification been checked by the originating organization?
- Is the drawing or specification complete with regard to the selected design inputs?
- Where appropriate, have adequate handling, storage cleaning, and shipping requirements been specified in the specification?
- Where appropriate, has adequate allowance been made for inservice inspection, maintenance, repair, and testing?

For each design area, the lead technical reviewer will supplement the generic checklist with appropriate questions for each drawing or specification being reviewed.

		Ρ	ROJECT IN	STRUCTIO	N
PI 00 REV.: 2	DATE:	5/18/83	SUBJECT:	Engineerir Midland Indep Construction V	ng Program Plan pendent Design and /erification Program
PAGE _93_	of 145	<u> </u>	PREPARED BY:	1144.	APPROVED BY:

#### 3.1.7 PLAN FOR ADDITIONAL SAMPLING AND VERIFICATION

Additional sampling or verification within the scope of the IDV or outside the scope into other systems will be conducted if discrepancies are found. The level of additional sampling or verification will be based upon the nature of the discrepancy. In all cases when discrepancies are found, an introspective evaluation will follow to identify the extent and root cause. The root cause may either be random or systematic (generic). The additional review will attempt to verify whether the discrepancy is restricted to the specific system, component, or structure under review; restricted to work by a specific design organization; or if the discrepancy cuts across many interfaces and applies to similarly designed systems, components, and structures. As a rule, mathematical errors will not precipitate additional sampling and verification unless these are found in significant numbers, leading to significant deficiencies or a compounding of errors. Judgement in making this assessment will be required on case-by-case basis.

As necessary, additional sampling or verification within the scope of the system sample selection boundaries identified in sections 3.1.3, 3.1.4 and 3.1.5 of this Plan will be undertaken by TERA and approved internally. All such actions will be documented in the Monthly Status Report. Additional sampling outside this scope is considered a substantive issue and will be discussed between TERA, CPC and NRC prior to initiation.

3.2 INDEPENDENT CONSTRUCTION VERIFICATION METHODOLOGY

The Independent Construction Verification (ICV) Program will consist of a review and evaluation of the quality of construction of selected components and

P	ROJECT INSTRUC	TION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engi Midland Construct	ineering Program Plan d Independent Design and ction Verification Program
PAGE 94 of 145	PREPARED BY:	APPROVED BY:

structures associated with the AFW, SEP and CR-HVAC systems. The construction activities to be reviewed include the major activities of the construction chain. These include the fabrication, storage, maintenance, installation or construction, and verification activities associated with the acceptance of the system or component, as further defined in Section 3.2.1 herein. The emphasis will be on making a determination of the overall quality of construction and an assessment of its compliance with licensing commitments. The review will be conducted to varying stages of construction completion depending upon the specific system, component, or structure under review. The methodology will include diverse approaches such as checking of records, hands-on inspection of hardware, and confirmatory testing. The basis for the sample selection is presented in Section 3.2.2. The definition of the scope of review is provided in the following sections:

System	Section
AFW	3.2.3
SEF	3.2.4
CR-HVAC	3.2.5

In many instances, a complete verification of the as-built configuration against design documents and other applicable requirements will be included. Where possible, systems and components selected for the Independent Design Verification Program will be utilized for review in the ICV Program, thereby providing verification of the complete chain from criteria and commitments through to the constructed and verified product.

P	ROJECT IN	STRUCTIC	N
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineer Midland Ind Construction	ing Program Plan ependent Design and Verification Program
PAGE 95 0 45	PREPARED BY:	LILL	APPROVED BY:

The ICV Program will be conducted utilizing detailed checklists described in Section 3.2.6. Additional sampling, verification, and testing activities that may be conducted as a result of the ICV Program are discussed in Section 3.2.7.

#### 3.2.1 CATEGORIES OF REVIEW: THE CONSTRUCTION CHAIN

The categories of review include the major construction activities identified in the construction chain. The ICV review categories included are:

- Review of supplier documentation
- Review of storage and maintenance documentation
- Review of construction/installation documentation
- Review of selected vesification activities
- Verification of physical configuration

It is necessary to reemphasize that the ICV review will be conducted to varying stages of construction completion depending upon the specific system, component or structure under review. As such, the ICV review categories will, as a minimum, include a detailed review of a static situation, or one which verifies the results of a completed activity, in addition to observations and reviews of a more dynamic environment where the construction activity being reviewed is actually in progress or has not been completed. The results of these types of reviews will be integrated with an assessment of selected, on-going over-inspection activities and the Construction Completion Program (see section 3.2.1.4). Proceeding in this manner allows an even-handed, objective appraisal of not only the quality of construction for completed items, but also permits an

Р	ROJECT INSTRUCTION
PI 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE of _145	PREPARED BY

evaluation of the outputs from on-going site activities undertaken to verify and confirm the quality of construction.

Each of these review categories is described in further detail in the following sections.

#### 3.2.1.1 Review of Supplier Documentation

For those components requiring fabrication or manufacture, selected supplier documentation and other associated information including shop inspection documentation will be reviewed against design output documents to ensure conformance with requirements. Supplier documentation will include such items as drawings, calculations, test reports, certified material property reports, storage and installation requirements, operations and maintenance requirements, and other major supplier documentation and data applicable to the component. For selected components, the review of supplier seismic and environmental qualification documentation against requirements defined in the design process will be included.

#### 3.2.1.2 Review of Storage and Maintenance Documentation

A review of site documentation will be performed to verify that requirements related to storage, including both in-storage and in-place maintenance have been met. Included will be the review of receipt inspection documentation. Requirements to be reviewed will include such parameters as temperature and humidity, cleanliness, lubrication, shaft rotation, energization, etc. Where possible, existing warehousing and maintenance documentation will be reviewed

P	ROJECT INSTRU	CTION
PI- <u>3201</u> .009 REV.: 2 DATE: 5/18/83	SUBJECT: En Midla Constru	gineering Program Plan nd Independent Design and uction Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

and associated activities (e.g. system layup associated with Construction Completion Program) observed to provide additional verification that components have been properly stored and maintained during the construction process.

#### 3.2.1.3 Review of Construction/Installation Documentation

A major factor in the evaluation of the quality of construction is the review of those items constructed or installed on site. The review of documentation associated with the construction/installation process will be conducted to verify that the applicable requirements have been met (e.g. conformance to construction specifications will be verified). Included in this review will be verification of the utilization of proper documents in the process such as design output requirements, construction specifications, erection specifications, installation requirements, construction procedures and other specified construction codes and standards, as applicable. Design changes, field modifications, and other input related to final as-built drawings will be reviewed. Included will be the review of documentation associated with such items as concrete materials, concrete, the welding process, bolting activities, NDE, etc. Inspection requirements, including personnel qualification and training, reports, and associated documentation will also be included in the review. Where possible, selected on-going construction/installation activities will be observed to provide additional information for the evaluation of this process.

#### 3.2.1.4 Review of Selected Verification Activities

Verification activities conducted subsequent to the construction/installation/ inspection activity will be reviewed and evaluated. Included will be overinspection activities associated with cable separation verification, bolt hardness testing verification, the pipe support reinspection program, the Construction DC-82-13

	P	ROJECT IN	ISTRUCTIO	N
PI-3201 0 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerir Midland Indep Construction V	ng Program Plan Dendent Design and Verification Program
PAGE _98	of 145	PREPARED BY:	1114	APPROVED BY:

Completion Program; as well as routine cold hydro testing, functional and preoperational testing, other specified preservice system and component testing programs and system walkdowns associated with turn-over. Associated requirements, plans, test reports, etc. will be reviewed and, where possible, these verification activities will be observed in order to provide additional information and data to support evaluations.

#### 3.2.1.5 Verification of Physical Configuration

Field verification of the as-built configuration of selected components of a portion of the systems under the scope of the ICV will be conducted to ensure conformance with requirements. Verification will address such aspects as identification, approximate physical dimensions, location, orientation, name plate data, grounding, use of proper materials, insulation, weld quality, and other features of the configuration as applicable to the component or system. Configuration verification will range from the review of general features for some components or systems to a 100% detailed dimensional verification of other selected components or systems, as defined further in subsequent sections herein.

#### 3.2.2 BASES FOR SAMPLE SELECTION

The selection of a sample for the ICV will generally follow the criteria discussed in Section 3.1.2 of this Plan for the IDV, with the exception that certain ICV activities may utilize statistical methods. These methods may be applied in establishing sample sizes and statistical levels of confidence for the assessment of repetitive production activities such as concrete and steel properties or

		PROJECT IN	ISTRUCTION
PI REV.: 2	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE 99	of 145	PREPARED BY:	LIFT APPROVED BY: 2

welding records. The efficacy of using these approaches will be determined. The decision to utilize or not utilize statistical techniques, including the basis for this decision will be appropriately documented.

The primary means of sample selection will be engineering judgment of the ICV reviewers. As with the IDV, the initial sample will be biased towards problems that have previously arisen in the industry. This sample will be refined by incorporating specific Midland project information to verify that the ICV encompasses previous problem areas and, thereby, serve as a verification that associated problems have been or are in the process of being adequately addressed and that they do not exist elsewhere in the same or similar form.

The ultimate sample selection program that is utilized, engineering judgement and/or statistics, will be clearly defined and differentiated in the case of a combined program.

#### 3.2.3 DEFINITION OF REVIEW SCOPE FOR THE AFW SYSTEM

The ICV review categories corresponding to the major activities of the construction chain were defined in Section 3.2.1. Presented in this section is an identification of the selected components and the associated level of construction completion of each to be reviewed. For the AFW system the scope of review is defined in the matrix in Figure 1.2-3, where the "X" designates the review scope applicable to each component. The criteria discussed in Sections 1.2 and 3.2.2 of the Plan were utilized to develop this initial matrix. The review areas (or topics) of the ICV are divided into major divisions by component type: mechanical, electrical, instrumentation and control, HVAC and structural.

P	ROJECT INSTRUCTIO	N
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 100 of 145	PREPARED BY:	APPROVED BY:

Another major element of the ICV is the NDE/Materials Testing Program. This program is discussed in section 3.2.3.6. The initial scope of review of each component within these major divisions is discussed in the sections that follow. The identified initial review scope as documented in Revisions 0 and 1 of this Plan has been modified based upon meeting one or more of the criteria documented in section 3.1.2. This change has been indicated on Figure 1.2-3 by the symbol "\*" for items added to scope.

#### 3.2.3.1 Mechanical Systems and Components

An evaluation of the quality of construction of selected mechanical systems and components will be conducted. Included in the scope of this portion of the review are selected mechanical equipment, piping and pipe supports associated with the AFW system.

#### 3.2.3.1.1 Mechanical Equipment - Topic I.I-Ic

A review of the complete construction chain including verification of the physical configuration will be conducted for the major mechanical components selected for detailed review in the IDV. The fabrication documentation review will encompass major supplier documentation, including functional requirement and environmental and seismic qualification documents. Included will be the review of the stresses in equipment and supports, including anchorages, as applicable. Storage/maintenance and construction/installation documentation will be reviewed and, where possible, selected associated activities will be observed. Verification documentation associated with major preservice equipment and related system testing programs will be reviewed and where

	P	ROJECT INSTR	UCTION
PI-3201 009 REV.: 2 DA	TE: 5/18/83	SUBJECT:	Engineering Program Plan dland Independent Design and
PAGE 101 of	145	PREPARED BY:	APPROVED BY:

possible verification activities including actual tests will be observed. The physical configuration review will include verification of equipment identity, principal features, name plate data, location, orientation, and support characteristics, as applicable. Conformance with design documents (including P&ID's, isometrics and equipment location drawings), supplier documents and associated installation requirements will be verified.

## 3.2.3.1.2 Piping - Topic 1.2-lc

This activity will include the review of major piping fabrication documentation associated with the portion of the AFW piping system selected for review in the IDV. Vendor drawings, material certification, shop welding and NDE documentation, as applicable will be reviewed. Major construction/installation documentation will be reviewed • including installation specifications, welding and NDE documentation and all associated inspection reports. Verification documentation related to preservice testing programs will be reviewed and where possible associated activities will be observed. A field survey of the physical configuration of the selected portion of the AFW system will be conducted to verify routing, location (to tape measure accuracy), piping diameter, cleanliness and other major piping characteristics. Conformance with the applicable design, supplier and other installation requirements will be confirmed.

## 3.2.3.1.3 Pipe Supports - Topic 1.3-lc

A review of the quality of construction will be conducted for the pipe supports associated with the portion of the AFW piping system selected for detailed

P	ROJECT INSTRUCTION	
PI_3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE 102 of 145	PREPARED BY:	

review in the IDV. For those supports selected for review in the IDV, fabrication and installation documentation will be reviewed. Verification documentation including that associated with the pipe support reinspection program will be reviewed and where possible these activities will be observed. Verification documentation associated with major preservice system testing will also be reviewed and will be observed where possible. Physical verification will include a 100% verification of the identity, location, and orientation of all pipe supports within the selected portion of the AFW piping system. In addition, complete dimensional verification of design details will be made for those supports selected for detailed review in the IDV. Dimensional verification will encompass weld size, quality and location, base plate size and thickness, anchor bolt size and location, and other principal features, as applicable.

#### 3.2.3.2 Electrical Systems and Components

An evaluation of the quality of construction of selected electrical systems and components will be conducted. Included in the scope of this review are selected electrical equipment, cable trays and supports, conduits and supports, and electrical cable associated with the AFW system.

#### 3.2.3.2.1 Electrical Equipment - Topic II.I-Ic

A review of the complete construction chain including verification of the physical configuration will be conducted for the major electrical components (e.g. motor control center, motor operated valve, electrical panel) and cable selected for detailed review in the IDV. The fabrication documentation review will encompass major supplier documentation, including functional requirement

	P	ROJECT INSTR	UCTION
PI00	DATE: 5/18/82	SUBJECT: Mid	Engineering Program Plan Iland Independent Design and
PAGE 103	of 145	PREPARED BY:	APPROVED BY:

0

and environmental and seismic qualification documents. Included will be the review of the stresses in equipment and supports, including anchorages, as applicable. Storage/maintenance and construction/installation documentation will be reviewed and, where possible, selected associated activities will be observed. Verification documentation associated with major preservice equipment and related system testing programs will be reviewed and, where possible, verification activities including actual tests will be observed. The physical configuration review will include verification of equipment identity, principal features, name plate data, location, orientation, and support characteristics, as applicable. Conformance with design documents (including single line diagrams, P&ID's, and equipment location drawings), supplier documents and associated installation requirements will be verified.

## 3.2.3.2.2 Cable Trays and Supports - Topic II.2-Ic

This activity will include a review of major fabrication and installation documentation, selected overinspection activities and physical verification of a selected portion of a cable tray and support system associated with a major AFW electrical system. Layout and installation drawings, material cerifications, and other applicable documentation will be reviewed to verify utilization of the most current and correct design and procedural requirements. A field survey of the selected portion will be conducted to verify location (to tape measure accuracy) routing, tray characteristics, and support location and configuration. Conformance with applicable design, supplier and other installation requirements will be confirmed. Proper cable assignment to trays, tray cleanliness and tray fill will be selectively verified. This verification will be supplemented with a detailed review of selected outputs and findings resulting from the ongoing cable separation overinspection program.

PL 3201 009	ROJECT INST	RUCTION	
EV: 2 005	SUBJECT:	Engineering Program Plan	
AGE 104 04 145	Co	Aidland Independent Design and Instruction Verification Program	

# 3.2.3.2.3 Conduits and Supports - Topic II.3-Ic

This activity will include a review of major fabrication and installation documentation, selected overinspection activities, and a field verification of the physical configuration of a selected portion of a conduit and support system associated with a major Ar & electrical system. The scope of review will be similar to that of the electrical tray and support review discussed in the preceding section. The conduit size and fill will be selectively verified.

## 3.2.3.2.4 Cable - Topic II.4-Ic

A review will be conducted of major supplier documentation associated with the cable of a selected portion of a major AFW electrical system. Overinspection activities associated with cable separation verification of selected cables in AFW electrical systems will be reviewed and evaluated. The supplier documentation review will encompass cable material certifications, insulation certifications, stranding and color coding characteristics and other applicable documentation. The physical configuration of a selected portion of the system will be verified including identification, visual inspection, routing, separation, tiedown, terminations and other principal characteristics as applicable. The cable terminations will be reviewed or proper lugging and lugging tool documentation. Cable puil documentation will be reviewed to verify compliance with pull tension limits. Cable wagger as continuity checks will be reviewed to verify installed cable integrity. Conformance with applicable design, supplier and other installation requirements will be confirmed.

P 3201 000	ROJECT IN	STRUCTI	ON
REV.: 2 DATE: 5/18/83	SUBJECT:	Enginee Midland Ind	ring Program Plan dependent Design and
PAGE 105 of 145	PREPARED BY:	Likt	APPROVED BY:

# 3.2.3.3 Instrumentation and Control Systems and Components

A review of the quality of construction of selected instrumentation and control (1 & C) systems and components will be conducted. This review will include selected instruments, piping and tubing, and wiring associated with the AFW system.

## 3.2.3.3.1 Instruments - Topic III.I-Ic

A complete review of the construction chain including verification of the physical configuration will be conducted for selected instruments of a major AFW I&C system. Major documentation will be reviewed including that received from the supplier, storage/maintenance (including calibration) and installation instructions. In addition, the verification documentation associated with preservice I&C system testing programs (e.g. calibration, response time, circuit continuity, trip set poir ts, etc.) will be reviewed and activities observed where possible. The physical configuration will be verified including instrument identity, name plate data, location, mounting conditions, and other principal characteristics, as applicable. Conformance with design documents and specifications, supplier requirements and installation requirements will be verified.

## 3.2.3.3.2 Piping/Tubing - Topic III.2-Ic

This activity will include a review of major fabrication documentation and a physical configuration verification of piping and tubing associated with a selected portion of a major AFW I&C system. Material certifications and other

P	ROJECT INSTR	UCTION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Mid Cons	Engineering Program Plan Iland Independent Design and truction Verification Program
PAGE 106 of 145	PREPARED BY:	APPROVED BY

applicable documentation will be reviewed against design requirements. A field survey of the selected portion will be conducted to verify routing, supports, size, slope and valve types. Conformance with applicable design, supplier and other installation requirements will be verified. Preservice hydro test results will be reviewed.

## 3.2.3.3.3 Cable - Topic III.3-Ic

A review will be conducted of major supplier and installation documentation associated with the cable of a selected portion of a major AFW I&C system. Overinspection activities associated with cable separation verification of selected cables in the AFW instrumentation systems will be reviewed and evaluated. The supplier documentation review will encompass cable material certifications, insulation certifications, stranding and color coding characteristics and other applicable documentation. The physical configuration of the selected portion of the system will be verified including routing and terminations (correct tools for lugging, proper crimp and lug size), visual inspection tie down, and cable pull documentation review. Conformance with applicable design, supplier and other installation requirements will be confirmed. Continuity test results will be reviewed to verify circuit integrity.

## 3.2.3.4 HVAC Systems and Components

An evaluation of the quality of construction of selected HVAC systems and components will be conducted. Included in the scope of this portion of the review are selected HVAC equipment, ducts and supports associated with the AFW system.

	P	ROJECT INSTRU	JCTION
PI-3201 009		SUBJECT: E	ngineering Program Plan
REV.: 2 D.	ATE: 5/18/83	Const	and Independent Design and ruction Verification Program
PAGE 107 of	145	PREPAREL CY:	APPROVED BY:

#### 3.2.3.4.1 HVAC Equipment - Topic IV.I-Ic

A review of the complete construction chain including verification of the physical configuration will be conducted for a major HVAC component. The fabrication documentation review will encompass major supplier documentation, including functional requirement and environmental and seismic qualification documents. Included will be the review of the stresses in equipment and supports, including anchorages, as applicable. Storage/maintenance and construction/installation documentation will be reviewed and, where possible, selected associated activities will be observed. Verification documentation associated with major preservice equipment and related system testing programs will be reviewed and, where possible, verification activities including actual tests will be observed. The physical configuration review will include verification of equipment identity, principal features, name plate data, location, orientation, and support characteristics, as applicable. Conformance with design documents (including P&ID's and equipment location drawings), supplier documents and associated installation requirements will be verified.

## 3.2.3.4.2 HVAC Ducts and Supports - Topic IV.2-Ic

This activity will include a review of major fabrication documentation and the physical configuration of a selected portion of a duct and support system associated with a major AFW HVAC system. Vendor drawings, material cerifications, and other applicable documentation will be reviewed. A field survey of the selected portion will be conducted to verify (to tape measure accuracy) routing, duct characteristics, and support location and configuration. Conformance with applicable design, supplier and other installation requirements will be confirmed.

P	ROJECT INSTRUC	TION
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT: Engi Midland Construct	ineering Program Plan d Independent Design and ction Verification Program
PAGE 108 of 145	PREPARED BY:	APPROVED BY

#### 3.2.3.5 Structural Components

The quality of construction of plant structures will be evaluated based upon a review of selected structural components. Included in the scope of this portion of the review are selected foundations, concrete structural elements and structural steel components of the structures which house the AFW system.

#### 3.2.3.5.1 Foundations - Topic V.I-Ic

This activity will include the review of fabrication and construction/installation documentation associated with building foundations selected for detailed review in the IDV. The fabrication documentation review will encompass major supplier documentation including material certifications, rebar placement drawings, and other applicable documentation. Construction/installation documentation to be reviewed will include concrete materials documentation, concrete cylinder test results, inspection reports and other applicable documentation. Conformance with design documents, supplier requirements and associated construction/installation requirements will be verified.

#### 3.2.3.5.2 Concrete Components - Topic V.2-Ic

A review of fabrication and construction/installation documentation will be conducted and the as-built configuration will be verified for major concrete structural elements selected for detailed review in the IDV. The documentation review will encompass major supplier and construction/installation documentation associated with reinforcing steel, inserts and penetrations, and concrete documentation of a selected portion of each component. A field survey will be

P	ROJECT INSTRUC	TION
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engi Midland Construc	neering Program Plan I Independent Design and tion Verification Pro-
PAGE 109 of 145	PREPARED BY:	APPROVED BY:

conducted to verify overall element dimensions (including thickness), location and size of major openings and selected penetrations, and principal characteristics of relected inserts. Conformance with applicable design, supplier and other installation requirements will be confirmed.

V

## 3.2.3.5.3 Structural Steel Components - Topic V.3-Ic

This octivity will include the review of major fabrication and construction/installation documentation and a physical configuration verification of the structural steel components selected for detailed review in the IDV. The fabrication documentation review will encompass shop detail drawings, material certifications, welding documentation, and other major supplier documentation. Construction/installation documentation will address field welding, bolting (torque) and other applicable documentation. A field survey will be conducted to verify, where possible, major element characteristics including member size, plate thickness, weld size, and bolt pattern and size for a selected connection of each member. Conformance with applicable design, fabricator and other installation requirements will be confirmed.

## 3.2.3.6 NDE/Materials Testing Program - Topic VI.I-Ic

As part of the review of supplier documentation for AFW system components, a review will be conducted to ascertain and verify conformance of vendor welding, NDE, and materials testing to applicable codes, standards, and procurement specification requirements. The intent of the NDE/Material Testing Program is to supplement the review of construction/installation documentation of welding, NDE, and material testing activities by establishing a program for the

	PROJECT IN	STRUCTIO	N
PI_3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 110 of 145	PREPARED BY:	1.111-4	APPROVED BY:

performance of NDE and material testing on selected material, components, and structures of the AFW system. The program will be conducted as an integral part of the ICV and will include over-inspection and testing of selected shopfabricated/vendor-supplied components in addition to the over-inspection and testing of on-site welding, weld repair, NDE and other site-material related testing and inspection programs. Results of the testing performed as part of the NDE/Materials Testing Program will be documented, reviewed, and compared against vendor supplied and site-generated material testing and NDE test data and against applicable codes and standards.

The direction and degree of testing performed as a part of the NDE/Materials Testing Program will be initiated and influenced by the results of the construction/installation documentation review as described in sections 3.2.3.1 through 3.2.3.5. The results of the documentation review will be integrated with the consideration of a statistical sampling approach and sound engineering judgment to arrive at the quantity and types of components and structures to be tested and the type of testing to be employed.

An intermediate output of the NDE/Materials Testing Program will be a listing defining the components/structures to be tested and the corresponding test to be performed. Rationale for component/structure selection will also be provided to enable reviewers to easily discern the derivation of the sample and the sample size. The NDE/Materials Testing Program will be documented in a Project Instruction to be issued prior to initiation of the program.

P	ROJECT INS	TRUCTION
PI	SUBJECT:	Engineering Program Plan
REV.: 2 DATE: 5/18/83		Midland Independent Design and Construction Verification Program
PAGE of	PREPARED BY:	41=1 APPROVED BY:

#### 3.2.4 DEFINITION OF REVIEW SCOPE FOR THE STANDBY ELECTRIC POWER SYSTEM

The scope of ICV review for the SEP system and for the CR-HVAC system (section 3.2.5) has been structured to enable a consistent methodology across all systems, components, and structures subject to ICV review. In other words, the methodology described in the previous sections and applied to the ICV review of the AFW system will also, to the extent practical, be applied to the SEP system and CR-HVAC system.

Consistency in application of the review methodology in the area of construction verification is as important as the scope of the review itself. By providing a consistent reference against which to gauge the construction process, a clear and precise picture of trends, discrepancies, and efficiencies in the Midland construction process will be made more easily discernible. Thus, to establish the necessary common reference, the ICV review categories (section 3.2.1) and scope of review for the AFW system (section 3.2.3) will also be applicable to the review conducted of the SEP system. Modifications to the scope of review have been deemed necessary to accommodate unique component or structural considerations pertinent to the SEP system. These unique considerations are provided in the following discussions and serve to supplement and modify the scope of review for the SEP system is defined as follows and in the matrix provided in Figure 1.2-5.

	P	ROJECT IN	STRUCTIC	N
PI	DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ing Program Plan ependent Design and Verification Program
PAGE	of 145	PREPARED BY:	山山二	APPROVED BY, Rul

#### 3.2.4.1 Mechanical Systems and Components

Included in the scope of this portion of the review are the diesel generator and selected mechanical equipment, piping and pipe supports associated with the diesel generator cooling water, starting, lubrication and fuel oil systems.

#### 3.2.4.1.1 Mechanical Equipment - Topic 1.1-2c

The diesel generator and selected mechanical components of the cooling water, starting, lubrication and fuel ail systems will be subject to the review methodology as defined in section 3.2.3.1.1. Similar to the methodology applied and described for the AFW system, the selection of the components in the cooling water, starting, lubrication, and fuel and fuel as performed to ensure that components subject to IDV review will also be reviewed as part of the construction verification review.

#### 3.2.4.1.2 Piping - Topic 1.2-2c

This activity will include the review of fabrication documentation submitted for piping associated with the diesel generator cooling water, starting, lubrication, and fuel oil systems selected for review in the IDV. Vendor drawings, material certification, shop welding and NDE documentation, as applicable, will be reviewed. Major construction/installation documentation will be reviewed, including installation specifications, welding and NDE documentation and associated inspection reports. A field survey of the physical configuration of the selected piping system will be conducted to verify routing, location, piping diameter, cleanliness and other piping characteristics.

P	ROJECT IN	ISTRUCTION
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE 113 of 145	PREPARED BY	APPROVED BY

#### 3.2.4.1.3 Pipe Supports - Topic 1.3-2c

A review will be conducted for the pipe supports associated with the piping system selected for detailed review in the SEP system IDV. For those supports selected for review in the IDV, fabrication and installation documentation will be reviewed. The physical configuration of selected pipe supports will also be verified. These activities will include verification of the identity, location, and orientation of selected pipe supports in addition to verification of design details and weld size, base plate size and thickness, anchor bolt size and location and other principal design features.

#### 3.2.4.2 Electrical Systems and Components

The sample selection boundary for the SEP system is provided in section 3.1.4 and defines the electrical systems subject to the IDV and ICV review process. Within the boundaries are power distribution systems and control systems of various classifications (IE and non IE) and voltages. The ICV electrical review will address the quality of construction of selected electrical and control systems which comprise the SEP system and, within the systems, selected electrical equipment, cable trays and supports, conduits and supports, and electrical cable.

#### 3.2.4.2.1 Electrical Equipment - Topic II.I-2c

A review of the complete construction process, as defined and described in section 3.2.3.2.1 for AFW system electrical components, will be conducted for selected components and cable of the SEP system. To ensure a consistent application of the review methodology, the components selected for detailed

PL 3201 009	ROJECT INSTRUCTION
REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent D
PAGE 114 of 145	PREPARED BY

review as part of the IDV will also comprise the nucleus of items subject to ICV review.

# 3.2.4.2.2 Cable Trays/Conduits and Supports - Topics 11.2-2c & 11.3-2c

The scope of activities described and defined in Sections 3.2.3.2.2 and 3.2.3.2.3 will be performed for a selected portion of the cable trays/conduits and supports utilized to route power distribution and control cables in the systems within the SEP system sample selection boundaries.

## 3.2.4.2.3 Cable - Topic II.4-2c

The scope of activities described and defined in Section 3.2.3.2.4 will be performed for a selected portion of power distribution cables within the SEP system sample selection boundaries.

# 3.2.4.3 Instrumentation and Control Systems and Components

Instrumentation and control (I&C) systems associated with the Standby Electric Power System are segregated into two distinct categories. The first category of I&C systems are comprised of controls, instruments, and alarms associated with the diesel generator cooling water, starting, lubrication and fuel oil systems. The second category of I&C systems relates to those instruments and controls which monitor and control the loading and distribution of power within the electrical sample selection boundaries of the SEP system (Section 3.1.4). The classification of instruments into two categories does not alter or affect the ICV review methodology. Rather, the distinction is made to highlight the fact that

P	ROJECT INSTRUC	TION
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT: Engir Midland Construct	neering Program Plan Independent Design and tion Verification Program
PAGE 115 of 145	PREPARED BY:	APPROVED BY

fluid process systems reside within the sample boundaries of the SEP system and as such the instruments and controls associated with the process systems are subject to the same review and selection as instruments and controls which monitor and control the loading and distribution of standby electric power.

#### 3.2.4.3.1 Instruments - Topic III.1-2c

Selected instruments from each of the previously defined categories of instruments will be subject to a detailed ICV review. The selection of instruments will be keyed to the IDV sample to enable a consistent and continuous evaluation of instruments utilized within the sample selection boundaries of the SEP system from design, through and including construction processes. The scope of ICV review activities to be undertaken for SEP system instruments are as indicated on Figure 1.2-5 and described in section 3.2.3.3.1.

#### 3.2.4.3.2 Piping/Tubing - Topic III.2-2c

Category IE instruments, and other instruments which monitor, control and alarm parameters in the diesel generator fluid process systems, comprise the sample boundaries for the selection of piping/tubing to be subject to ICV review. The scope of activities described and defined in section 3.2.3.3.2 will be performed for a selected portion of the piping and tubing associated with instruments in the SEP system, specifically including piping and tubing utilized in the diesel generator cooling water, starting, lubrication and/or fuel oil systems. The activities defined in section 3.2.3.2 will be supplemented to verify utilization of the most current and correct design, construction, and installation documentation and to include design changes and field modifications.

Р	ROJECT IN	STRUCTIC	N
PI	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 116 of 145	PREPARED BY:	Hich :	APPROVED BY:

#### 3.2.4.3.3 Cable - Topic 111.3-2c

The scope of activities described and defined in section 3.2.3.3.3 will be performed for a selected portion of the instrumentation cabling and wiring utilized within the sample selection boundaries of the SEP system. Additionally, a review will be conducted to ascertain and verify that the cabling utilized was properly stored and, once installed, was adequately and correctly handled and maintained.

#### 3.2.4.4 HVAC Systems and Components

An evaluation of the quality of construction of selected diesel generator building HVAC systems and components will be conducted. Selected fans, coolers, dampers and HVAC ducts and supports located in, and associated with, the diesel generator building will be subject to ICV review as described and defined as follows.

#### 3.2.4.4.1 HVAC Equipment - Topic, IV.1-2c

The documentation associated with the manufacture of selected fans, coolers, and dampers located in the diesel generator building will be reviewed. The review will seek to verify that correct and adequate design and documentation requirements are transmitted to vendors for the diesel generator HVAC equipment and to ascertain and verify that the vendor's documentation submittals are complete, correct, and responsive to the procurement documentation requirements.

	PROJECT INSTRU	ICTION
PI-3201 009 REV.: 2 DATE: 5/11	SUBJECT: Er Midle S/83 Constr	ngineering Program Plan and Independent Design and ruction Verification Program
PAGE of	PREPARED BY:	APPROVED BY:

Once having characterized and identified selected HVAC equipment by means of the above supplier documentation review, the physical configuration of the installed equipment within the diesel generator building will be verified. The asbuilt configuration of selected HVAC equipment will be checked and verified against vendor-supplied documentation and against installation requirements as derived from applicable codes, industry standards, and pertinent installation/erection documentation.

#### 3.2.4.4.2 HVAC Ducts and Supports - Tupic IV.2-2c

The ICV review activities to be undertaken to verify the quality of construction for selected HVAC ducts and supports in the diesel generator building are defined and described in section 3.2.3.4.2 and are similar to the scope of review activities to be conducted for HVAC ducts and supports associated with the AFW system.

#### 3.2.4.5 Structural Components

The quality of construction of the diesel generator building and foundations will be evaluated based upon a review of selected structural components. Selected foundations, concrete, and structural steel used in the erection of the diesel generator building, and major component foundations within the diesel generator building, will be reviewed.

#### 3.2.4.5.1 Foundations - Topic V.1-2c

This activity will include the review of fabrication and construction/installation documentation associated with the diesel generator building and major

	P	ROJECT INSTRUC	TION
PI320100	DATE: 5/18/83	SUBJECT: Engi Midland Construct	ineering Program Plan d Independent Design and ction Verification Program
PAGE	of _145_	PREPARED BY: LIAL	APPROVED BY:

equipment foundations selected for detailed review in the IDV. The description and definition of activities to be undertaken are provided in Section 3.2.3.5.1.

3.2.4.5.2 Concrete Components - Topic V.2-2c

A review of fabrication and construction/installation documentation will be conducted for major concrete structural elements of the diesel generator building selected for detailed review in the IDV. The documentation review will encompass major supplier and construction/installation documentation associated with reinforcing steel, inserts and penetrations, and concrete documentation of a selected portion of each structure subject to ICV review.

3.2.4.5.3 Structural Steel Components - Topic V.3-2c

The scope of review to be applied to the structural steel components of the diesel generator building will be similar to the scope of activities to be undertaken in verifying diesel generator building concrete components as defined in the previous section. Emphasis will be placed upon the review of shop detail drawings, material certifications, shop and field welding documentation, bolting and other applicable vendor-supplied and site-generated documentation.

3.2.5 DEFINITION OF REVIEW SCOPE FOR THE CONTROL ROOM HVAC SYSTEM

To establish and maintain a common reference against which to gauge the quality of construction, the scope of ICV review for the CR-HVAC system has been developed to be similar to the scope of ICV review previously defined for

P	ROJECT INSTRUCTION
REV: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and
PAGE 119 of 145	PREPARED BY: APPROVED BY:

the AFW and SEP systems. Additionally, the ICV review methodology defined and described in previous sections of this program plan (Sections 3.2.1 through 3.2.3) will be retained and applied to components and structures of the CR-HVAC system. Modifications to the scope of review will be made as necessary to accommodate unique component or structural considerations of the CR-HVAC system. These unique considerations are provided in the following discussions and serve to supplement and modify the scope of activities previously described in Sections 3.2.3.1 through 3.2.3.6. The scope of review for the CR-HVAC system is defined as follows and in the matrix provided in Figure 1.2-7.

## 3.2.5.1 Mechanical Systems and Components

Included in the scope of this portion of the review are selected mechanical equipment, piping, and pipe supports included within the sample selection boundaries of the CR-HVAC system (see section 3.1.5). Selected components such as fans, coolers, dampers, filters, cooling water piping and pipe supports comprise the items to be subject to the ICV review of CR-HVAC mechanical systems and components.

## 3.2.5.1.1 Mechanical Equipment - Topic 1.1-3c

A review of the complete construction chain including verification of the physical configuration will be conducted for fans, coolers, dampers, and filters selected for detailed review in the IDV. The selected CR-HVAC system mechanical equipment will be subject to the review methodology and scope of review activities as previously defined in Section 3.2.3.1.1.

Р	ROJECT IN	STRUCTIC	N
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 120 of 145	PREPARED BY:	ビーシー	APPROVED BY:

#### 3.2.5.1.2 Piping - Topic 1.2-3c

Selected piping, exclusive of ductwork, within the CR-HVAC system will be subject to a review of fabrication and installation documentation and a verification of the physical configuration. Vendor drawings, material certification, shop welding and NDE documentation, as applicable, will be reviewed. Major construction/installation documentation will be reviewed including installation specifications, welding and NDE documentation and associated inspection reports. A field survey of the physical configuration of the selected piping system will be conducted to verify routing, location, piping diameter, cleanliness, and other piping characteristics.

#### 3.2.5.1.3 Pipe Supports - Topic 1.3-3c

A review will be conducted of the pipe supports associated with the piping system selected for review under Topic 1.2-3c. For those supports selected fabrication and installation documentation will be reviewed and the physical configuration will be verified. The physical configuration verification of selected pipe supports will include verification of design details, weld, size, base plate size and thickness, anchor bolt size and location, and other principal design features.

#### 3.2.5.2 Electrical Systems and Components

A detailed review to verify the quality of construction will be conducted for selected electrical equipment, trays/conduits and supports and cables within the sample selection boundaries of the CR-HVAC system. In addition to reviews of

	P	ROJECT IN	ISTRUCTIC	DN .
PI0 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineer Midland Inde Construction	ing Program Plan ependent Design and Verification Program
PAGE	of _145	PREPARED BY:	List -	APPROVED BY:

vendor supplied documentation and documentation utilizied to record the implementation of installation activities, a verification of the physical configuration will be conducted for selected trays, conduits, equipment and supports.

#### 3.2.5.2.1 Electrical Equipment - Topic II.I-3c

Electrical equipment, such as fan motors, switchgear, and motor control centers, selected as part of the IDV review will be subject to a detailed ICV review. The review will comprise a verification of fabrication- and installation-related documentation and a verification of the physical configuration of selected electrical equipment. Activities will include review of vendor-supplied drawings, calculations, test reports, certified material property reports and other major supplier documentation. Documentation utilizied to record the performance of construction installation, orientation, name plate data and grounding of the installed and as-built electrical equipment selected for review. Preoperational and other specified preservice component and system testing and test programs will be reviewed. This review will be conducted to verify that associated test requirements, plans, reports, etc. provide adequate assurance of electrical component operation in accordance with system design requirements.

## 3.2.5.2.2 Trays/Conduit and Supports - Topics 11.2-3c and 11.3-3c

This activity will include a review of major fabrication and installation documentation of cable trays/conduit and supports associated with the electrical

P	ROJECT INSTRUCTION
PI_3201_009 REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program
PAGE 122 of 145	PREPARED BY:

system selected for IDV review. Layout and installation drawings, material certifications, and other applicable documentation will be reviewed. A field survey will also be conducted to verify location, routing and routing characteristics, and support location and configuration. Proper cable assignment to trays and conduit will be selectively verified in addition to cleanliness and cable fill.

#### 3.2.5.2.3 Cable - Topic II.4-3c

An ICV review will be conducted of major fabrication and installation documentation associated with the CR-HVAC electrical cable selected as part of the IDV review. Overinspection activities associated with cable separation verification of selected cables in the CR-HVAC electrical system will be reviewed, evaluated and factored into the ICV review. The physical configuration of a selected portion of the cable network will be verified including identification, visual inspection, routing, tiedown, terminations, and other principal characteristics as appropriate and applicable. Conformance with applicable design, supplier, and other installation requirements will be confirmed.

#### 3.2.5.3 Instrumentation and Control Systems and Components

Selected instrumentation and controls essential for isolation of the control room and operation of the CR-HVAC system will be subject to the ICV review activities described in the following sections. Similar to the review of the AFW and SEP systems, this review will include selected instrument, piping and tubing, and wiring associated with the CR-HVAC system.

	P	ROJECT INSTRUCTION	DN
PI		SUBJECT: Engineering Program Plan	
REV.: 2	DATE: 5/18/83	Midland Independent Design and Construction Verification Program	
PAGE 123	of 145	PREPARED BY:	APPROVED BY:

#### 3.2.5.3.1 Instruments - Topic III.1-3c

A complete review of the construction chain including verification of the physical configuration will be conducted for selected instruments essential for isolation and habitability of the control room. The scope of activities which comprise this review will be as defined and described in Section 3.2.3.2.4.

#### 3.2.5.3.2 Piping and Tubing - Topic III.2-3c

The ICV review to be conducted for selected piping and tubing within the Control Room HVAC instrumentation and control systems will include those activities as described and defined in Section 3.2.3.3.2. The scope of review activities enumerated in the aforementioned section will be supplemented with a review of documentation utilized to record, document, and control the installation of the instrumentation- and control-related piping and tubing. Included in this review will be verification of the utilization of proper documents in the installation process such as design output requirements, erection specifications, installation requirements, and other specified construction codes and standards as applicable.

#### 3.2.5.3.3 Cable - Topic III.3-3c

The activities to be undertaken to verify the quality of construction of instrumentation cables will be as described and defined in Section 3.2.5.2.3 for the CR-HVAC electrical system cables.
	F	ROJECT INS	STRUCTIO	N
PI	DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde	ng Program Plan pendent Design and Verification Pro
PAGE	of 145	PREPARED BY:	think in	APPROVED BY:

#### 3.2.5.4 HVAC

Included in the scope of this portion of the ICV review of the Control Room HVAC system are selected ducts and supports which channel air to, from, and within the control room under normal and accident conditions.

#### 3.2.5.4.1 HVAC Ducts and Supports - Topic IV.2-3c

This activity will include a review of major fabrication and installation documentation and a physical configuration verification of a selected portion of the ducts and supports of the CR-HVAC System. The scope of activities to be undertaken as part of this review are as described and defined in Section 3.2.3.4.2 for the AFW system. The scope of review activities enumerated in the aforementioned section will be supplemented with a review of storage and maintenance documentation and documentation utilized to record and control the installation of selected ducts and supports. These supplemental reviews will be performed to verify that requirements related to storage, including both instorage and in-place maintenance have been, and are being, met and that design and procedural requirements controlling the installation and erection of the ducts and supports were correctly implemented and utilized.

#### 3.2.5.5 Structural Components

The focus of ICV review activities will be upon those portions of the control room structure which form the control room pressure boundary including penetrations and doors. Also included in the scope of this portion of the review are selected concrete structural elements and structural steel components of the structures which form the control room pressure boundary.

	P	ROJECT INS	TRUCTION
PI	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Midland Independent Design and construction Verification Program
PAGE	of _145_	PREPARED BY:	APPROVED BY:

### 3.2.5.5.1 Concrete Components - Topic V.2-3c

The scope of activities described and defined in Section 3.2.3.5.2 will be performed for a selected portion of the concrete components which comprise the control room pressure boundary.

# 3.2.5.5.2 Structural Steel Components - Topic V.3-3c

The cope of activities described and defined in Section 3.2.3.5.3 will be performed for a selected portion of the structural steel components, including penetrations and doors, which comprise the control room pressure boundary.

# 3.2.5.6 NDE/Material Testing Program - Topic VI.I-3c

The methodology and scope of activities to be undertaken in executing the methodology for the NDE/Material Testing Program is as described and defined in Section 3.2.3.6. The methodology and activities will be applied to selected components, structures, and duct work within the sample selection boundaries of the CR-HVAC system. Similarly, the NDE/Materials Testing Program for the CR-HVAC system will be documented in a Project Instruction which will be issued prior to initiation of the program.

# 3.2.6 DEVELOPMENT OF ICV PROGRAM CHECKLISTS

Generic checklists exist for each of the review scope categories discussed in section 3.2.1, utilizing guidance contained in applicable ANSI documents, the construction review program guidelines published by INPO and other industry

PROJECT INSTRUCTION				
PI00 REV.: 2	DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program		
PAGE	of _145_	PREPARED BY:		

standards. Similarly to the IDV, these checklists have been developed to assist in documenting engineering evaluations, and providing uniformity of review from topic to topic and between reviewers. For each of the construction review scope areas within the scope shown in Figures 1.2-3, 1.2-5 and 1.2-7, the reviewer may utilize the generic checklists or develop a specific checklist incorporating portions of the generic checklists as appropriate. In most cases, the specific checklist will be derived from the generic checklist by the addition of specific requirements applicable to the construction area being reviewed. In some cases, it may be appropriate to use only a portion of the generic checklist or to develop a unique checklist. The specific instructions for the use of construction verification checklists are contained in Project Instruction, PI-320I-007, Use of Construction Verification Checklists.

In each case that a generic checklist has been modified, the checklist prepared by the reviewer will be checked by the Lead Technical Reviewer for the area. (Note that if the LTR prepares a checklist, it is permissible for him to both originate and check the contents of the checklist). During their review process, the LTRs examine the checklist for interfaces with other ICV areas and perform a general review of the completeness and adequacy of the proposed checklist. The LTR's review is to be coordinated with the Project Manager as necessary to resolve questions which cut across discipline lines. In the event that the Project Manager or the LTR has comments on the checklist, the checklist preparer and those having comments will discuss the comments and reach an appropriate resolution. After reaching concurrence in the adequacy of the checklist, the LTR will indicate his approval and the checklist will be available for use by the reviewer.

PROJECT INSTRUCTION					
PI-3201 - 0 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineering Program Plan Nidland Independent Design and Instruction Verification Program		
PAGE	of 145	PREPARED BY:	APPROVED BY:		

The reviewer, having an approved checklist, can then proceed with the review process for this specified area, in accordance with Project Instruction PI-3201-001, Engineering Evaluation Preparation and Control. In performing the evaluation, the reviewer will document the information which he used in order to complete the checklist. Such information will include component identification, the date or revision number of the associated documents, the document number, and an indication of the source of the information (i.e., where data and any associated documents were obtained).

#### 3.2.6.1 Development of Checklists for Review of Supplier Documentation

The generic checklist for review of supplier documentation was developed considering questions such as:

- What is the identity of the supplier documentation being reviewed (including P.O. number, supplier name, component name and identification number)?
- Has the documentation been reviewed and accepted by the appropriate organization?
- Is the documentation complete?
- Does the documentation comply with purchase specification requirements?
- Where appropriate, does seismic and environmental qualification documentation comply with purchase specification requirements?
- Have the necessary shipping, handling, storage, installation, and maintenance requirements been specified by the supplier and are these consistent with purchase specification requirements?

	P	ROJECT IN	ISTRUCTIO	N
PI_3201_009 REV.: 2 D	ATE: 5/18/83	SUBJECT:	Engineerin Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 128 of	145	PREPARED BY:	11+1-	APPRUVED BY:

For each type of system, component or structural element the lead technical reviewer will supplement the generic checklist with appropriate additional questions, as applicable.

#### 3.2.6.2 Development of Checklists for Review of Storage and Maintenance Documentation

The generic checklist for review of storage and maintenance documentation was developed considering questions such as:

- What is the identity of the storage and maintenance documentation being reviewed, including document type (receipt inspection, in-storage/in-place maintenance records, etc.) and document identification (document title, revision, date)?
- What is the identity of the component being reviewed (name, identification number)?
- Does the documentation for the receiving process include component review against purchase specification requirements?
- Are nonconforming items properly identified, processed and closed out?
- Does the maintenance program meet the necessary requirements specified for the component relative to humidity, cleanliness, lubrication, shaft rotation, energization, etc., as applicable?

For each type of system, component or structural element the lead technical reviewer will supplement the generic checklist with appropriate additional questions, as applicable.

	P	ROJECT INSTRU	CTION
PI00 REV.: 2	DATE: 5/18/83	SUBJECT: En Midla Constr	gineering Program Plan nd Independent Design and uction Verification Program
PAGE	of145_	PREPARED BY:	APPROVED BY

#### 3.2.6.3 Development of Checklists for Review of Construction and Installation Documentation

The generic checklist for review of construction and installation documentation was developed considering questions such as:

- What is the identity of the construction/installation documentation being reviewed, including type (concrete, welding, bolting, NDE, etc.) and identification (title, revision, date)?
- What is the identity of the system, component or element and its physical location in the plant?
- Are all appropriate construction/installation procedures and instructions identified?
- Are the current revisions of drawings, specifications and other requirements utilized in the work including those specified in Field Change Requests?
- Does the documentation include verification that the work has been performed by properly qualified personnel?
- For those activities observed, do the construction/installation activities conform to requirements?
- Have the necessary inspections been performed?
- Has the work been performed utilizing the proper tools/equipment? Have such tools/equipment been properly calibrated in accordance with procedures?
- Have rework activities including Field Change Requests been performed in accordance with requirements and appropriately closed-out?
- Have deviations from design/supplier requirements been properly documented, processed and closed out?

DC-82-13

PROJECT INSTRUCTION				
PI00	19	SUBJECT:	Engineerin Midland Inde	ng Program Plan
REV.: 2	DATE: 5/18/83		Construction V	/erification Program
PAGE 130	of 145	PREPARED BY:	11+2-	APPROVED BY

For each type of system, component or structural element the lead technical reviewer will supplement the generic checklist with appropriate additional questions, as applicable.

## 3.2.6.4 Development of Checklists for Review of Selected Verification Activities

The generic checklist for review of selected verification activities was developed considering questions such as:

- What is the identity of the verification activity being reviewed (cable separation verification, pipe support reinspection, bolting study, pre-service test, including type, etc.)?
- What is the identity of the system, component or element(s) included in the verification activity under review?
- What is the identity of the verification activity documentation being reviewed (program plan, procedures, instructions, etc.)?
- What is the quality-related objective of the verification activity and does the activity as specified/documented meet the objective?
- Where verification activities are observed, do the activities comply with requirements and are they properly documented?
- Are nonconformances properly identified, processed and closed out?

For each type of system, component or structural element the lead technical reviewer will supplement the generic checklist with appropriate additional questions, as applicable.

DC-82-13

	P	ROJECT INSTRUCTION	
PI- 3201 _ 00 REV.: 2	DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and Construction Verification Program	
PAGE	of 145	PREPARED BY:	

#### 3.2.6.5 Development of Checklists for Review of Verification of Physical Configuration

The generic checklist for review of verification of physical configuration was developed considering questions such as:

- What is the identity of the system, component or structural element being reviewed (name, identification number, location in plant, reference design documents)?
- Has the system, component or element been properly tagged/marked for identification in accordance with requirements?
- On the basis of visual inspection, has the component been properly constructed/installed and has it been maintained and protected during the construction process in accordance with requirements?
- Does the configuration comply with design requirements, including physical dimensions, location, orientation, name plate data, grounding, use of proper materials, insulation, routing, etc., as applicable?
- Have deviations from design requirements been properly identified, processed and closed out?

For each type of system, component or structural element the lead technical reviewer will supplement the generic checklist with appropriate additional questions, as applicable.

PROJECT INSTRUCTION				
PI320100 REV.: 2	DATE: 5/18/83	SUBJECT:	En Mic Cons	ng Program Plan ependent Design and Verification Program
PAGE	of	PREPARED BY	114	APPROVED BY:

# 3.2.7 PLAN FOR ADDITIONAL SAMPLING, VERIFICATION, AND TESTING

The initial sampling and verification within the scope of the ICV is primarily based upon an evaluation of project generated documentation and data to verify the quality of both inaccessible (e.g. rebar placement) and accessible systems, components and structures. The only exception to this is the incorporation of the NDE/Materials Testing Program (reference sections 3.2.3.6 and 3.2.5.6) into the ICV scope. Under this program, data are collected by TERA as a confirmatory check. The quality of accessible items will be further verified by visual inspection or measurement as appropriate.

Additional sampling or verification within the scope of the ICV systems or outside the scope into other systems will be conducted if discrepancies are found. As indicated in section 3.1.7, additional sampling or verification within the scope of the system cample selection boundaries identified in sections 3.1.3, 3.1.4 and 3.1.5 of this Plan will be undertaken by TERA, approved internally and documented in the Monthly Status Report. Additional sampling outside this scope is considered a substantive issue and will be discussed between TERA, CPC and NRC prior to initiation. The level of additional sampling or verification will be based upon the nature of the discrepancy. In all cases when discrepancies are found, an introspective evaluation will follow to identify the extent and root cause. The root cause may either be random or systematic (generic). The additional review will attempt to verify whether the discrepancy is restricted to the specific system, component, or structure under review; restricted to work by a specific construction organization; or if the discrepancy cuts across many interfaces and applies to similarly constructed systems, components, and structures.

DC-82-13

	P	ROJECT INSTRU	JCTION
PI 00 REV.: 2	DATE: 5/18/83	SUBJECT: E Midl Const	ngineering Program Plan and Independent Design and ruction Verification Program
PAGE 133	of 145	PREPARED BY:	APPROVED BY

At first, the additional sampling and verification will be directed at an evaluation of additional documentation; however, if this documentation is incomplete or insufficient to identify the extent and root cause of discrepancies, further inspection or testing will be considered, as appropriate. If required to supplement internal resources, TERA may consider subcontracting a portion of any required inspection or testing services (e.g. non-destructive examination, materials testing, etc.) to a qualified organization that meets the independence requirements of Section 1.4 of this Plan.

P	ROJECT INSTRUCT	ION
PI- <u>3201</u> 009 REV.: 2 DATE: 5/18/83	SUBJECT: Engine Midland I Constructi	eering Program Plan Independent Design and on Verification Program
PAGE 134 of 145	PREPARED BY: LIACH	APPROVED BY:

#### 4.0 DOCUMENTATION

Auditable records must be maintained to document substantive elements of the IDCV review and evaluation process, to document technical conclusions including the status of disposition of items associated with the review process leading to Findings, to document the revision of records, and to establish quality assurance measures necessary to provide adequate confidence and assurance of the quality of services. The following sections establish documentation requirements for engineering evaluations, calculations, field verification, and external communications. Section 5.0 of this Plan establishes the requirements for reporting documentation. Section 6.0 of this Plan establishes the QA documentation requirements.

# 4.1 DOCUMENTATION OF ENGINEERING EVALUATIONS, CALCULATIONS, AND FIELD VERIFICATION RESULTS

Engineering evaluations, calculations, and field verification results provide the bases for all substantive conclusions reached in the IDCV. These items provide the "trail" of information which supports IDCV conclusions, both positive and negative, as the case may be. While the reporting mechanism established in Section 5.0 of this Plan addresses the documentation of reporting requirements which are generally applicable to negative conclusions, it is equally vital that positive conclusions are documented in an auditable form as well.

The requirements for preparation and control of engineering evaluation documentation required for the Midland IDCV are contained in Project Instruction

PI_ 3201 _ 009	ROJECTINSTRUCTION
REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent
PAGE 135 of 145	Construction Verification Program

PI-3201-001, Engineering Evaluation Preparation and Control. Engineering evaluations are required for tasks such as design criteria evaluation, commitment compliance evaluation, design evaluation, construction records evaluation,

The requirements for preparation and control of calculation documentation, including computer analyses documentation, required for the Midland IDCV are contained in Engineering Control Procedure ECP-5.2, Calculation Preparation and Control. Calculations are prepared as required to verify designs, design parameters, design criteria, performance parameters, evaluare data, and otherwise provide quantitative information in accordance with accepted analytical and mathematical methods. Calculations are intended to assist IDCV reviewers in reaching necessary conclusions relative to the quality of the Midland plant

# 4.2 DOCUMENTATION

COMMUNICATIONS

PROTOCOL

FOR EXTERNAL

The requirements for the preparation and control of documentation for external communications including the protocol for communications are contained in Project Instruction PI-3201-010, External Communications, Protocol and the Preparation of Contact Log Sheets. Under prescribed circumstances, oral communications and meetings that include discussions with parties external to the IDCV review organization must be documented to provide an auditable record of information which may have an impact on IDCV conclusions and the preservation of an independent process in reaching these conclusions. Accordingly, all oral communications, meetings and exchanges of written

AND

DC-82-13

PI_ 3201 009	ROJECT INSTRUCTION
REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and
PAGE 136 of 145	PREPARED BY:

documents with parties external to the IDCV review organization that include discussion of any subjects material to the scope of the Midland IDCV Program, Status reporting, Findings and Findings resolution, including recommendations, evaluations, correspondence, interim and final reporting are documented and controlled consistent with the provisions of PI-3201-010.

The protocol governing communications between CPC and TERA is in accordance with the provisions documented in a letter from James G. Ke pler, Administrator, NRC Region III to James W. Cook, Vice President, CPC, dated

# 5.0 PROGRAM REPORTING

#### TYPES OF REPORTS 5.1

The following types of reports will be prepared in the IDCV:

- Open, Confirmed, and Resolved (OCR) Item Reports
- Observations
- Finding Reports
- Finding Resolution Reports
- Draft and Final Reports
- Interim Technical Reports
- Monthly Status Reports

OCR reports document the disposition of the IDCV review process leading to either Findings or the resolution of items which have surfaced during the review, but have been resolved after considering additional information. Observations

DC-82-13

PI_ 3201 0	09	PROJECT INSTRUCTION
REV.: 2	DATE: 5/18/83	SUBJECT: Engineering Program Plan
PAGE 137	of 145	PREPARED BY

are deficiencies that are not sufficiently serious to warrant classification as OCRs or Findings, yet cannot be dismissed directly as Resolved Items, but should be reviewed and corrected by CPC during the completion of the Midland project.

Finding Reports document verified deviations in the implementation of design criteria, design, or construction commitments and design or construction procedures in areas such as: quality assurance, design or construction control, analysis, design, engineering evaluation, specification, design or construction implementation or field installation. Findings may fall into two categories: those affecting the ability of systems, components, or structures to meet their intended safety function and those without an impact on safety functions.

Finding Resolution Reports document the conclusions of the review process which has been undertaken to resolve Findings and completely close out any concern about the Findings. Finding resolution may require additional analysis, design, or construction changes or procedural changes. Full resolution requires the identification of root cause and extent and a plan for corrective action if required.

The IDCV Final Report documents all substantive conclusions reached in the IDCV, including the process leading to these conclusions. Both positive and negative conclusions will be identified to provide a balanced perspective and to document a complete record. While the overall IDCV objective is to verify the quality of the Midland project design and construction efforts identifying any deficiencies, it is necessary to have a record which documents items that have been dismissed (i.e., positive conclusions) because the bases for these conclusions

PI_ 3201 009	ROJECT INSTRUCTION
REV.: 2 DATE: 5/18/83	SUBJECT: Engineering Program Plan Midland Independent Design and
PAGE 138 of 145	PREPARED BY:

Interim Technical Reports may be written from time to time, as appropriate to document details of review and conclusions reached on specific technical issues (e.g. seismic design) or potentially to document the conclusion of the review for any of the three systems within the IDCV scope. Monthly Status Reports provide outside parties up-to-date information relative to program progress and important issues identified during the reporting period. The following items are included:

- Tracking System Summary for OCRs, Finding Reports and Finding Resolution Reports
- Current Confirmed Item Reports .
- IDCV Program Status Summary
- Financial Report (CPC only)
- 5.2 REPORTING PROCESS
- 5.2.1 REPORTING SYS EM

The system for IDCV reporting is shown graphically in Figure 5.2-1. This figure provides a diagram or flow chart of the report generation process and a summary of the sequence.

Upon initial technical review, Potential Open Items may be identified by an IDCV reviewer. This determination will be based upon his judgment that a potential deviation exists in implementation of design criteria, design or construction commitments, and design or construction procedures, thus requiring additional investigation or confirmatory analysis by the IDCV review team.

DC-82-13



VERIFICATION PROGRAM

M

F	ROJECT INSTRU	CTION
PI	SUBJECT: En Midla Constru	gineering Program Plan nd Independent Design and uction Verification Program
PAGE 140 of 145	PREPARED BY:	APPROVED BY:

Upon documenting his determination, the IDCV reviewer forwards a preliminary OCR Report or Potential Open Item Report to his Lead Technical Reviewer (LTR) who reviews it with the project team (Project Manager and all LTRs). If the project team concurs with the reviewer's determination, the Potential Open Item becomes an Open Item which is formally controlled. The project team may resolve the Potential Open Item, thus requiring reclassification of the item as a Resolved Item and modification of the OCR Report reflecting this change. Both the Resolved Item Report and superceded Potential Open Item Report are then formally controlled. Alternatively, the LTR or project team may judge the identified deficiency as not being sufficiently serious to warrant classification as an OCR, yet it must be documented and corrected by CPC during the completion of the Midland Project. Accordingly, the item will be classified as an Observation and processed in accordance Project Instruction PI-320I-005, Documentation of Observations.

Open Items will be reviewed further by the review team until such a point that available information has been depleted. At this time, the IDCV reviewer will prepare a Resolved Item Report or a Confirmed Item Report which documents his determination after further review. A Confirmed Item is judged to be an apparent finding by the review team and requires further action to provide documentation that may not have been available to the IDCV review team. His recommendation is forwarded to his LTR who reviews the classification and makes a recommendation to the project team. The project team may agree with the LTR's recommendation at which point the Resolved Item Report or Confirmed Item Report becomes final. Alternatively, the project team may review the classification and require further work by the IDCV reviewers. All final OCR Reports are forwarded to the Principal-in-Charge (PIC) for his

P	ROJECT INS	TRUCTIC	DN
PI-3201 009 REV.: 2 DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde	ing Program Plan pendent Design and Verification Program
PAGE 141 of 145	PREPARED BY:	ホー	APPROVED BY:

concurrence, disposition, and determination whether a formal review is required by the Senior Review Team (SRT). In all cases, the SRT receives a copy of the OCR Report irrespective of whether they are requested to undertake a formal review and at the descretion of the Chairman of the SRT, the SRT may independently undertake a review.

The PIC may agree with the project team's classification and recommend that the Project Manager forward Confirmed Item Reports to CPC, NRC and other outside parties on the IDCV program service list as part of the monthly status report, with copies to the appropriate design or construction organizations; or he may request a review by the SRT to assist him in making his determination. Alternatively, or in parallel, he may request that the project team or review team conduct further review.

Additional information will be solicited from CPC/original design or construction organization at which point the LTRs and IDCV reviewers will then review any information received and make a determination whether the Confirmed Item becomes a Resolved Item or a Finding. The LTRs will make the recommendation to the project team who will review the classification. The project team may agree with the LTR's recommendation, at which point the Resolved Item Report or Finding Report becomes final. Alternatively, the project team may review the classification and require further work by the IDCV reviewers. Upon completion of this process, the OCR Report or Finding Report is forwarded to the PIC by the Project Manager for a similar review process as has been previously described. After his review and any required review by the SRT, the PIC will direct the Project Manager to forward Finding Reports to CPC, NRC and outside parties on the IDCV program service list as part of the Monthly Status Report.

DC-82-13

	P	ROJECT IN	STRUCTIC	N
PI00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ing Program Plan ependent Design and Verification Program
PAGE _142	of _145_	PREPARED BY:	山山	APPROVED BY:

CPC/original design or construction organization will respond with an action plan for resolution of the issues identified. The project team will review the response and determine whether the issue has been resolved. If so, a Finding Resolution Report will be issued by the project team for review by the PIC in c similar fashion as has been previously described. Alternatively, the Finding may not be resolved, at which point it will remain open and be documented in the Final Report. It must be noted that this eventuality is not anticipated since closure must be sought by the involved organizations. Finding Resolution Reports will also be forwarded as part of the Monthly Status Reports. The Final Report will document all IDCV conclusions as discussed previously.

# 5.2.2 REPORT PREPARATION AND DISTRIBUTION

The preparation and control of OCR Reports, Finding Reports, and Finding Resolution Reports is addressed in Project Instruction PI-3201-008, Preparation and Control of Open, Confirmed, and Resolved Item Reports, Finding Reports, and Finding Resolution Reports. Section 3.0 of PI-3201-008 provides instructions for report preparation, Section 4.0 identifies the review and approval chain and Section 6.0 addresses the distribution of these reports.

The Final Report will include documentation of all conclusions, including references to applicable documents that support these conclusions. A draft Final Report will be transmitted to CPC and NRC for their review. Resolution of their comments will be documented in an auditable manner. A copy of the draft Final Report will be sent to outside parties on the IDCV service list. It should be noted that CPC and NRC comments are intended to be of a clarification nature e: to correct misinformation. Upon TERA resolution of the comments, the Final

	P	ROJECT IN	ISTRUCTIO	N
PI00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineeri Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE 143	of 145	PREPARED BY:	111	APPROVED BY:

Report will be issued and distributed to CPC, NRC, and outside parties on the IDCV service list.

### 5.2.3 INTERCHANGE OF INFORMATION

The requirements of Section 4.2 are not intended to prohibit the informal interchange of information between IDCV personnel and external parties. These communications are essential to the IDCV review process. However, the items in Section 4.2 require documentation for the reasons cited. Furthermore, to preserve the independence of the IDCV review process, it is important that IDCV personnel maintain discretion in the dissemination of information bearing on findings to outside parties until such a time that this information is final. This procedure will prevent confusion and foster credibility to the IDCV review process.

## 5.3 IDENTIFICATION AND EVALUATION OF DESIGN/CONSTRUCTION PROBLEMS

It is the duty of each IDCV team member to dentify any deficiency known to him that may be significant to the public health and safety. He shall be permitted to conduct all reasonable evaluations necessary to make a determination of the significance of suspected items. IDCV personnel are responsible for presenting their conclusions in a manner that other technically qualified personnel may understand and independently verify. Furthermore, it is the responsibility of IDCV personnel to assess the significance of their conclusion and attempt to understand the extent and root cause of findings. Any deviation from the above should be brought to the attention of the Project Manager.

	P	ROJECT IN	STRUCTIO	N
PI-3201 - 00 REV.: 2	DATE: 5/18/83	SUBJECT:	Engineerin Midland Inde Construction	ng Program Plan pendent Design and Verification Program
PAGE	of 145	PREPARED BY:	14:1-	APPROVED BY: Real

#### 6.0 QUALITY ASSURANCE

#### 6.1 APPLICABLE REQUIREMENTS

The Midland IDCV shall be performed in accordance with applicable quality assurance requirements of the NRC's regulation 10 CFR 50, Appendix B. Furthermore, the IDCV will comply with:

- NRC Regulatory Guide 1.28 (6/7/72) including Sections 1, 2, 3, 5, 7, 17, end 18 of ANSI N45.2-1971
- NRC Regulatory Guide 1.64 (Revision 1, 2/75) including Sections 1, 2, and 6 of ANSI N45.2.11-1974

These requirements are implemented by the TERA Corporate Quality Assurance Plan (QAP), Revision 3 (January 1, 1980) and the Midland IDCV Project Quality Assurance Plan (PQAP), Revision 0 (November 11, 1982).

#### 6.2 VERIFICATION OF COMPUTER CODES

All computer codes utilized by IDCV analysts shall be verified as follows:

- Program Verification The quality of the code should be determined from a comparison of the code generated solutions with known solutions of selected problems.
- Facility Verification Given that the generic quality of the code has been determined, the capability to reproduce known results utilizing hardware and software available to TERA must be determined.

PL 3201 009	ROJECT INSTRUCTION
PEV. a	SUBJECT: Engineering Program Plan
DATE: 5/18/83	Midland Independent Design and Construction Verification D
PAGE 145 of 145	PREPARED BY:

Program verification may be completed by external parties; however, facility verification is the responsibility of TERA and must be so demonstrated.

# 6.3 AUDITS

Quality assurance audits of project operations are conducted in accordance with ECP-5.6, "Quality Assurance Audits". For the Midland IDCV program, audits will be performed at least every 90 days of active project work.



SERKELEY . DALLAS . BETHESDA . BATON ROUGE . DEL MAR . NEW YORK . SAN ANTONIO . DENVER . LOS ANGELES

#### May 27, 1983

Mr. James W. Cook Vice President Consumers Power Company 1945 West Parnall Road Jackson, Michigan 49201

6 K . . .

Mr. J. G. Keppler Administrator, Region III Office of Inspection and Enforcement U.S. Nuclear Regulatory Commission 799 Roosevelt Road Glen Ellyn, IL 60137

Mr. D. G. Eisenhut Director, Division of Licensing Office of Nuclear Reactor Regulatory U.S. Nuclear Regulatory Commission Washington, D.C. 20555

FRINCIPAL STAFF RA D/RA A/RA DPRP DRMA DRMSPI DE CL

Re: Docket Nos. 50-329 OM, OL and 50-330 OM, OL Midland Nuclear Plant - Units 1 and 2 Independent Design and Construction Verification (IDCV) Program First Monthly Status Report

Attached is our first Monthly Status Report covering the period from project inception through May 27, 1983. Included in this report are:

- General background information on the Midland IDCV program and details related to Monthly Status Reports
  - Introduction and Purpose Section 1.0
  - Midland IDCV Program Background Section 2.0
  - Scope Section 3.0
  - Reporting Period and Issuance Section 4.0
- IDCV Program Status Summary Section 5.0
  - Project Chronology Attachment I

MAY 3 1 1983

TERA CORPORATION ATTA 7101 WISCONSIN AVENUE BETHESDA, MARYLAND 20814 301.654.8960 BY FEDERAL EXPRESS 8365314198

ATTACHMENT 2

Mr. James W. Cook Mr. J. G. Keppler Mr. D. G. Eisenhut

T. 4

5

Tracking System Summary for Open, Confirmed and Resolved (OCR) Item Reports, Finding Reports and Finding Resolution Reports - Section 6.0 and Attachment 2

-2-

Current Confirmed Item Reports - Attachment 3

Financial Status Report (CPC only) - Attachment 4

Sincerely,

livin une.

Howard A. Levin Project Manager Midland IDCV Program

Enclosures

cc: L. Gibson, CPC F. Buckman, CPC D. Hood, NRC J. Clements, Bechtel F. Levandoski, B&W IDCV Service List

HAL/sl



NRR Service List for Midland Independent Design and Construction Verification Program

Mr. Howard Levin, Project Manager TERA Corporation 7101 Wisconsin Avenue Bethesda, Maryland 20814

s"

. '

cc: James G. Keppler, Regional Administrator
U.S. Nuclear Regulatory Commission,
 Region III
 799 Roosevelt Road
 Glen Ellyn, Illinois 60137

U.S. Nuclear Regulatory Commission Resident Inspectors Office Route 7 Midland, Michigan 48640

Mr. J. W. Cook Vice President Consumers Power Company 1945 West Parnall Road Jackson, Michigan 49201

Michael I. Miller, Esq. Isham, Lincoln & Beale Three First National Plaza, 51st floor Chicago, Illinois 60602

James E. Brunner, Esq. Consumers Power Company 212 West Michigan Avenue Jackson, Michigan 49201

Ms. Mary Sinclair 5711 Summerset Drive Midland, Michigan 48640

Cherry & Flynn Suite 3700 Three First National Plaza Chicago, Illinois 60602

Ms. Barbar Stamiris 5795 N. River Freeland, Michigan 48623

Mr. Wendell Marshall Route 10 Midland, Michigan 48640 Mr. Steve Gadler 2120 Carter Avenue St. Paul, Minnesota 55108

Billie Pirner Garde Director, Citizens Clinic for Accountable Government Government Accountability Project Institute for Policy Studies 1901 Que Street, N.W. Washington, D. C. 20009

Charles Bechhoefer, Esq. Atomic Safety & Licensing Board U.S. Nuclear Regulatory Commission Washington, D. C. 20555

Dr. Frederick P. Cowan Apt. B-125 6125 N. Verde Trail Boca Raton, Florida 33433

Jerry Harbour, Esq. Atomic Safety and Licensing Board U.S. Nuclear Regulatory Commission Washington, D. C. 20555

Mr. Ron Callen Michigan Public Service Commission 6545 Mercantile Way P.O. Box 30221 Lansing, Michigan 48909

Mr. Paul Rau Midland Daily News 124 McDonald Street Midland, Michigan 48640

Ms. Lynne Bernabei Government Accountability Project 1901 Q Street, N.W. Washington, D. C. 20009

# MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM MONTHLY STATUS REPORT NUMBER I PERIOD INCEPTION THROUGH MAY 27, 1983

Prepared by:

ć

. "

1/shard

**Project Manager** 

Approved by:

Reviewed by:

Prinqigal-in-charge

# MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM (IDCV) MONTHLY STATUS REPORT NUMBER I PERIOD INCEPTION THROUGH MAY 27, 1983

#### 1.0 Introduction and Purpose

. "

Monthly Status Reports have been instituted by agreement between the Consumers Power Company (CPC), the Nuclear Regulatory Commission (NRC) and TERA to provide parties external to TERA's IDCV project team with up-todate information relative to program progress and any important issues identified during the reporting period. This initial report covers the period from project inception through May 27, 1983. A description of the scope, reporting periods and report issuance dates for Monthly Status Reports, as well as a summary of the background of the IDCV program are presented in this initial report. Subsequent reports will include only those ite ns discussed in section 3.0.

#### 2.0 Midland IDCV Program Background

The Nuclear Regulatory Commission (NRC) issued a letter on July 9, 1982 which requested that Consumers Power Company (CPC) provide for an independent assessment of the design adequacy of the Midland plant. CPC responded to this request on October 5, 1982 by submitting an outline of the scope of a proposed independent review program. A public meeting was held an October 25, 1982 at the NRC's Bethesda, Maryland offices to discuss details of the proposed program, the scope of which included an evaluation of the Midland Unit 2 Auxiliary Feedwater (AFW) system. During this meeting, the NRC requested that the scope of the independent design assessment program be expanded, including an assessment of the quality of construction. The NRC requested that CPC identify three candidate systems for scope expansion based upon their contribution to plant risk, from which one system would be selected.

1

8305316203

CPC responded to NRC by a letter dated December 3, 1982 which identified the Standby Electric Power system (diesel generator), Safeguards Chilled Water system and Containment Isolation system as candidate systems. A public meeting was held on February 8, 1983 at Midland, Michigan to discuss details of the program related to the evaluation of the AFW system and to provide status.

On March 22, 1983 the NRC selected the Standby Electric Power system and the Control Room HVAC system for scope expansion. Proposed elements of the scope of evaluation for these systems as well as the AFW system were discussed at another public meeting held on April 13, 1983 at the NRC's Bethesda, Maryland offices.

TERA Corporation has been selected by CPC to scope, manage, and implement the Midland Independent Design and Construction Verification (IDCV) Program. By a letter dated May 3, 1983, the NRC approved the selection of TERA. The selection is based upon the firm's technical qualifications, experience, and independence from the Midland project. Such independence includes al! individuals who may contribute to the IDCV Program.

The Engineering Program Plan (EPP), Revision 2, dated May 18, 1983, has been established to outline the scope, philosophy of review, methodology, independence requirements, organization, control, documentation, reporting, and quality assurance requirements for the Midland IDCV Program. The Project Quality Assurance Plan (PQAP), Revision 3, dated May 18, 1983, has been established to define the documented, auditable, control measures necessary to ensure the quality of services provided by TERA.

#### 3.0 Scope

e 11

The following items are included in Monthly Status Reports:

- IDCV Program Status Summary
- Tracking System Summary for Open, Confirmed and Resolved (OCR) Item Reports, Finding Reports and Finding Resolution Reports

- Current Confirmed Item Reports, Finding Reports and Finding Resolution Reports
- Financial Status Report (CPC only)

#### 4.0 Reporting Period and Issuance Dates

\*º

. .

The reporting period shall generally be on a calendar month basis with issuance of the corresponding Monthly Status Report around mid-month of the month following the end of the reporting period. The reporting period for this initial Monthly Status Report is from project inception through May 27, 1983, the date of this report. The second Monthly Status Report will be issued in mid-July, covering the period from May 27, 1983 through June 30, 1983.

#### 5.0 IDCV Program Status Summary

#### 5.1 Programmatic Activities

Attachment I provides the chronology for major project milestones during the reporting period. This chronology will be maintained up-to-date and included in future reports.

Several milestones warrant special highlight. On March 22, 1983, the NRC selected the Standby Electric Power (SEP) system and the Control Room HVAC (CR-HVAC) system for inclusion within the IDCV program scope. This selection along with the previously identified Auxiliary Feedwater (AFW) system completes the scope identification process for the IDCV program. A public meeting was held on April 13, 1983 to discuss details of TERA's AFW system review and conceptual plans for the SEP system and CR-HVAC system reviews. Comments were assimulated from CPC, NRC and interested members of the public. TERA responded to this direction by further development of the existing program to incorporate the revised scope. On May 18, 1983, TERA issued Revision 2 of the Engineering Program Plan and Revision 3 of the Project Quality Assurance Plan, reflecting the full scope of the IDCV program.

During the period of March-April, TERA transmitted information to the NRC relative to corporate and individual independence and professional qualifications. The NRC reviewed this information and on May 3, 1983 documented their formal acceptance of TERA to conduct the IDCV program and acceptance of the scope of the AFW system review. The NRC is currently reviewing TERA's proposed scope of review for the SEP system and CR-HVAC system as defined in Revision 2 of the Engineering Program Plan.

5.2 Design Verification Activities

2

5.2.1 Introduction and Background

Independent Design Verification (IDV) review activities during the reporting period of this status report focused upon the development and establishment of resources, programs, and organizational interfaces necessary to execute the IDV review methodology and making substantial progress in the IDV review for the AFW system. The methodology, as described in the IDCV Engineering Program Plan, strives to establish a consistent set of review activities applicable to systems, components, structures, and materials subject to IDV review. These review activities have been categorized into five areas as follows:

- Review of Design Criteria and Commitments
- Review of Implementing Documents
- Check of Calculations or Evaluations
- Confirmatory Calculation or Evaluation
- Check of Drawings and Specifications

The intent of this portion of the status report is to present and summarize important IDV activities undertaken during the reporting period relative to review progress made in the above five categories for each of the 45 design topics within the scope of the AFW system review. Future reports will be limited to significant activities on topics which have been completed during the month or on which substantial progress has been made. The programatic development was completed for the Standby Electric Power (SEP) system and the Control Room HVAC (CR-HVAC) system during the reporting period. Preliminary review activities were also initiated and will be reported in the next Monthly Status Report.

It is estimated that the AFW system IDV review is 60-75% complete relative to the initial scope defined in Revision 0 of the Engineering Program Plan. This estimate does not include any efforts required to resolve existing issues identified in section 6.0.

#### 5.2.2 IDV Topic Summaries

The IDV Topics and summaries of the scope for the AFW system are presented in section 3.1.3 of Revision 2 of the Engineering Program Pian. The corresponding Initial Sample Review Matrices are presented in Figure 1 for convenience. The following sections provide a topic-by-topic summary of progress:

#### 1.1-1 SYSTEM OPERATING LIMITS

Applicable operating limits for various components of the AFW system have been extracted from documents such as the FSAR and the Babcock and Wilcox (B&W) Balance-of-Plant Criteria Document. The review includes a check for completeness of specified parameters and bounding values and a check for consistency from document to document.

A check of appropriate calculations and evaluations is being conducted to verify that the specified limits are either capable of being met or are used correctly as input to assure proper system or component operation.

The limits identified in this review are being utilized in the review of other topics related specifically to component operability.

# INITIAL SAMPLE REVIEW MATRIX FOR THE AUXILIARY FEEDWATER SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM

			T	s	COPE OF	REVIEW	1
TOPICAL	DESIGN AREA	REVIEW OF DE	REVIEW OF	CHECK OF CLIMENTS	CONFIRMATIONS	CHECK OF	SPECIFICATIONS AND
	AFW SYSTEM PERFORMANCE REQUIREMENTS						
1.1-1	SYSTEM OPERATING LIMITS	x	x	×	1.000		
1.2-1	ACCIDENT ANALYSIS CONSIDERATIONS	x		1			
1.3-1	SINGLE FAILURE	x	×	x			
1.4-1	TECHNICAL SPECIFICATIONS	×	×				
1.5-1	SYSTEM ALIGNMENT/SWITCHOVER	1 x	x	1.11.00		1.33	
1.6-1	REMOTE OPERATION AND SHUTDOWN	X					
1.7-1	SYSTEM ISOLATION/INTERLOCKS	x	x				
1.8-1	OVERPRESSURE PROTECTION	×				•	
1.9-1	COMPONENT FUNCTIONAL REQUIREMENTS						
1.10-1	SYSTEM HYDRAULIC DESIGN	12	12	1 0		×	
1.11-1	SYSTEM HEAT REMOVAL CAPABILITY	12	12	10		1.1.1	
1.12-1	COOLING REQUIREMENTS	x		^	•		
1.13-1	WATER SUPPLIES	×		1.1.1	200	1.1.1	
1.14-1	PRESERVICE TESTING/CAPABILITY FOR OPERATIONAL TESTING						
1.15-1	POWER SUPPLIES	Î	×				
1.16-1	ELECTRICAL CHA' CTERISTICS	ÎŶ	-			•	
1.17-1	PROTECTIVE DEVICES/SETTINGS	x	×			x	
1.18-1	INSTRUMENTATION	x	×	×			
1.19-1	CONTROL SYSTEMS	x	×	Ŷ	4	^	
1.20-1	ACTUATION SYSTEMS	X	-	^	133		
1.21-1	NDE COMMITMENTS	x		1.1			
1.22-1	MATERIALS SELECTION	x	×				
1.23-1	FAILURE MODES AND EFFECTS			1			
KEY	NOTE						'

X - INITIAL SCOPE OF REVIEW

. . .

1.

X. DELETED SCOPE OF REVIEW

. - ADDED SCOPE OF REVIEW

INITIAL SAMPLE DOCUMENTED IN REV. 0 AND I OF THIS PLAN HAS BEEN MODIFIED EFFECTIVE 4/13/83

FIGURE |

# INITIAL SAMPLE REVIEW MATRIX FOR THE AUXILIARY FEEDWATER SYSTEM MIDLAND INDEPENDENT DESIGN VERIFICATION PROGRAM (CONTINUED)

Topicas	DESIGN AREA	REVIEW OF DEL	REVIEW COMMITMENTS	CHECK OF CAL	CONFIRMATIONS ONS AD	CHECK OF DO	JECHERANINGS AND
	AFW SYSTEM PROTECTION FEATURES						
il.1-1	SEISMIC DESIGN	×			10.11		
11.2-1	PRESSURE BOUNDARY	×	×	×	×	×	
11.3-1	PIPE/EQUIPMENT SUPPORT     EQUIPMENT OF ALLENDERT	X	X	×	×	×	
11.4-1	· EQUIPMENT QUALIFICATION	×	×	×		×	
11.5-1	HIGH ENERGY LINE BREAK ACCIDENTS	x					
11.6-1	· PIPE WHIP	x	x	×	1.1	x	
11.7-1	· JET IMPINGEMENT	×					
11.8-1	ENVIRONMENTAL PROTECTION	x					
11.9-1	. ENVIRONMENTAL ENVELOPES	x	x	x	x	x	
11.10-1	. EQUIPMENT QUALIFICATION	x	x	x	1 ^	1 x	
11.11-1	HVAC DESIGN	x		-			
11.12-1	FIRE PROTECTION						
11.13-1	MISSILE PROTECTION	1 2	×	×	1.1.1	1.12	
11.14-1	SYSTEMS INTERACTION				10.0		
		^	· ·	×	1.54		
	STRUCTURES THAT HOUSE THE AFW SYSTEM						
11.1-1	SEISMIC DESIGN/INPUT TO FOURPLENT				1212		
11.2-1	WIND & TORNADO DESIGN/MISSI E PROTECTION	1 2	*	×		×	
11.3-1	FLOOD PROTECTION	1 0	1.000				
11.4-1	HELBALOADS	1 2					
		^					
111.5-1	CIVIL/STRUCTURAL DESIGN CONSIDERATIONS	×	12.13			1.000	
111.6-1	• FOUNDATIONS	x	×	x			
111.7-1	CONCRETE/STEEL DESIGN	x	×	x		x	
and the second second	A TANKS	Ø	0	(Q)			88

X DELETED SCOPE OF REVIEW

. - ADDED SCOPE OF REVIEW

.

INITIAL SAMPLE DOCUMENTED IN REV. 0 AND I OF THIS PLAN HAS BEEN MODIFIED EFFECTIVE 4/13/83

۱.

#### 1.2-1 ACCIDENT ANALYSIS CONSIDERATIONS

The FSAR has been reviewed to determine those events for which the AFW system would be expected to play a role either in mitigation or recovery. The system was also reviewed to determine if there were any plausible means by which it could cause an accident or exacerbate an existing accident.

A meeting was held with Babcock and Wilcox to gather information related to the design requirements for the auxiliary feedwater system. Further review of CPC/Bechtel actions in response to the B&W-developed Anticipated Transient Operation Guidelines document has been deemed necessary and will be accomplished.

The review scope also was expanded somewhat to review calculations regarding the required system heat removal capability under accident conditions. This subject is being considered further under Topic 1.11-1, System Heat Removal Capability.

#### 1.3-1 SINGLE FAILURE

Applicable criteria have been extracted from the FSAR, NRC Regulations, and the B&W Balance-of-Plant Criteria document. Applicable documents such as piping and instrumentation diagrams and electrical schematics have been reviewed to determine whether the system can meet these criteria.

It has been determined that two complementary actions are necessary to verify the design relative to the capability of the AFW system to withstand a single failure. First, a confirmatory evaluation of the system is being conducted to verify the design from a single-failure-proof standpoint, especially regarding power supplies. This effort will concentrate mainly on the portions of the system comprising the pumps' suction and the steam discharge to the steamdriven turbine. Concurrently, a Failure Modes and Effects Analysis will be performed, as documented under Topic 1.23-1.

#### 1.4-1 TECHNICAL SPECIFICATIONS

The draft Midland Technical Specifications contained in the FSAR have been reviewed as they relate to the AFW system. The finalization of these specifications is on-going as well as the NRC's review. TERA is monitoring this process and when complete, the IDCV review will verify that the specifications are complete, consistent with NRC Standard Technical Specifications, and reflect commitments made in the FSAR.

#### 1.5-1 SYSTEM ALIGNMENT/SWITCHOVER

Applicable criteria have been drawn from such sources as the NRC Regulations, FSAR, B&W Balance-of-Plant Criteria document and the NRC Standard Review Plan and applicable Branch Technical Position.

The pertinent Piping and Instrumentation Diagram was reviewed to ascertain whether the criteria had been implemented. In addition, a CPC letter regarding specific switchover design capabilities, and the process by which they were derived, was reviewed. Finally, available procedures were reviewed to determine what guidance will be available to operators regarding alignment and switchover. These procedures are in draft form; further review will be undertaken later in the IDCV process.

The switchover of AFW control from the main control room to the auxiliary shutdown panel is under review as part of the control systems topic and also will be covered as part of the fire protection review.

#### 1.6-1 REMOTE OPERATION AND SHUTDOWN

Applicable criteria are included in the NRC Regulations, the FSAR, and the B&W Balance-of-Plant Criteria document. These criteria have been reviewed to determine their completeness and consistency. Results of the review also included several systems capabilities requiring further review under other topics. For example, the capability to control the system and shut down the plant from the auxiliary shutdown panel, and the regulatory guidance for manual actuation and control, are under review as part of the applicable electrical, instrumentation and control topics.

### 1.7-1 SYSTEM ISOLATION/INTERLOCKS

Criteria for this topic are contained in the NRC Regulations, the FSAR, the B&W Balance-of-Plant Criteria document, and the NRC Standard Review Plan. The applicable piping and instrumentation diagram was reviewed to determine whether the criteria had been implemented into the design.

Further review is being devoted to specific aspects of the design process, including a Design Change Approval Request relating to AFW pump low suction pressure trips.

#### 1.8-1 OVERPRESSURE PROTECTION

The criteria for this topic review were drawn from the FSAR and applicable codes and standards. Independent confirmatory calculations were performed for selected sections of piping to determine whether overpressure protection devices were needed. Attention was given to resolution of Management Corrective Action Report 65 and its related updates and submittals to the NRC. These deal with a potential AFW system suction piping overpressure problem discovered at an operating plant and applicable to the Midland design. The IDCV team will continue to follow the corrective action taken.

Site-requested changes to piping design pressure ratings are under review. This is an active review topic.
#### 1.9-1 COMPONENT FUNCTIONAL REQUIREMENTS

The component functional requirements review is progressing in parallel with reviews in several other topics as AFW system design criteria are translated into corresponding component specifications for parameters such as flow rates, allowable pressure drops, NPSH, voltage, device settings and similar characteristics. The review has also included IDV confirmation of functional requirement parameters. For example, the functional requirements for the AFW pumps are being independently confirmed as confirmatory calculations related to the topic reviews of System Hydraulic Design and System Heat Removal are completed. Reviews of test data are also in progress to confirm that specific components can meet their specified functional requirements. The components shown in Table I have been initially selected for this review. Because of its dependency on many topic reviews, this topic will be among the last to be completed.

## 1.10-1 SYSTEM HYDRAULIC DESIGN

Significant progress has been made in the System Hydraulic Design review area. The identification of design criteria and confirmatory calculations which are part of this review are essentially complete. Several Bechtel calculations have received preliminary reviews to date. Completion of the reviews of these calculations and selection of those calculations to complete the sample is currently in progress. An initial identification of implementing documents to be reviewed her been made.

## 1.11-1 SYSTEM HEAT REMOVA! CAPABILITY

Progress in the System Heat Removel Capability review area parallels that of the System Hydraulic Design review area. Identification of design criteria and development of confirmatory calculations is essentially complete. A B&W calculation concerning heat removal requirements has been reviewed. An initial identification of implementing documents to be reviewed has been made.

## 1.12-1 COOLING REQUIREMENTS

The criteria for cooling requirements have been identified and reviewed. This review has provided input to the selection of calculations and other documents to

## TABLE I

.

## SUPPLIER MIDLAND IDCV DOCUMENTATION REVIEW March 8, 1983

Item		Component ID				Fact			Weld			
No.	Туре	ID No.	P.O. No.	Cmpl	pl Dwgs	wgs Reqs	EQ	SQRT	QA	Props	Misc	Comments
1.	Pump	2P-005A	M-14	×	x	x		x	x	x	×	
2.	Motor	2P-005A	M-14	x	x	x		x	x	x	x	
3.	Pump	2P-005B	M-14	x	x	x		x	x	x	×	
4.	Turbine	2G-005B	M-14	x	x	x			x	x	×	
5.	Valve	2LV-3975AIV	J-255	x	x	x	x	x	x	x	x	
6.	Operator	2LV-3975AI	J-255	x	x	x	x	x		~	x	
7.	Valve	2MO-3965AV	M-117	x	x	x	x		x	×	×	
8.	Operator	2MO-3965A	M-117	x	x	x	x			~	Ŷ	
9.	Valve	2MC-3993A2V	M-398		x	x		x			^	
10.	Operator	2MO-3993A2	M-398		x	x		x				
11.	Valve	2XV-3989	M-118	x	x	x						
12.	Operator	2XV-3989A1	M-118		x	x						
13.	Valve	25V-3969A	J-256	x	x	x			×	~	~	
14.	Valve	2MO-3226V	M-117		x	x	x		^	^	^	
15.	Operator	2MO-3226	M-117		x	x	x					
16.	Valve	2MO-3277AV	M-117		x	x	x	×	×			
17.	Operator	2MO-3277A	M-117		x	x	x	×	^			
18.	Heat-X	2E-105A	M-14		x	x	^	x				

INDLE I (CUNINUED)	TABLE	(CUNTINUED)
--------------------	-------	-------------

Item	em Component ID			Gen		Foct			Weld	Mat		
No.	Туре	ID No.	P.O. No.	Cmpl	Dwgs	Reqs	EQ	SORT	QA	Props	Misc	Comments
19.	Panel	2C-114	J-202		x	x		x				
20.	MCC	2BP-03	E-45		x	х.	x	x				
21.	SwGear	2A-05	E-205	х	x	x		x			x	
22.	Cable		E-26A		×	x	x				x	600V
23.	Transmitter	2PT39000B1	J-245		x	x	х					
24.	Transmitter	2FT3969A	J-245	x	x	x					x	
25.	Transmitter	2FT3975AB	J-245		x	x	x	x				
26.	Transmitter	2LT3298	J-245		x	x						
27.	Transmitter	2LT3975AA2	J-245	X	x	x	x	x			x	
28.	Indicator	2LIK3975AA2	J-204		x	x						
29.	Switch	2ZS3975A1	J-255X	х	x	x		x			x	
30.	Cable		E-60		x	x	x					Instru
31.	Air Cooler	2VM-54A	M-149	x	x	x		x			x	mano.
32.	Elec. Penet.		E-20A				x				<b>^</b>	
33.	Piping				x				x	x		
34.	Pipe supports	5			x				x	x		
35.	Cable Tray				x					x		
36.	Tray Support	5			x					x		
37.	Conduit				x							

## TABLE I (CONTINUED)

.

.

Item	Component ID			Gen		Fact			Weld	Mat		
No.	Туре	Cmpl	Dwgs	Reqs	EQ	SORT	QA	Props	Misc	Comments		
38.	Conduit S	upports			x					x		
39.	Instru. Pip	ping								x		
40.	HVAC Du	cts (later)										
41.	HVAC Sup	oports (later)										
42.	Rebar									x		
43.	Str. Steel								x	x		
44.	Inserts								x	x		

be reviewed in the Equipment Qualification and Component Functional Requirements review areas.

#### 1.13-1 WATER SUPPLIES

The criteria for the AFW water supplies have been identified and reviewed. This review has provided input to the selection of calculations and other documents to be reviewed in the System Hydraulic Design and Component Functional Requirements review areas. For example, the criteria for switchover from condensate storage to service water have been used as an input to reviewing calculations in the System Hydraulic Design area. Implementing documents for review of the Water Supply area have been identified.

## 1.14-1 PRESERVICE TESTING/CAPABILITY FOR OPERATIONAL TESTING

Criteria for the review of preservice testing requirements and operational testing capability are being identified in conjunction with other review areas, including the Technical Specification Review Area. The scope of review in this area has been expanded to include a review of implementing documents and engineering evaluations supporting test programs. This will serve as input to the ICV review. This expansion is based upon the desire to further verify system conformance with design criteria and commitments through an evaluation of tests that serve to establish the adequacy of the design and the capability of the system to function as planned.

#### 1.15-1 POWER SUPPLIES

The applicable design criteria for AFW power supplies have been identified from NSSS vendor, regulatory and industry requirements. The Midland FSAR is the primary implementing document design which has been checked to verify the proper consideration of the design criteria determined from the criteria review. The AFW system logic and schematic diagrams have been reviewed to ensure that requirements relative to the quality of power supplies (diversity and redundancy) are met. In particular the review included the assurance that the AFW system is operable in the event of loss of offsite power and station blackout.

### 1.16-1 ELECTRICAL CHARACTERISTICS

Design criteria relevant to the electrical characteristics of cable physical separation, system electrical separation, cable and raceway sizing and terminal voltage on power circuits have been identified. The Midland FSAR sections implementing these criteria have been reviewed to verify that the criteria have been considered in the design process. Cable sizing calculations have been reviewed as applied to seven power circuits in the AFW system. The cable routing design process is being reviewed to ensure consideration of cable separation criteria in that process.

### 1.17-1 PROTECTIVE DEVICES/SETTINGS

Design criteria relevant to this topic have been identified. The Midland FSAR has been reviewed to ensure that the criteria have been documented and that commitments have been made to meet the criteria. The schematic diagrams for all motor-operated values in the AFW system have been reviewed to ensure incorporation of thermal overload and opening torque swtich bypass features. The AFW pump motor schematic is being reviewed against the committed design criteria. The evaluation of the electrical penetration assembly protection scheme are under review to ensure compliance with design criteria.

## 1.18-1 INSTRUMENTATION

The instrumentation and alarms required to operate, monitor and protect the AFW system, as determined by design criteria, commitments and expected plant operations, have been reviewed against those specified for the AFW system to verify the adequacy of the instrumentation. Selected instrument accuracies under applicable plant operating conditions have been reviewed and evaluated. Instrument loop diagrams for steam generator water level indication have been reviewed for proper circuit electrical design. The calculation for steam generator low water level setpoint has been reviewed for compliance with design criteria. Major instrument package procurement specifications have been reviewed to verify that the design criteria have been considered in the purchase of the instrument hardware.

#### 1.19-1 CONTROL SYSTEMS

Design criteria and commitments governing the steam generator water level and AFW turbine control systems have been checked to verify the inclusion of necessary regulatory, industry, and system performance requirements. The Midland FSAR has been reviewed to verify that the necessary requirements were used as input to the control system design. An evaluation of control system characteristics such as time response, component characteristics, and separation from actuation systems has been performed. A very limited FMEA review has been made (See Topic 1.23-1, Failure Mode and Effects). Control system circuitry design (voltages, currents, polarity) has been reviewed to verify that selected components will function as intended in the steam generator water level control system. The circuitry design review has included instrument loop diagrams, logic diagrams, and valve and motor schematic diagrams.

## 1.20-1 ACTUATION SYSTEMS

The auxiliary feedwater actuation system (AFWAS - which includes FOGG, "Feed Only Good Generator") design criteria and commitments have been reviewed to verify the proper consideration of regulatory requirements, industry codes and standards, and plant operational requirements. AFW system logic diagrams and schematic diagrams for all motor operated valves and the AFW pump motor have been reviewed against the design commitments. In addition, the AFWAS procurement specification is being reviewed against the design criteria and commitments.

### 1.21-1 NDE COMMITMENTS

Design criteria, commitments and implementing documents related to nondestructive examination have been identified and are under review against applicable industry codes and standards. A detailed checklist has been developed to assist in this activity. As commitments and proper translation into specifications and field procedures are verified, this input is being factored directly into the ICV review process to verify that these have been properly implemented. The review of implementing documents and specifications was added to the scope of the IDV to support the expanded NDE/Material Testing program documented in section 5.3.1 of this report.

### 1.22-1 MATERIAL SELECTION

This topic will be initiated in June, 1983 and will be reported upon in future status reports.

### 1.23-1 FAILURE MODES AND EFFECTS

This topic has been added to the scope of the IDV to verify conclusions reached about system and component failure modes and effects under various operating conditions.

The topic review will be initiated by continuing where the FSAR evaluation ended. It is intended, at the present time, that emphasis will be placed on components of the electrical, instrumentation and control systems. Criteria from other review areas will be consolidated as an initial step in preparing the planned confirmatory evaluation.

#### II.J-I SEISMIC DESIGN

The seismic design chain, criteria and commitments applicable to the design of the Midland plant were identified and reviewed with particular emphasis on specific aspects of the criteria applicable to AFW components and systems and structures that house these components and systems. In view of several major perturbations during the design process, a significant portion of time was devoted to the identification and understanding of the seismic design chronology for the plant. The knowledge gaired from this activity was utilized to assist IDV reviewers in the selection of issues and methodologies on which to concentrate the review. The selection of specific structural elements/features, components and systems was also influenced by this activity.

### 11.2-1 SEISMIC DESIGN - PRESSURE BOUNDARY

Progress on this topic has been made in two principal areas. A confirmatory seismic stress analysis is nearing completion for a portion (i.e. one piping problem) of AFW piping and supports on the "B" train inside the Unit 2 containment building. The line evaluated runs from the containment penetration to the first anchor which is approximately midway along the "B" train line on its paths from the containment penetration to the steam generator ring header for the AFW discharge. IDV analysts will soon be in the process of comparing the results of their analyses with Bechtel's analyses to independently confirm the adequacy of implementation of the design methodology and results. The comparison includes the contribution of seismic stress at critical locations, predicted support loads for all supports along the line and a design verification for representative support types. The model was developed by the IDV analysts without prior benefit or knowledge of Bechtel's methodology and in particular, specific modeling assumptions. The IDV analysts utilized the dimensional asbuilt data that was independently compiled through the ICV field verification program related to the program activity, Verification of Physical Configuration (see sections 5.3.1 and 5.3.6 of this report). In a separate activity, IDV reviewers identified and initiated a review of pertinent criteria, implementing documents, calculations and specifications applicable to ASME Code considerations associated with the pressure boundary integrity of a portion of the AFW discharge piping located in the auxiliary building. Future activities will include a review of Bechtel's recent configuration changes associated with the AFW piping and supports inside containment as well as a review of field engineering for small bore piping.

## 11.3-1 SEISMIC DESIGN - PIPE/EQUIPMENT SUPPORT

This topic closely parallels that of Topic II.2-I which is associated with pressure boundary integrity and ASME Code considerations. As discussed, piping supports are chosen for evaluation consistent with the selection of piping lines to permit an integrated evaluation of the seismic design capabilities of the total system. Progress to date has been discussed for piping supports. The anchorage and support for AFW equipment is under evaluation as part of Topic II.4-1. For components selected for evaluation under this topic (see Table I), selected calculations, dravings and specifications are being checked to verify adequate seismic capability in accordance with seismic design criteria and commitments.

## 11.4-1 SEISMIC DESIGN - EQUIPMENT QUALIFICATION

In addition to a review of seismic equipment qualification design criteria and commitments and implementing documents, the principal progress on this topic has been to select a sample of components for review (see Table I) and to acquire existing SQRT qualification "packages" from Bechtel. Progress has been slowed because Bechtel's seismic equipment qualification process is in early stages of completion. Complete SQRT packages are being reviewed along with the process for completing additional packages.

II.5-1 HELB/PIPE WHIP/JET IMPINGEMENT (Including II.6-1 and II.7-1)

Criteria for this group of review area have been identified and preliminary reviews conducted. Implementing documents, calculations, and drawings will be reviewed upon completion of the confirmatory calculation in the Seismic Design review area.

II.8-1 ENVIRONMENTAL PROTECTION/ENVIRONMENTAL ENVEL-(Including OPES/EQUIPMENT QUALIFICATION/HVAC DESIGN II.10-1 & II.11-1)

The criteria and commitments for this group of review areas have been identified and reviewed. A sample of equipment for the review of calculations and evaluations, primarily associated with the Equipment Qualification Report, has been made as shown in Table I. Reviews of the selected equipment qualification packages have been initiated. A confirmatory calculation in the environmental envelopes review area has been initiated. HVAC design criteria have been identified.

## II.12-1 FIRE PROTECTION

."

Steps have been completed to organize the review of fire protection for the AFW system into subtopics. These topics are:

- Safe shutdown analyses
- Associated circuits analyses
- Fire hazards analyses
- Remote shutdown transfer switches/isolation devices
- Fire barriers
- Fire detection systems
- Suppression systems
- Emergency lighting

FSAR commitments, documentation of the fire protection program, and CPC submittals to NRC related to a comparison to 10CFR50 Appendix R and to BTP CMEB 9.5-1 have been reviewed. Interactions with Bechtel personnel have taken place to identify and collect design documentation pertaining to the AFW fire protection features, and to discuss fire protection program status and approaches in key areas. Detailed design and analysis information has been received. Verifications and reviews were initiated for two of the eight fire protection subtopics, namely fire barriers and emergency lighting. It is expected that these two subtopics and the remaining six will be completed in the next reporting period.

#### 11.13-1 MISSILE PROTECTION

The review scope for the Missile Protection review area consists of a review of criteria and commitments. This review is currently in progress.

### II.14-1 SYSTEMS INTERACTION

Criteria for this review are defined in the Bechtel/CPC program for determination and resolution of potential systems interactions. This program was obtained for review after discussion with key Bechtel personnel involved in the program.

The program will be reviewed for completeness and consistency. System walkdowns in selected areas will be observed, and selected data sheets and recommendations will be reviewed.

## III.I-I SEISMIC DESIGN/INPUT TO EQUIPMENT

In parallel with discussions and reviews associated with the seismic design chronology, substantial progress has been made relative to the understanding and review of modeling procedures and techniques utilized to generate in-structure seismic input (e.g. floor response spectra). This activity has taken more effort than anticipated to identify the complex history associated with the seismic design chain and verify that the various perturbations were adequately handled by the project designers and analysts. Particular attention has been focused on the acquisition and review of information related to the effects of floor flexibility on predicted floor response spectra. Emphasis is being placed on the proper specification, use, and transfer of floor response spectra between interfacing groups both internal and external to Bechtel.

111.2-1 WIND AND TORMADO/MISSILE PROTECTION

III.3-I FLOOR PROTECTION

III.4-I HELBA LOADS

The criteria and commitments associated with these topics have been identified and the review commenced. Progress will be reported in future reports.

## 11.5-1 CIVIL-STRUCTURAL DESIGN CONSIDERATIONS

Progress has been made on this topic in two principal areas. First efforts to identify design criteria such as that incorporated within Bechtel's Civil/Structural Design Criteria document and the FSAR have been completed and the review is continuing. Secondly, efforts are continuing in a review of project experience within the civil/structural discipline to identify important issues that have surfaced during the project, review how these have been resolved and verify that these do not exist in the same or similar form elsewhere.

### III.6-I FOUNDATIONS

The concentration of this topic is on structural aspects of foundation design verses soil mechanics aspects. Accordingly, a portion of the auxiliary building foundation has been selected for detailed structural review. Efforts to date have focused on an identification of foundation design criteria, a review of project experience to understand the design chronology and important loading conditions and the collection of pertinent calculations. The detailed structural review is just being initiated and will be reported upon in future reports.

## 111.7-1 CONCRETE/STEEL DESIGN

Specific structural elements (e.g. shear walls, floor diaphram) have been selected for detailed review and evaluation. Emphasis is being placed upon an evaluation of the project's capability to transfer loading information both internally and externally from one organization (e.g. analytical groups) to another (e.g. design groups) and on the proper identification and interpretation of this information. Input from other IDV topics is important relative to information gained in the review of the various loading conditions that affect structural elements. The specific use and implementation of this information is being verified through a review of design calculations. These calculations are being reviewed to verify the design organization's capability to properly size and detail concrete and steel structural elements.

#### 5.3 Construction Verification Activities

#### 5.3.1 Introduction and Background

Independent Construction Verification (ICV) review acitvities during the reporting period of this status report focused upon the development and establishment of resources, programs, and organizational interfaces necessary to execute the ICV review methodology and initiation of the ICV review. The methodology, as described in the IDCV Engineering Program Plan, strives to establish a consistent set of review activities applicable to systems, components, structures, and materials subject to ICV review. These review activities have been categorized into five areas as follows:

- Review of Supplier Documentation
- Review of Storage and Maintenance Documentation
- Review of Construction/Installation Documentation
- Review of Selected Verification Activities
- Verification of Physical Configuration

The intent of this portion of the status report is to present and summarize important ICV activities undertaken during the reporting period and to categorize these activities using the above five review categories. Sections 5.3.2 through 5.3.6 address each of these review categories respectively. The ICV review categories and Topics for the AFW System are presented in section 3.2.3 of Revision 2 of the Engineering Program Plan. The corresponding Initial Sample Review Matrix is presented in Figure 2 for convenience.

Events external to the ICV review program have had significant impact on the program. Accordingly, the following discussion summarizes the background of events which have had an influence on where the ICV review is today and where it is to be directed in the future.

In a letter to the NRC dated October 5, 1982, CPC outlined a proposed scope for the planned Midland independent design review program. In addition to a design

## INITIAL SAMPLE REVIEW MATRIX FOR THE AUXILIARY FEEDWATER SYSTEM MIDLAND INDEPENDENT CONSTRUCTION VERIFICATION PROGRAM<sup>1</sup>

	//		/	S	COPE OF	REVIEW	
TopICMIN	SYSTEM/COMPONENT	REVIEW OF	MARTENER OF TION	REVIEW OF CONNENTAND	REVIEW OCCUMENTATION	VERIFICATION ACTIVITIES CONFICURATION SICAL	/
	MECHANICAL						
1.1-1c	. EQUIPMENT	×	x	x	x	x	
1.2-1c	· PIPING	x		x	x	×	
13-1c	PIPE SUPPORTS	×		×	x	x	
	ELECTRICAL						
II.I-Ic	. EQUIPMENT	x	×	×	×	×	
11.2-1c	. TRAYS AND SUPPORTS	×				×	
11.3-1c	CONDUIT AND SUPPORTS	x		•		x	
11.4-1c	• CABLE	×	×	x	×	x	
	INSTRUMENTATION AND CONTROL				Sala a		
III.1-1c	• INSTRUMENTS	×	x	x	×	× 1000	
III.2-1c	· PIPING/TUBING	×				x	
111.3-1c	• CABLE	x		•		x	
	HVAC						
V.I-Ic	. EQUIPMENT	×		*	*		
IV.2-Ic	. DUCTS AND SUPPORTS	x		^	^	x	1
	STRUCTURAL						1
Valate	. FOUNDATIONS						
V.2-1c	• CONCRETE	÷	10.00	×		. 1	
V.3-Ic	. STRUCTURAL STEEL	x		Ŷ		: /	
VI.1-1c	NDE/MATERIAL TESTING PROGRAM	-		-		./	
EY	NOTE						

X- DELETED SCOPE OF REVIEW
 ADDED SCOPE OF REVIEW

.

. .

INITIAL SAMPLE DOCUMENTED IN REV. 0 AND I OF THIS PLAN HAS BEEN MODIFIED EFFECTIVE 4/13/83

### FIGURE 2

verification component, this program included a verification of physical configuration of selected structures and components for the AFW system. A public meeting was held on October 25, 1982 at NRC's Bethesda, Maryland offices where the details of this program were discussed. The NRC indicated that they would like the proposed program to be expanded to include a review of an additional system with increased emphasis on the verification of the quality of construction including additional verification of physical configuration.

TERA responded to NRC and CPC direction by developing an expanded Independent Construction Verification (ICV) program centered around the five previously discussed review categories. The scope of this revised program was documented in Revision 0 of the EPP dated November 29, 1982. Details of the ICV and IDV were discussed at public meetings held on February 8, 1983 at Midland, Michigan and April 13, 1983 at NRC's Bethesda, Maryland offices.

TERA's initial field verification activities were initiated the week of November 29, 1982 with a physical configuration verification of the AFW system piping and supports inside containment. In early December 1982, CPC instituted their Construction Completion Program (CCP). Under direction from NRC and CPC, TERA was asked to hold certain portions (in particular, physical configuration verification) of the ICV review in obeyance pending resolution of critical interfaces with the CCP and other on-going construction related programs. Accordingly, only reviews of supplier documentation, storage and maintenance documentation and selected verification activities proceeded.

On March 22, 1983, the NRC selected the Standby Electric Power system and the HVAC system assuring control room habitability as additional systems for IDCV review. Revision 2 of the EPP dated May 18, 1983 incorporates these systems into the scope of the ICV as well as the IDV.

During the April 13, 1983 public meeting, the NRC, CPC and TERA agreed that the scope of ICV activities within the prescribed sample selection boundaries could proceed irrespective of the stage of construction completion. This direction enables the ICV review to obtain better insight into the quality of:

- Completed construction activities
- On-going construction processes from the standpoint of how these will impact future completed construction products
- Remedial and corrective actions taken in response to ongoing construction review efforts such as the CCP

At the current time, the ICV scope has been fully defined and the review process is gearing up to full speed, consistent with critical interfaces with on-going construction related programs.

The events described above have enabled the initiation of all planned ICV review activities which are described below and in the following sections.

- The sample selection boundaries for the ICV review of the AFW system were firmly established and implemented into the ICV review program. Development of the AFW System sample selection boundaries was performed through the joint efforts of IDV and ICV reviewers. Additional, detailed discussions were undertaken by Lead IDV and ICV personnel to identify which components, structures, and material within the sample selection boundaries would be subject to detailed ICV review. The selection process employed the sample selection criteria as defined in the EPP and resulted in the designation of the items shown in Table I as being subject to initial ICV review.
- The ICV review activities associated with the AFW System were expanded in scope. The additional review activities and the reason these activities were factored into the ICV review program are as follows:

System/Component	Scope of Review Added	Reason(s)
- Electrical Cable Trays & Supports Conduit & Supports I&C Cable	- Review of Construc- tion/Installation Documentation & Review of Selected Verification Activities	<ul> <li>Project experience</li> <li>Monitor the outputs of the on-going over inspection program for cable separation as directed by NRC</li> </ul>
<ul> <li>NDE/Material Testing Program</li> </ul>	- Verification of Physical Configura- tion	- Project experience - NRC direction

- As a result of adding the NDE/Material Testing Program as an integral part of the AFW system ICV review, Lead ICV personnel commenced the development of this program. Program execution will involve first the selection of the sample and sample size, selection of the specific components and material to be tested, determination of the type(s) of testing to be performed, testing, and evaluation and documentation of the test results. To assist in executing the NDE/Material Testing Program, Lead ICV program personnel initiated the solicitation and review of proposals from material testing firms who have exhibited the capability to accomplish required testing in a professional, objective manner. Selection of a material testing firm has not been completed; review of proposals and identification of material testing firm capabilities continues.
- Important interfaces between the Lead ICV program personnel and reviewers and IDV personnel have been tested and utilized to ensure their effectiveness and efficacy. Additionally, critical interfaces with site-construction personnel have evolved to the point where ICV reviewers can acquire needed information and are afforded the flexibility and latitude necessary to be effective in the ICV review program.

## 5.3.2 Review of Supplier Documentation

. \*

The overwhelming majority of resources expended in executing the ICV review activities has been devoted to defining the detailed steps of the Supplier Documentation Review and performing the review steps. These activities are of substantial importance to the remaining portions of the ICV review, because they establish the documented resource which is used as initial input to evaluating remaining construction activities. Additionally issues and trends determined as a result of performing the review of supplier documentation have alerted, and will continue to alert, ICV reviewers to outputs in the construction process which require a greater degree of scrutiny. In essence, the results of the review of supplier documentation establishes the reference for the effective continuance of the ICV review process.

During the period of this status report the following important activities have been undertaken as part of the review of supplier documentation. Detailed review matrices for components within the AFW system sample selection boundary were developed as a joint effort with IDV reviewers and serve to direct the activities of the ICV reviewers performing the review of supplier documentation. The review of supplier documentation has been broken down into discrete review categories as follows:

- General Completion Overall review of documentation to ensure that the supplier package is generally complete for Document Categories required by specification for the component.
- Drawings Review of supplier drawings for conformance to specification requirements for the component, subcomponent or part.
- Functional Requirements Review of supplier documentation for conformance of major functional requirements to specifications.
- Environmental Qualification Review of supplier documentation for conformance to specification requirements.
- Seismic Qualification Review of supplier documentation for conformance to specification requirements.
- Welding, NDE, QA Review of supplier documentation for conformance to specification requirements for the component, subcomponent or part.
- Material Properties Review of supplier certified material property reports for conformance to specification requirements for the component, subcomponent or part.
- Miscellaneous Review of instruction manuals, cleaning and coating procedures, storage and handling instructions and shipping procedures for conformance to specification requirements for the component, subcomponent or part.

In practice, an ICV or IDV reviewer is assigned one or more of these review categories for a specific component or group of components identified for ICV review.

As of the writing of this report, the majority of the activities necessary to perform the following documentation reviews for the AFW system have been completed:

- General Completion
- Drawings

. \*

.

- Functional Requirements
- Miscellaneous

As a result of conducting the above reviews, approximately 1,000 documents have been reviewed for applicability, catalogued, and categorized as to the type of document - i.e., drawing, welding procedure, seismic qualification report, etc.

The "Environmental" and "Seismic Qualification" reviews are tied closely to the IDV review process and have progressed to the stage of completion identified for selected components in the IDV review portion of this status report.

The "Welding, NDE, QA" documentation review has focused upon identifying the derivation of the requirements, the completeness and consistency of the requirements and the cataloguing of vendor-supplied documentation which satisfies the requirements for welding, NDE, and QA aspects of selected fabricated components. Further, more detailed review of the vendor-supplied documentation has not been aggressively pursued pending finalization of the degree of involvement of an outside material testing firm (see Section 5.3.1 of this status report) in the ICV review program.

The review necessary to verify the adequacy of Material Properties by reviewing certified material property reports has most recently been initiated and, as a result, not much progress has been made toward completing this review during the current reporting period.

To ensure that a consistent method and set of data are used and collected during the review of supplier documentation, detailed checklists were prepared and implemented. The checklists, and associated implementing Project Instruction (PI-3201-007), direct the ICV reviewer to sources of information and direct the recording of required information onto a standardized form. As of the writing of this report, five checklists have been prepared and used to conduct the review of vendor supplied documentation. The title and a brief description of each checklist used in this protion of the ICV review are as follows:

Documentation Verification Form (DVF)

Checkoff list utilized to record those requirements imposed upon suppliers and vendors which define the specific documents to be submitted to fulfill and satsify procurement and specification requirements;

Documentation Availability Checklist (DAC)

The DAC is used to document the process and sources of information used to complete the Documentation Verification Form and to provide a consistent, standard format for documenting the results of evaluating the completeness of vender documentation submittals;

Supplier Documentation Functional Review (SDFR) Form

The SDFR provides the format and directs the recording of data relevant to the following of specific categories of vendor-supplied documentation:

- a) Instructions (operating, maintenance, etc.)
- b) Cleaning & Coating Procedures
- c) Certified Material Reports
- d) Supplier Shipping Procedures:
- Supplier Documentation Adequacy (SDA) Verification Form

This form is used in conjunction with the SDFR to evaluate the adequacy of the vendor's documentation submittal; and

Time-Base Evaluation (TBE) Form for Vendor Documentation Submittals

This form provides the format for establishing a method to evaluate the timeliness of certain vendor documentation submittals associated with a specific component. Vendor documentation submittals are compared on a time-base against two key events in the construction process:

- a) Date component is received at the site
- b) Date component is withdrawn from storage for installation.
- Commencement of the supplier documentation review required a greater-than-anticipated scope of task initiation activities. These activities were necessary to develop an understanding of the following:

- Relationship of site vendor files to vendor files retained in Ann Arbor;
- Distinctions made between supplier documentation included as part of a QA data package and that documentation included as part of the vendor document control system;
- Location of different document control centers and their principal file holdings and scope of responsibilities;
- Information required to access needed documents and records; and
- Location and operation of systems utilized to index needed information.
- As of the writing of this report, activities undertaken with regard to supplier documentation reviews, have been focused upon the collection and assimilation of vendorsupplied information. Current and near term activities of the ICV reviewers are and will be directed toward a thorough evaluation and assessment of the significance of findings resulting from the review of supplier documentation.

# 5.3.3 Review of Storage and Maintenance Documentation

This review is intended to ascertain the stored and as-installed condition of selected components of the systems selected as part of the IDCV program. Discrete activities which constitute this review include the following:

- Documentation Review and Observation of Receipt Inspections;
- Documentation Review and Observation of Warehouse Storage Practices;
- Documentation Review and Observation of In-place Maintenance Practices; and
- Visual Inspection of Installed/Stored Components.

The progress made to date in conducting this review has all been associated with the components selected in the AFW System. Activities undertaken to date include the following:

- Checklists have been prepared and implemented which direct the acquisition and recording of information and data which characterize the receipt inspection, storage and maintenance activities. Detailed Project Instructions (PI-3201-007) have been prepared which provide ICV reviewers with an explanation in the use of the following checklists:
  - Receipt Inspection Checklist; and
  - Storage and Maintenance Checklist.
- Data required by the checklists have been collected and completed for the components selected with the AFW System sample selection boundaries. The components selected for this review are shown in Table 1.

Current and near-term activities involve the evaluation of the collected data and an assessment and recording of the significance of any issues resulting from the evaluation.

 ICV reviewers, in a joint effort with the IDV reviewers, prepared the review matrices for the Control Room HVAC and Standby Electric Power Systems. The matrices require a review of storage and maintenance documentation applicable to the following categories of components within the sample boundaries of the indicated systems.

Standby Electric Power System

- Mechanical Equipment
- Electrical Equipment and Cable
- Instruments and Instrument Cable

Control Room HVAC System

- Mechanical Equipment
- Instruments
- HVAC Ducts & Supports

Specific components within each of the above categories are currently being identified.

## 5.3.4 Review of Construction/Installation Documentation

. \*

As of the writing of this report, no resources have been expended in performing the actual review of construction/installation documentation. Activities undertaken to date have been directed toward the selection of specific components within the AFW System sample selection boundaries whic', will be subject to this review.

### 5.3.5 Review of Selected Verification Activities

During this reporting period ICV reviewers commenced the review of selected outputs from the cable separation and pipe support over-inspection program which relate directly to cables and pipe supports within the ICV review sample selection boundaries of the AFW System. These activities were conducted at the site and focused upon the collection of required documentation, including procedures and drawings, and the evaluation of the procedures to discern the methodology employed by the over-inspection programs. This evaluation is necessary to identify those outputs of the program which are most representative of the final products of the over-inspection process and therefore those products which should be subject to ICV review. Evaluation of selected outputs v as initiated and continues. Near term activities relate to continued detailed evaluation of selected outputs from the program that relate to the AFW system and the extension of these evaluations to include the Control Room HVAC and Standby Electric Power Systems.

## 5.3.6 Verification of Physical Configuration

As a first and important review associated with the verification of the physical configuration of selected components within the sample selection boundaries of the AFW system, ICV reviewers conducted a review of selected AFW System pipe, hangers, and supports. This review involved not only the careful selection of those pipes, hangers and supports to ensure a comparative basis for other, similar reviews and extrapolation to similar items, but also extensive field verification and measurement.

The review involved the field measurement of pipe, hangers, and supports of the "B" Auxiliary Feedwater train, inside the Midland Unit 2 containment building. Subsequent to acquisition of field measurements and verification of identity and orientation, the collected data were compared against design documentation and documentation used as input to representative stress and seismic design calculations. The results of these efforts have been summarized into an engineering evaluation report which highlights the salient findings of the review and evaluation and documents the methodology utilized in conducting the physical configuration verification.

Near term activities relate to completing the review of issues arising from the physical configuration verification of selected AFW System pipe, hangers, and supports and selecting similar samples associated with the Control Room HVAC and Standby Electric Power systems.

6.0 Summary of Open, Confirmed and Resolved (OCR) Item Reports, Finding Reports and Finding Resolution Reports

Attachment 2 provides TERA's Tracking System Summary for Open, Confirmed and Resolved (OCR) Item Reports, Finding Reports and Finding Resolution Reports. This tool assists TERA in tracking the disposition of issues as they progress through the review process. Attachment 3 provides re-typed copies of all existing Confirmed Item Reports. To date no items have progressed to the Findings stage of the reporting process which is documented in Project Instruction PI-320I-008 and can be found as part of Appendix B of the Project Quality Assurance Plan.

A meeting will be held on June 3, 1983 at Bechtel's Ann Arbor, Michigan offices to obtain additional information reltaive to the Confirmed Items presented in Attachment 3.

## PROJECT CHRONOLOGY

## MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM TERA PROJECT 3201 THROUGH 5/27/83

Date	Milestone.
September 2, 1982	TERA proposal to CPC for Midland Independent Design Verification (IDV) Program
September 20, 1982	CPC letter of intent to use TERA for Midland IDV
September 24, 1982	TERA identification of IDV goals, objectives, system selection criteria, methodology, tasks, and schedule (outline presented to CPC on 9/28/82)
September 28, 1982	<ul> <li>Meeting of CPC, TERA, and MAC in Jackson to develop submittal to NRC addressing IDV and INPO evaluation programs. TERA selects can- didate system for IDV program</li> </ul>
September 30, 1982	TERA submittal of corporate Quality Assurance Plan to CPC for their review and acceptance
October 5, 1982	CPC submittal of Midland Independent Review Program to NRC
October 12, 1982	CPC approval of TERA corporate Quality Assurance Plan
October 25, 1982	Presentation on Midland IDV and INPO pro- grams to NRC at NRC's Bethesda offices
October 27, 1982	TERA conceptual development of IDV program modifications to further address the quality of construction (telecopy to CPC)
October 28, 1982	CPC decision to separate IDV and INPO evalu- ation programs

Date	Milestone
November 2, 1982	Introductory meeting at the Midland site to initiate IDV and INPO programs
November 3, 1982	Midland site tour and walkdown of the AFW system
November 4, 1982	TERA project team meetings in Jackson to review Midland project experience (e.g., 50.55e reports, NRC inspection reports, etc.); identi- fication of information needs
November 5, 1982	Meeting of TERA, CPC and Bechtel manage- ment in Ann Arbor to discuss programmatic de- tails of the IDV program, logistics for TERA- Bechtel interaction on the IDV; review of Bechtel organization, interfaces, etc.; identi- fication of information needs

NRC issues meeting summary for October 25, 1982 meeting

TERA issues Revision 0 of the Midland Independent Design and Construction Verification - (IDCV) Project Quality Assurance Plan

CPC approval of TERA Project Quality Assurance Plan

TERA issues draft Engineering Program Plan for interim use and comments

TERA field verification team is on-site conducting physical configuration verification of AFW system piping and supports inside containment

CPC submittal to NRC of response to NRC comments during October 25, 1982 meeting; CPC commits to separate IDV and INPO evaluation, identifies candidate systems for adding an additional system to the IDV scope, expansion of IDV program to include a verification of the quality of construction of the IDV systems; details of IDV interactions and INPO reporting

2

November 11, 1982

November 15, 1982

November 23, 1982

November 29, 1982

November 29 -December 3, 1982

December 3, 1982

Date	Milestone
December 6, 1982	TERA project team meets individually with Bechtel group supervisors and group leaders to give a programmatic overview of the expanded IDCV; identify elements of the design process, interfaces, logistics for conducting the IDCV review; identify information, etc.
December 8-15, 1982	Lead technical reviewers interview Bechtel personnel as part of the IDCV review process; identification of information needs
December 10, 1982	Agreement reached with Bechtel on proprietary information
December 16, 1982	TERA completes Engineering Program Plan
January 17-21, 1983	TERA design review team in Ann Arbor
January 24, 1983	TERA begins ICV program — review of supplier documentation, storage, and maintenance docu- mentation
Januarý 24–26, 1983	TERA construction review team on-site review- ing supplier documentation and storage and maintenance documentation
January 25-27, 1983	TERA design review team in Ann Arbor
February 7-11, 1983	TERA construction review team on-site
February 8, 1983	Public meeting on Midland Construction Com- pletion Program and Independent Design and Construction Verification Program
February 9, 1983	TERA transmits Engineering Program Plan (EPP) and Project Quality Assurance Plan (PQAP) to the NRC
February 17, 1983	TERA issues Revision I of the EPP and Revision 2 of PQAP

11

vi

Date	Milestone	
February 28 - March 4, 1983	TERA construction review team on-site and design review team at Ann Arbor	
February 28, 1983	TERA meeting with B&W in Lynchburg	
March 1, 1983	TERA meets with Bechtel management in Ann Arbor to clarify requests for information	
March 2, 1983	Project team meeting; Ann Arbor	
March 11, 1983	Project quality assurance audit conducted by the Project Quality Assurance Engineer	
March 18, 1983	TERA transmits information to NRC regarding corporate and individual independence, profes- sional qualifications, scope of review, reporting and auditability, and program status	The second se
March 21-25, 1983	TERA construction review team on-site and TERA design raview team at Ann Arbor	
March 22, 1983	NRC selects Standby Electric Power System as the second system and the HVAC system assur- ing control room habitability as the third system for the IDCV program	
March 24, 1953	NRC provides TERA with a service list for Midland IDCV program	
March 28, 1983	NRC issues the protocol for the Midland IDCV program	
March 30, 1983	TERA transmits supplemental information to NRC regarding affidavits of independence and professional qualifications, including additional affidavits by individuals previously employed by NRC	

Date	Milestone
April 8, 1983	Project quality assurance audit report issued by the Project Quality Assurance Engineer
April 9, 1983	Senior Review Team meets to review project status, review OCRs, and develop recommenda- rions for the project team
Apri' 13, 1983	Meeting at NRC, Bethesda, including TERA, CPC, GAP, and NRC. TERA presents synopsis of progress to date of AFW system review, plus discussion of topics to be reviewed for the two additional systems (Standby Electric Power; Control Room HVAC) selected by NRC. All parties discuss protocol for Midland IDCV Pro- gram
April 21, 1983	TERA transmits supplemental information to NRC regarding affidavits of independence for individuals previously employed by NRC
May 3, 1983	NRC letter, Novak to Cook (CPC) stating acceptance of TERA Corporation to conduct IDCV Program and acceptance of Engineering Program Plan for the Auxiliary Feedwater System
May 18, 1983	TERA issues general Revision 2 of the EPP and Revision 3 of the PQAP to incorporate the addition of the Standby Electric Power System and Control Room HVAC System to the IDCV scope, update personnel qualifications, add project instructions and reference new protocol for communications
May 18, 1983	TERA meets with NRC, I&E HQ management to discuss consideration of the Midland IDCV program within NRC's response to the Ford Amendment legislation.
May 27, 1983	TERA issues first Monthly Status Report.

.

Attachment 2

## OCR, FINDING REPORT, AND FINDING RESOLUTION REPORT TRACKING SYSTEM MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM

OCR No.	Resp. LTR	Potential Open Item	Open Item	Confirmed Item	Resolved Item	Finding Report	Finding Resolution Report	Topic	•	Comments
001	RPS	12/21/83	3/4/83	3/4/83		•		1.4-*	Tech Specs	
002	RPS	12/21/83	3/4/83	3/4/83 `				1.4-1	Tech Specs	
003	RPS	1/3/83	3/4/83		3/4/83			1.8-1	Overpressure Protection	
904	RPS	1/3/83	3/4/83		3/4/83			1.8-1	Overpressure Protection	
005	RPS	1/4/83	3/4/83	3/4/83				1.1-1	System Operating Limits	
006	RPS	1/12/83	3/4/83		3/4/83			1.2-1	Accident Analysis Considerations	
007	RPS	1/12/83	3/4/83		3/4/83			1.2-1	Accident Analysis Considerations	
008	LB	1/19/83	3/4/83				•	1.19-1	Control Systems	
009	CS	1/20/83	3/4/83		3/4/83			11.1-1	Selsmic Design	
010	FAD	1/20/83	3/4/83	4/14/63				1.10-1	Hydraulic Design	
011	LB	1/27/83	3/4/83	3/4/83				1.19-1	Control Systems	
012	LB	2/7/83	3/4/83	3/4/83				1.15-1	Power Supplies	
013	RPS	2/8/83	3/4/83					1.5-1	Syst. Align./Switchover	

.

#### OCR, FINDING REPORT, AND FINDING RESOLUTION REPORT TRACKING SYSTEM

٩.

#### MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM

#### (Continued)

OCR No.	Resp. LTR	Potential Open Item	Open Item	Confirmed Item	Resolved Item	, Report	Finding Resolution Report	. Topic		Comments
014	RPS	2/8/83	3/4/83	() · ·				1.5-1	Syst. Align./Switchover	
015	CS	2/10/83	3/4/83					111.1-1	Selsmic Design/Input to Equipment	
016	CS	2/10/83	3/4/83					111.5-1	Civil/Stu Design Consid.	
017	FAD	2/17/83	3/4/83	3/4/83				1.11-1	Heat Removal Cap	
								1.10-1	Hydraulic Design	
018	FAD	2/17/83	3/4/83	3/4/83				1.11-1	Heat Removal Cap.	
019	LB	2/21/83	3/4/83					1.18-1	Instrumentation	
020	FAD	2/24/83	3/4/83	3/4/83				1.11-1	Heat Removal Cap.	
								1.9-1	Comp. Func. Req.	
021	FAD	2/24/83	3/4/83					11.10-1	Eq. Qual.	Rev. 1, 4/14/83
022	LB	2/24/83	3/4/83					1.19-1	Control Syst.	
023	LB	2/28/83	3/4/83					1.18-1	Instrumentation	
								1.19-1	Control	

## OCR, FINDING REPORT, AND FINDING RESALUTION REPORT TRACKING SYSTEM MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM

٩.

#### (Continued)

OCR No.	Resp. LTR	Potential Open Item	Open Item	Confirmed Item	Resolved Item	Finding Report	Finding Resolution Report	Topic		<u>Comments</u>
024	RPS	3/1/83	3/4/83					1.2-1	Acc. Anal. Consid.	
025	RPS	3/1/83	3/4/83	3/4/83				1.2-1	Acc. Anol. Consid.	
026	RPS	3/1/83	3/4/83					1.8-1	Overpress. Prot.	
027	FAD	3/1/83	3/4/83	3/4/83				1.9-1	Comp. Func. Req.	
								11.9-1	Env. Eng.	
028	FAD	3/2/83	3/4/83	4/14/83				1.9-1	Comp. Func. Req.	
029	LB	2/22/83	3/4/83		3/4/83			1.18-1	Instrumentation	
								1.19-1	Control System	
030	LB	1/19/83	3/4/83		3/4/83			1.19-1	Control System	
031	CS	2/11/83	3/4/83	3/4/83				1.3-lc	Pipe Supports	
032	CS	2/11/83	3/4/83	3/4/83				1.3-ic	Pipe Supports	

## OCR, FINDING REPORT, AND FINDING RESOLUTION REPORT TRACKING SYSTEM MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION PROGRAM

..

#### (Continued)

OCRN	<u>lo.</u> <u>Resp. LTR</u>	Potential Open Item	Open Item	Confirmed Item	Resolved	Finding Report	Finding Resolution Report	Tople		Comments
033	cs	2/11/83	3/4/83	3/4/83				1.3-lc	Pipe Supports	
034	CS	2/11/83	3/4/83	3/4/83				1.3-lc	Pipe Supports	
035	cs	2/11/83	3/4/83	3/4/83				1.3-lc	Pipe Supports	Rev. I. 5/25/83
036	CS	2/11/83	3/4/83	3/4/83				11.2-1	Pressure Boundary	Rev. 1, 5/25/83
037	CS	1/20/83	3/4/83	3/4/83				111.1-1	Selsmic Design/Input to Equipment	
038	LB	3/1/83	3/4/83	3/4/83				1.15-1	Power Supplies	같이 많은 것이.
039	LB	3/30/83	4/14/83			1.		11.10-1	Env. Eq. Qual.	
040	LB	3/8/83	4/14/83					1.16-1	Elec. Characteristics	
041	LB	3/25/83	4/14/83					1.15-1	Power Supplies	
042	LB	3/31/83	4/14/83					1.10-1	Env. Eq. Qual.	
043	FAD	3/15/83	4/14/83					1.10-1	System Hydraulic Design	
044	FAD	3/15/83	4/14/83					11.10-1	Env. Eq. Qual.	
045	Tulo	3/17/83	4/14/83 .	5/25/83				II.1-IC	Electrical Equipment/ Storage & Maintenance	
346	Tulo	3/17/83	4/14/83	5/25/83				I.I-IC	Mechanical Equipment/ Storage & Maintenance	

CURRENT CONFIRMED ITEM REPORTS

			EILE NO	3201.000
TYPE OF REPORT: OP			DOC NO. REV. NO	<u>3201-008-C -001</u>
DATES REPORTED TO:	LTR 3/3/83 PRINCIPAL-IN-CH	SRT PROJE	CT TEAM/PROJECT	MGR. 3/3/83
STRUCTURE(S), SYSTEM	(S), OR COMPONEN	T(S) INVOLVED:		
AFW system operabi	ility and surv	eillance requirement	s in Technical	Specifications
IDCV PROGRAM AREA C	OR TASK (IF APPLI	CABLE):		
Topic I.4-1, Techr	nical Specific	ations		
DESCRIPTION OF CONC A commitment made Midland Technical recommendation GS- is not clear that dual valve lineup incorporate a requ	ERN: in response to Specification of regarding to the Technical after mainter uirement for to	to NRC requests has n is. That commitment verification of prope Specifications do i nance. Also, the ass valve lineup verifica	ot been incorpo involved NUREG- r AFW system va ncorporate the ociated draft p tion (See OCR-0	orated into the 0611, Appendix II lve lineup. It means to assure procedure does not 14).
	FRN.			
RECOMMENDATION	OR RESOLU	TION		
RECOMMENDATION	OR RESOLU	TION: ject Quality Assurance	e Plan.	
RECOMMENDATION Process in accorda	OR RESOLU	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accorda	OR RESOLU	ITION ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accorda	OR RESOLU	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accorda	OR RESOLU	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accorda	OR RESOLU ance with Pro:	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accords	OR RESOLU ance with Pro: REGUIRED):	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accorda	OR RESOLU ance with Pro;	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accords	OR RESOLU ance with Pro:	ITION ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accords COMMENTS BY SRT (IF F	COR RESOLU ance with Pro; REGUIRED):	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accorda COMMENTS BY SRT (IF F REFERENCES (INCL. REI FSAR, REV. 47	COR RESOLU ance with Pro; REGUIRED):	TION: ject Quality Assuranc	e Plan.	
RECOMMENDATION Process in accords COMMENTS BY SRT (IF F REFERENCES (INCL. REL FSAF., REV. 47 SIGNATURE(5): DBC	COR RESOLU ance with Pro: REGUIRED):	TION: ject Quality Assurance	e Plan.	
RECOMMENDATION Process in accorda COMMENTS BY SRT (IF F REFERENCES (INCL. REL FSAF., REV. 47 SIGNATURE(S): <u>RPS</u> OCR ITEM REPORT	REGUIRED):	TION: ject Quality Assurance	e Plan.	N/A JWB
RECOMMENDATION Process in accords COMMENTS BY SRT (IF F REFERENCES (INCL. REL FSAR, REV. 47 SIGNATURE(S): RPS OCR ITEM REPORT ORIGINATOR	COR RESOLU ance with Pro: REGUIRED):	REPORT NO.):	e Plan. JWB PRINCIPAL- IN-CHARGE	N/A JWB SRT (IF REQUIRED)
RECOMMENDATION Process in accords COMMENTS BY SRT (IF F REFERENCES (INCL. REL FSAF., REV. 47 SIGNATURE(S): RPS OCR ITEM REPORT ORIGINATOR 3/3/83	REGUIRED): REGUIRED): REGUIRED): LATED OCR ITEM F	REPORT NO.): HAL PROJECT MANAGER FOR PROJECT TEAM 3/4/83	e Plan. JWB PRINCIPAL- IN-CHARGE 3/14/83	N/A JWB SRT (IF REQUIRED)
MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION				
--				
OPEN, CONFIRMED AND RESOLVED (OCR) ITEM REPORT				

•

• \*

TYPE OF REPORT: OPE RES		RMED X	FILE NO. DOC NO. REV. NO.	3201-008 3201-008-C-002 0
DATES REPORTED TO:	LTR 3/3/83 S	RGE 3/7/83 CPC/DE	CT TEAM/PROJECT M	GR. 3/3/83
STRUCTURE(S), SYSTEM( AFW system operabi	s), OR COMPONENT( lity and survei	S) INVOLVED: illance requirement:	s in Technical S	pecifications.
IDCV PROGRAM AREA O	R TASK (IF APPLICA	BLE):		
Topic I.4-1, Techn	ical Specificat	tions		
DESCRIPTION OF CONCE	RN:			
Midland Technical in that:	Specifications	do not meet NRC B&N	W Standard i	al Specificatio
An action sta systems are i	tement is neede noperable.	ed to require immed	iate action if b	oth AFW
	EBN-			
		•		
RECOMMENDATION X	OR RESOLUTION	ONt		
Process in accorda	nce with Project	ct Quality Assurance	e Plan.	
COMMENTS BY SRT (IF F	EQUIRED):			
REFERENCES (INCL. REI	ATED OCR ITEM RE	PORT NO.):		
REFERENCES (INCL. REL Midland Technical S	ATED OCR ITEM RE	PORT NO.): (Rev.33) in FSAR; N	UREG-0103, REV.	4, FALL 1980
REFERENCES (INCL. REL Midland Technical S SIGNATURE(S): RPS	ATED OCR ITEM RE pecifications RPS	PORT NO.): (Rev.33) in FSAR; N HAL	UREG-0103, REV. JWB	4, FALL 1980 N/A JWB
REFERENCES (INCL. REL Midland Technical S SIGNATURE(S): RPS OCR ITEM REPORT ORIGINATOR	ATED OCR ITEM RE Specifications RPS LTR	PORT NO.): (Rev.33) in FSAR; N HAL PROJECT MANAGER FOR PROJECT TEAM	UREG-0103, REV. JWB PRINCIPAL- IN-CHARGE	4, FALL 1980 N/A JWB SRT (IF REQUIRED)
REFERENCES (INCL. REL Midland Technical S SIGNATURE(S): RPS OCR ITEM REPORT ORIGINATOR 	ATED OCR ITEM RE specifications RPS LTR 3/3/83	PORT NO.): (Rev.33) in FSAR; N HAL PROJECT MANAGER FOR PROJECT TEAM 3/4/83	UREG-0103, REV. JWB PRINCIPAL- IN-CHARGE 3/14/83	4, FALL 1980 N/A JWB SRT (IF REQUIRED)

TYPE OF REPORT: CP	SOLVED COM	ITEM	FILE NO. DOC NO. REV. NO.	<u>3201-008</u> <u>3201-008-C - 005</u> 0
DATES REPORTED TO:	LTR 3/3/83 PRINCIPAL-IN-CH	SRT PROJEC	T TEAM/PROJECT A	AGR. 3/3/83
STRUCTURE(S), SYSTEM Entire AFW system	n(S), OR COMPONED	VT(S) INVOLVED:		
IDCV PROGRAM AREA	OR TASK (IF APPLI	CABLE):		
Topic I.1-1, Syst	tem Operating	Limits		
DESCRIPTION OF CONC	ERN:			
requirements and and 0-028 also ap	other design pply.	parameters. OCRs C-0	ard to AFW syst 17, C-018, C-02	tem flowrate 20, C-027
	CERNA			
Nuclear steam sup be adequately or	oply system pe consistently	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not asign.
Nuclear steam sup be adequately or	oply system pe consistently	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not sign.
Nuclear steam sup be adequately or	X OR RESOLU	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not sign.
Nuclear steam sup be adequately or RECOMMENDATION	X OR RESOLU	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not sign.
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU dance with Pro	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not sign.
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU dance with Pro	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not sign.
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU dance with Pro	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not sign.
Nuclear steam sup be adequately or RECOMMENDATION Process in accord	X OR RESOLU Ance with Pro	rformance requirement reflected in the bala	s for the AFW s nce of plant de	system may not sign.
Nuclear steam sup be adequately or RECOMMENDATION Process in accord COMMENTS BY SRT (IF ) REFERENCES (INCL. RE SAR, REV. 47; B&W ICRS	X OR RESOLU Ance with Pro REQUIRED): LATED OCR ITEM I BOP Criteria	rformance requirement reflected in the bala /TION: ject Quality Assuranc	e Plan. REV. 01 (6/25/	system may not asign. (82)
Nuclear steam sup be adequately or RECOMMENDATION Process in accord COMMENTS BY SRT (IF ) REFERENCES (INCL. RE SAR, REV. 47; B&W DCRS SIGNATURE(S):	X OR RESOLU Ance with Pro REGUIRED): LATED OCR ITEM I I BOP Criteria	rformance requirement reflected in the bala /TION: ject Quality Assuranc	s for the AFW s nce of plant de e Plan. REV. 01 (6/25/	system may not esign. (82)
RECOMMENDATION Process in accord COMMENTS BY SRT (IF I REFERENCES (INCL. RE SAR, REV. 47; B&V CRS SIGNATURE(S): RPS OCR ITEM REPORT	X OR RESOLU A OR RESOLU dance with Pro REQUIRED): LATED OCR ITEM I N BOP Criteria RPS	REPORT NO.): Document 36-1004477,	s for the AFW s nce of plant de e Plan. REV. 01 (6/25/ JWB	N/A JWB
RECOMMENDATION Process in accord COMMENTS BY SRT (IF I REFERENCES (INCL. RE SAR, REV. 47; B&V CRS SIGNATURE(S): RPS OCR ITEM REPORT ORIGINATOR	X OR RESOLU Ance with Pro REQUIRED): LATED OCR ITEM I N BOP Criteria RPS LTR	rformance requirement reflected in the bala	s for the AFW s nce of plant de e Plan. REV. 01 (6/25/ JWB PRINCIPAL- IN-CHARGE	(82) N/A JWB SRT (IF REQUIRED

			FUEN	0 0001 000
TYPE OF REPORT:	OPEN CON RESOLVED	FIRMED X	DOC N REV. N	0. <u>3201-008</u> C_010 10
DATES REPORTED TO	PRINCIPAL-IN-CH	ARGE 416 83 C	ROJECT TEAM/PROJECT	MGR. 3/3/83
STRUCTURE(S), SYSTE	EM(S), OR COMPONEN	T(S) INVOLVED:		
AFW - piping and	d valves			
IDCV PROGRAM ARE	A OR TASK (IF APPLIC	ABLE):		
Hydraulic Design	n (I.10-1)			
DESCRIPTION OF CO	NCERN:			
In calculation storage to serve was full of wate coming available	the volume of wa ice water suctio er. However, th e because of the	ter available du n source it was a e water might le lack of Category	ring the transfer assumed that all us ak out prior to the y I check valves.	from the condensat Category I piping he service water be
The recommendat: AFW pumps could	ion of OCR-3201- have a loss of	008-0-010 was im suction during s	olemented. It was witchover to serv	s determined that t ice water.
of events it may	y be possible fo	r the AFW pump to	lose suction.	and some sequence
of events it may The AFW pumps co	y be possible fo buld be damaged	r the AFW pump to by running dry.	o lose suction.	
of events it may The AFW pumps control RECOMMENDATION	y be possible fo ould be damaged X OR RESOLUT	by running dry.	o lose suction.	
of events it may The AFW pumps control RECOMMENDATION_ 1. Process per	y be possible fo buld be damaged X OR RESOLUT PQAP.	by running dry.	o lose suction.	
of events it may The AFW pumps control RECOMMENDATION_ 1. Process per 2. Review seism analysis of upstream of	y be possible fo ould be damaged <u>X</u> OR RESOLUT PQAP. mic analysis of the switchover Category I/non-	r the AFW pump to by running dry.	o lose suction. D evaluate assump that credit can face.	tion in Bechtel's be taken for piping
of events it may The AFW pumps control RECOMMENDATION_ 1. Process per 2. Review seism analysis of upstream of COMMENTS BY SRT (IF	y be possible fo buld be damaged <u>X</u> OR RESOLUT PQAP. nic analysis of the switchover Category I/non-	r the AFW pump to by running dry.	o lose suction.	tion in Bechtel's be taken for piping
of events it may The AFW pumps control RECOMMENDATION 1. Process per 2. Review seison analysis of upstream of COMMENTS BY SRT (IF	y be possible fo buld be damaged <u>X</u> OR RESOLUT PQAP. nic analysis of the switchover Category I/non-	r the AFW pump to by running dry.	o lose suction.	tion in Bechtel's be taken for piping
of events it may The AFW pumps control RECOMMENDATION 1. Process per 2. Review seison analysis of upstream of COMMENTS BY SRT (IF	y be possible fo buld be damaged <u>X</u> OR RESOLUT PQAP. nic analysis of the switchover Category I/non- FREGUIRED):	r the AFW pump to by running dry.	o lose suction.	tion in Bechtel's be taken for piping
of events it may The AFW pumps control RECOMMENDATION 1. Process per 2. Review seiss analysis of upstream of COMMENTS BY SRT (IF REFERENCES (INCL. F	y be possible fo buld be damaged <u>X</u> OR RESOLUT PQAP. nic analysis of the switchover Category I/non- F REGUIRED):	r the AFW pump to by running dry. TION; suction piping to to service water Category I inter EPORT NO.):	o lose suction.	tion in Bechtel's be taken for piping
of events it may The AFW pumps control RECOMMENDATION 1. Process per 2. Review seison analysis of upstream of COMMENTS BY SRT (IF REFERENCES (INCL_R SIGNATURE(S):	y be possible fo buld be damaged <u>X</u> OR RESOLUT PQAP. nic analysis of the switchover Category I/non- F REGUIRED):	r the AFW pump to by running dry. TION: suction piping to to service water Category I inter EPORT NO.):	o lose suction.	tion in Bechtel's be taken for piping
of events it may The AFW pumps con- RECOMMENDATION 1. Process per 2. Review seison analysis of upstream of COMMENTS BY SRT (IF REFERENCES (INCL. R SIGNATURE(S): FAD OCR ITEM RESOLUTION	y be possible fo build be damaged <u>X</u> OR RESOLUT PQAP. nic analysis of the switchover Category I/non- FREQUIRED): RELATED OCR ITEM R	r the AFW pump to by running dry. TION; suction piping to to service water Category I inter EPORT NO.):	o lose suction.	tion in Bechtel's be taken for piping
of events it may The AFW pumps control RECOMMENDATION 1. Process per 2. Review seist analysis of upstream of COMMENTS BY SRT (IF REFERENCES (INCL. R SIGNATURE(S): FAD OCR ITEM REPORT ORIGINATOR	y be possible fo build be damaged X OR RESOLUT PQAP. nic analysis of the switchover Category I/non- F REGUIRED): RELATED OCR ITEM R FAD LTR	r the AFW pump to by running dry. TION: suction piping to to service water Category I inter EPORT NO.J: PROJECT MANAGE FOR PROJECT TEA	PRINCIPAL- INCHARCE	tion in Bechtel's be taken for piping : : SRT (IF REQUIRED)
of events it may The AFW pumps control RECOMMENDATION 1. Process per 2. Review seist analysis of upstream of COMMENTS BY SRT (IF REFERENCES (INCL. R SIGNATURE(S): FAD OCR ITEM REPORT ORIGINATOR 3/29/83	y be possible fo build be damaged X OR RESOLUT PQAP. nic analysis of the switchover Category I/non- F REQUIRED): RELATED OCR ITEM R FAD LTR 3/29/83	r the AFW pump to by running dry. TION	R PRINCIPAL- M PRINCIPAL- IN-CHARGE	tion in Bechtel's be taken for piping : : SRT (IF REQUIRED)

OF	PEN, CONFIRMI	ED AND RESOLVED (O	CR) ITEM REPO	RT
TYPE OF REPORT: OP RE	EN CON	FIRMED X	FILE NO. DOC NO. REV. NO	<u>3201-008</u> <u>3201-008-C-011</u> 0
DATES REPORTED TO:	LTR 3/4/83 PRINCIPAL-IN-CH	SRT PROJEC	CT TEAM/PROJECT	MGR. 3/4/83
STRUCTURE(S), SYSTEM	(S), OR COMPONEN	T(S) INVOLVED:		
AFW "Feed Only I	Good Generator	" (FOGG) Control		
IDCV PROGRAM AREA	R TASK (IF APPLIC	ABLE):		
Topic I.19-1, Con	ntrol Systems			
The B&W BOP critic control for FOGG The FOGG interlo controllable from	ERN: eria document be available cks are contro m the Auxiliar	(36-1004477-01- Draf at both the MCR and llable (invertable) y Shutdown Panel.	t) section 3.12 the Auxiliary S from the MCR bu	2 requires that Shutdown Panel. It are not
•				
SIGNIFICANCE OF CONC	ERN:			
B&W BOP criteria not met.	regarding con	trol of FOGG from Au	xiliary Shutdow	n Panel are
B&W BOP criteria not met.	regarding con	trol of FOGG from Au	xiliary Shutdow	n Panel are
B&W BOP criteria not met.	regarding con	trol of FOGG from Au:	xiliary Shutdow	n Panel are
B&W BOP criteria not met. RECOMMENDATION	OR RESOLUT	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel s	COR RESOLUT	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel :	OR RESOLUT	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel :	C OR RESOLUT	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel :	COR RESOLUT	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel :	COR RESOLUT OR RESOLUT firms concern should be revis	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel :	COR RESOLUT OR RESOLUT firms concern should be revi	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel :	COR RESOLUT OR RESOLUT firms concern should be revis	trol of FOGG from Au:	xiliary Shutdow	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel s	COR RESOLUT COR RESOLUT firms concern should be revi	trol of FOGG from Au:	hat design inte	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel : COMMENTS BY SRT (IF F	COR RESOLUT firms concern should be revia	trol of FOGG from Au:	hat design inte	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel s COMMENTS BY SRT (IF F	COR RESOLUT COR RESOLUT firms concern should be revia	TION: and has determined thewed further.	hat design inte	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel : COMMENTS BY SRT (IF F	COR RESOLUT firms concern should be revia	TION: and has determined thewed further.	hat design inte	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel s COMMENTS BY SRT (IF F REFERENCES (INCL. REI SIGNATURE(S):	COR RESOLUT firms concern should be revia	trol of FOGG from Au:	hat design inte	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel : COMMENTS BY SRT (IF F REFERENCES (INCL. REI SIGNATURE(S): RPS	COR RESOLUT firms concern should be revia REQUIRED): ATED OCR ITEM R RPS	TION: and has determined thewed further. EPORT NO.):	hat design inte	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel : COMMENTS BY SRT (IF F COMMENTS BY SRT (IF F SIGNATURE(S): REFERENCES (INCL. REI SIGNATURE(S): REFERENCES (INCL. REI OCR ITEM REPORT ORIGINATOR	COR RESOLUT firms concern should be revia	TION: and has determined thewed further. EPORT NO.): HAL PROJECT MANAGER	JWB PRINCIPAL-	erface between
B&W BOP criteria not met. RECOMMENDATION Project team cont B&W and Bechtel : COMMENTS BY SRT (IF F COMMENTS BY SRT (IF F SIGNATURE(S): REFERENCES (INCL. REI SIGNATURE(S): RPS OCR ITEM REPORT ORIGINATOR 3/4/83	COR RESOLUT firms concern should be revia REQUIRED): LATED OCR ITEM R RPS LTR 3/4/92	TION: and has determined thewed further. EPORT NO.): HAL PROJECT MANAGER FOR PROJECT TEAM	JWB PRINCIPAL- IN-CHARGE	erface between

OF	EN, CONFIRME	DESIGN AND CONST AND RESOLVED (O	RUCTION VERIFICE) ITEM REPO	RT
TYPE OF REPORT: OPI RE:		RMED X	FILE NO. DOC NO. REV. NO.	<u>3201-008</u> <u>3201-008</u> -C_012 0
DATES REPORTED TO:	LTR 2/7/83 PRINCIPAL-IN-CHAI	RGE 3/7/83 PROJEC	T TEAM/PROJECT N	MGR. 3/3/83
STRUCTURE(S), SYSTEM	(S), OR COMPONENT(	S) INVOLVED:		
IDCV PROGRAM AREA C	R TASK (IF APPLICA	BLE):		
Topic I.15-1, Pov	ver Supplies			
require that the blackout condition and BA are powers valves 2MO-3277A loss of AFW funct	AFW system be on on (loss of all ed from Class le and B to shut, tion during blac	AC). The FOGG inte AC). The FOGG inte AC (lost during bl cutting off steam t ckout.	for two hours rlock relays f ackout). This to the AFW turb	in a station for channel AA would cause tine and causing
SIGNIFICANCE OF CONC	ERN:			
The AFW system ma	ay not be functi	ional during station	blackout cond	litions.
The AFW system ma	ay not be functi	ional during station	blackout cond	litions.
The AFW system ma	x OR RESOLUTION	ional during station	blackout cond	litions.
The AFW system ma RECOMMENDATION Although limited AFW, a systematic conditions.	X OR RESOLUTION Failure Modes E analysis shoul	Ional during station	EAs) have been siders all app	performed on licable plant
The AFW system ma RECOMMENDATION Although limited AFW, a systematic conditions.	X OR RESOLUTION Failure Modes E analysis shoul	Ional during station	EAs) have been siders all app	performed on licable plant
The AFW system ma RECOMMENDATION Although limited AFW, a systematic conditions.	X OR RESOLUTION Failure Modes E analysis shoul	Ional during station	EAs) have been siders all app	performed on licable plant
The AFW system ma RECOMMENDATION Although limited AFW, a systematic conditions. COMMENTS BY SRT (IF F REFERENCES (INCL. REL OCR 3201-008-0-03 Drawings E-1580 S	X OR RESOLUTION Failure Modes E analysis shoul REQUIRED):	Ional during station	EAs) have been siders all app	performed on licable plant
The AFW system ma RECOMMENDATION Although limited AFW, a systematic conditions. COMMENTS BY SRT (IF F  REFERENCES (INCL. REI OCR 3201-008-0-03 Drawings E-1580 S SIGNATURE(S): LB	X OR RESOLUTION Failure Modes E analysis shoul REGUIRED): ATED OCR ITEM RED 8 & C-038 5H41, 42, 24, 25	ON: CON: Effects Analyses (FM Id be done which con  PORT NO.): 5 HAI	EAs) have been siders all app	performed on licable plant
The AFW system ma RECOMMENDATION Although limited AFW, a systematic conditions. COMMENTS BY SRT (IF F COMMENTS BY SRT (IF F COMMEN	X OR RESOLUTION Failure Modes E analysis shoul REQUIRED): LATED OCR ITEM REP 8 & C-038 SH41, 42, 24, 25 LB LTR	ON: CON: Effects Analyses (FM Id be done which con  PORT NO.): 5 HAL PROJECT MANAGER FOR PROJECT TEAM	EAs) have been siders all app JWB PRINCIPAL- IN-CHARGE	N/A JWB SRT (IF REQUIRED)

## MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION OPEN, CONFIRMED AND RESOLVED (OCR) ITEM REPORT

-

TYPE OF REPORT: OPEN RESO		TRMED X	-	FILE NO. DOC NO. BEV. NO.	<u>3201-008</u> <u>3201-008</u> C - 017
DATES REPORTED TO: L	TR 3/3/83	SRT	PROJECT CPC/DES	TEAM/PROJECT	MGR. 3/3/83
STRUCTURE(S), SYSTEM(S AFW Pumps	), OR COMPONENT	(S) INVOLVED:			
IDCV PROGRAM AREA OR Component Function System Heat Remova	TASK (IF APPLICA al Requirement 1 Capability	ABLE): ts (I.9-1) (I.11-1)	System H (Criteri	ydraulic Desi a & Commitmen	gn (I.10-1) ts/Review of Calcs
DESCRIPTION OF CONCEI	RN:				
Rev. 1, (Ref. 1) 1 (Ref. 2) requires value, although (a with appropriate d water to remove the Document (i.e. 30	ists values of B50 gpm and a s reported in esign paramete e heat being s sec after rea	f 500 gpm an B&W calcula other OCRs) ers. The 85 generated at ctor trip).	d 720 gpm tion (Ref this cal 0 gpr fig the time	. The B&W BO . 3) is consis- culation may ure may not p specified in	P Criteria Documen stent with this not be consistent rovide enough the B&W Criteria
	RN.				
RECOMMENDATION X	OR RESOLUT	10N :			
Process per Project	t Quality Assi	urance Plan.			
			Sec. Sec.		
COMMENTS BY SRT (IF RE	QUIRED):				
REFERENCES (INCL. REL	TED OCR ITEN OF	PORT NO. (	Concord	ual Decies Ct	udy for Auxilian
	ad Data Contra	al for REW	77-Fuel A	Luar Design St	udy for Auxiliary
Feedwater System Fe (2) BOP Criteria - 32-0525, Rev. 00.	Aux Feedwater	Sys (36-100	04477, Rev	(.1). (3) B&W	AFW Calculation
Feedwater System Fe (2) BOP Criteria - 32-0525, Rev. 00. SIGNATURE(S): FD	Aux Feedwater	Sys (36-100	04477, Rev	JWB	AFW Calculation
Feedwater System Fe (2) BOP Criteria - 32-0525, Rev. 00. SIGNATURE(S): FD OCR ITEM REPORT ORIGINATOR	Aux Feedwater FD LTR	HAL FOR PROJECT MA	ANAGER	JWB PRINCIPAL- IN-CHARGE	AFW 1612, Rev. AFW Calculation N/A JWB SRT (IF REQUIRED)
Feedwater System Fe (2) BOP Criteria - 32-0525, Rev. 00. SIGNATURE(S): FD OCR ITEM REPORT ORIGINATOR 3/3/83	FD Aux Feedwater FD LTR 3/3/83	HAL PROJECT MA FOR PROJEC 3/4/	ANAGER	JWB PRINCIPAL- IN-CHARGE 3/14/83	AFW Calculation N/A JWB SRT (IF REQUIRED)

C	PEN, CONFIRM	MED AND RESOLVED (O	CR) ITEM REPO	RT
TYPE OF REPORT: O	PENCO		FILE NO. DOC NO. REV. NO	<u>3201-008</u> <u>3201-008-C -018</u>
DATES REPORTED TO	ELTR 3/3/83 PRINCIPAL-IN-C	SRT PROJE	CT TEAM/PROJECT	MGR. 3/3/83
STRUCTURE(S), SYSTE AFW System (ge	M(S), OR COMPONE eneral)	NT(5) INVOLVED:		
IDCV PROGRAM AREA	OR TASK (IF APPL	ICABLE):		
System heat re	emoval capabil	ity (I.11-1)		
be capable of decay heat cur heat curve whe the method of ment BAW 1612 Reference 3.	removing. The reas FSAR page the NRC's Brar (Ref. 4) uses Ref. 3 require	AFW calculation perf a 10A-17 item (e) stat a 10.4-37 states that a 10.4-37 states that a 10.4-37 states that a 10.4-37 states that a 20% margin to be a 20% margin to be	ormed by B&W (I es that 1.0 x) the design is APCSB 9.2 (Ref % which is cons added to the AM	Ref. 1) uses a B& ANS 5.1 (Ref. 2) in conformance wi f. 3). B&W Docu- sistent with NS curve. The ac
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result	ad used for an ated heat remo	malysis is less than t oval requirement will ng the AFW pumps.	he ANS curve (F be too low and	Ref. 2) plus could conse-
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result	ACERN: design to ad used for an ated heat remo in undersizin	ution:	he ANS curve (F be too low and	Ref. 2) plus could conse-
SIGNIFICANCE OF COM If the heat lo 20% the calcul quently result RECOMMENDATION_ Process per Pro	ACERNE design to ad used for an ated heat remo in undersizin X OR RESOLU- oject Quality	ution:	he ANS curve (F be too low and	Ref. 2) plus could conse-
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result RECOMMENDATION Process per Pro	ACERNE design to ad used for an ated heat remo in undersizin X OR RESOLU	ution:	he ANS curve (F be too low and	Ref. 2) plus could conse-
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result RECOMMENDATION Process per Pro	ACERNE design to ad used for an ated heat remo in undersizin X OR RESOLU- oject Quality	ution:	he ANS curve (F be too low and	Ref. 2) plus could conse-
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result RECOMMENDATION Process per Pro	ACERN: design to ad used for an ated heat remo in undersizin X OR RESOLU oject Quality REQUIRED):	UTION:	he ANS curve (F be too low and	Ref. 2) plus could conse-
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result RECOMMENDATION Process per Pro	ACERN: design to ad used for an ated heat remo in undersizin X OR RESOLU oject Quality	UTION:	he ANS curve (F be too low and	Ref. 2) plus could conse-
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result RECOMMENDATION Process per Pro Process per Pro COMMENTS BY SRT (IF REFERENCES (INCL. RI (2) American Nucle APCSB 9.2. (4) 880	ACERN: design to ad used for an ated heat remo in undersizin X OR RESOLU oject Quality REQUIRED): ELATED OCR ITEM ear Society St W 1612(Rev. 1)	Assumance Plan REPORT NOJ: (1) B&W Ca andard 5.1-1979. (3) I , Conceptual Design Si	alculation for NRC Branch Tech	AFW 32-0525, Rev.
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result RECOMMENDATION Process per Pro Process per Pro COMMENTS BY SRT (IF REFERENCES (INCL. RI (2) American Nucle APCSB 9.2. (4) B&V SIGNATURE(S): FAD	ACERN: design to ad used for an ated heat remo in undersizin X OR RESOLU oject Quality REQUIRED): ELATED OCR ITEM ear Society St N 1612(Rev. 1) FAD	Assis is not clearly i halysis is less than t byal requirement will ng the AFW pumps. UTION: Assurance Plan REPORT NO.J: (1) B&W Ca andard 5.1-1979. (3) I , Conceptual Design Si HAL	Alculation for WRC Branch Tech tudy.	AFW 32-0525, Rev. nical Position
SIGNIFICANCE OF CON If the heat lo 20% the calcul quently result RECOMMENDATION Process per Pro Process per Pro COMMENTS BY SRT (IF REFERENCES (INCL. RI (2) American Nucle APCSB 9.2. (4) B&V SIGNATURE(S): FAD OCR ITEM REPORT ORIGINATOR 3/2/92	ACERN: design to ad used for an ated heat remo- in undersizin X OR RESOLU oject Quality REQUIRED): ELATED OCR ITEM ear Society St M 1612(Rev. 1) FAD LTR	Assis is not clearly i halysis is less than t byal requirement will ng the AFW pumps. UTION: Assurance Plan REPORT NO.J: (1) B&W Ca andard 5.1-1979. (3) I , Conceptual Design Si HAL PROJECT MANAGER FOR PROJECT TEAM	Alculation for NRC Branch Tech tudy. JWB PRINCIPAL- IN-CHARGE	AFW 32-0525, Rev. nical Position

MIDLAND	INDEPENDEN EN, CONFIRM	IT DESIGN AND CONST ED AND RESOLVED (C	RUCTION VERI	FICATION
TYPE OF REPORT: OPE RES	NCON	FIRMED X	FILE NO. DOC NO. REV. NO	<u>3201-008</u> <u>3201-008-</u> C- 020
DATES REPORTED TO:	LTR 3/3/83 PRINCIPAL-IN-CH	SRT PROJE	CT TEAM/PROJECT	MGR
STRUCTURE(S), SYSTEM	S), OR COMPONEN	T(S) INVOLVED:		
AFW System (general	1)			
IDCV PROGRAM AREA O System Heat Removal Component Functiona	R TASK (IF APPLIC Capability al Requirement	CABLE): System (I.11-1) ts (I.9-1)	Hydraulic Des	ign (I:10-1)
AFW analyses. The Ba comperature for AFW AFW Control System" (section 2.1) makes FSAR contains analysis Bechtel calculation *(Document #36-1004)	W criteria* system design document (410 use of a 100 ses indicating FM 4117-28 us 177 Rev 1)	(section 2.14) requir n. B&W's "Specific D DO) describes 90°F as °F value in calculati g a maximum service w ses a max. SW tempera	the use of 96 design Criteria "typical". B/ ng minimum flow ater temperatur ture of 108°F.	D°F inlet water for Safety Grade AW 1612, Rev. 1 W requirements. Th re of 105°F.
basis and the sizing	g of component	tem neat ramoval capa ts.	bility, its hyd	iraulic design
RECOMMENDATION	OR RESOLU			
Process per Project	Quality Assur	rance Plan		
COMMENTS BY SRT (IF R	EQUIRED):			
REFERENCES (INCL. REL (2) B&W Balance o Design Study (BAW Control System (8)	ATED OCR ITEM F f Plant Crite -1612, Rev. 1). 6-1119130. da	REPORT NO.): (1) Bechter ria for AFW (36-1004) (4) B&W Specific De ted 4/80)	el Calculation 477,Rev.Ol). (3 sign Criteria f	FM4117-28 (Rev.O). ) B&W Conceptual or Safety Grade AFW
SIGNATURE(S):	EAD	UAL	11.10	N /A 110
OCR ITEM REPORT	LTR	PROJECT MANAGER	PRINCIPAL-	SRI (IF REQUIRED)
3/3/92	3/3/03	FOR PROJECT TEAM	IN-CHARGE	
3/3/03	3/ 3/ 61	1/4/91	3/1//02	

OF				
TYPE OF REPORT: OPI	EN CON	ITEM	FILE NO. DOC NO. REV. NO.	3201-008 3201-008- C- 025
DATES REPORTED TO:	LTR 3/3/83 PRINCIPAL-IN-CH	SRT PROJEC	T TEAM/PROJECT N	AGR. 3/3/83
STRUCTURE(S), SYSTEM	(S), OR COMPONEN	T(S) INVOLVED:		
AFW system operabi may function in de	lity under pos trimental mann	stulated accident conc ner	litions - "FOGG	" system
IDCV PROGRAM AREA C	R TASK (IF APPLIC	CABLE):		
Topic I.2-1, Accid	ent Analysis (	Considerations		
mental manner under offsite power. It only because FOGG based upon a delta action, the steam- leaking primary co- mentioned) to "inv ture is recognized	r conditions of s design would logic directs pressure mean driven pump co olant. The FS ert" FOGG and & mitigated	of steam generator tub d force it to direct if feed to the steam gen surement between thw if buld be flooded and re SAR analysis assumes of send flow to good gen in sufficient time.	be failure foll feed to the "ba herator with th two SGs. Witho endered inopera operator action herator such th The basis for t	owed by loss of d" steam generate higher pressure ut prompt operate ble as a result of (no time delay at the SG tube re his assumption is
SIGNIFICANCE OF CONC	ERN. not clean	r. With a single fail	lure of the mot	or driven AFW put
Failure of operato (taking into accou	r to take act nt single fai	nay be rendered inopen ion quickly could resulure).	rable. ult in total lo	ss of AFW
Failure of operato (taking into accou	r to take act nt single fai	nay be rendered inopen ion quickly could resu lure).	rable. ult in total lo	ss of AFW
Failure of operato (taking into accou RECOMMENDATION	r to take act nt single fai	ion quickly could resulure).	able. Ult in total lo	ess of AFW
Failure of operato (taking into accou RECOMMENDATION	r to take act nt single fai	ion quickly could resulure).	rable. ult in total lo	oss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW r r to take act nt single fai X OR RESOLU	ion quickly could resulure).	Plan.	oss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW r r to take act nt single fai X OR RESOLU nce with Proj	ion quickly could resulure).	rable. ult in total lo Plan.	ess of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW or to take act nt single fai	ion quickly could resulure).	rable. ult in total lo Plan.	ss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW r r to take act nt single fai X OR RESOLU	ion quickly could resulure).	<u>able.</u> It in total lo Plan.	ess of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW r r to take act nt single fai <u>X</u> OR RESOLU nce with Proj	ion quickly could resulure).	rable. ult in total lo Plan.	ess of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW r r to take act nt single fai X OR RESOLU nce with Proj	ion quickly could resulure).	nable. Alt in total lo	ess of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW r r to take act nt single fai X OR RESOLU nce with Proj	ion quickly could resulure).	nable. ult in total lo Plan.	ess of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	all AFW r r to take act nt single fai X OR RESOLU nce with Proj	ion quickly could resulure).	Plan.	iss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda	All AFW of the action of the single failed of the s	ion quickly could resulure).	rable. ult in total lo Plan.	iss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda COMMENTS BY SRT (IF F	ATED OCR (TEM)	<pre>ion quickly could resulure). ITION: ect Quality Assurance</pre>	Plan.	iss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda COMMENTS BY SRT (IF F	All AFW or to take act nt single fai X OR RESOLU NCE with Proj REQUIRED): LATED OCR (TEM )	nay be rendered inopen ion quickly could resulure). THON: ect Quality Assurance	Plan.	iss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda COMMENTS BY SRT (IF F REFERENCES (INCL. REI Topic 1.2-1 Engine SIGNATURE(S):	All AFW r r to take act nt single fai X OR RESOLU NCE with Proj REGUIRED):	nay be rendered inopen ion quickly could resulure). THON: ect Quality Assurance	<pre>rable. ult in total lo Plan</pre>	iss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda COMMENTS BY SRT (IF F REFERENCES (INCL. RE Topic 1.2-1 Engine SIGNATURE(S): RPS	all AFW mention of the single failed of the single	In quickly could result ion quickly could result ure). THON: ect Quality Assurance ect Quality Assurance 	<pre>rable. ult in total lo Plan JWB</pre>	nss of AFW
Failure of operato (taking into accou RECOMMENDATION Process in accorda COMMENTS BY SRT (IF F COMMENTS BY SRT (IF F REFERENCES (INCL. REI Topic 1.2-1 Engine SIGNATURE(S): RPS OCR ITEM REPORT ORIGINATOR 2/2/22	ATED OCR (TEM ) REQUIRED): LATED OCR (TEM ) ering Evaluat RPS LTR 2/2/02	In quickly could result ion quickly could result ure). THON: ect Quality Assurance ect Quality Assurance 	Plan.	N/A JWB

0	PEN, CONFIRM	ED AND RESOLVED (C	CR) ITEM REPO	RT
TYPE OF REPORT: OF	ENCO	NFIRMED X	FILE NO DOC NO REV. NO	<u>3201-008</u> <u>3201-008</u> C_ 027 0
DATES REPORTED TO:	LTR 3/3/83 PRINCIPAL-IN-CH	SRT PROJE	CT TEAM/PROJECT	MGR. 3/3/83
STRUCTURE(S), SYSTEM AFW (general)	A(S), OR COMPONE	NT(S) INVOLVED:		
IDCV PROGRAM AREA Component Functio Environmental Env	OR TASK (IF APPL) onal Requireme velopes (II.9-	CABLE): nts (I.9-1) 1)		
DESCRIPTION OF CONC (a) 2452 MWt - li core inventories analysis.	CERN: The FSA cense power 1 for accident	R contains references evel, (b) 2552 MWt - analyses, (c) 2603 MW	to the follow power level for t - power level	ing power levels: r calculation of for containment
The 2552 MWt powe 102% of 2552. FS used for AFW anal	er was used in AR page 10A-1 ysis. Thus t	the B&W AFW calculat 7 (Item a) states tha he power level for AF	ion (Ref. 1). t 102% of maxim W analysis shou	The 2603 MWt is num power level i ald be 2603 MWt.
If 2552 MWt was u underestimated co resulting in unde need to be perfor	mpared to the rsizing of AFI med at 2603 M	heat load associated & components. Furthe Wt.	with operation rmore, other ar	at 2603 MWt alyses may
If 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION	mpared to the rsizing of AFI med at 2603 M	heat load associated w components. Furthe wt.	with operation rmore, other ar	at 2603 MWt halyses may
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje	X OR RESOLU	heat load associated w components. Furthe wt.	with operation more, other ar	at 2603 MWt halyses may
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje	X OR RESOLU	heat load associated A components. Furthe At. JTION: surance Plan.	with operation rmore, other ar	at 2603 MWt halyses may
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje	X OR RESOLU	heat load associated components. Furthe wt.	with operation rmore, other ar	at 2603 MWt halyses may
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje PIC	X OR RESOLU	heat load associated A components. Furthe At. JTION: surance Plan.	with operation rmore, other ar	at 2603 MWt halyses may
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje PIC COMMENTS BY SR# (IF )	X OR RESOLU OR RESOLU Ct Quality As:	heat load associated w components. Furthe wt.	with operation rmore, other ar	at 2603 MWt halyses may
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje PIC COMMENTS BY SRT (IF I Before doing any rationale for the used with project	X OR RESOLU REGUIRED): confirmatory / use of 2552 M manager and f	AFW flow requirements Wt by B&W, and disc PIC.	analyses, dete uss core power	ermine the level to be
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje PIC COMMENTS BY SRI (IF I Before doing any rationale for the used with project	X OR RESOLU REQUIRED): confirmatory / use of 2552 M manager and f	AFW flow requirements Wt by B&W, and disc JWB	analyses, dete	ermine the level to be
IT 2552 MWt was u underestimated co resulting in unde need to be perfor RECOMMENDATION Process per Proje PIC COMMENTS BY SR# (IF I Before doing any rationale for the used with project REFERENCES (INCL. RE ef 1: B&W AFW Calc	X OR RESOLU REQUIRED): confirmatory / use of 2552 M manager and f LATED OCR ITEM F culation 32-05	AFW flow requirements Multiple Second Secon	analyses, dete	ermine the level to be
PIC PIC Process per Proje Process per Proje Process per Proje Pic COMMENTS BY SR# (IF I Before doing any rationale for the used with project REFERENCES (INCL. RE ef 1: B&W AFW Calc SIGNATURE(S):	X OR RESOLU REQUIRED): confirmatory / use of 2552 N manager and f LATED OCR ITEM F culation 32-05	AFW flow requirements MW flow requirements MW flow requirements MW by B&W, and disc PIC. JWB REPORT NO.): 25, Rev. 00	analyses, dete	ermine the level to be
PIC COMMENTS BY SRT (IF I Before doing any rationale for the used with project REFERENCES (INCL. RE If 1: B&W AFW Calc SIGNATURE(S): FAD COR ITS OF COMMENTS	X OR RESOLU REQUIRED): confirmatory / use of 2552 M manager and f LATED OCR ITEM F culation 32-05 FAD	AFW flow requirements MUTION: Surance Plan. AFW flow requirements MUTION: AFW flow requirements AFW flow requirements AF	analyses, dete	ermine the level to be
PIC RECOMMENDATION Process per Proje Process per Proje PIC COMMENTS BY SRF (IF I Before doing any rationale for the used with project REFERENCES (INCL. RE ef 1: B&W AFW Calc SIGNATURE(S): FAD OCR ITEM REPORT ORIGINATOR 3/3/83	X OR RESOLU REQUIRED): confirmatory / use of 2552 M manager and F LATED OCR ITEM F culation 32-05 FAD LTR 3/3/83	AFW flow requirements WTION: Surance Plan. AFW flow requirements Wt by B&W, and disc PIC. JWB REPORT NOJ: 25, Rev. 00 HAL PROJECT MANAGER FOR PROJECT TEAM 3/4/83	analyses, dete uss core power	ermine the level to be <u>N/A JWB</u> SRT (IF REQUIRED)

	OFEN, CONFIRMED AND RES	SOLVED (OCR) TIEM REP	URI
TYPE OF REPORT:	OPEN CONFIRMED X		0. <u>3201-008</u> 0. <u>3201-008</u> C_ 028 0.
DATES REPORTED T	OF LTR 3/29/83 SRT	PROJECT TEAM/PROJECT	MGR. 3131183
STRUCTURE	EM(S), OR COMPONENT(S) INVOLVED		
AFW System			
IDCV PROGRAM ARE Component Funct (Review of Crit	A OR TASK (IF APPLICABLE): tional Requirements (I.9-1) teria and Commitments)		
DESCRIPTION OF CO	NCERN:		
water temperature a 40°F minimum for reactor coo coolant system temperature as the original wa	AFW temperature. This crit olant system analysis (Ref. components. Bechtel calclu a worst case winter tempera is implemented, but no addit	5 BOP criteria for AFW terion is consistent with 2) which is used in and lation FM-4117-28 (Ref. ature. The recommend tion analyses were identication	(Ref. 1) requires (Ref. 1) requires th the B&W document alysis of reactor 3) uses a 32°F dation contained in tified.
SIGNIFICANCE OF CO	DNCERN		
If the interface	e requirement is not met, a d become invalid	analyses of the reactor	coolant system
If the interfac components coul	ce requirement is not met, a d become invalid.	analyses of the reactor	coolant system
If the interfac components coul	A CR RESOLUTION	analyses of the reactor	coolant system
RECOMMENDATION	x OR RESOLUTION	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	X OR RESOLUTION	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	X OR RESOLUTION	analyses of the reactor	coolant system
If the interfac components coul RECOMMENDATION Process per PQA	x OR RESOLUTION	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	A CR RESOLUTION	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	<pre>CERNE ce requirement is not met, a d become invalid. XOR RESOLUTION; P. IF REGUIRED);</pre>	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	<pre>PNCERNé te requirement is not met, a d become invalid.  X OR RESOLUTION P.  F REGUIRED):</pre>	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	<pre>PNCERNé ce requirement is not met, a d become invalid.  X OR RESOLUTION P.  F REGUIRED):</pre>	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	<pre>PNCERNé ce requirement is not met, a d become invalid.  X OR RESOLUTION P.  F REGUIRED):</pre>	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA	<pre>PNCERNé ce requirement is not met, a d become invalid.  XOR RESOLUTION: P.  F REGUIRED): </pre>	analyses of the reactor	coolant system
RECOMMENDATION Process per PQA COMMENTS BY SRT ( REFERENCES (INCL. B&W Criteria B&W Functional Bechtel Calcul	Contract Specification for Afunction FM-4117-29	Reactor Coolant System	coolant system
RECOMMENDATION Process per PQA COMMENTS BY SRT ( REFERENCES (INCL. B&W Criteria f B&W Criteria f B&W Functional B&W Functional B&W Functional	X       OR RESOLUTION         X       OR RESOLUTION         P.	Reactor Coolant System	coolant system
RECOMMENDATION Process per PQA COMMENTS BY SRT ( B&W Criteria B&W Functional Bechtel Calcul SIGNATURE(S): FAD	A OR RESOLUTION	Reactor Coolant System	coolant system
RECOMMENDATION Process per PQA COMMENTS BY SRT ( REFERENCES (INCL. ) B&W Criteria ) B&W Criteria ) B&W Functional Bechtel Calcul SIGNATURE(S): FAD OCR ITEM REPORT ORIGINATOR	PNCERNE         ise requirement is not met, a         d become invalid.         X       OR RESOLUTION         P.         IF REGUIRED):         IF REGUIRED):         RELATED OCE ITEM REPORT NO.): 1)         On TAFW (36-100447/7, Rev. 1)         Contract Specification for ation FM-4117-28	Reactor Coolant System	coolant system
RECOMMENDATION Process per PQA COMMENTS BY SRT ( BAW Criteria BAW Functional Bechtel Calcul SIGNATURE(S): FAD OCR ITEM REPORT ORIGINATOR 3/29/83	A       OR RESOLUTION         X       OR RESOLUTION         P.         IF REGUIRED):         RELATED OCB ITEM REPORT NO.): 1)         Or AFW (36-1004477, Rev.' 1)         Contract Specification for ation FM-4117-28	Reactor Coolant System	coolant system

1

-----

. ....

•

TYPE OF BEBORT	EN, CONFIRM	ED AND RESOLVED (C	CR) ITEM REPO	RT
RESC	N CON	FIRMED X	FILE NO. DOC NO. REV. NO.	3201-008 3201-008- C- 031 0
DATES REPORTED TO: 1	RINCIPAL-IN-CH	SRT PROJE	CT TEAM/PROJECT A	AGR. 3/3/83
STRUCTURE(S), SYSTEM(S AFW System Pipe Su	), OR COMPONEN pports	IT(5) INVOLVED:		
DCV PROGRAM AREA OF Topic I.3.1c - Pip Verification of Ph	R TASK (IF APPLIC e Supports ysical Config	CABLE):		
DESCRIPTION OF CONCE	RN:			
SIGNIFICANCE OF CONCE	RN:			
		TION		
RECOMMENDATION X 1. Review further of breakdown.	OR RESOLUT	TION: tion deviation contr	ol process to d	etermine extent
RECOMMENDATION X 1. Review further of breakdown.	OR RESOLUT	TION: tion deviation contr	ol process to d	etermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pro-	OR RESOLUT the construct oject Quality	TION: tion deviation contr Assurance Plan.	ol process to d	letermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pro-	OR RESOLUT the construct oject Quality	TION: tion deviation contr Assurance Plan.	ol process to d	letermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pro-	OR RESOLU the construc	TION: tion deviation contr Assurance Plan.	ol process to d	etermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pro- COMMENTS BY SRT (IF RE	OR RESOLUT the construct oject Quality EQUIRED):	TION: tion deviation contr Assurance Plan.	ol process to d	letermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pro- COMMENTS BY SRT (IF RE	OR RESOLUT the construct oject Quality EQUIRED):	TION: tion deviation contr Assurance Plan.	ol process to d	etermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pro- COMMENTS BY SRT (IF RE	OR RESOLU the construc oject Quality EQUIRED):	TION: tion deviation contr Assurance Plan.	ol process to d	etermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pr COMMENTS BY SRT (IF RE	OR RESOLUT the construct oject Quality	TION: tion deviation contr Assurance Plan.	ol process to d	letermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pr COMMENTS BY SRT (IF RE REFERENCES (INCL. REL) Wg 7220-H-639 SH T Spec 7220-M-326 (Q)	OR RESOLUT the construct oject Quality EQUIRED): ATED OCR ITEM R 4 (Q), Rev 1 Rev 8 "Inst	TION: tion deviation contr Assurance Plan. REPORT NO.): 1 all., Inspect. & Doc	ol process to d	etermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pr COMMENTS BY SRT (IF RE REFERENCES (INCL. REL) Wg 7220-H-639 SH 1 Spec 7220-M-326 (Q) SIGNATURE(S).	OR RESOLUT the construct oject Quality EQUIRED): ATED OCR ITEM R 4 (Q), Rev 1 (Q), Rev 1 (Rev 8 "Inst	TION: tion deviation contr Assurance Plan. Assurance Plan.	ol process to d	etermine extent
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pr COMMENTS BY SRT (IF RE REFERENCES (INCL. REL/ Ng 7220-H-639 SH 1 Spec 7220-M-326 (Q) SIGNATURE(S): CS	OR RESOLUT the construct oject Quality EQUIRED): ATED OCR ITEM F 4 (Q), Rev 1 Rev 8 "Insta	TION: tion deviation contr Assurance Plan. REPORT NO.): 1 all., Inspect. & Doc HAL	ol process to d	rts"
RECOMMENDATION X 1. Review further of breakdown. 2. Process per Pr COMMENTS BY SRT (IF RE REFERENCES (INCL. REL) Wg 7220-H-639 SH 1 Spec 7220-H-639 SH 1 Spec 7220-M-326 (Q) SIGNATURE(S): CS OCR ITEM REPORT ORIGINATOR 2/2/02	OR RESOLUT the construct oject Quality EQUIRED): ATED OCR ITEM R 4 (Q), Rev 1 Rev 8 "Insti- CS LTR	TION: tion deviation contr Assurance Plan. REPORT NO.): 1 all., Inspect. & Doc HAL PROJECT MANAGER FOR PROJECT TEAM	ol process to d . of Pipe Suppo JWB PRINCIPAL- IN-CHARGE	rts"

TYPE OF REPORT: OPE		DAND RESOLVED (OC	or or or	
RES	N CON	FIRMED X	FILE NO. DOC NO. REV. NO.	3201-008 3201-008- C- 032 0
DATES REPORTED TO:	LTR 3/3/83 PRINCIPAL-IN-CH	SRT PROJEC	T TEAM/PROJECT M	GR. 3/3/83
STRUCTURE(S), SYSTEM( AFW System Pipe Sup	s), OR COMPONEN oports	T(S) INVOLVED:		
IDCV PROGRAM AREA O	R TASK (IF APPLIC	ABLE):		
Topic 1.3-1c - Pipe Verification of Phy	Supports	ration		
DESCRIPTION OF CONCE	RN:			
from its design loc allowable tolerance not forwarded for a	ation (along for snubbers opproval and p	the direction of the of O'-6". Construct rocessing by engineer	pipe axis) which ion deviation ing as required	ch exceeds the information was d by procedures.
SIGNIFICANCE OF CONC	ERN			
<ol> <li>The piping anal of this change calculated.</li> </ol>	ysis for this leading to hi	portion of the syste gher support loads an	m may be affect d piping stress	ted as a result ses than
<ol> <li>The construction for this case (</li> </ol>	n deviation c refer to sepa	ontrol process does n rate OCR for recommen	ot appear to be dation).	e functioning
1. Input this info	ormation to th	e TERA confirmatory p	iping analysis	for further
evaluation.	ject Quality	Assurance Plan		
evaluation. 2. Process per Pro	and the second se			
evaluation. 2. Process per Pro				
evaluation. 2. Process per Pro				
evaluation. 2. Process per Pro				
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R	EQUIRED):			
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R	EQUIRED):			
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R	EQUIRED):			
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R	ATED OCR ITEM			
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R REFERENCES (INCL. REL Dwg 7220-H-639 SH 1 Spec 7220-M-326 (Q)	ATED OCR ITEM F 14 (Q), Rev. 1 ), Rev. 8 "In:	REPORT NO.):	. of Pipe Supp	orts"
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R REFERENCES (INCL. REL Dwg 7220-H-639 SH 1 Spec 7220-M-326 (Q) SIGNATURE(S):	ATED OCR ITEM F 14 (Q), Rev. 1 ), Rev. 8 "In:	REPORT NO.):	. of Pipe Supp	orts"
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R REFERENCES (INCL. REL Dwg 7220-H-639 SH 1 Spec 7220-M-326 (Q) SIGNATURE(S): CS OCR ITEM REPORT	ATED OCR ITEM F 14 (Q), Rev. 1 ), Rev. 8 "In: CS	REPORT NO.): 11 stall., Inspect. & Doc HAL	. of Pipe Supp JWB	orts" N/A JWB
evaluation. 2. Process per Pro COMMENTS BY SRT (IF R REFERENCES (INCL. REL Dwg 7220-H-639 SH 1 Spec 7220-M-326 (Q) SIGNATURE(S): CS OCR ITEM REPORT ORIGINATOR	ATED OCR ITEM P 14 (Q), Rev. 1 ), Rev. 8 "In: CS LTR	HAL PROJECT MANAGER FOR PROJECT TEAM	JWB PRINCIPAL- IN-CHARGE	orts" N/A JWB SRT (IF REQUIRED)

MIDLAND INDEPENDENT DESIGN AND CONSTRUCTION VERIFICATION OPEN, CONFIRMED AND RESOLVED (OCR) ITEM REPORT

TYPE OF REPORT: OPE	NCON	FIRMED X	FILEN	O. 3201-008	-033
RES		TEM	REV. N	NO. 0	
DATES RE ORTED TO:	PRINCIPAL-IN-CH	SRT	PROJECT TEAM/PROJEC CPC/DESIGN ORG.	T MGR3	/3/83
STRUCTURE(S), SYSTEM( AFW System Pipe Su	s), OR COMPONENT pports	(S) INVOLVED:			
DCV PROCRAM AREA O	R TASK (IF APPLIC	ABLE):	•		
Verification of Ph	ysical Configu	uration			
DESCRIPTION OF CONCE	RN				
Hanger H-7, a vert from its design lo allowable toleranc for approval and p	ical rigid har cation (along e of l'-O". ( rocessing by e	nger, was field the direction Construction de engineering as	measured by TERA of the pipe axis) viation informatio required by proced	to be about which excee n was not f ures.	t 3'-0" eds the forwarded
SIGNIFICANCE OF CONCI 1. The piping ana of this change	ERN: lysis for this leading to hi	portion of thigher support 1	e system may be af oads and piping st	fected as a resses than	result
calculated.	and and the second				
for this case	(refer to sepa	arate OCR for r	ecommendation).		oning
RECOMMENDATION	OR RESOLUT	ION:			
. Input to TERA	confirmatory p	iping analysis	for further evaluation	ation.	
2. Process per Pr	niact Quality	Accumance 07			
L. HOLESS PET FR	Ject quartey	Assurance Flan			
COMMENTS BY SRT (IF RE	EQUIRED):				
REFERENCES (INCL. REL	TED OCR ITEM RE	PORT NO.)			
wg 7220-H-639 SH 1 pec 7220-M-326 (Q)	4 (Q), Rev. 11 , Rev. 8 "Inst	tall., Inspect.	& Doc. of Pipe Su	pports"	
IGNATURE(S)					
	CS	HAL	JWB	N/A	JWB
ORIGINATOR	LTR	PROJECT MANA	AM INCHARCE	SRT (IF F	EQUIRED)
3/3/83	3/3/83	3/4/83	3/14/83		
DATE	DATE	DATE	DATE		ATE

1

11

	, CONFIRMED A	ND RESOLVED (O	RUCTION VERIF	ICATION
TYPE OF REPORT: OPEN RESOL		ED X	FILE NO. DOC NO. BEV. NO.	<u>3201-008</u> <u>3201-008-</u> C- 034
DATES REPORTED TO: LT	R 3/3/83 SRT	PROJEC	T TEAM/PROJECT N	GR. 3/3/83
STRUCTURE(S), SYSTEM(S),	OR COMPONENT(S)	NVOLVED:		
AFW System Pipe Sup	ports			
IDCY PROGRAM AREA OR	TASK (IF APPLICABLE	5:		x t
Verification of Phy	sical Configurat	tion		
DESCRIPTION OF CONCER	Ne			
allowable tolerance approval and proces	. Construction sing by engineer	deviation inform ring as required l	ation was not f by procedures.	orwarded for
SIGNIFICANCE OF CONCEP	RN:			
<ol> <li>The piping anal result of this stresses than c</li> </ol>	ysis for this po change leading t alculated.	ortion of the system to a higher suppo	tem may be affe rt loads and pi	cted as a ping
<ol> <li>The construction for this case (</li> </ol>	n deviation cont refer to separa	trol process does te OCR for recomm	not appear to endation).	be functioning
RECOMMENDATION Y	OR RESOLUTION			
ALCOMMENTON A		ing analysis for	further evaluat	ion.
1. Input to TERA c	ontirmatory pip	ing analysis ior	The second s	
1. Input to TERA c	inst Ourlity for			
<ol> <li>Input to TERA c</li> <li>Process per Pro</li> </ol>	ject Quality As	surance Plan.		
<ol> <li>Input to TERA c</li> <li>Process per Pro</li> </ol>	ject Quality As	surance Plan.		
1. Input to TERA c 2. Process per Pro	ject Quality As	surance Plan.	•	
1. Input to TERA c 2. Process per Pro	onfirmatory pip ject Quality As	surance Plan.		
1. Input to TERA c 2. Process per Pro	onfirmatory pip ject Quality As	surance Plan.		
1. Input to TERA c 2. Process per Pro	onfirmatory pip ject Quality As	surance Plan.	•	
1. Input to TERA c 2. Process per Pro COMMENTS BY SRT (IF REC Process (INCL- RELA Dwg 7220-H-639 SH 14 Spec 7220-M-326 (Q)	TED OCR TEM REPOR	RT NO.):	c. of Pipe Sup	ports"
1. Input to TERA c 2. Process per Pro COMMENTS BY SRT (IF REC 	DUIRED): TED OCR TEM REPOR	surance Plan.	c. of Pipe Sup	ports"
1. Input to TERA c 2. Process per Pro COMMENTS BY SRT (IF REC 	DURED): TED OCR TEM REPOR (Q), Rev TI (Q), Rev TI Rev 8 "Install CS	RT NO.): ., Inspect., & Do	c. of Pipe Supp JWB	ports" JWB N/
1. Input to TERA c 2. Process per Pro COMMENTS BY SRT (IF REC 	DUIRED): TED OCR TEM REPOR (Q), REV TI (Q), REV TI Rev 8 "Install CS LTR	RT NOJ: ., Inspect., & Do HAL PROJECT MANAGER FOR PROJECT TEAM	JWB PRINCIPAL- IN-CHARCE	Dorts" JWB N/ SRT (IF REQUIRED
1. Input to TERA c 2. Process per Pro COMMENTS BY SRT (IF REC 	DUIRED): TED OCR TEM REPOR (Q), Rev TI (Q), Rev TI (Q)	RT NOJ: ., Inspect., & Do HAL PROJECT MANAGER FOR PROJECT TEAM 3/4/83	JWB PRINCIPAL- IN-CHARGE 3/14/83	Dorts" JWB N/ SRT (IF REQUIRED

Sec. 2

1

1

S ....

5

 $\tau_{ij}$ 

-

10.00

2

Ì?

5 108

a start

1

N.

TYPE OF REPORT. OP	and the second		FILENO	2201.000
RE:	EN CONFIRME	DX	DOC NO. REV. NO.	<u>3201-008</u> <u>3201-008-C-035</u> (1) One
DATES REPORTED TO:	LTR 5/10/83 SRT PRINCIPAL-IN-CHARGE	5726783 PROJEC	T TEAM/PROJECT A	AGR. 5/20/83
STRUCTURE(S), SYSTEM	(S), OR COMPONENT(S) IN	VOLVED:		
IDCV PROGRAM AREA O Topic 1.3-1c Pipe Verification of Ph	R TASK (IF APPLICABLE) Supports sysical Configurati	on		•
DESCRIPTION OF CONC Hanger H-11, a ver elevation but mis- Further measurement elevations shown f	ERN: tical rigid hanger located by l'-3" a ots show DP-260 at	was field measu ccording to draw proper elevation	ired by TERA to ing dimensions , but dimensio	be at the prope from DP-260. ns do not match
support elevations	as measured.	oreer rocarrons	and penetration	on locations
SIGNIFICANCE OF CONC	ERN:			
<ol> <li>Drawing errors indicate the p higher than de</li> </ol>	of this nature ar robability of othe sign levels.	e not consistent r drawing errors	with pipe ana that would de	lysis and may velope loading
<ol> <li>The constructi does not appea</li> </ol>	on deviation contr r to be functionin	ol process and d g.	rawing checking	g process
RECOMMENDATION X	OR RESOLUTION	'		
RECOMMENDATION X 1. Investigate qu procedures and of acceptance	OR RESOLUTION ality paperwork to feed back of resu resolution.	determine effec lts of design gr	tiveness of acc oup for determ	ceptance ination
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of	OR RESOLUTION ality paperwork to feed back of resu resolution. op drawing approva dimension/elevation	determine effec lts of design gr l and establish on nonconformanc	tiveness of acc oup for determ feed back to de e.	ceptance ination asign
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of COMMENTS BY SRT (IF R	OR RESOLUTION ality paperwork to feed back of resu resolution. op drawing approva dimension/elevation REQUIRED):	determine effec lts of design gr destablish	tiveness of acc oup for determ feed back to de e.	ceptance ination esign
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of COMMENTS BY SRT (IF R	OR RESOLUTION ality paperwork to feed back of resu resolution. op drawing approva dimension/elevation REQUIRED):	determine effec lts of design gr	tiveness of acc oup for determ feed back to de e.	ceptance ination esign
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of COMMENTS BY SRT (IF R	OR RESOLUTION ality paperwork to feed back of resu resolution. op drawing approva dimension/elevation REQUIRED):	determine effec lts of design gr l and establish on nonconformanc	tiveness of acc oup for determ feed back to de e.	ceptance ination esign
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of COMMENTS BY SRT (IF R REFERENCES (INCL. REL Drawing 7220-H639	OR RESOLUTION ality paperwork to feed back of resu resolution. op drawing approva dimension/elevation REGUIRED): ATED OCR ITEM REPORT Sh. 14(Q), Rev. 11	determine effec lts of design gr l and establish on nonconformanc	tiveness of acc oup for determ feed back to de e. valuation 3201-	ceptance ination esign
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of COMMENTS BY SRT (IF R REFERENCES (INCL. REL Drawing 7220-H639 SIGNATURE(S):	OR RESOLUTION ality paperwork to feed back of resu resolution. op drawing approva dimension/elevation REQUIRED): ATED OCR ITEM REPORT Sh. 14(Q), Rev. 11	determine effec lts of design gr l and establish on nonconformanc	tiveness of acc oup for determ feed back to de e. valuation 3201-	ceptance ination esign
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of COMMENTS BY SRT (IF F COMMENTS BY SRT (IF F REFERENCES (INCL. REL Drawing 7220-H639 SIGNATURE(S): <u>RCS</u> OCR ITEM REPORT	OR RESOLUTION ality paperwork to feed back of resures resolution. op drawing approva dimension/elevation EGUIRED): ATED OCR ITEM REPORT Sh. 14(Q), Rev. 11	determine effec lts of design gr l and establish on nonconformanc NO.): & Engineering E	tiveness of acc oup for determ feed back to de e. valuation 3201-	esign
RECOMMENDATION X 1. Investigate qu procedures and of acceptance Investigate sh and drawing of COMMENTS BY SRT (IF F COMMENTS BY SRT (IF F REFERENCES (INCL. REL Drawing 7220-H639 SIGNATURE(S): <u>RCS</u> OCR ITEM REPORT O'RIGINATOR	OR RESOLUTION ality paperwork to feed back of resur- resolution. op drawing approva dimension/elevation EGUIRED): ATED OCR ITEM REPORT Sh. 14(Q), Rev. 11 DBT DBT FC	i determine effec lts of design gr l and establish on nonconformanc NO.): δ Engineering E HAL ROJECT MANAGER R PROJECT TEAM	tiveness of acc oup for determ feed back to de e. valuation 3201- JB PRINCIPAL- IN-CHARGE	-001-001, Pgs 7 &

0	PEN, CONFIRME	ED AND RESOLVED (O	R) ITEM REPO	RI
TYPE OF REPORT: OF			FILE NO. DOC NO. REV. NO	<u>3201-008</u> 3201-008_C_036
DATES REPORTED TO:	LTR 5/11/8 PRINCIPAL-IN-CH	ARGE 5/26/83 CPC/DE	T TEAM/PROJECT	MGR. 5/20/83
AFW System Pip	N(S), OR COMPONEN	T(S) INVOLVED:		
IDCV PROGRAM AREA Topic 11.2-1 P Drawing Review	OR TASK (IF APPLIC ressure Boundar	ABLE): Y		
DESCRIPTION OF CON	CERN:			
given along pip 285, 300 and 30 signed have not	be centerline a D6. Difference t been adequate	s follows. Distances s range from 5/16 and ly checked.	between DP 27 7/16. Drawin	70 and 280, 280 an ngs that have been
Inconsistencie: structures, sy:	s in design dra stems and compo	wings could lead to o nents from design ass	eviation of co umptions.	onstructed
Inconsistencie: structures, sys	x OR RESOLUT	wings could lead to o nents from design ass	eviation of co umptions.	onstructed
Inconsistencies structures, sys RECOMMENDATION 1. Investigate and feed ba	X OR RESOLUT or shop drawing ock to design a	wings could lead to o nents from design ass TION: approval system to es nd drafting.	eviation of co umptions. tablish method	onstructed
Inconsistencies structures, sys RECOMMENDATION 1. Investigate and feed ba	X OR RESOLUT or shop drawing ock to design a	wings could lead to o nents from design ass TION: approval system to es nd drafting.	eviation of co umptions. tablish method	onstructed
Inconsistencies structures, sys RECOMMENDATION 1. Investigate and feed ba	X OR RESOLUT or shop drawing ock to design a	wings could lead to o nents from design ass TION: approval system to es nd drafting.	eviation of co umptions. tablish method	onstructed
Inconsistencies structures, sys RECOMMENDATION 1. Investigate and feed ba	X OR RESOLUT schop drawing ack to design a	wings could lead to o nents from design ass TION: approval system to es nd drafting.	eviation of co umptions. tablish method	onstructed
Inconsistencies structures, sys RECOMMENDATION 1. Investigate and feed ba	X OR RESOLUT a shop drawing ack to design a	wings could lead to d nents from design ass TION: approval system to es nd drafting.	eviation of co umptions. tablish method	onstructed
RECOMMENDATION 1. Investigate and feed back COMMENTS BY SRT (IF	X OR RESOLUT a shop drawing ack to design a REQUIRED):	Wings could lead to d nents from design ass TION: approval system to es nd drafting.	eviation of co umptions. tablish method	onstructed
Inconsistencies structures, sys RECOMMENDATION 1. Investigate and feed ba COMMENTS BY SRT (IF  REFERENCES (INCL. RE Drawing 7220-H-	X OR RESOLUT a shop drawing ack to design a REQUIRED): CLATED OCR ITEM R 639 (Q), Sh.	Wings could lead to o nents from design ass TION: approval system to es nd drafting. EPORT NO.): 14, Rev. 11 & Eng. Ev	eviation of co umptions. tablish method	onstructed d of resolution
RECOMMENDATION 1. Investigate and feed ba COMMENTS BY SRT (IF COMMENTS BY SRT (IF Drawing 7220-H- SIGNATURE(S): RCS	X OR RESOLUT a shop drawing ack to design a REQUIRED): LATED OCR ITEM RI 639 (Q), Sh.	Wings could lead to d nents from design ass TION: approval system to es nd drafting. EPORT NO.): 14, Rev. 11 & Eng. Ev	eviation of co umptions. tablish method	onstructed
RECOMMENDATION 1. Investigate and feed ba COMMENTS BY SRT (IF REFERENCES (INCL. RE Drawing 7220-H- SIGNATURE(S): RCS OCR ITEM REPORT	X OR RESOLUT a shop drawing ack to design a REQUIRED): CLATED OCR ITEM R 639 (Q), Sh. DBT LTR	Wings could lead to o nents from design ass TION: approval system to es nd drafting. EPORT NO.): 14, Rev. 11 & Eng. Ev HAL PROJECT MANAGER	al. 3201-001-0	onstructed
RECOMMENDATION 1. Investigate and feed ba COMMENTS BY SRT (IF REFERENCES (INCL. RE Drawing 7220-H- SIGNATURE(S): RCS OCR ITEM REPORT ORIGINATOR 5/10/83	X OR RESOLUT a shop drawing ack to design a REQUIRED): CLATED OCR ITEM R 639 (Q), Sh. DBT LTR 5/20/83	Wings could lead to d nents from design ass TION: approval system to es nd drafting. EPORT NO.): 14, Rev. 11 & Eng. Ev HAL PROJECT MANAGER FOR PROJECT TEAM 5/25/82	al. 3201-001-0 JB PRINCIPAL- IN-CHARGE	onstructed d of resolution 001, page 9 SRT (IF REQUIRED)

			CIVITEM REFU	
TYPE OF REPORT: OPE RES		MED X	FILE NO. DOC NO. REV. NO	<u>3201-008</u> <u>3201-008</u> C. 037
DATES REPORTED TO:	LTR 3/3/83 SR	T PROJEC	T TEAM/PROJECT	MCR. 3/3/83
STRUCTURE(S), SYSTEM( AFW System - A11	S), OR COMPONENT(S)	INVOLVED:		
IDCV PROGRAM AREA OF Topic III.1-1 - S Review of Design	R TASK (IF APPLICABL eismic Design Criteria	£);		
DESCRIPTION OF CONCE	RN:			
FSAR text nor the building. The FS	models and resp AR updating proc	onse spectra for ess is not consis	the containment tent nor timely	and auxiliary
SIGNIFICANCE OF CONC	ERN:			
RECOMMENDATION X	OR RESOLUTION			
RECOMMENDATION X	OR RESOLUTION	·:		
RECOMMENDATION X 1. Review furthe	OR RESOLUTION	garding the FSAR	updating proces	5.
RECOMMENDATION X 1. Review furthe 2. Process per P	OR RESOLUTION r information re roject Quality A	garding the FSAR sourance Plan.	updating proces	55.
RECOMMENDATION X 1. Review furthe 2. Process per P	OR RESOLUTION r information re roject Quality A	garding the FSAR ssurance Plan.	updating proces	55.
RECOMMENDATION X 1. Review furthe 2. Process per P	OR RESOLUTION r information re roject Quality A	garding the FSAR ssurance Plan.	updating proces	55.
RECOMMENDATION X 1. Review furthe 2. Process per P	OR RESOLUTION r information re roject Quality A	garding the FSAR ssurance Plan.	updating proces	is.
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF R	OR RESOLUTION r information re roject Quality A	garding the FSAR ssurance Plan.	updating proces	55.
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF R	OR RESOLUTION r information re roject Quality A	garding the FSAR ssurance Plan.	updating proces	55.
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF R	OR RESOLUTION r information re roject Quality A	garding the FSAR ssurance Plan.	updating proces	55.
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF RI REFERENCES (INCL. PEL	OR RESOLUTION r information re roject Quality A EQUIRED):	garding the FSAR ssurance Plan.	updating proces	iS.
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF RI REFERENCES (INCL. REL SAR, Rev. 46, Sect pec. 7220-G-6, Rev	OR RESOLUTION r information re roject Quality A EQUIRED): ATED OCR ITEM REPO tion 3.7 7.7 and G-7, Rev	garding the FSAR ssurance Plan.	Aux. Bldg. R	esponse Spectra
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF RI REFERENCES (INCL. REL SAR, Rev. 46, Sect pec. 7220-G-6, Rev SIGNATURE(S):	OR RESOLUTION r information re roject Quality A EQUIRED): ATED OCR ITEM REPO tion 3.7 7. 7 and G-7, Rev	garding the FSAR ssurance Plan.	apdating proces	esponse Spectra
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF R REFERENCES (INCL. REL SAR, Rev. 46, Sect pec. 7220-G-6, Rev SIGNATURE(S): CS	OR RESOLUTION r information re roject Quality A EQUIRED): ATED OCR ITEM REPO tion 3.7 7. 7 and G-7, Rev CS	garding the FSAR ssurance Plan.	updating proces & Aux. Bldg. Re JWB	esponse Spectra
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF R REFERENCES (INCL. REL SAR, Rev. 46, Sect SIGNATURE(S): CS OCR ITEM REPORT ORIGINATOR	CS LTR OR RESOLUTION Project Quality A Project	garding the FSAR ssurance Plan. RT NO.): 9, Containment HAL PROJECT MANAGER FOR PROJECT TEAM	Aux. Bldg. Re JWB PRINCIPAL- IN-CHARGE	esponse Spectra N/A JWB SRT (IF REQUIRED
RECOMMENDATION X 1. Review furthe 2. Process per P COMMENTS BY SRT (IF RI IEFERENCES (INCL. REL SAR, Rev. 46, Sect Dec. 7220-G-6, Rev IGNATURE(S): CS OCR ITEM REPORT ORIGINATOR 2/000	OR RESOLUTION r information re roject Quality A EQUIRED): ATED OCR ITEM REPO tion 3.7 7. 7 and G-7, Rev CS LTR	garding the FSAR ssurance Plan. RT NO.): 9, Containment HAL PROJECT MANAGER FOR PROJECT TEAM	Aux. Bldg. Re JWB PRINCIPAL- IN-CHARGE	esponse Spectra N/A JWB SRT (IF REQUIRED

Press Branch Vielant Brancharts and and and	and the second se	and the second		
TYPE OF REPORT: C	PENCON	TEM	FILE NO. DOC NO. REV. NO.	3201-008 3201-008-C -038
DATES REPORTED TO	PRINCIPAL-IN-CI	SRT PROJEC	T TEAM/PROJECT N	AGR. 3/3/83
STRUCTURE(S), SYSTE	MIS), OR COMPONE	VT(S) INVOLVED:		
IDCV PROCRAM ARE		CARLES.		
Topic I.15-1, Co	ntrol/Power Sur	oplies		
DESCRIPTION OF CON Under condition valve 257-3969B power. The Midl that AFW be oper of time flow thr to the pump.	CERN: of loss of all would not be op and FSAR and Ba able for two ho ough the minim	AC (station blackout) perable because it is SW BOP crimeria docume ours under station bla um flow line may be ne	, the AFW pump powered from C ent (36-1004477 ickout. During cessary to pre	minimum flow lass IE AC ) both require this period vent damage
SIGNIFICANCE OF CO	NCERN:			
	and an and a second			
RECOMMENDATION	X OR RESOLU	J710N3		1
RECOMMENDATION _	X OR RESOLU	JTIONsurance Plan.		
RECOMMENDATION _	X OR RESOLU	Surance Plan.		
RECOMMENDATION _	X OR RESOLU	JTIONs surance Plan.		*
RECOMMENDATION _ Process per Proj	X OR RESOLU ect Quality As REQUIRED:	Surance Plan.		
RECOMMENDATION Process per Proj	X OR RESOLU ect Quality As: REQUIRED:	Surance Plan.		
RECOMMENDATION Process per Proj COMMENTS BY SRT (IF	X OR RESOLU ect Quality As: REQUIRED):	Surance Plan.		
RECOMMENDATION Process per Proj COMMENTS BY SRT (IF REFERENCES (INCL. F OUR 3203-008-0-1)	X OR RESOLU ect Quality As: REQUIRED: REQUIRED: NELATED OCR ITEM	MEPORT NO.J: rawing E-158(Q) SH 29,	29A, 29B, 29C	
RECOMMENDATION Process per Proj COMMENTS BY SRT (IF REFERENCES (INCL. F OCR 3203-008-0-1 SIGNATURE(S): LB	X OR RESOLU ect Quality As: REQUIRED: REQUIRED: NELATED OCR ITEM 12 & C-012 ; Dr LB	TTION surance Plan. REPORT NOJ: rawing E-158(Q) SH 29, HAL	29A, 29B, 29C JWB	N/A JWB
RECOMMENDATION Process per Proj COMMENTS BY SRT (IF REFERENCES (INCL. F OUR 3203-008-0-K SIGNATURE(S): LB OCR ITEM REPORT ORIGINATOR 2/1/02	X OR RESOLU ect Quality As: REQUIRED: REQUIRED: ULL & C-012; Dr LB LTR	HAL PROJECT MANAGER FOR PROJECT TEAM	29A, 29B, 29C JWB PRINCIPAL- IN-CHARGE	N/A JWB SRT (IF REQUIRED

ð.,

	OPEN, CONFIRM	NED AND RESOLVED (C	CR) ITEM REPO	IFICATION DRT
TYPE OF REPORT:	OPEN COM		FILE NO DOC NO REV. NO	0. <u>3201-008</u> 0. <u>3201-008</u> C. 045
DATES REPORTED	TO: LTR 3/17/83 PRINCIPAL-IN-CH	SRT PROJE	CT TEAM/PROJECT	MGR. 5/20/83
STRUCTURE(S), SYS	STEM(S), OR COMPONE	NT(S) INVOLVED:		
uxiliary Feedwa	ater System: AFW	Pump Motor 2P005A		
IDCV PROGRAM AR	REA OR TASK (IF APPLI	CABLE):		
CV: Review of	Storage and Main	tenance Documentatio	n	
DESCRIPTION OF C	ONCERN:			
. Manufacture every two we	r's recommended s eeks while motor	torage instructions is in storage (Ref:	Vendor Doc. No	shaft rotation . 7220-M14-68).
<ul> <li>Bechtel proc of motor sha rotations, a</li> </ul>	cedure governing aft every 90 days as recommended by	in-place maintenance , exceeding the maxim the vendor, by a fac	(F-10-247) rec num duration be ctor of 6.	uires rotation otween shaft
SIGNIFICANCE OF C	ONCERN:			
Failure to co for the motor shaft bearing	omply with manufa may have a dele s, and rotating	cturer's recommended terious effort upon a elements of the motor	shaft rotation the shaft beari r.	ng surfaces,
Failure to co for the motor shaft bearing	omply with manufa r may have a dele gs, and rotating	cturer's recommended terious effort upon f elements of the motor	shaft rotation the shaft beari r.	schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION	omply with manufa r may have a dele gs, and rotating	cturer's recommended terious effort upon in elements of the motor	shaft rotation the shaft beari r.	schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa	mply with manufa may have a dele gs, and rotating <u>X</u> OR RESOLU for inspection by nees.	cturer's recommended terious effort upon re elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari r. and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa	omply with manufa r may have a dele as, and rotating <u>X</u> OR RESOLU for inspection by aces.	cturer's recommended terious effort upon s elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari r. and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa	mply with manufa may have a dele s, and rotating <u>X</u> OR RESOLU or inspection by ces.	cturer's recommended terious effort upon re elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari r. and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa	The second secon	cturer's recommended terious effort upon re elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa	omply with manufa r may have a dele rs, and rotating (IF REQUIRED):	cturer's recommended terious effort upon te elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa	mply with manufa r may have a dele s, and rotating <u>X</u> OR RESOLU or inspection by dees.	cturer's recommended terious effort upon re elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa	mply with manufa may have a dele s, and rotating (IF REQUIRED): RELATED OCR ITEM R	cturer's recommended terious effort upon in elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa COMMENTS BY SRT	mply with manufa may have a dele s, and rotating (IF REQUIRED): RELATED OCR ITEM R Procedure F-10-21 No. 7220-M14-68	cturer's recommended terious effort upon re elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa COMMENTS BY SRT REFERENCES (INCL echtel Storage endor Document SIGNATURE(S):	mply with manufa may have a dele s, and rotating (IF REQUIRED): RELATED OCR ITEM R Procedure F-10-21 No. 7220-M14-68	cturer's recommended terious effort upon in elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa COMMENTS BY SRT COMMENTS BY SRT REFERENCES (INCL echtel Storage indor Document SIGNATURE(S): MBJ OCR ITEN STORES	mply with manufa may have a dele s, and rotating (IF REQUIRED): RELATED OCR ITEM R Procedure F-10-21 No. 7220-M14-68 DBT	cturer's recommended terious effort upon re elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review	e schedule ng surfaces,
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa COMMENTS BY SRT COMMENTS BY SRT REFERENCES (INCL echtel Storage endor Document SIGNATURE(S): MBJ OCR ITEM REPORT ORIGINATOR	mply with manufa may have a dele ps, and rotating <u>X</u> OR RESOLU or inspection by dess. (IF REQUIRED): RELATED OCR ITEM R Procedure F-10-22 No. 7220-M14-68 DBT LTR	Cturer's recommended terious effort upon in elements of the motor TION: manufacturer's rep.	shaft rotation the shaft beari and ICV review JB PRINCIPAL- IN-CHARGE	er of motor
Failure to co for the motor shaft bearing RECOMMENDATION Recommend mot bearing surfa COMMENTS BY SRT COMMENTS BY SRT REFERENCES (INCL echtel Storage endor Document SIGNATURE(S): MBJ OCR ITEM REPORT ORIGINATOR 3/17/83	mply with manufa may have a dele s, and rotating (IF REQUIRED): RELATED OCR ITEM R Procedure F-10-21 No. 7220-M14-68 DBT LTR 	Cturer's recommended terious effort upon in elements of the motor TION: manufacturer's rep.	JB PRINCIPAL- IN-CHARGE 5/27/83	er of motor SRT (IF REQUIRED)

A STATE OF THE OWNER	OPEN, CONFIRM	NT DESIGN AND CON	OCR) ITEM REP	IFICATION DRT
TYPE OF REPORT:	COEN CO		FILE NO DOC NO REV. N	0. <u>3201-008</u> 0. <u>3201-008</u> C_ 046
DATES REPORTED T	O: LTR 3/17/83 PRINCIPAL IN 4	SRT PRO. HARGE 5726783 CPC/	ECT TE M/PROJECT	MGR. 5/20/83
STRUCTURE(S), SYST Auxiliary Free	EM(S), OR COMPONE	ATCO PANOLVED:		
IDCV PROCEAM ADE	A OR TASK //2 ADDI			1
CV: Review	of Storage + Ma	intenance Documentat	tion	
DESCRIPTION OF CO . Pump manufact under vacuum than 50%.	NCERN: urgr's recommen with VPI crysta	ded storage instruct is (dessicant) to ma	ions require pu intain Relative	mp to be stored Humidity at less
2. Bechtel Proce nor humidity	dure for storag check per item	a of pumps, Proc. #	-10-118, does n	ot require vacuum
. Surther to co flooding & ot	ncera, review o har damage, & s	f records indicates	pump have been	open, subject to
SIGNIFICANCE OF CO	NCERN: indicatin	e maintenance proble d out.	ms which have n	ot been addressed
arehouse and in	place) raise con	ncerns as to the exi	stence of inter	both in the nal damage to
arehouse and in the pumps and tur	place) raise con bine resulting	from rust, corrosion	stence of inter , and foreign m	both in the nal damage to aterials.
RECOMMENDATION Recommend pumps	X OR RESOLU	from rust, corrosion	stence of inter , and foreign m	both in the nal damage to aterials.
RECOMMENDATION Recommend pumps Disassembly and	A constant of the second secon	TION: Isassembly and inspe	stence of inter , and foreign m ction. manufacturer's	both in the nal damage to aterials. rep. and ICV revie
RECOMMENDATION Recommend pump Disassembly and COMMENTS BY SRT (1)	A CR RESOLU A CR RESOLU A A CR RESOLU A A A A A A A A A A A A A A A A A A A	TION: isassembly and inspe	stence of inter , and foreign m ction. manufacturer's	both in the nal damage to aterials. rep. and ICV revie
RECOMMENDATION Recommend pump Disassemply and COMMENTS BY SRT (7	X OR RESOLU s and turbine d d inspection sho	TION: Isassembly and inspe	stence of inter , and foreign m ction. manufacturer's	both in the nal damage to aterials. rep. and ICV revie
RECOMMENDATION Recommend pump Disassembly and COMMENTS BY SRT () REFERENCES (INCL. P euhtel Procedure	X OR RESOLU S and turbine d d inspection sho REQUIRED): REQUIRED): RELATED OCR ITEM F F-10-118 and St	REPORT NO.): corage and Maintenand	et Checklist GN-	oth in the nal damage to aterials. rep. and ICV revie
RECOMMENDATION Recommend pump Disassembly and COMMENTS BY SRT () REFERENCES (INCL. P subtel Procedure SI INATURE(S): MBJ	X OR RESOLU S and turbine d d inspection sho REQUIRED): REQUIRED): CELATED OCR ITEM F F-10-118 and St	REPORT NO.): corage and Maintenand	e Checklist GN-	ooth in the nal damage to aterials. rep. and ICV revie
REFERENCES (INCL. P COMMENTS BY SRT (I) REFERENCES (INCL. P cubtel Procedure SUNATURE,S): MBJ DCR ITEM REPORT URIGINATOR	Place) raise combine resulting         X       OR RESOLUTION         S and turbine did         d inspection show         REQUIRED):         REQUIRED):         REQUIRED):         CBT         LTR	REPORT NO.): Corage and Maintenand HAL PROJECT MANAGER FOR PROJECT TEAM	tion. manufacturer's Checklist GN- JB PRINCIPAL-	oth in the nal damage to aterials. rep. and ICV revie 3-118 SRT (IF REQUIRED)

F03 026583



LONE TREE COUNCIL

P. O. Box 421

Essexville, Michigan 48732

Advisory Board

CONSTANCE SMITH PRESIDENT COMMUNICATION WORKERS OF AMERICA LOCAL 4108

QUINTER BURNETT, M D \*FATHER JOHN GUSSENBAUER JOSEPH SHEERAN. ATTORNEY JACK WESTON NASH. D.D.S JOAN SABOURIN ASSOCIATE PROFESSOR OF CHEMISTRY. DELTA DR DAVID DALGARN. ASSOCIATE PROFESSOR OF BIOLOGY SVSC May 31, 1983

James Keppler U.S. Nuclear Regulatory Commission Region III 799 Roosevelt Road Glen Ellyn, Illinois 60137

Dear Mr. Keppler, Thank you for the prompt reply regarding an independent audit of the planned Midland Nuclear Power Plant. Unfortunately, we are disturbed by your ending paragraph in which you imply that letting Consumers Power Company reinspect its own work does not make a mockery of the NRC's commitment to ensuring safe construction. You add, "particularly in view of third party inspections and other actions being taken under the Construction

Let us look at these "third party" inspections and the CCP. As we understand it, an audit is an examination for the purpose of verification--in this case a safely constructed nuclear plant. Our understanding, and please correct us if we are wrong, is that Consumers Power selects what TERA Corporation will inspect (with NRC approval). To use an analogy, if we are audited by the Internal Revenue Servic; we get to choose those parts of our finances we would like disclosed. Of course, this is ludicrous. Yet Consumers seemingly has that power. In addition, it has selected, and the NRC has approved, the Auxiliary Feedwater System, which has been reviewed and approved recently.

An audit is usually thought to be complete and very thorough. While it is reassuring that the heating, ventilation, and air conditioning system, and the emergency power system might be reviewed, it appears that TERA will be focusing mainly on the design of these systems rather than the construction--a very distinct difference.

Concerning the CCP, your letter disguises the fact that there would probably be no CCP had it not been forcibly suggested by the NRC. The plan was not a result of the utility's initiative. This does not create a feeling of confidence in the

"ORGANIZATIONS LISTED FOR IDENTIFICATION PURPOSES ONLY



PATRICIA HEARRON, CHILD DEVELOPMENT SPECIALIST utility's commitment to do the job properly. In a related matter, Consumers Power announced that, unless told otherwise by the NRC, they would begin their Construction Implementation Overview (CIO) on April 18, 1983. Publicly, the NRC has remained silent.

Regarding the selection of outside firms as third party inspectors and citizen input, you have previously stated that the public will not have a vote in this since you "don't believe in the shared process of decision making." (Midland Daily News, May 4, 1983) We find it ironic that the licensee has chosen the areas for re-inspection as well as the inspectors, and obviously is able to share in this process, yet citizens are denied it.

In summary, there are several questions that are raised:

- What new light will be shed by a re-evaluation of the Auxiliary Feedwater System?
- Would you explain the extent to which TERA Corporation will examine construction as well as design?
- 3. Regarding the CIO-
  - a. Did it begin April 18, 1933, as announced by Consumers?
  - b. If so, have you approved of the plan?
  - c. Has Stone and Webster, therefore, been approved?
  - d. If so, will it include a 100% review as promised by Consumers in December, 1982?

We have different ideas on what a third party audit should encompass. It does not seem unreasonable that a truly independent audit should:

- Include a full scope overview of completed construction done by the third party rather that the utility.
- Consist of a thorough inspection of as-is construction, as well as the design of the plant.
- Be selected solely by the NRC (or allow the public the same voice as the utility)

It would seem that this approach would totally assure our community that the plant has been constructed safely. Would you explain why this method is not possible?

Mr. Keppler, we know that the Midland plants have become an albatross for you. You have indicated several times that your role is that of regulator, not builder. We understand your position, but one cannot ignore the many guality assurance breakdowns and poor construction record of Consumers Power. To use your own words, "You wonder after so many screw-ups whether the utility is capable of doing the job right." (Interview, WXYZ-TV, Fill, 1982)



UNITED STATES JUCLEAR REGULATORY COMMISSION **REGION III** 799 ROOSEVELT ROAD GLEN ELLYN. ILLINOIS 60137

()FLL ()FLL F03041:563

OCT 1 8 1983

Docket No. 50-329 Docket No. 50-330

Lone Tree Council ATTN: Mr. Leo R. Romo Corresponding Secretary P. O. Box 421 Essexville, Michigan 48732

Dear Mr. Romo:

8310240107

This is in reply to your September 25, 1983, letter to Mr. James Keppler demanding that an investigation of Stone and Webster be conducted and that the Construction Completica Program not be approved pending such investigation.

The NRC has reviewed Consumers Power Company's (CPCo) proposal to have Stone and Webster perform the third party independent overview of the Construction Completion Program (CCP). We recognize that A-E's have job sites that are relatively problem free while they have other job sites that have problems. In our evaluation we considered the qualifications of both the S&W organization and the individuals proposed as team members to conduct the Construction Implementation Overview (CIO) of Consumers' CCP. We reviewed S&W's experience in assessing nuclear construction projects, particularly its performance in independent reviews of design, construction, and quality assurance undertaken for utilities as input to the NRC's operating license reviews. We also reviewed the qualifications of the key persons proposed for the project. We verified their experience and competence by reference checks and/or discussions with NRC staff members familiar with them.

Based on our review of the documentation submitted by CPCo and S&W, followup checks, and consideration of comments by members of the public, we concluded that S&W meets the independence and competence criteria for third party reviewers and that S&W's proposed CIO program is adequate to provide for an assessment of the CCP. In my letter to CPCo dated September 29, 1983, I approved S&W to perform the CIO. A copy of the approval letter and the staff's evaluation of CPCo's proposal to use S&W are enclosed. A further investigation of S&W is not planned.

I have enclosed a copy of the NRC's October 6, 1983 response to the letter dated June 13, 1983, from Billie Pirner Garde of the Government Accountability Project on behalf of the Lone Tree Council and others. In addition I have enclosed a copy of the NRC's Confirmatory Order dated October 6, 1983, to CPCo approving the CCP.

Lone Tree Council

2

OCT 1 8 1983

If I can be of further assistance, let me know.

Sincerely, signed by James G. Keppler

James G. Keppler Regional Administrator

Enclosures:

-

 Letter from J. G. Keppler to CPCo dated September 29, 1983, approving S&W to perform CIO w/encl
 NRC letter to Billie P. Garde dated

October 6, 1983

 NRC Confirmatory Order dated October 6, 1983

cc w/ltr dtd 9/25/83: DMB/Document Control Desk (RIDS) Mr. James W. Cook, Vice President Midland Project Resident Inspector, RIII The Honorable Charles Bechhoefer, ASLB The Honorable Jerry Harbour, ASLB The Honorable Frederick P. Cowan, ASLB The Honorable Ralph S. Decker, ASLB William Paton, ELD Michael Miller Ronald Callen, Michigan Public Service Commission Myron M. Cherry Barbara Stamiris Mary Sinclair Wendell Marshall Colonel Steve J. Gadler (P.E.) Howard Levin (TERA) Billie P. Garde, Government Accountability Project Lynne Bernabei, Government Accountability Project

11.4

RIII Harrison/ls 09/29/83

RIII RIII RFW Lewis's Warnick Communds Lewis 10/14 marry rule EFW

RIII



1.1.1

UNITED STATES NUCLEAR REGULATORY COMMISSION REGION III 795 ROOSEVELT ROAD GLEN ELLYN, ILLINOIS 60137

inh!

# SEP 2 9 1983

Docket No. 50-329 Docket No. 50-330

Consumers Power Company ATTN: Mr. James W. Cook Vice President Midland Project 1945 West Parnall Road Jackson, MI 49201

Gentlemen:

(Jan

831005034L

We have reviewed your proposal to have the Stone and Webster Corporation (S&W) perform the third party independent overview of the Construction Completion Program (CCP). Our evaluation is enclosed.

The staff has considered the qualifications of both the S&W organization and the individuals proposed as team members to conduct the Construction Implementation Overview (CIO) of Consumers Power Company's (CPCo) Construction Completion Program. Inputs to this review included the information supplied by S&W, as set forth in the April 6, 1983, April 11, 1983, and May 19, 1983 submittals, the staff's existing knowledge of S&W performance at other nuclear power plants, and information as to S&W personnel competence.

The CIO program described by S&W in the August 30, 1983, and September 9, 1983, submittals and at the August 25, 1983, meeting has been reviewed by the NRC staff and found to constitute an acceptable third party overview program. The NRC staff has reviewed the CIO activities performed to date and has found this overview to have been adequate.

Based on NRC review of the documentation submitted by CPCo and S&W, followup checks, and consideration of comments by members of the public, we conclude that S&W meets the independence and competence criteria for third party reviewers and that S&W's proposed CIO program is adequate to provide for an assessment of the Construction Completion Program (CCP).

This letter constitutes NRC approval of S&W to perform the CIO.

Consumers Power Company

SEP 2 9 1983

Should you have any questions regarding this letter please contact Mr. R. F. Warnick of my staff.

Sincerely,

2

Original signed by Januar C.

James G. Keppler Regional Administrator

## Enclosure: As stated

1210 1 1

cc w/encl: DMB/Document Control Desk (RIDS) Resident Inspector, RIII The Honorable Charles Bechhoefer, ASLB The Honorable Jerry Harbour, ASLB The Honorable Frederick P. Cowan, ASLB The Honorable Ralph S. Decker, ASLB William Paton, ELD Michael Miller Ronald Callen, Michigan Public Service Commission Myron M. Cherry Barbara Stamiris Mary Sinclair Wendell Marshall Colonel Steve J. Gadler (P.E.) Howard Levin (TERA) Billie P. Garde, Government Accountability Project Lynne Bernabei, Government Accountability Project



STAFF EVALUATION OF CONSUMERS POWER COMPANY PROPOSAL TO USE STONE AND WEBSTER MICHIGAN, INC. TO CONDUCT THE THIRD PARTY CONSTRUCTION IMPLEMENTATION OVERVIEW OF THE MIDLAND NUCLEAR PLANT

#### Purpose and Background

The purpose of this document is to provide an evaluation of the Consumers Power Company's (CPCo) proposal to use Stone and Webster 'S&W), Michigan, Inc. to conduct the third party overview of the Construction Completion Program at Midland. Consumers' proposal is documented in their letter of April 6, 1983, in response to the NRC's March 28, 1983, request for additional information. The CPCo commitment to provide for an independent third party Construction Implementation Overview (CIO) has been reviewed and found acceptable. This evaluation provides the basis of the NRC's acceptance of Consumers proposal.

The purpose of the CIO is to provide an independent overview of the Construction Completion Program (CCP) to assure the program is adequate and will be properly implemented. This is to ensure that the construction of the facility can be completed in conformance with the Commission's regulations and the construction permits.

The S&W overview of the CCP will be independent from and supplemental to the normal NRC inspection program. As part of their inspection program, the NRC inspectors will monitor and review the S&W CIO.

The use of S&W as the third party overviewer will provide additional assurance of proper implementation of the quality program. In addition, it will function as a mechanism to allow members of the public and the NRC to regain confidence in the program.

The results of the overview program will be submitted to the Regional Administrator in a weekly report of CCP activities overviewed and any problems identified.

The NRC has required communications between CPCo and S&W to follow a protocol to assure S&W's independence is being maintained and to assure public and NRC knowledge of S&W activities and correspondence. It should be noted that the protocol provides for a monthly meeting, open to the public for observation, to review S&W activities for the month and to discuss problems identified by the overview.

## CPCo's Proposel Third Party Reviewer

CPCo has proposed that Stone and Webster perform an independent overview of the Midland project CCP. The NRC staff has considered CPCo's submittal of April 6, 1983, and responses to Region III questions, public comments, and the clarification of submitted comments and additional comments received at public meetings held in Midland, Michigan on February 8, 1983, and August 11, 1983. In addition, the staff conducted numerous meetings and telephone conversations with representatives of the Government Accountability Project (GAP) and the intervenors. In considering CPCo's proposal, the staff has used as guidance the letter of February 1, 1982, from Chairman Palladino to Congressmen Ottinger and Dingell, (attached) which sets forth the "competence and independence" standards that have been applied by the Commission in determining the acceptability of proposed third-party reviewers.

#### S&W Competence

The staff has considered the qualifications of both the S&W organization and the individuals proposed as team members to conduct the independent overview of the Midland project. Input to the staff's review included the information supplied in CPCo's submittal, the responses to the staff's inquiries, the S&W submittals, and the staff's existing knowledge of S&W performance at other nuclear power plants.

The staff has reviewed S&W's experience in assessing nuclear construction projects, particularly its performance in independent reviews of design, construction, and quality assurance undertaken for utilities as input to the NRC's operating license reviews.<sup>1</sup>

The staff has also reviewed the qualifications of the key persons proposed for the project, as set forth in the April 6, 1983, April 11, 1983, and May 19, 1983, submittals, and has concluded that the team has significant stated experience in QA/QC matters, nuclear plant construction, and management systems. These are the skills which we find necessary to carry out the third party overview. Through reference checks and/or discussions with NRC staff members familiar with the key personnel, we have verified their experience and competence in these areas.

Based upon its review, the staff concludes that the S&W organization and the individual overview team members are competent to conduct the Construction Implementation Overview and meet the technical competence standards set forth in the Ottinger/Dingell letter.

#### S&W Independence

The staff believes that for an organization to be acceptable to conduct this program the organization must be independent of the utility which owns Midland and independent of contractors whose work will be subject to the third party overview. Independence has been defined by the Comm.ssion as being the ability "... to provide an objective, dispassionate technical judgement, provided solely on the basis of technical merit...." (Page 1 of Response to Questions, attached to Ottinger/Dingell letter.) The Commission further defined the term by stating that the company approved to conduct an independent review must be one "...not previously involved with the activities...that they will now be reviewing..." Id.

<sup>1</sup>Reference Secy 82-414, "Diablo Canyon Design Verification Program Phase II Recommendations"

The staff has reviewed the information provided by CPCo and S&W regarding previous work performed by S&W for the Midland site and the principal contractors for the Midland project. Previous work at Midland consisted of limited activities (one person) in the planning phase of providing interface controls going from construction/preoperation testing into operations and is not considered to violate the independence criteria.

To the best of our knowledge, all the professional personnel assigned to work on the Midland Construction Implementation Overview have provided the NRC with sworn statements regarding their independence. S&W has stated that none of the staff expected to be assigned to the Midland review has any prior work experience with CPCo or on Midland.

Based on this information and the assessment of S&W to perform work as defined in Secy 82-414, the staff has no basis to believe that S&W is not sufficiently independent of CPCo.

The staff concludes that S&W and the key personnel who have been identified for the conduct of the review meet the standards of independence outlined in the Ottinger/Dingell letter.

#### S&W's Overview Program

The purpose of the independent third party overview is to provide additional assurance that the CCP is adequate and will be properly implemented. This overview requirement was necessitated by the loss of NRC staff confidence in CPCo to implement sucressfully the Quality Assurance Program. The CIO will remain in place at the Midland site until the necessary confidence level has been restored to the satisfaction of the NRC staff. CPCo also has the option to continue the CIO as an additional system of checks and balances, beyond any period of time required by the staff.

The written CIO program is controlled by site originated program documents and by S&W corporate program documents as follows:

- A. The documents written expressly for the CIO include:
  - CIO Program Document dated April 1, 1983
  - CIO Quality Assurance Plan
  - Third Party CIO Plan
  - CIO Assessment Procedure, 10.01
  - 2
- Nonconformance Identification and Reporting Procedure, 15.01
- A detailed attribute checklist for each CPCo Project Quality Control Instruction (PQCI)

- A detailed checklist to review generic types of requirements (for non-PQCI activities); e.g., QA Audits and Surveillances
- Additional Quality Control Instructions as needed to provide adequate overview control
- ₿.

The following S&W corporate master program documents will also be utilized for the CIO, as required:

- QA Topical Report SWSQAP 1-74A, S&W Standard Nuclear Quality Assurance Program
- S&W Quality Standards; e.g., for quality sampling
- S&W Quality Assurance Directives

The NRC met with S&W on August 25, 1983, to gain additional insight into the total S&W program. This meeting was held in Midland, Michigan and was open to the public. Questioning by the public on the CIO was permitted at the end of the meeting. Subsequent to this meeting, S&W submitted on August 30, 1983, to the NRC copies of the material presented at the August 25, 1983, public meeting and on September 9, 1983, submitted a summary of the program presented at that same meeting.

The program described by S&W in the above documents and at the August 25, 1983, meeting has been reviewed by the NRC staff and found to constitute an acceptable third party overview program. The CIO program will be audited independently by the S&W corporate QA staff from Boston and on a routine inspection effort by the NRC.

S&W personnel onsite for the CIO will vary with the demand of the work activities to be overviewed. S&W's CIO staffing plan currently has nine people assigned at the Midland site and there are currently planned increases to 32 people as work activities dictate. These numbers, however, are only estimates and S&W will commit whatever personnel is necessary to conduct the CIO. The number of personnel used is not subject to limitation by CPCO.

The S&W overview activities of the CCP to date have been somewhat limited, since the CCP has not yet been approved and work in progress is therefore limited. Activities being overviewed were pre-Phase I. The activities being overviewed have included the following CCP and non-CCP activities:

- Program and procedure review
- -

Review of MPQAD QA/QC personnel training and certification

-

Review of general training of CPCo and Bechtel personnel, including construction craftspersons

Review of CCP Management Reviews

## Review of System Interaction Walkdowns

Review of Design Documents

The above reviews have identified various concerns and one nonconformance that required CPCo actions to resolve. The NRC staff has reviewed the CIO activities performed to date and has found this overview, including actions taken by CPCo, to have been adequate.

#### Summary and Conclusion

.

Based on NRC review of the documentation submitted by CPCo and S&W, followup checks, and consideration of comments by members of the public, we conclude that S&W meets the independence and competence criteria for third party reviewers and that S&W's proposed CIO program is adequate to provide for an assessment of the Construction Completion Program (CCP).

Consumers Power Company

- 1

Should you have any questions regarding this letter please contact Mr. R. F. Warnick of my staff.

Sincerely,

2

mes & Keppler James G. Keppler Regional Administrator

di.

· P. E' %

Enclosure: As stated

-

cc w/encl: DMB/Document Control Desk (RIDS) Resident Inspector, RIII The Honorable Charles Bechhoefer, ASLB The Honorable Jerry Harbour, ASLB The Honorable Frederick P. Cowan, ASLB The Honorable Ralph S. Decker, ASLB William Paton, ELD Michael Miller Ronald Callen, Michigan Public Service Commission Myron M. Cherry Barbara Stamiris Mary Sinclair Wendell Marshall Colonel Steve J. Gadler (P.E.) Howard Levin (TERA) Billie P. Garde, Government Accountability Project Lynne Bernabei, Government Accountability Project



UNITED STATL NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

DCT 06 523

Docket Nos 50-329

50-330

(10 CFR 2.206)

Ms. Billie Pirner Garde Government Accountability Project Institute for Policy Studies 1091 Que Street, N.W. Washington, D.C. 20009

Dear Ms. Garde:

This is in response to your letter of June 12, 1983 on behalf of the Lone Tree Council and others, requesting that the Commission take a number of actions with respect to the Midland Plant. Your letter was treated as a request for action under 10 CFR 2.206 of the Commission's regulations.

For the reasons set forth in the enclosed "Director's Decision" under 10 CFR 2.206, your request has been granted in part and denied in part. A copy of the decision will be referred to the Secretary for the Commission's review in accordance with 10 CFR 2.206. For your information, I have also enclosed a mopy of the notice filed with the Office of the Federal Register for publication.

Sincerely.

. Richard C. DeYoung, Effector . Office of Inspection and Enforcement

=1 .

Enciosures: as stated

cc w/encl.: Consumers Power Company Michael Miller, Esq.

Dune a

#### UNITED STATES OF AMERICA NUCLEAR REGULATORY COMMISSION

OFFICE OF INSPECTION AND ENFORCEMENT Richard C. DeYoung, Director

In the Matter of CONSUMERS POWER COMPANY

Docket Nos. 50-329 50-330

=1

(Midland Nuclear Power Plant, Units 1 and 2)

(10 CFR 2.206)

## DIRECTOR'S DECISION UNDER 10 CFR 2.206

#### Incroduction

By letter to the Nuclear Regulatory Commission (NRC) dated June 13, 1983, Billie Pirner Garde of the Government Accountability Project, on behalf of the Lone Tree Council and others (hereinafter referred to as the petitioners), requested that, among other relief, the NRC take immediate action with regard to the Midland project. The letter was referred to the Director of the Office of Inspection and Enforcement for treatment as a request for action pursuant to 10 CFR 2.206 of the Commission's regulations.

On July 22, 1983, Edward L. Jordan, Acting Director of the Office of Inspection and Enforcement, acknowledged receipt of the petition and informed the petitioners that their request for immediate action was denied. Mr. Jordan noted that safety-related work at the Midland site had been stopped, with the exception of certain specified activities, and that the NRC staff was closely

831.0200386

following the current activities at the Midland site. Mr. Jordan further noted that Consumers Power Company had agreed not to proceed with implementation of a construction completion program until such a program had been reviewed by the NRC. The staff expected to be able to complete its evaluation of the request before final action was taken on that program. Consequently, Mr. Jordan concluded that "continuation of currently authorized activities at " Midland should not affect the staff's ability to grant the requested relief." Letter from Edward L. Jordan, Acting Director, Office of Inspection and Enforcement to Billie Pirner Garde (July 22, 1983). The staff has now completed its evaluation of the petition, and for the reasons stated herein, the request is granted in part and denied in part.

#### **Issues Raised**

1210 1 6

Petitioners requested that the following six actions be taken by the Commission:

Modify the Construction Permit (Midland Nuclear Power Plant, Units 1 and 2) to include mandatory "hold points" on the balance-of-plant (BOP) work and incorporate the current Atomic Safety and Licensing Board (ASLB or Board) ordered "hold points" on the soils remedial work into the Midland Construction permit (sic).

Require a management audit of Consumers Power Company (CPCo) by an independent, competent management auditing firm that will determine the causes of the management failures that have resulted in the soils settlement disaster and the recently discovered Quality Assurance breakdown.

Reject the Construction Completion Plan (CCP) as currently proposed, including a rejection of Stone and Webster to conduct the third party audit of the plant. Instead a truly independent, competent, and credible third party auditor should be selected with public participation in the process.

2
Remove the Quality Assurance/Quality Control function from the Midland Project Quality Assurance Department (MPQAD) and replace them with an independent team of QA/QC personnel that reports simultaneously to the NRC and CPCo management.

Increase the assignment of NRC personnel to include additional technical and inspection personnel as requested by the Midland Section of the Office of Special Cases.

Require a detailed review of the soils settlement resolution as outlined in the Supplemental Safety Evaluation Report, incorporating a technical analysis of the implementation of the underpinning project at the current stage of completion.

Petition at 1. The fifth issue relates to a matter of internal Commission organization and staffing, namely the allocation of staff to inspection of facilities. The staff is expecting to augment inspection personnel available to work on Midland. However, the creation of positions within the Office of Special Cases is a matter that will be determined by the Commission budget process. For these reasons, the staff is not considering this aspect of the request in this decision.

Background

-1

The Consumers Power Company (CPCo or licensee) holds Construction Permits No. CPPR-81 (Unit 1) and No. CPPR-82 (Unit 2), issued by the Atomic Energy Commission in 1972, which authorized construction of the Midland Plant. The Midland nuclear plant is located in Midland, Michigan, and consists of two pressurized water reactors of Babcock and Wilcox design and related facilities for use in the commercial generation of electric power.

Since the start of construction, Midland has experienced significant construction problems attributable to deficiencies in implementation of

its quality assurance (QA) program.  $\frac{1}{2}$  Following the identification of these problems, the licensee took action to identify the cause and correct each problem. Steps were also taken to upgrade the Midland QA program. Nevertheless, the licensee continued to experience problems in the implementation of its quality assurance program.

In 1980, the licensee reorganized its QA department so as to increase the involvement of high level CPCo management in onsite QA activities. Among its other tasks, the reorganized QA department, called the Midland Project Quality Assurance Department (MPQAD), was given the responsibility for quality control (QC) of heating, ventilation and air conditioning (HVAC) work in place of the HVAC contractor, Zack Company.

In May 1981, the NRC conducted a special, in-depth team inspection of the Midland site to examine the status of implementation and effectiveness of the QA program. Based on this inspection, Region III concluded that the newly

1/ Significant construction problems identified to date include:

13/2 -	DI 9WD52	splicing	deficie	ncies
1076	and the second second			

- 1976 rebar omissions
- 1977 bulge in the Unit 2 Containment Liner Plate
- 1977 tendon sheath location errors
- 1978 discovery of soil settlement problem
- 1980 Zack Company heating, ventilation, and air conditioning deficiencies
- 1980 reactor pressure vessel anchor stud failures
- 1981 piping suspension system installation deficiencies
- 1982 electrical cable misinstallations
- 1100014

Several of these deficiencies resulted in the Commission taking escalated enforcement action.

organized QA program was acceptable. See Inspection Reports 50-329/81-12; 50-330/81-12. The special team did, however, identify deficiencies in previous QC inspections of piping supports and restraints, and electrical cable installations.<sup>2/</sup> QC functions were further reorganized by the licensee's integration of the QC organization of its architect-engineer, Bechtel Power Corporation, into MPQAD in September 1982. This reorganization reflected the recommendations of the NRC staff. As part of this change, the licensee also undertook to retrain and recertify all previously certified Bechtel QC inspectors.

Nevertheless, construction difficulties continued to be identified at the Midland site. An inspection conducted during the period of October 1982 through January 1983 found significant problems with equipment in the diesel generator building. The subsequent identification of similar findings by CPCo in other portions of the plant prompted the licensee to halt the majority of the safety related work activities in December 1982. In view of the history of QA problems at the Midland plant and the lack of effectiveness of corrective actions to implement an adequate quality assurance program, the NRC indicated to the licensee that it was necessary to develop a comprehensive program to verify the adequacy of previous construction activities and to assure the adequacy of future construction. In view of the licensee's performance history, such an

2/2 As a result of staff discussions about the seriousness of such findings and of similar indications of deficiencies as identified in the Systematic Assessment of Licensee Performance Report issued in April 1982, a special Midland Section in Region III was formed in July 1982. The Midland Section devoted increased attention to inspection of the Midland facility, including upgrading the QC program of the project's constructor, the Bechtel Power Corporation.

effort was necessary to restore staff's confidence in CPCo's ability to properly construct the Midland plants.

Consequently, CPCo discussed with the NRC the concept of a construction completion program which would address the concerns raised by the staff. These discussions were followed by a formal submittal of the Midland Construction Completion Program (CCP).

The CCP is the licensee's program for the planning and management of the construction and quality activities necessary for its completion of the construction of the Midland facility. An important aspect of the CCP is the third party overview, which is designed to provide additional assurance as to the effectiveness of the CCP. In response to comments from the NRC and members of the public, the CCP underwent several revisions. As revised and submitted by the licensee on August 26, 1983,<sup>4/</sup> the CCP includes: (1) NRC hold points; (2) the requirement for 100% reinspection of accessible installations; (3) the integration of Bechcel's QC program with MPQAD; (4) the retraining and recertification of QC inspectors; (5) the general training of licensee and contractor personnel in quality requirements for nuclear work, requirements of the CCP, safety orientation and inspection, and work procedures; (6) the revision, as necessary, of Project Quality Control Instructions (PQCI's); (7) CCP team training; and (8) an independent third party overview of CCP activities.

<sup>4/</sup> The Petition was apparently based upon the June 3, 1983 version of the CCP. Subsequent versions of the CCP, as described in this decision, address a number of issues raised by petitioners.

The CCP is divided into two phases. Phase 1 consists of a systematic review of the safety-related systems and areas of the plant. This review will be conducted on an area-by-area basis and will be done by teams with responsibility for particular systems. Phase 1 is intended to provide a-clear identification of remaining installation work, including any necessary rework and an up-to-date inspection to verify the quality of existing work.

Phase 2 will take the results of the Phase 1 review and complete any necessary work or rework, thereby bringing the project to completion. The teams organized for Phase 1 activities will continue as the responsible organizational units to complete the work in Phase 2.

It should be noted that the CCP does not include the remedial soils program, nuclear steam supply system installation, HVAC installation, and the reinspection of pipe hangers and electrical cable. The remedial soils act: ities are being closely inspected under the conditions of the construction permits which implement the Atomic Safety and Licensing Board's April 30, 1982, order and under a work authorization procedure. Therefore, the staff does not consider it necessary to require the remedial soils activities to be included in the CCP. Controls over the soils work have been implemented under a separate program. Similarly, reinspection of the pipe hangers and electrical cable were not included in Phase I of the CCP because that reinspection is being done under a separate commitment to the NRC. See letters from James G. Keppler, Regional Administrator, NRC Region III to James W. Cook, Consumers Power Company (August 30, September 2, 1982). Nuclear Steam Supply System installation and HVAC installation were not drawn into question by the diesel generator building inspection.

The staff has not developed facts to indicate that installation of these systems should be included in the CCP. However, these activities will be included in the construction implementation overview to be conducted by the third party overviewer.

The CCP is designed to address the generic applicability of the problems identified by the NRC's inspection of the diesel generator building. The objective of the CCP is to look at the plant hardware and equipment, identify existing problems, correct these problems and complete construction of the plant.

## Consideration of Issues Raised

-1

# 1. Modification of Midland Construction Permits

Petitioners request that the Commission modify the Midland construction permits in two respects: 1) require "hold points" at various stages of the construction completion process; and, 2) incorporate those hold points concerning remedial soils work previously authorized by the Atomic Safety and Licensing Board panel with jurisdiction over the Midland proceeding.

The hold points are fundamental elements of the Midland CCP. As used by both the staff and petitioners, hold points refer to predetermined stages beyond which activities cannot proceed until authorized. Only when such prior work is found to be satisfactory will new work be authorized under the CCP. In this regard, the petitioners requested that three specific hold points be incorporated into the CCP to require NRC or third party review prior to continuation of work.

Based on their review of an early version of the CCP, petitioners asserted that the Midland project had been detrimentally affected by the lack of organizational freedom for its QA staff. See Petition at 13. Accordingly. the petitioners requested that a hold point be incorporated into the CCP whereby the success of the proposed program for the retraining and recertification of QA/QC personnel would be evaluated before any actual work was authorized under Phase 1 of the CCP. Id. at 13, 15. Subsequent to its initial discussions with the staff concerning development of a comprehensive construction completion program,  $\frac{5}{}$  the licensee began preliminary work, such as team training and recertification of QC inspectors in preparation for its anticipated Phase 1 activities, quality verification program and status assessments. The NRC was informed when training and recertification of QA/QC personnel and CCP team training would begin, and conducted a review of the licensee's actions. The staff suggested that the licensee undertake additional work before proceeding with some of its training effort. Consequently, ... e retraining hold point requested by petitioners has already been satisfied by the staff.

- Review and approval of training and recertification of QC inspectors before beginning Phase 1;
- 2. neview and approval of CCP team training before beginning Phase 1;
- 11.00
- Review and approval of the Quality Verification Program (QVP) and status assessments before beginning Phase 1;
- 4. Review and approval of the program for rework or systems completion work before beginning Phase 2.

<sup>5/</sup> On December 2, 1982, when CPCo first discussed a construction completion plan with the NRC staff, CPCo was informed by Region III staff that it would be necessary to incorporate NRC hold points. The staff identified four points at which it would require NRC inspectors to review completed work before the next activity could be undertaken. These hold points were identified as:

The petitioners also viewed the proposed CCP as lacking in comprehensiveness. To remedy this deficiency, petitioners proposed that "either a third party or NRC 'hold point' be contained in the reinspection Phase I activities [of the CCP] to determine the adequacy of the 'accessible systems' approach." $\frac{6}{}$  Petition at 13.

As described in section three, <u>infra</u>, a third party will be conducting an extensive overview of the CCP and other construction completion activities. The fact that the third party overviewer will also have hold point controls over the licensee should provide additional assurance that construction is proceeding in accordance with all applicable requirements. <u>See</u> Consumers Power Company, Construction Completion Program (August 26, 1983) at 34. The NRC and the third party will monitor the reinspection activities. The staff believes that these monitoring activities will provide the control sought by the petitioners in their request to establish a hold point during Phase 1 reinspection to determine the adequacy of the accessible systems approach.

The third hold point requested by petitioners derives from another criticism of the proposed CCP - the failure of that plan to specify inspection procedures and evaluation criteria. <u>See</u> Petition at 10-11. Accordingly, petitioners request a systematic and thorough review of the construction and quality work packages which will be completed as a prerequisite to initiation of new construction work under Phase 2 of the CCP. <u>Id.</u> at 11.

<sup>6/</sup> The accessible systems approach refers to the extent of reinspection under the CCP. Inaccessible areas of the plant will be reinspected by utilizing a records review and destructive and non-destructive testing as required. See Consumers Power Company, Construction Completion Program (August 26, 1983) at 22-23.

The CCP requires that representative construction and quality work packages be reviewed to assure that any completed work is consistent with statements made by the licensee in both its Final Safety Analysis Report and Quality Assurance Fopical Report. In addition, the third party overviewer will be using sampling techniques and reviewing selected work and quality packages prior to and during Phase II. Should the results of this sampling approach identify inadequate work packages, the sampling size will be increased as necessary to provide the needed assurance that work packages are adequately reviewed. Moreover, the NRC staff, in performing its inspection activities, will overview this entire process, including reviewing selected quality and work packages.

In summary, the staff believes that those hold points it has incorporated into the CCP, when viewed in the aggregate, substantially satisfy the hold points requested by petitioners. The licensee is required to adhere to these hold noints as part of the CCP in conformance with the Confirmatory Order for Modification of Construction Permits (Effective Immediately).

With respect to the second aspect of the requested relief, incorporation of NRC hold points authorized by the Licensing Board's April 30, 1982, Memorandum and Order, the petitioners' request has been satisfied by previous action of the Commission. By amendment dated May 26, 1982, the hold points ordered by the Board were incorporated into the construction permits. See 47 Fed. Reg. 23999 (June 2, 1982). Accordingly, the construction permits already prohibit CPCo from performing the following activities without "explicit prior approval" from the staff:

 (a) any placing, compacting, excavating, or drilling soil materials around safety-related structures and systems;

- (b) physical implementation of remedial action for correction of soil-related problems under and around safety-related structures and systems, including but not limited to:
  - (i) dewatering systems
  - (ii) underpinning of service water building
- (iii) removal and replacement of fill beneath the feedwater isolation valve pit areas, auxiliary building electrical penetration areas and control tower, and beneath the turbine building
- (iv) placing of underpinning supports beneath any of the structures listed in (iii) above
- (v) compaction and loading activities;
- (c) construction work in soil materials under or around safety-related structures and systems such as field installation, or rebedding, of conduits and piping.

Construction Permits No. CPPR-81 and CPPR-82, Amendment No. 3 (May 26, 1982).

## 2. Management audit of CPCo

- 1

The petitioners request that the NRC require a management audit of CPCo's performance on the Midland project. The staff does not believe that a management audit is necessary at this time as a condition for going forward with the CCP. The staff expects that the CCP, with its built-in hold points and third party overview, should provide an effective process to satisfactorily complete construction at Midland, without the previous quality assurance problems. The third party overview together with the planned staff inspection activities should provide information to determine the adequacy of the licensee's implementation of the CCP. Nevertheless, the staff will continue to review information concerning the licensee's performance in other'areas to determine whether an audit is required.

## Rejection of Construction Completion Program and Third Party Overview Organization

In requesting that the Commission reject the Midland construction completion plan, petitioners based their position on the unacceptability of the Stone and Webster Engineering Corporation (S&W) to conduct the third party overview of the CCP. Petitioners raised three objections to the selection of S&W: the failure of S&W to meet the Commission's criteria for the independence required of a third party see Petition at 19; the failure of S&W to submit a minimally adequate audit proposal, <u>id</u>. at 18-19; and the lack of public participation in the selection of S&W as the third party review organization for the Midland project. <u>Id</u>. at <u>18-20</u>.

In support of fis argument that S&W is not sufficiently independent to monitor implementation of the CCP, the petitioners asserted that "under both a literal and realistic reading of the Commission's primary financial criteria, ... the third party not have any direct previous involvement with the Company." Petition at 19. In order to evaluate whether an audit organization is sufficiently independent to conduct a third party review, the Commission generally utilizes the guidance originally set forth in a letter from Chairman Palladino to Representatives Ottinger and Dingell. The Commission's standard does not require that a proposed third party reviewer have had no previous involvement with the utility whose program it will be reviewing. Rather, the criteria require that the audit organization, including those employees who will be participating in the third party review, will not be reviewing specific

activities in which they were previously involved. See Letter from Chairman Palladino to Representatives Ottinger and Dingell (Feb. 1, 1982), Attachment 1, at 1. Petitioners stated that S&W's role as the overviewer of remedial Soils work at Midland prohibits that organization from serving in the same capacity for the CCP. The staff disagrees. Since the remedial soils activities are outside the scope of the CCP, S&W will not be called upon to review its own work. Consequently, the staff does not agree that S&W's overview activities will conflict with the established independence criteria. $\frac{7}{}$ 

The petitioners questioned why TERA was disqualified from consideration as the overviewer under the CCP while S&W was not disqualified on the ground of independence. See Petition at 19. TERA's disqualification was based on the potential for conflict that could be raised by TERA overview under the CCP of determinations that TERA had previously made under the Independent Design and Construction Verification Program (IDCVP) of the adequacy of the construction of the Auxiliary Feedwater System, the onsite emergency AC power supplies and the HVAC system for the control room. Since TERA has been approved by the NRC to perform the IDCVP, the staff determined that TERA would not satisfy the Commission independence criteria for the third party overview of the CCP. See letter from James G. Keppler, Regional Administrator, Region III to James W. Cook, Consumers Power Company (March 28, 1983) at 3.

The written program documents being utilized to directly control and implement the Construction Implementation Overview (CIO) program  $\frac{8}{2}$  and the applicable S&W corporate master program documents  $\frac{9}{2}$  have been reviewed by the staff. These documents are representative of the scope and depth of the S&W overview. The NRC staff also met with S&W on August 25, 1983, in Midland, Michigan in order to gain additional insight into the total S&W program. Based upon its document review and discussions with S&W at the August 25, 1983, meeting, the staff has found the S&W proposal to constitute an acceptable third party overview program. To provide additional assurance that the third party audit is being properly implemented, the CIO program will also be audited independently by the S&W corporate cuality assurance staff. NRC inspectors will also monitor the adequacy of the CIO program.

8/ The documents written expressly for the CIO include:

- CIO Program Document dated April 1, 1983.
- 2. CIO Quality Assurance Plan.
- 3. Third Party CIO Plan.

ç.ee

- 4. CIO Assessment Procedure, 10.01.
- 5. Nonconformance Identification and Reporting Procedure, 15.01.
- A detailed attribute checklist for each CPCo Project Quality Control Instruction (PQCI).
- A detailed checklist to review generic types of requirements (for non-PQCI activities); e.g., QA Audits and Surveillances.
- Additional Quality Control Instruction as needed to provide adequate overview control.
- 9/2 The following S&W corporate master program documents will also be utilized for the CIO, as required:
  - QA Topical Report SWSQAP 1-74A, S&W Standard Nuclear Quality Assurance Program.
  - 2. Saw Quality Standards; e.g., for quality sampling.
  - 3. S&W Quality Assurance Directives.

Of particular concern to the petitioners was the number of personnel which S&W had assigned to the Midland overview. <u>See</u> Petition at 18. The number of qualified people will vary with the demand of the work activities to be overviewed. S&W's CIO staffing plan currently has nine people assigned at the Midland site and there are planned increases to 32 people as work activities progress. These numbers, however, are only estimates and S&W has represented that it will commit whatever personnel are necessary to conduct the CIO. Furthermore, the number of personnel utilized by S&W is not subject to limitation by CPCo.

Saw has already begun to review preliminary activities of the licensee in preparation for initiation of the CCP. $\frac{10}{}$  This effort has identified various concerns and one nonconformance that required CPCo action to resolve. The NRC staff has reviewed the CIO activities performed to date and has found this overview, including actions taken by CPCo, to be of the quality expected of a third party overview.

- . Program and procedure reviews.
- Review of PQCI's.

(Jag

- . Review of MPQAD QA/QC personnel training and certification.
- Review of general training of CPCo and Bechtel personnel, including construction craftspersons.
- . Review of CCP Management Reviews.
- . Review of System Interaction Walkdowns.
- . Review of Design Documents.

<sup>10/</sup> The activities being overviewed have included the following CCP and non-CCP activities:

The purpose of the independent third party overview is to provide additional assurance that the CCP is adequate and will be properly implemented. This overview requirement was necessitated by the loss of NRC staff confidence in CPCo to successfully implement a quality assurance program for the Midland project. The CIO will remain in place at the Midland site until the necessary level of confidence in the ability of the licensee to construct the Midland project has been restored to the satisfaction of the NRC staff.  $\frac{11}{}$  Given that the third party overview is expected to continue until NRC confidence in the Midland project is restored, petitioners' criticism that the CIO is of insufficient duration appears unfounded.

Opportunity has been provided to the public to participate in the selection of S&W as the third party overviewer, and to comment on the CCP itself. A meeting was held on February 8, 1983, between CPCo and the staff to discuss the CCP. On August 11, 1983, the staff met with the intervernors, representatives of the Government Accountability Project (GAP) and the Lone Tree Council to discuss the CCP and the CIO. Subsequently, on August 25, 1983, the staff met with S&W to discuss the CIO. These meetings were conducted in Midland, Michigan and were open to public observation. Evening sessions to receive public comments regarding the CCP were held on February 8, and August 11, 1983. Similarly, public comments were received following the August 11 and August 25, 1983, meetings. Several additional meetings between the staff, intervenors and a representative of GAP to discuss the CCP and CIO have also been held.

11/ The staff anticipates that the third party overview will be a long term. effort.

The petitioners' reference in its request to "closed door" meetings appears to refer to working level meetings that have been held principally between the Midland section of the Region III staff and CPCo site personnel, and, in some cases, S&W onsite personnel. <u>See</u> Petition at 19. Such meetings continue to be necessary to enable the NRC staff to achieve a full understanding of the CCP, including the CIO, and to discharge its inspection duties.

For the reasons set forth above, petitioners' request to reject the selection of S&W to conduct the CIO, and to reject the CCP, is denied.  $\frac{12}{}$ 

# 4. <u>Removal of the Licensee from Primary Responsibility for the Midland</u> <u>Quality Assurance Program</u>

The petitioners request that MPQAD be relieved of responsibility for the QA/QC function at the Midland plant and that an independent team of QA/QC personnel be created which would report simultaneously to the NRC staff and CPCo. In support of their request, petitioners cite much of the same history of QA/OC deficiencies that the staff summarized in the background section of this decision. See Petition at 20.

1

1. 1 . 1. 1.

<sup>12/</sup> The staff has approved S&W to conduct the CIO. See Staff Evaluation of Consumers Power Company Proposal to Use Stone and Webster Michigan, Inc. to Conduct the Third Party Construction Implementation Overview of the Midland Nuclear Plant (Sept. 29, 1983).

The changes that CPCo has most recently instituted through development of the CCP should improve its capability to discharge its responsibility under applicable Commission regulations, such as 10 CFR 50.34(a)(7) and Appendix B to 10 CFR Part 50, which require the establishment and execution of a QA/QC program. While Criterion I of Appendix B permits a construction permit holder to delegate to other organizations the detailed execution of the QA/QC program, the history of the Midland project makes it clear that the licensee has retained too little control over the QA/QC program. CPCo seems to be proceeding in a positive direction by integrating the implementation of the QC function formerly under the control of Bechtel into the MPQAD. This consolidation of quality control and quality assurance functions should reinforce the separation between the QC function, which will be assumed by MPQAD, and the construction function, which will remain with Bechtel.

While it might be permissible under Appendix B to 10 CFR Part 50 for CPCo to retain an independent organization to execute the QA/QC program, the licensee remains ultimately responsible for the establishment and execution of the program. As stated above, the staff considers the strengthening of MPQAD to be a positive step in improving CPCo's capability to assure the quality of construction of the Midland facility. In view of the relatively short existence of the MPQAD, there does not currently exist any justification for requiring CPCo to retain an outside organization to execute the QA/QC program. Therefore, this aspect of petitioners' request is denied.

Petitioners also requested that the independent QA/QC team report simultaneously to the NRC and to CPCo management. The petitioners apparently intended that

the NRC would be involved in making management decisions regarding construction of the facility based upon the reports of the independent QA/QC team. There appears to be no basis for this extraordinary departure from the NRC's regulatory function. Accordingly, this aspect of the petition is denied.

# 5. Detailed Review of Soils Settlement Resolution

The petitioners requested that the staff conduct a detailed review of the resolution of the soils settlement problems, including a technical analysis of the implementation of the underpinning project at the current stage of completion. Petition at 23. In its supporting discussion, the petition focused upon the questionable structural integrity of the diesel generator building.

A detailed review of the program for resolution of the soils settlement problem has previously been conducted by the NRC staff and its consultants. In 1979 the U.S. Army Corps of Engineers was contracted to assist the staff in the safety review of the Midland project in the field of geotechnical engineering. After the soils problem became known, additional assistance to the staff in specialized engineering fields (structural, mechanical, and underpinning) was obtained from the U.S. Naval Surface Weapons Center, Harstead Engineering Associates, Geotechnical Engineers, Inc., and Energy Technology Engineering Center. These consultants assisted in the review of technical studies, participated in design audits, visited the site, provided input to the Safety Evaluation Report, and provided expert testimony before the Atomic Safety and

Licensing Board. Thus, the approach to the resolution of the soils settlement issue has been thoroughly studied by the staff and its consultants.

The implementation of the remedial soils activities is being closely followed as part of the NRC's inspection program. This inspection effort includes ongoing technical review of the remedial soils program and its implementation by a Region III soils specialist. Technical expertise to evaluate implementation is also provided by the NRC's Office of Nuclear Reactor Regulation. Additionally, the NRC is utilizing Geotechnical Engineers Inc. in assessing aspects of the remedial soils and underpinning activities. In addition, the soils settlement question has been in litigation for over two years before an Atomic Safety and Licensing Board. Consequently, the relief requested with regard to the soils settlement issue has been substantially satisfied by prior action of the Commission.

Along with review of the soils settlement issue, petitioners requested that another study of the seismic design deficiencies of the Midland plant, with emphasis on the diesel generator building, be conducted. The petitioners further requested that this review would be conducted by a "non-nuclear construction consultant." See Petition at 23.

The NRC staff has initiated a task force study by consultants from Brookhaven National Laboratory (BNL) and NRC structural engineers to evaluate concerns about the structural integrity of the diesel generator building faised by a NRC Region III inspector in testimony before the Subcommittee on Energy and the Environment of the House Committee on Interior and Insular Affairs. Following their review, a report will be issued addressing the concerns raised by the inspector. Decisions on whether further actions are required will be

made based upon that report. Additional details on the task force were provided to the Government Accountability Project by letter dated August 10, 1983, and in Board Notifications 83-109 and 83-142, which were transmitted to GAP on July 27 and September 22, 1983, respectively.

As to the request that a review of the diesel generator building be conducted by a "non-nuclear construction consultant", BNL has established an expert team to resolve the concerns raised by the Region III inspector. Expertise rather than the label "non nuclear construction consultant" should be the governing criteria. The staff has reviewed the qualifications of the team members and is satisifed with their experience. The task force study currently in progress substantially satisfies this aspect of the petition.

The petition also appears to be requesting an additional review of the seismic design of structures other than the diesel generator building. Petitioners have not, however, stated any basis why additional reviews beyond those reflected in the Safety Evaluation Report and Supplements are necessary. The staff does not believe that an additional review by an outside organization of the facility's seismic design is required at this time.

## Conclusion

Based upon the foregoing discussion, I have granted the petition in part and denied it in part.

A copy of this decision will be filed with the Office of the Secretary of the Commission for the Commission's review in accordance with 10 CFR 2.206(c) of the Commission's regulations. This decision will become the final action of the Commission twenty-five days after date of issuance unless the Commission, on its own motion, institutes a review of the decision within that time.

Se.

Richard C. DeYoung, Director Office of Inspection and Enforcement

Dated at Bethesda, Maryland, this 6thday of October 1983

omit La

P. C. 4

[7590-01] -

NUCLEAR REGULATORY COMMISSION [Docket Nos. 50-329 and 50-330] CONSUMERS POWER COMPANY (Midland Plant, Units 1 and 2)

## ISSUANCE OF DIRECTOR'S DECISION UNDER 10 CFR 2.206

Notice is hereby given that the Director, Office of Inspection and Enforcement, has issued a decision concerning a petition dated June 13, 1983, filed by Billie Pirner Garde of the Government Accountability Project on behalf of the Lone Tree Council and others. The petitioners had requested that the Commission take a number of actions with respect to the Midland Plant. The Director, Office of Inspection and Enforcement, has decided to grant in part and deny in part the petitioners' request.

The reasons for this decision are explained in a "Director's Decision" under 10 CFR 2.206 (DD-83-16), which is available for public inspection in the Commission's Public Document Room, 1717 H Street, N.W., Washington, D.C. 20555, and in the Local Public Document Room for the Midland Plant, located at the Grace Dow Memorial Library; 1910-K. St. Andrews Road, Midland, Michtgan, 48640.

Dated at Bethesda, Maryland this 6th day of October, 1983.

\$310200392

FOR THE NUCLEAR REGULATORY COMMISSION

Richard C. Peyoung, Director Office of Inspection and Enforcement



UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON; D. C. 20555

-OCTOBER 6 1983

Docket No. 50-329 50-330

Mr. James W. Cook. Vice President Consumers Power Company 1945 West Parnall Road Jackson, Michigan 49201

Dear Mr. Cook:

0.40

-

23102005

Enclosed please find a Confirmatory Order for Modification of Construction Permits (Effective Immediately) for the Midland Plant issued this day. In accordance with 10 CFR 2.790 of the NRC's "Rules of Practice," Part 2: Title 10, Code of Federal Regulations, a copy of this letter and the enclosure will be placed in the NRC's Public Document Room.

Sincerely:

Richard C: Defoung, Director Office of Inspection and Enforcement

1.1.1

End caure: Confirmatory Order

cc: Michael Miller, Esq. Billie Pirner Garde, Government Accountability Project

## UNITED STATES NUCLEAR REGULATORY COMMISSION

In the Matter of

CONSUMERS POWER COMPANY (Midland Plant Units 1 and 2) Docket No. 50-329 50-330 EA-83-109

## CONFIRMATORY ORDER FOR MODIFICATION OF CONSTRUCTION PERMITS (EFFECTIVE IMMEDIATELY)

I

Consumers Power Company (the "licensee") is the holder of construction permits CPPR-81 and CPPR-82 issued by the Atomic Energy Commission (now the Nuclear Regulatory Commission, hereafter "Commission"), which authorize the construction of the Midland Plant, Units 1 and 2 (the "facility"). The facility is under construction in Midland, Michigan.

#### II

Since the start of construction, the facility has experienced significant quality assurance ("QA") problems. Although the licensee took corrective actions in each case, problems continued to be experienced in the implementation of its QA program.

An NRC Region III inspection, commenced in October 1982 and completed in January 1983, identified significant problems with the QA inspection process and with the conformance to design documents of installed components in the Diesel Generator Building ("DGB"). These findings were identified to the licensee in an exit meeting following the inspection in November 1982. The licensee subsequently made similar findings in other areas of the facility. In view of 1) the widespread nature of the problems identified, 2) the history of QA problems at the facility,

- 33102 pp 380

and 3) the ineffectiveness of past corrective actions to resolve these problems, the NRC staff requested the licensee to develop a comprehensive program to verify the adequacy of previous construction and to assure the adequacy of future construction. On December 2, 1982, the licensee directed that the majority of safety related work at the site be halted and presented to the staff the outlines of a Construction Completion Program ("CCP"). By letter dated December 30, 1982, the NRC confirmed the licensee's stopping work and other commitments undertaken by the licensee. In accordance with those commitments, the CCP was formally submitted to the staff on January 10, 1983.

The CCP is a program to provide guidance in the planning and management of the construction and QA activities necessary for completion of the facility in accordance with Commission regulations. The CCP has undergone revisions in response to questions and comments raised by the staff and by members of the public and was submitted in final form on August 26, 1983.

Part of the CCP is a Construction Implementation Overview ("CIO") to be conducted by an independent third party. The CIO effort is described in the CCP and documents provided to NRC on April 6 and 11, May 19, August 30 and September 9, 1983.

6.17.4

- 2 -

The CIO was necessitated by the NRC staff's loss of confidence in the licensee alone to implement an effective QA program. In response to this concern, the licensee has committed to keep the CIO in effect until the licensee has demonstrated to the NRC staff that a third party overview is no longer necessary to provide reasonable assurance that the facility can be constructed in compliance with the Commission's QA criteria (10 CFR Part 50, Appendix B). The licensee has proposed and the staff has approved, by letter dated September 29, 1983, Stone and Webster Engineering Corporation to perform the CIO.

#### III

The NRC staff has conducted a review of the CCP and has concluded that it constitutes a program which provides reasonable assurance that the facility can be satisfactorily completed in accordance with Commission requirements. I have concluded that the activities halted by the licensee on December 2, 1982, may resume provided they are conducted in accordance with the CCP. I, therefore, find that the public health, safety and interest requires that any continuation of construction be in accordance with the CCP and that the CCP be confirmed by order made immediately effective.

#### IV

Accordingly, pursuant to Sections 103 and 161i of the Atomic Energy Act of 1954, as amended, and the Commission's regulations in 10 CFR Parts 2 and 50, Construction Permits CPPR-81 and CPPR-82 are hereby modified to include the following provisions:

- 3 -

a. The licensee shall adhere to the Construction Completion Program, dated August 26, 1983, for the duration of construction of the facility.

- 4 -

- b. The licensee shall maintain in effect the Construction Implementation Overview provision of the Construction Completion Program with the Stone and Webster Engineering Corporation as the third party overviewer until the Regional Administrator, NRC Region III, finds in writing that the third party overview is no longer necessary to provide reasonable assurance that the facility can be constructed in compliance with 10 CFR Part 50.
- c. The licensee may make changes to the Construction Completion Program provided such changes (1) do not decrease its effectiveness, (2) are submitted to the Regional Administrator with appropriate justification, and (3) are approved in writing by the Regional Administrator prior to their implementation.

٧

The licensee may request a hearing on this Order within 25 days of the Edate of this Order. Any request for hearing shall be submitted to the Director, Office of Inspection and Enforcement, U.S. Nuclear Regulatory

Commission, Washington, D.C. 20555. A copy of the request shall also be sent to the Executive Legal Director at the same address and to the Regional Administrator, NRC Region III, 799 Roosevelt Road, Glen Ellyn, - Illinois 60137. A REQUEST FOR HEARING SHALL NOT STAY THE IMMEDIATE EFFECTIVENESS OF SECTION IV OF THIS ORDER.

If a hearing is to be held concerning this Order, the Commission will issue an order designating the time and place of hearing. If a hearing is held, the issue to be considered at such hearing shall be whether this Order should be sustained.

FOR THE NUCLEAR REGULATORY COMMISSION

Richard C. DeYoung, Director

Office of Inspection and Enforcement

Dated at Bethesda, Maryland, this 6<sup>24</sup> day of October, 1983

3

- 5

#### UNITED STATES OF AMERICA NUCLEAR REGULATORY COMMISSION

## OFFICE OF INSPECTION AND ENFORCEMENT Richard C. DeYoung, Director

In the Matter of CONSUMERS POWER COMPANY

Docket Nos. 50-329 50-330

(Midland Nuclear Power Plant, Units 1 and 2)

(10 CFR 2.206)

## DIRECTOR'S DECISION UNDER 10 CFR 2.206

Introduction

By letter to the Nuclear Regulatory Commission (NRC) dated June 13, 1983, Eillie Pirner Garde of the Government Accountability Project, on behalf of the Lone Tree Council and others (hereinafter referred to as the petitioners), requested that, among other relief, the NRC take immediate action with regard to the Midland project. The letter was referred to the Director of the Office of Inspection and Enforcement for treatment as a request for action pursuant to 10 CFR 2.206 of the Commission's regulations.

On July 22, 1983, Edward L. Jordan, Acting Director of the Office of Inspection and Enforcement, acknowledged receipt of the petition and informed the petitioners that their request for immediate action was denied. Mr. Jordan noted that safety-related work at the Midland site had been stopped, with the exception of certain specified activities, and that the NRC staff was closely



following the current activities at the Midland site. Mr. Jordan further noted that Consumers Power Company had agreed not to proceed with implementation of a construction completion program until such a program had been reviewed by the NRC. The staff expected to be able to complete its evaluation of the request before final action was taken on that program. Consequently, Mr. Jordan concluded that "continuation of currently authorized activities at Midland should not affect the staff's ability to grant the requested relief." Letter from Edward L. Jordan, Acting Director, Office of Inspection and Enforcement to Billie Pirmer Garde (July 22, 1953). The staff has now completed its evaluation of the petition, and for the reasons stated herein, the recuest is granted in part and denied in part.

#### Issues Raised

Petitioners requested that the following six actions be taken by the Commission:

Modify the Construction Permit (Midland Nuclear Power Plant, Units 1 and 2) to include mandatory "hold points" on the balance-of-plant (BOP) work and incorporate the current Atomic Safety and Licensing Board (ASLB or Board) ordered "hold points" on the soils remedial work into the Midland Construction permit (sic).

Require a management audit of Consumers Power Company (CPCo) by an independent, competent management auditing firm that will determine the causes of the management failures that have resulted in the soils settlement-disaster and the recently discovered Quality Assurance breakdown.

Reject the Construction Completion Plan (CCP) as currently proposed, including a rejection of Stone and Webster to conduct the third party audit of the plant. Instead a truly independent, competent, and credible third party auditor should be selected with public participation in the process.

Remove the Quality Assurance/Quality Control function from the Midland Project Quality Assurance Department (MPQAD) and replace them with an independent team of QA/QC personnel that reports simultaneously to the NRC and CPCo management.

Increase the assignment of NRC personnel to include additional technical and inspection personnel as requested by the Midland Section of the Office of Special Cases.

Require a detailed review of the soils settlement resolution as outlined in the Supplemental Safety Evaluation Report, incorporating a technical analysis of the implementation of the underpinning project at the current stage of completion.

Petition at 1. The fifth issue relates to a matter of internal Commission organization and staffing, namely the allocation of staff to inspection of facilities. The staff is expecting to augment inspection personnel available to work on Midland. However, the creation of positions within the Office of Special Cases is a matter that will be determined by the Commission budget process. For these reasons, the staff is not considering this aspect of the request in this decision.

## Background

The Consumers Power Company (CPCo or licensee) holds Construction Permits No. CPPR-81 (Unit 1) and No. CPPR-82 (Unit 2), issued by the Atomic Energy Commission in 1972, which authorized construction of the Midland Plant. The Midland nuclear Plant is Tocated in Midland, Michigan, and consists of two pressurized water reactors of Babcock and Wilcox design and related facilities for use in the commercial generation of electric power.

Since the state of construction, Midland has experienced significant construction problems attributable to deficiencies in implementation of

its quality assurance (QA) program.  $\frac{1}{}$  Following the identification of these problems, the licensee took action to identify the cause and correct each problem. Steps were also taken to upgrade the Midland QA program. Nevertheless, the licensee continued to experience problems in the implementation of its quality assurance program.

In 1980, the licensee reorganized its QA department so as to increase the involvement of high level CPCo management in onsite QA activities. Among its other tasks, the reorganized QA department, called the Midland Project Quality Assurance Department (MPQAD), was given the responsibility for quality control (OC) of heating, ventilation and air conditioning (HVAC) work in place of the HVAC contractor, Zack Company.

In May 1981, the NRC conducted a special, in-depth team inspection of the Midland site to examine the status of implementation and effectiveness of the QA program. Based on this inspection, Region III concluded that the newly

Significant construction problems identified to date include:

1/

1973	-	cadweld splicing deficiencies
197.6	-	rebar omissions
1977	-	bulge in the Unit 2 Containment Liner Plate
1977	-	tendon sheath location errors
1978	-	discovery of soil settlement problem
1980	-	Zack-Company-heating, ventilation, and air conditionin
		deficiencies
1980	- 1	reactor pressure vessel anchor stud failures
=1981	-	piping suspension system installation deficiencies
1982	-	electrical cable misinstallations

Several of these deficiencies resulted in the Commission taking escalated enforcement action.

organized QA program was acceptable. See Inspection Reports 50-329/81-12; 50-330/81-12. The special team did, however, identify deficiencies in previous QC inspections of piping supports and restraints, and electrical cable installations.<sup>2/</sup> QC functions were further reorganized by the Ticensee's integration of the QC organization of its architect-engineer, Bechtel Power Corporation, into MPQAD in September 1982. This reorganization reflected the recommendations of the NRC staff. As part of this change, the licensee also undertook to retrain and recertify all previously certified Bechtel QC inspectors.

Nevertheless, construction difficulties continued to be identified at the Micland site. An inspection conducted during the period of October 1982 through January 1983 found significant problems with equipment in the diesel generator building. The subsequent identification of similar findings by CPCo in other portions of the plant prompted the licensee to halt the majority of the safety related work activities in December 1982. In view of the history of QA problems at the Midland plant and the lack of effectiveness of corrective actions to implement an adequate quality assurance program, the NRC indicated to the licensee that it was necessary to develop a comprehensive program to verify the adequacy of previous construction activities and to assure the adequacy of future construction. In view of the licensee's performance history, such an

2/a As a result of staff discussions about the seriousness of such findings and of similar indications of deficiencies as identified in the Systematic Assessment of Licensee Performance Report issued in April 1982, a special Midland Section in Region III was formed in July 1982. The Midland Section devoted increased attention to inspection of the Midland facility, including upgrading the QC program of the project's constructor, the Bechtel Power Corporation.

effort was necessary to restore staff's confidence in CPCo's ability to properly construct the Midland plants.

Consequently, CPCo discussed with the NRC the concept of a construction completion program which would address the concerns raised by the staff. These discussions were followed by a formal submittal of the Midland Construction Completion Program (CCP).

The CCP is the licensee's program for the planning and management of the construction and quality activities necessary for its completion of the construction of the Midland facility. An important aspect of the CCP is the third party overview, which is designed to provide additional assurance as to the effectiveness of the CCP. In response to comments from the NRC and members of the public, the CCP underwent several revisions. As revised and submitted by the licensee on August 26, 1983,<sup>4/</sup> the CCP includes: (1) NRC hold points; (2) the requirement for 100% reinspection of accessible installations; (3) the integration of Bechtel's QC program with MPQAD; (4) the retraining and recertification of QC inspectors; (5) the general training of licensee and contractor personnel in quality requirements for nuclear work, requirements of the CCP, safety orientation and inspection, and work procedures; (6) the revision, as necessary, of Project Quality Control Instructions (PQC1's); (7) CCP team training; and (8) an independent third party overview of CCP activities.

<sup>4/</sup> The Petition was apparently based upon the June 3, 1983 version of the CCP. Subsequent versions of the CCP, as described in this decision, address a number of issues raised by petitioners.

The CCP is divided into two phases. Phase 1 consists of a systematic review of the safety-related systems and areas of the plant. This review will be conducted on an area-by-area basis and will be done by teams with responsibility for particular systems. Phase 1 is intended to provide a clear identification of remaining installation work, including any necessary rework and an up-to-date inspection to verify the quality of existing work.

Phase 2 will take the results of the Phase 1 review and complete any netessary work or rework, thereby bringing the project to completion. The teams organized for Phase 1 activities will continue as the responsible organizational units to complete the work in Phase 2.

It should be noted that the CCP does not include the remedial soils program, nuclear steam supply system installation, HVAC installation, and the reinspection of pipe hangers and electrical cable. The remedial soils activities are being closely inspected under the conditions of the construction permits which implement the Atomic Safety and Licensing Board's April 30, 1982, order and under a work authorization procedure. Therefore, the staff does not consider it necessary to require the remedial soils activities to be included in the CCP. Controls over the soils work have been implemented under a separate program. Similarly, reinspection of the pipe hangers and electrical cable were not included in Phase I of the CCP begause that reinspection is being done under a separate commitment to the NRC. See letters from James G. Keppler, Regional Administrator NRC Region III to James W. Cook, Consumers Power Company (August 30, September 2, 1982). Auclear Steam Supply System installation and HV/C installation were not drawn into question by the diesel generator building inspection.

2.0

The staff has not developed facts to indicate that installation of these systems should be included in the CCP. However, these activities will be included in the construction implementation overview to be conducted by the third party overviewer.

The CCP is designed to address the generic applicability of the proclems identified by the NRC's inspection of the diesel generator building. The objective of the CCP is to look at the plant hardware and equipment, identify existing problems, correct these problems and complete construction of the plant.

## Consideration of Issues Raised

= :

# 1. Modification of Midland Construction Permits

Petitioners request that the Commission modify the Midland construction permits in two respects: 1) require "hold points" at various stages of the construction completion process; and, 2) incorporate those hold points concerning remedial soils work previously authorized by the Atomic Safety and Licensing Board panel with jurisdiction over the Midland proceeding.

The hold points are fundamental elements of the Midland CCP. As used by both the staff and petitioners, hold points refer to predetermined stages beyond which activities cannot proceed until authorized. Only when such prior work is found to be satisfactory will new work be authorized under the CCP. In this regard, the petitioners requested that three specific hold points be incorporated into the CCP to require NRC or third party review prior to continuation of work.
Eased on their review of an early version of the CCP, petitioners asserted that the Midland project had been detrimentally affected by the lack of organizational freedom for its QA staff. See Petition at 13. Accordingly, the petitioners requested that a hold point be incorporated into the CCP whereby the success of the proposed program for the retraining and recertification of QA/QC personnel would be evaluated before any actual work was authorized under Phase 1 of the CCP. Id. at 13, 15. Subsequent to its initial discussions with the staff concerning development of a comprehensive construction completion program, 5/ the licensee began preliminary work, such as team training and recertification of OC inspectors in preparation for its anticipated Phase 1 activities, quality verification program and status assessments. The NRC was informed when training and recertification of QA/QC personnel and CCP team training would begin, and conducted a review of the licensee's actions. The staff suggested that the licensee undertake additional work before proceeding with some of its training effort. Consequently, the retraining hold point requested by petitioners has already been satisfied by the staff.

5/ On December 2, 1982, when CPCo first discussed a construction completion plan with the NRC staff, CPCo was informed by Region III staff that it would be necessary to incorporate NRC hold points. The staff identified four points at which it would require NRC-inspectors to review-completed work before the next activity could be undertaken. These hold points were identified as:

- Review and approval of training and recertification of QC inspectors before beginning Phase 1;
- 2. Review and approval of CCP team training before beginning Phase 1;
- Review and approval of the Quality Verification Program (QVP) and status assessments before beginning Phase 1;
- Review and approval of the program for rework or systems completion work before beginning Phase 2.

The petitioners also viewed the proposed CCP as lacking in comprehensiveness. To remedy this deficiency, petitioners proposed that "either a third party or NRC 'hold point' be contained in the reinspection Phase I activities [of the CCP] to determine the adequacy of the 'accessible systems' approach.  $\frac{6}{-1}$  Petition at 13.

As described in section three, <u>infra</u>, a third party will be conducting an extensive overview of the CCP and other construction completion activities. The fact that the third party overviewer will also have hold point controls over the licensee should provide additional assurance that construction is proceeding in accordance with all applicable requirements. <u>See</u> Consumers Fower Company, Construction Completion Program (August 26, 1983) at 34. The NRC and the third party will monitor the reinspection activities. The staff believes that these monitoring activities will provide the control sought by the petitioners in their request to establish a hold point during Phase 1 reinspection to determine the adequacy of the accessible systems approach.

The third hold point requested by petitioners derives from another criticism of the proposed CCP - the failure of that plan to specify inspection procedures and evaluation criteria. See Petition at 10-11. Accordingly, petitioners request a systematic and thorough review of the construction and quality work packages which will be completed as a prerequisite to initiation of new construction work under Phase 2 of the CCP. Id. at 11.

<sup>6/</sup> The accessible systems approach refers to the cont of reinspection under the CCP. Inaccessible areas of the plant will be reinspected by utilizing a records review and destructive and non-destructive testing as required. See Consumers Power Company, Construction Completion Program (August 26, 1983) at 22-23.

The CCP requires that representative construction and quality work packages be reviewed to assure that any completed work is consistent with statements made by the licensee in both its Final Safety Analysis Report and Quality Assurance Topical Report. In addition, the third party overviewer will be using sampling techniques and reviewing selected work and quality packages prior to and during Phase II. Should the results of this sampling approach identify inadequate work packages, the sampling size will be increased as necessary to provide the needed assurance that work packages are adequately reviewed. Moreover, the NRC staff, in performing its inspection activities, will overview this entire process, including reviewing selected quality and work packages.

In summary, the staff believes that those hold points it has incorporated into the CCP, when viewed in the aggregate, substantially satisfy the hold points requested by petitioners. The licensee is required to adhere to these hold points as part of the CCP in conformance with the Confirmatory Order for Modification of Construction Permits (Effective Immediately).

With respect to the second aspect of the requested relief, incorporation of NRC hold points authorized by the Licensing Board's April 30, 1982, Memorandum and Order, the petitioners' request has been satisfied by previous action of the Commission. By amendment dated May 26, 1982, the hold points ordered by the Board were incorporated into the construction permits. <u>See</u> 47 Fed. Rep. 23999 (June 2, 1982). Accordingly, the construction permits already prohibit CPCo from performing the following activities without "explicit prior approval" from the staff:

> (a) any placing, compacting, excavating, or drilling soil materials around safety-related structures and systems;

- (b) physical implementation of remedial action for correction of soil-related problems under and around safety-related structures and systems, including but not limited to:
  - (i) dewatering systems
  - (ii) underpinning of service water building
- (iii) removal and replacement of fill beneath the feedwater isolation valve pit areas, auxiliary building electrical penetration areas and control tower, and beneath the turbine building
- (iv) placing of underpinning supports beneath any of the structures listed in (iii) above
- (v) compaction and loading activities;
- (c) construction work in soil materials under or around
  safety-related structures and systems such as field installation, or rebedding, of conduits and piping.

Construction Permits No. CPPR-81 and CPPR-82, Amendment No. 3 (May 26, 1982).

#### 2. Management audit of CPCo

The petitioners request that the NRC require a management audit of CPCo's performance on the Midland project. The staff does not believe that a management audit is necessary at this time as a condition for going forward with the CCP. The staff expects that the CCP, with its built-in hold points and third party overview, should provide an effective-process to satisfactorily complete construction at Midland, without the previous quality assurance problems. The third party overview together with the planned staff inspection activities should provide information to determine the adequacy of the licensee's implementation of the CCP. Nevertheless, the staff will continue to review information concerning the licensee's performance impother areas to determine whether an audit is required.

### Rejection of Construction Completion Program and Third Party Overview Organization

In requesting that the Commission reject the Midland construction completion plan, petitioners based their position on the unacceptability of the Stone and Webster Engineering Corporation (S&W) to conduct the third party overview of the CCP. Petitioners raised three objections to the selection of S&W: the failure of S&W to meet the Commission's criteria for the independence required of a third party, see Petition at 19; the failure of S&W to submit a minimally acequate audit proposal, <u>id</u>. at 18-19; and the lack of public participation in the selection of S&W as the third party review organization for the Midland project. Id. at 19-20.

In support of its argument that S&W is not sufficiently independent to monitor implementation of the CCP, the petitioners asserted that "under both a literal and realistic reading of the Commission's primary financial criteria, ...the third party not have any direct previous involvement with the Company." Petition at 19. In order to evaluate whether an audit organization is sufficiently independent to conduct a third party review, the Commission generally utilizes the guidance originally set forth in a letter from Chairman Palladino to Representatives Ottinger and Dingell. The Commission's standard does not require that a proposed third party reviewer have had no previous involvement with the utility whose program it will be reviewing. Rather, the criteria require that the audit organization, including those employees who will be participating in the third party review, will not be reviewing specific

activities in which they were previously involved. See Letter from Chairman Palladino to Representatives Ottinger and Dingell (Feb. 1, 1982), Attachment 1, at 1. Petitioners stated that S&W's role as the overviewer of remedial soils work at Midland prohibits that organization from serving in the same capacity for the CCP. The staff disagrees. Since the remedial soils activities are outside the scope of the CCP, S&W will not be called upon to review its own work. Consequently, the staff does not agree that S&W's overview activities will conflict with the established independence criteria. $\frac{7}{}$ 

7/ The petitioners questioned why TERA was disqualified from consideration as the overviewer under the CCP while Saw was not disqualified on the ground of independence. See Petition at 19. TERA's discualification was based on the potential for conflict trat could be raised by TERA overview under the CCP of determinations that TERA had previously made under the Independent Design and Construction Verification Program (IDCVP) of the adequacy of the construction of the Auxiliary Feedwater System, the onsite emergency AC power supplies and the HVAC system for the control room. Since TERA has been approved by the NRC to perform the IDCVP, the staff determined that TERA would not satisfy the Commission independence criteria for the third party overview of the CCP. See letter from James G. Keppler, Regional Administrator, Region III to James W. Cook, Consumers Power Company (March 28, 1983) at 3.

14

2...

The written program documents being utilized to directly control and implement the Construction Implementation Overview (CIO) program<sup>8/</sup> and the applicable S&W corporate master program documents<sup>9/</sup> have been reviewed by the staff. These documents are representative of the scope and depthof the S&W overview. The NRC staff also met with S&W on August 25, 1983, in Midland, Michigan in order to gain additional insight into the total S&W program. Based upon its document review and discussions with S&W at the August 25, 1983, meeting, the staff has found the S&W proposal to constitute an acceptable third party overview program. To provide additional assurance that the third party audit is being properly implemented, the CIO program will also be audited independently by the S&W corporate cuality assurance staff. NRC inspectors will also monitor the adequacy of the CIO program.

2/ The documents written expressly for the CIO include:

- 1. CIO Program Document dated April 1, 1983.
- 2. CIO Quality Assurance Plan.
- 3. Third Party CIO Plan.
- CIO Assessment Procedure, 10.01.
- 5. Nonconformance Identification and Reporting Procedure, 15.01.
- A detailed attribute checklist for each CPCo-Project Quality--Control Instruction (PQCI).
- A detailed checklist to review generic types of requirements (for non-PQCI activities); e.g., QA Audits and Surveillances.

 Additional Quality Control Instruction as needed to provide adequate overview control.

- 9/1 The following S&W corporate master program documents will also be utilized for the CIO, as required:
  - QA Topical Report SWSQAP 1-74A, S&W Standard Nuclear Guality Assurance Program.
  - S&W-Guality Standards; e.g., for quality sampling.
  - S&W Quality Assurance Directives.

Of particular concern to the petitioners was the number of personnel which . S&W had assigned to the Midland overview. <u>See</u> Petition at 18. The number of qualified people will vary with the demand of the work activities to be overviewed. S&W's CIO staffing plan currently has nine people assigned at the Midland site and there are planned increases to 32 people as work activities progress. These numbers, however, are only estimates and S&W has represented that it will commit whatever personnel are necessary to conduct the CIO. Furthermore, the number of personnel utilized by S&W is not subject to limitation by CPCo.

Say has already begun to review preliminary activities of the licensee in preparation for initiation of the CCP. $\frac{10}{}$  This effort has identified various concerns and one nonconformance that required CPCo action to resolve. The NRC staff has reviewed the CIO activities performed to date and has found this overview, including actions taken by CPCo, to be of the quality expected of a third party overview.

10/ The activities being overviewed have included the following CCP and non-CCP activities:

- . Program and procedure reviews.
- . Review of MPQAD QA/QC personnel training and certification.
- Review of general training of CPCo and Bechtel personnel, including construction craftspersons.
   Review of CCP Management Reviews.
   Review of System Interaction Walkdowns.
- Review of Design Documents.

17

auf's

1.0

The purpose of the independent third party overview is to provide additional assurance that the CCP is adequate and will be properly implemented. This overview requirement was necessitated by the loss of NRC staff confidence in CPCo to successfully implement a quality assurance program for the Midland project. The CIO will remain in place at the Midland site until the necessary level of confidence in the ability of the licensee to construct the Midland is project has been restored to the satisfaction of the NRC staff.  $\frac{11}{}$  Given that the third party overview is expected to continue until NRC confidence in the Midland project is restored, petitioners' criticism that the CIO is of insufficient duration appears unfounded.

Cppertunity has been provided to the public to participate in the selection of S&W as the third party overviewer, and to comment on the CCP itself. A meeting was held on February 8, 1983, between CPCo and the staff to discuss the CCP. On August 11, 1983, the staff met with the intervernors, representatives of the Government Accountability Project (GAP) and the Lone Tree Council to discuss the CCP and the CIO. Subsequently, on August 25, 1983, the staff met with S&W to discuss the CIO. These meetings were conducted in Midland, Michigan and were open to public observation. Evening sessions to receive public comments regarding the CCP were held on February 8, and August 11, 1983. Similarly, public comments were received following the August 11 and August 25, 1983, meetings. Several additional meetings between the staff, intervenors and a representative of GAP to discuss the CCP and CIO have also been held.

7 The staff anticipates that the third party overview will be a long term effort.

The petitioners' reference in its request to "closed door" meetings appears to refer to working level meetings that have been held principally between the Midland section of the Region III staff and CPCo site personnel, and, in some cases, S&W onsite personnel. <u>See</u> Petition at 19. Such meetings continue to be necessary to enable the NRC staff to achieve a full understanding of the CCP, including the CIO, and to discharge its inspection duties.

For the reasons set forth above, petitioners' request to reject the selection of S&W to conduct the CIO, and to reject the CCP, is denied.  $\frac{12}{}$ 

## 4. Removal of the Licensee from Primary Responsibility for the Midland Uselity Assurance Program

The petitioners request that MPQAD be relieved of responsibility for the QA/QC function at the Midland plant and that an independent team of QA/QC personnel be created which would report simultaneously to the NRC staff and CPCo. In support of their request, petitioners cite much of the same history of QA/QC deficiencies that the staff summarized in the background section of this decision. See Petition at 20.

12/ The staff has approved S&W to conduct the CIO. See Staff Evaluation of Consumers Power Company Proposal to Use-Stone-and Webster Nichigan, Inc. to Conduct the Third Party Construction Implementation Overview of the Midland Nuclear Plant (Sept. 29, 1983).

The petitioners' reference in its request to "closed door" meetings appears to refer to working level meetings that have been held principally between the Midland section of the Region III staff and CPCo site personnel, and, in some cases, S&W onsite personnel. See Petition at 19. Such meetings continue to be necessary to enable the NRC staff to achieve a full understanding of the CCP, including the CIO, and to discharge its inspection duties.

For the reasons set forth above, petitioners' request to reject the selection of S&W to conduct the CIO, and to reject the CCP, is denied.  $\frac{12}{}$ 

## 4. <u>Removal of the Licensee from Primary Responsibility for the Midland</u> Luality Assurance Program

The petitioners request that MPQAD be relieved of responsibility for the QA/QC function at the Midland plant and that an independent team of QA/QC personnel be created which would report simultaneously to the NRC staff and CPCo. In support of their request, petitioners cite much of the same history of QA/QC deficiencies that the staff summarized in the background section of this decision. See Petition at 20.

12/ The staff has approved S&W to conduct the CIO. See Staff Evaluation of Consumers Power Company Proposal to Use-Stone-and Webster Nichigan, Inc. to Conduct the Third Party Construction Implementation Overview of the Midland Nuclear Plant (Sept. 29, 1983).

The changes that CPCo has most recently instituted through development of the CCP should improve its capability to discharge its responsibility under applicable Commission regulations, such as 10 CFR 50.34(a)(7) and Appendix B to 10 CFR Part 50, which require the establishment and execution of a QA/QC program. While Criterion I of Appendix B permits a construction permit holder to delegate to other organizations the detailed execution of the QA/QC program, the history of the Midland project makes it clear that the licensee has retained too little control over the QA/QC program. CPCo seems to be proceeding in a positive direction by integrating the implementation of the QC function formerly under the control of Bechtel into the MPQAD. This consolidation of quality control and quality assurance functions should reinforce the separation between the QC function, which will be assumed by MPQAD, and the construction function, which will remain with Bechtel.

While it might be permissible under Appendix B to 10 CFR Part 50 for CPCo to retain an independent organization to execute the QA/QC program, the licensee remains ultimately responsible for the establishment and execution of the program. As stated above, the staff considers the strengthening of MECAD to be a positive step in improving CPCo's capability to assure the cuality of construction of the Midland facility. In view of the relatively short existence of the MPQAD, there does not currently exist any justification for requiring CPCo to retain an outside organization to execute the QA/QC program. Therefore, this aspect of petitioners' request is denied.

Petitioners also requested that the independent QA/QC team report simultaneously to the NRC and to CPCo management. The petitioners apparently intended that

the NRC would be involved in making management decisions regarding construction of the facility based upon the reports of the independent QA/QC team. There appears to be no basis for this extraordinary departure from the NRC's regulatory function. Accordingly, this aspect of the petition is denied.

#### 5. Detailed Review of Soils Settlement Resolution

The petitioners requested that the staff conduct a detailed review of the resolution of the soils settlement problems, including a technical analysis of the implementation of the underpinning project at the current stage of completion. Petition at 23. In its supporting discussion, the petition focused upon the questionable structural integrity of the diesel generator building.

A detailed review of the program for resolution of the soils settlement problem has previously been conducted by the NRC staff and its consultants. In 1979 the U.S. Army Corps of Engineers was contracted to assist the staff in the safety review of the Midland project in the field of geotechnical engineering. After the soils problem became known, additional assistance to the staff in specialized engineering fields (structural, mechanical, and underpinning) was obtained from the U.S. Naval Surface Weapons Center, Harstead Engineering Associates, Geotechnical Engineers, Inc., and Energy Technology Engineering Center. These consultants assisted in the review of technical studies, participated in design audits, visited the site, provided input to the Safety Evaluation Report, and provided expert testimony before the Atomic Safety and

Licensing Board. Thus, the approach to the resolution of the soils settlement issue has been thoroughly studied by the staff and its consultants.

The implementation of the remedial soils activities is being closely\_followed as part of the NRC's inspection program. This inspection effort includes ongoing technical review of the remedial soils program and its implementation is by a Region III soils specialist. Technical expertise to evaluate implementation is also provided by the NRC's Office of Nuclear Reactor Regulation. Additionally, the NRC is utilizing Geotechnical Engineers Inc. in assessing aspects of the remedial soils and underpinning activities. In addition, the soils rettlement question has been in litigation for over two years before an Atomic Safety and Licensing Board. Consequently, the relief requested with regard to the soils settlement issue has been substantially satisfied by prior action of the Commission.

Along with review of the soils settlement issue, petitioners requested that another study of the seismic design deficiencies of the Midland plant, with emphasis on the diesel generator building, be conducted. The petitioners further requested that this review would be conducted by a "non-nuclear construction consultant." See Petition at 23.

The NRC staff has initiated a task force study by consultants from Brookhaven National Laboratory (BNL) and NRC structural engineers to evaluate concerns about the structural integrity of the diesel generator building raised by a NRC Region III inspector in testimony before the Subcommittee on Energy and the Environment of the House Committee on Interior and Insular Affairs. Following their review, a report will be issued addressing the concerns raised by the inspector. Decisions on whether further actions are required will be

made based upon that report. Additional details on the task force were provided to the Government Accountability Project by letter dated August 10, 1983, and in Board Notifications 83-109 and 83-142, which were transmitted to GAP on July 27 and September 22, 1983, respectively.

As to the request that a review of the diesel generator building be conducted by a "non-nuclear construction consultant", BNL has established an expert team to resolve the concerns raised by the Region III inspector. Expertise rather than the label "non nuclear construction consultant" should be the governing criteria. The staff has reviewed the qualifications of the team removes and is satisifed with their experience. The task force study currently in progress substantially satisfies this aspect of the petition.

The petition also appears to be requesting an additional review of the seismic design of structures other than the diesel generator building. Petitioners have not, however, stated any basis why additional reviews beyond those reflected in the Safety Evaluation Report and Supplements are necessary. The staff does not believe that an additional review by an outside organization of the facility's seismic design is required at this time.

### Conclusion

Based upon the foregoing discussion, I have granted the petition in part and defied it in part.

A copy of this decision will be filed with the Office of the Secretary of the Commission for the Commission's review in accordance with 10 CFR 2.206(c) of the Commission's regulations. This decision will become the final action of the Commission twenty-five days after date of issuance unless the Commission, on\_its own motion, institutes a review of the decision within that time.

Richard C: DeYoung, Dimector Office of Inspection and Enforcement

Dated at Bethesda, Maryland, this 6thday of October 1983

14 1 1

NUCLEAR REGULATORY COMMISSION [Docket Nos. 50-329 and 50-330] CONSUMERS POWER COMPANY (Midland Plant, Units 1 and 2)

# ISSUANCE OF DIRECTOR'S DECISION UNDER 10 CFR 2.206

Notice is hereby given that the Director, Office of Inspection and Enforcement, has issued a decision concerning a petition dated June 13, 1983, filed by Billie Pirner Garde of the Government Accountability Project on behalf of the Lone Tree Council act others. The petitioners had requested that the Commission take a number of actions with respect to the Midland Plant. The Director, Office of Inspection and Enforcement, has decided to grant in part and deny in part the petitioners' request.

The reasons for this decision are explained in a "Director's Decision" under 10 CFR 2.206 (DD-83-16), which is available for public inspection in the Commission's Public Document Room, 1717 H Street, N.W., Washington, D.C. 20555, and in the Local Public Document Room for the Midland Plant, located at the Grace Dow Memorial Library, 1910 W. St. Andrews Road, Midland, Michigan, 48640.

Dated at Bethesda, Maryland this 6th day of October, 1983.

310200392

FOR THE NUCLEAR REGULATORY COMMISSION

T7590-017

Richard C. Peyoung, Director Office of Inspection and Enforcement



UNITED STATES NUCLEAR REGULATORY COMMISSION REGION III 799 ROOSEVELT ROAD GLEN ELLYN, ILLINOIS 60137 LIJS NO. FO30 19973 Lore In Coural

file

MAY 1 3 1083

Lone Tree Council ATTN: Mr. Leo R. Romo Corresponding Secretary P. O. Box 421 Essexville, MI 48732

Dear Mr. Romo:

This is in response to the Lone Tree Council's letters of April 21, 1983 to Chairman Palladino, D. G. Eisenhut, and me regarding reinspections and independent audits to identify problems at the Midland Nuclear Power Plant.

During the public meeting in Midland on February 8, 1983, between the NRC and Consumers Power Company, the licensee described its proposed Construction Completion Program which is designed to identify and correct past deficiencies at the plant while establishing a program for future construction. The Construction Completion Program will include two independent third-party firms. On May 3, 1983, the NRC staff approved TERA Corporation to perform an independent review of the design and construction of the Auxiliary Feedwater System. The NRC is still reviewing Consumers Power Company's proposal to have TERA review the design and construction of part of the emergency power system and part of the heating, ventilation, and air conditioning system. The second independent third-party will overview the Construction Completion Program, including the licensee's reinspections. Consumers Power Company has proposed Stone and Webster Engineering Company to conduct this overview and the NRC is currently reviewing the acceptability of this proposal.

Although separate from the Construction Completion Program, Consumers Power has also employed Stone and Webster to overview the remedial work on the soil settlement problems at Midland. The NRC has approved Stone and Webster for this overview.

We believe that the plans for these third party inspections are appropriate and sufficient at this time.

8305170392

Lone Tree Council

As a point of clarification, the NRC has not ordered a reinspection at Midland. That is a voluntary commitment made by the licensee as part of their Construction Completion Program.

We disagree with your position that letting Consumers Power Company reinspect their own work makes a mockery of the NRC's commitment to ensuring safe construction, particularly in view of the third party inspections and other actions being taken under the CCP to improve performance. Furthermore, the NRC will be inspecting this effort.

I trust that this has been responsive to your comments. If you have any further comments or questions, please do not hesitate to contact me.

> Original signed by James G. Keppler

James G. Keppler Regional Administrator

cc: V. Stello, EDO H. Denton, IE R. DeYoung, NRR D. Eisenhut, NRR DME/Document Control Desk (RIDS) Resident Inspector, RIII The Honorable Charles Bechhoefer, ASLB The Honorable Jerry Harbour, ASLB The Honorable Frederick P. Cowan, ASLB The Honorable Ralph S. Decker, ASLB William Paton, ELD Michael Miller Ronald Callen, Michigan Public Service Counission Myron M. Cherry Barbara Stamiris Mary Sinclair Wendell Marshall Colonel Steve J. Gadler (P.E.) Howard Levin, TERA Billie P. Garde, Government Accountability Project

OFFICE	RULA		NRR 114	XAZ	211	A	X
SURNAME	Harison/ls	Watnick	Hood Or Gr	Lewis	Strama	Davis	Keppler
DATE	.5/.11/.83	5.1183	51283	5/12/83	51283	5/12	5/13/83